UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 10-Q

☑ QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT
OF 1934

For the quarterly period ended September 30, 2013

TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from ______ to _____

Commission File Number: 001-34778

QEP RESOURCES, INC.

(Exact name of registrant as specified in its charter)

STATE OF DELAWARE (State or other jurisdiction of incorporation or organization) 87-0287750 (I.R.S. Employer Identification No.)

1050 17th Street, Suite 500, Denver, Colorado 80265 (Address of principal executive offices)

Registrant's telephone number, including area code (303) 672-6900

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes \boxtimes No o

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate website, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (Section 232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes \boxtimes No o

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer," and "smaller reporting company" in Rule 12b-2 of the Exchange Act:

Large accelerated filer⊠Accelerated fileroNon-accelerated filero (Do not check if a smaller reporting company)Smaller reporting companyo

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes o No 🗵

At September 30, 2013, there were 179,281,102 shares of the registrant's common stock, \$0.01 par value, outstanding.

QEP Resources, Inc. Form 10-Q for the Quarter Ended September 30, 2013

TABLE OF CONTENTS

			Page
PART I. I	FINANCIAL II	<u>NFORMATION</u>	<u>2</u>
	ITEM 1.	FINANCIAL STATEMENTS	<u>2</u>
		UNAUDITED CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS FOR THE THREE AND	
		NINE MONTHS ENDED SEPTMEBER 30, 2013 AND 2012	<u>2</u>
		VIVA VIDITED CONDENSED CONSOLIDATED STATENCENTS OF COMPRESSOR TO THE PARTY.	
		UNAUDITED CONDENSED CONSOLIDATED STATEMENTS OF COMPREHENSIVE INCOME FOR THE THREE AND NINE MONTHS ENDED SEPTEMBER 30, 2013 AND 2012	<u>3</u>
		THREE AND MINE MONTHS ENDED SETTEMBER 30, 2013 AND 2012	2
		UNAUDITED CONDENSED CONSOLIDATED BALANCE SHEETS AT SEPTEMBER 30, 2013 AND	
		DECEMBER 31, 2012	<u>4</u>
			_
		UNAUDITED CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS FOR THE NINE	
		MONTHS ENDED SEPTEMBER 30, 2013 AND 2012	<u>5</u>
		UNAUDITED NOTES ACCOMPANYING THE CONDENSED CONSOLIDATED FINANCIAL	
		<u>STATEMENTS</u>	<u>6</u>
	ITEM 2.	MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF	
		<u>OPERATIONS</u>	<u>27</u>
	ITEM 2	OLIANITETATIVE AND OLIALITATIVE DISCLOSUDES ADOLIT MADVET DISC	F1
	ITEM 3.	QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK	<u>51</u>
	ITEM 4.	CONTROLS AND PROCEDURES	FF
	11 EWI 4.	CONTROLS AND PROCEDURES	<u>55</u>
DA DT II	OTHER INFO	DMATION	
PARI II.	OTHER INFO	<u>RMATION</u>	<u>55</u>
	ITEM 1.	LEGAL PROCEEDINGS	<u>55</u>
	II LIVI I.	LEGAL FROCEEDINGS	<u>55</u>
	ITEM 1A.	RISK FACTORS	<u>55</u>
	11Livi 1/1,	MONTHOTORD	<u>55</u>
	ITEM 2.	UNREGISTERED SALES OF EQUITY SECURITIES AND USE OF PROCEEDS	<u>55</u>
	1111112.	ONALGISTERED STEED OF EQUIT OF ORGANITED WAY COLD OF TROOLED OF	<u>55</u>
	ITEM 3.	DEFAULTS UPON SENIOR SECURITIES	<u>55</u>
	1121115.	<u>BEMIELIO OI ON BEMION OLIGONITIED</u>	<u>55</u>
	ITEM 4.	MINE SAFETY DISCLOSURES	<u>56</u>
	ITEM 5.	OTHER INFORMATION	<u>56</u>
			<u>50</u>
	ITEM 6.	EXHIBITS	<u>57</u>
SIGNATI	LIRES		58

PART I. FINANCIAL INFORMATION

ITEM 1. FINANCIAL STATEMENTS QEP RESOURCES, INC. CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS (Unaudited)

	Three Months Ended September 30,				Nine Months Ended September 30,			
		2013	2012		2013		2012	
REVENUES			(in millions, excep	t per	share amounts)		
Natural gas sales	\$	194.8	\$ 170.3	\$	610.5	\$	470.4	
Oil sales		253.8	117.7		656.3		335.7	
NGL sales		71.3	67.5		215.0		247.0	
Gathering, processing and other		46.9	46.3		135.1		141.9	
Purchased gas, oil and NGL sales		206.0	140.6		603.4		449.9	
Total Revenues		772.8	542.4		2,220.3		1,644.9	
OPERATING EXPENSES								
Purchased gas, oil and NGL expense		205.0	142.6		608.8		455.9	
Lease operating expense		45.0	42.2		127.4		122.8	
Natural gas, oil and NGL transportation and other handling costs		34.5	36.3		105.8		111.5	
Gathering, processing and other		22.4	22.1		66.5		66.4	
General and administrative		49.3	41.7		136.2		114.5	
Production and property taxes		42.5	24.3		117.7		68.4	
Depreciation, depletion and amortization		253.1	234.6		757.1		648.3	
Exploration expenses		1.8	2.2		9.5		6.3	
Impairment		3.8	9.0		4.0		70.9	
Total Operating Expenses		657.4	555.0		1,933.0		1,665.0	
Net gain from asset sales		12.8			113.0		1.5	
OPERATING INCOME (LOSS)		128.2	(12.6)		400.3		(18.6)	
Realized and unrealized (losses) gains on derivative contracts (See Note 8)		(27.8)	36.1		51.6		334.7	
Interest and other income (loss)		4.0	(0.2)		9.1		2.4	
Income from unconsolidated affiliates		0.8	2.3		3.7		5.6	
Loss from early extinguishment of debt		_	_		_		(0.6)	
Interest expense		(41.7)	(30.0)		(122.5)		(82.9)	
INCOME (LOSS) BEFORE INCOME TAXES		63.5	(4.4)		342.2		240.6	
Income tax (provision) benefit		(22.4)	2.3		(125.0)		(86.5)	
NET INCOME (LOSS)		41.1	(2.1)		217.2		154.1	
Net income attributable to noncontrolling interest		(3.8)	(1.0)		(5.8)		(2.7)	
NET INCOME (LOSS) ATTRIBUTABLE TO QEP	\$	37.3	\$ (3.1)	\$	211.4	\$	151.4	
	_			_				
Earnings Per Common Share Attributable to QEP								
Basic total	\$	0.21	\$ (0.02)	\$	1.18	\$	0.85	
Diluted total	\$	0.21	\$ (0.02)		1.18	\$	0.85	
Weighted-average common shares outstanding								
Used in basic calculation		179.3	177.9		179.2		177.6	
Used in diluted calculation		179.5	177.9		179.4		178.6	
Dividends per common share	\$	0.02	\$ 0.02	\$	0.06	\$	0.06	

See notes accompanying the condensed consolidated financial statements.

QEP RESOURCES, INC. CONDENSED CONSOLIDATED STATEMENTS OF COMPREHENSIVE INCOME (Unaudited)

		Three Mor Septen				Nine Mon Septen	-	
		2013		2012		2013		2012
				(in mi	llions)			
Net income (loss) attributable to QEP	\$	41.1	\$	(2.1)	\$	217.2	\$	154.1
Other comprehensive income (loss), net of tax:								
Reclassification of previously deferred derivative gains ⁽¹⁾		(19.7)		(42.1)		(60.4)		(133.8)
Pension and other postretirement plans adjustments:								
Amortization of net actuarial loss (2)		0.4		0.5		1.1		0.7
Amortization of prior service cost (3)		0.8		0.9		2.5		2.6
Total pension and other postretirement plans adjustments		1.2		1.4		3.6		3.3
Other comprehensive loss		(18.5)		(40.7)		(56.8)		(130.5)
Comprehensive income (loss)		22.6		(42.8)		160.4		23.6
Comprehensive income attributable to noncontrolling interests				(1.0)		(5.8)		(2.7)
Comprehensive income (loss) attributable to QEP	\$	18.8	\$	(43.8)	\$	154.6	\$	20.9

⁽¹⁾ Presented net of income tax benefit of \$11.7 million and \$35.8 million during the three and nine months ended September 30, 2013 and \$24.9 million and \$79.2 million during the three and nine months ended September 30, 2012, respectively.

See notes accompanying the condensed consolidated financial statements.

Presented net of income tax expense of \$0.2 million and \$0.7 million during the three and nine months ended September 30, 2013 and \$0.2 million and \$0.4 million during the three and nine months ended September 30, 2012, respectively.

Presented net of income tax expense of \$0.5 million and \$1.5 million and during the three and nine months ended September 30, 2013 and \$0.5 million and \$1.6 million during the three and nine months ended September 30, 2012, respectively.

QEP RESOURCES, INC. CONDENSED CONSOLIDATED BALANCE SHEETS (Unaudited)

	September 30, 2013	December 31, 2012
ASSETS	(in m	illions)
Current Assets		
Cash and cash equivalents	\$ 123.0	\$ —
Accounts receivable, net	321.9	387.5
Fair value of derivative contracts	39.4	188.7
Gas, oil and NGL inventories, at lower of average cost or market	13.1	13.1
Prepaid expenses and other	49.4	68.0
Deferred income taxes		
Total Current Assets	546.8	657.3
Property, Plant and Equipment (successful efforts method for gas and oil properties)		
Proved properties	11,038.8	10,234.3
Unproved properties	911.8	937.9
Midstream field services	1,670.4	1,634.9
Marketing and resources	80.4	64.6
Material and supplies	63.4	61.9
Total Property, Plant and Equipment	13,764.8	12,933.6
Less Accumulated Depreciation, Depletion and Amortization	'	
Exploration and production	4,747.4	4,258.1
Midstream field services	394.7	357.9
Marketing and resources	20.3	18.1
Total Accumulated Depreciation, Depletion and Amortization	5,162.4	4,634.1
Net Property, Plant and Equipment	8,602.4	8,299.5
Investment in unconsolidated affiliates	39.0	41.2
Goodwill	59.5	59.5
Fair value of derivative contracts	5.8	4.1
Other noncurrent assets	53.2	46.9
TOTAL ASSETS	\$ 9,306.7	\$ 9,108.5
	ψ 3,300.7	3,100.3
LIABILITIES AND EQUITY Current Liabilities		
	\$ 1.6	\$ 39.7
Checks outstanding in excess of cash balances Accounts payable and accrued expenses	\$ 1.6 593.6	\$ 39.7 643.4
Production and property taxes	58.0	41.8
Interest payable	34.6	36.9
Fair value of derivative contracts	10.1	2.6
Deferred income taxes	9.5	5.0
	707.4	769.4
Total Current Liabilities		
Long-term debt	2,882.3	3,206.9
Deferred income taxes	1,494.7	1,493.5
Asset retirement obligations	168.4	191.4
Fair value of derivative contracts Other long term liabilities	129.0	3.6
Other long-term liabilities	129.0	130.0
Commitments and contingencies (see Note 11)		
EQUITY Common stock - par value \$0.01 per share; 500.0 million shares authorized;		
179.3 million and 178.5 million shares issued, respectively	1.8	1.8
Treasury stock - 0.4 million and 0.1 million shares, respectively	(14.7)	(3.7)
Additional paid-in capital	489.3	462.1
Retained earnings	2,973.7	2,773.0
Accumulated other comprehensive (loss) income	(24.0)	32.8
Total Common Shareholders' Equity	3,426.1	3,266.0
Noncontrolling interest	498.8	47.7
Total Equity	3,924.9	3,313.7
TOTAL LIABILITIES AND EQUITY	\$ 9,306.7	\$ 9,108.5

See notes accompanying the condensed consolidated financial statements.

QEP RESOURCES, INC. CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (Unaudited)

	Nine N	Ended	
	Sep	r 30,	
	2013		2012
	(in	millio	ons)
OPERATING ACTIVITIES			
Net income	\$ 217.	2 \$	154.1
Adjustments to reconcile net income to net cash provided by operating activities:			
Depreciation, depletion and amortization	757.	1	648.3
Deferred income taxes	39.	3	54.7
Impairment	4.	0	70.9
Share-based compensation	20.	0	19.5
Amortization of debt issuance costs and discounts	4.	7	3.7
Net gain from asset sales	(113.	0)	(1.5)
Income from unconsolidated affiliates	(3.	7)	(5.6)
Distributions from unconsolidated affiliates and other	5.	9	6.1
Unrealized loss (gain) on derivative contracts	55.	5	(32.8)
Changes in operating assets and liabilities	(8.	1) _	54.6
Net Cash Provided by Operating Activities	978.	9	972.0
INVESTING ACTIVITIES			
Property acquisitions	(39.	3)	(1,400.3)
Property, plant and equipment, including dry exploratory well expense	(1,089.	6)	(1,040.7)
Proceeds from disposition of assets	208.	3	5.3
Net Cash Used in Investing Activities	(920.	6)	(2,435.7)
FINANCING ACTIVITIES			
Checks outstanding in excess of cash balances	(38.	1)	(1.9)
Long-term debt issued	<u> </u>	_	1,450.0
Long-term debt issuance costs paid	(3.	0)	(17.0)
Long-term debt repaid	_	_	(6.7)
Proceeds from credit facility	1,075.	0	933.5
Repayments of credit facility	(1,400.	0)	(876.0)
Treasury stock repurchases	(8.	7)	_
Net proceeds from the issuance of QEPM common units	449.	6	_
Other capital contributions	3.	6	(4.2)
Dividends paid	(10.	8)	(10.7)
Excess tax benefit on share-based compensation	1.	3	2.0
Distribution to noncontrolling interest	(4.	2)	(5.3)
Net Cash Provided by Financing Activities	64.	7	1,463.7
Change in cash and cash equivalents	123.	0	_
Beginning cash and cash equivalents	_	_	_
Ending cash and cash equivalents	\$ 123.	0 \$	
	-	<u> </u>	
Supplemental Disclosures:			
Cash paid for interest, net of capitalized interest	\$ 120.	1 \$	79.8
Cash paid for income taxes	49.		28.0
Non-cash investing activities:			
Change in capital expenditure accrual balance	\$ 53.	7 \$	97.5

Nine Months Ended

See notes accompanying the condensed consolidated financial statements.

QEP RESOURCES, INC. NOTES ACCOMPANYING THE CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

Note 1 - Nature of Business

QEP Resources, Inc. (QEP or the Company) is a holding company with three major lines of business: natural gas and crude oil exploration and production; midstream field services; and energy marketing. These businesses are conducted through the Company's three principal subsidiaries:

- QEP Energy Company (QEP Energy) acquires, explores for, develops, and produces natural gas, oil, and natural gas liquids (NGL);
- QEP Field Services Company (QEP Field Services, which includes the ownership and operations of QEP Midstream Partners, LP (QEP Midstream)),
 provides midstream field services, including natural gas gathering, processing, compression, and treating services, for affiliates and third parties; and
- QEP Marketing Company (QEP Marketing) markets affiliate and third-party natural gas and oil, and owns and operates an underground gas-storage reservoir.

QEP's operations are focused in two major regions: the Northern Region (primarily in North Dakota, Wyoming and Utah) and the Southern Region (primarily in Oklahoma, Louisiana and the Texas Panhandle) of the United States. QEP's corporate headquarters are located in Denver, Colorado.

Shares of QEP's common stock trade on the New York Stock Exchange under the ticker symbol "QEP".

Note 2 – Basis of Presentation of Interim Consolidated Financial Statements

The interim condensed consolidated financial statements contain the accounts of QEP and its majority-owned or controlled subsidiaries, including QEP Midstream (see Note 3 - QEP Midstream). The condensed consolidated financial statements were prepared in accordance with United States Generally Accepted Accounting Principles (GAAP) and with the instructions for quarterly reports on Form 10-Q and Regulations S-X and S-K. All significant intercompany accounts and transactions have been eliminated in consolidation.

The condensed consolidated financial statements reflect all normal recurring adjustments and accruals that are, in the opinion of management, necessary for a fair statement of financial position and results of operations for the interim periods presented. Interim condensed consolidated financial statements do not include all of the information and notes required by GAAP for audited annual consolidated financial statements. These condensed consolidated financial statements should be read in conjunction with the consolidated financial statements and notes thereto included in the Company's Annual Report on Form 10-K for the year ended December 31, 2012.

The preparation of the condensed consolidated financial statements and notes in conformity with GAAP requires that management make estimates and assumptions that affect revenues, expenses, assets and liabilities, and disclosure of contingent assets and liabilities. Actual results could differ from estimates. The results of operations for the three and nine months ended September 30, 2013, are not necessarily indicative of the results that may be expected for the year ending December 31, 2013.

New accounting pronouncements

In February of 2013, the FASB issued ASU 2013-02, *Other Comprehensive Income (Topic 220: Reporting of Amounts Reclassified Out of Accumulated Other Comprehensive Income)*, which seeks to improve the reporting of entities by requiring an entity to report the effect of significant reclassifications out of accumulated other comprehensive income on the respective line items in net income if the amount being reclassified is required under GAAP to be reclassified in its entirety to net income. For other amounts that are not required under GAAP to be reclassified in their entirety to net income in the same reporting period, an entity is required to cross-reference other disclosures required under GAAP that provide additional detail about those amounts. The amendments are effective prospectively for reporting periods beginning on or after December 15, 2012. The Company adopted this standard in the first quarter of 2013 and noted that it did not have a significant impact on the Company's consolidated financial statements.

In December of 2011, the FASB issued ASU 2011-11, *Disclosures about Offsetting Assets and Liabilities*, which enhances disclosure requirements regarding an entity's financial instruments and derivative instruments that are offset or subject to a master netting arrangement. This information about offsetting and related netting arrangements will enable users of financial

statements to understand the effect of those arrangements on the entity's financial position, including the effect of rights of setoff. Additionally, the FASB issued ASU 2013-01, *Clarifying the Scope of Disclosures about Offsetting Assets and Liabilities*, which clarifies the implementation of ASU 2011-01. The amendments are required for annual reporting periods beginning after January 1, 2013, and interim periods within those annual periods. The Company adopted this standard effective January 1, 2013. It did not have a significant impact on the Company's consolidated financial statements.

In July of 2012, the FASB issued ASU 2012-02, *Intangibles - Goodwill and Other: Testing Indefinite-Lived Intangible Assets for Impairment*, which revises the way an entity can test indefinite-lived intangible assets for impairment by allowing an entity to first assess qualitative factors to determine whether the existence of events and circumstances indicates that it is more likely than not that the indefinite-lived intangible asset is impaired. If there is no indication of impairment from the qualitative impairment test, the entity is not required to complete a quantitative impairment test of determining and comparing the fair value with the carrying amount of the indefinite-lived asset. Under the guidance in this ASU, an entity also has the option to bypass the qualitative assessment in any period and proceed directly to performing the quantitative impairment test, while retaining the ability to resume performing the qualitative assessment in any subsequent period. The amendments are effective for annual and interim impairment tests performed for fiscal years beginning after September 15, 2012. The Company adopted this standard January 1, 2013, which has allowed the Company to more efficiently complete the annual goodwill impairment test but has not had a significant impact on the Company's consolidated financial statements.

Note 3 – QEP Midstream

QEP Midstream is a publicly traded master limited partnership that was formed by QEP to own, operate, acquire and develop midstream energy assets. QEP Midstream's assets currently consist of ownership interests in four gathering systems and two Federal Energy Regulatory Commission (FERC) regulated pipelines, which provide natural gas and crude oil gathering and transportation services. These assets are located in, or within close proximity to, the Green River Basin located in Wyoming and Colorado, the Uinta Basin located in eastern Utah, and the portion of the Williston Basin located in North Dakota.

Initial Public Offering

On August 14, 2013, QEP Midstream completed its initial public offering (the Offering) of 20,000,000 common units, representing limited partner interests in QEP Midstream, at a price to the public of \$21.00 per common unit. QEP Midstream received net proceeds of \$390.7 million from the sale of the common units, after deducting underwriting discounts and commissions, structuring fees and offering expenses of approximately \$29.3 million. Following the Offering, the underwriters exercised their over-allotment option to purchase an additional 3,000,000 common units, at a price of \$21.00 per common unit, providing additional net proceeds of \$58.9 million, after deducting \$4.1 million of underwriters' discounts and commissions and structuring fees, to QEP Midstream

QEP Midstream used the net proceeds to repay its outstanding debt balance with QEP, which was assumed with the assets contributed to QEP Midstream, pay revolving credit facility origination fees and make a cash distribution to QEP, a portion of which was used to reimburse QEP for certain capital expenditures it incurred with respect to assets contributed to QEP Midstream. The following table is a reconciliation of proceeds from the initial public offering (in millions):

Total proceeds from the Offering	\$ 483.0
Offering costs	(33.4)
Net proceeds from the Offering	449.6
Revolving credit facility origination fees	(3.0)
Repayment of outstanding debt with QEP	(95.5)
Net proceeds distributed to QEP from the Offering	\$ 351.1

QEP Midstream Partners GP, LLC (the General Partner), a wholly-owned subsidiary of QEP, serves as the general partner of QEP Midstream. QEP owns a 57.8% interest in QEP Midstream and consolidates QEP Midstream for financial reporting purposes with the portion not owned by QEP reflected as a reduction to net income and equity as noncontrolling interest.

Contribution, Conveyance and Assumption Agreement

On August 14, 2013, in connection with the closing of the Offering, QEP entered into a Contribution, Conveyance and Assumption Agreement (the Contribution Agreement) with QEP Field Services, the General Partner and QEP Midstream Partners Operating, LLC (the Operating Company). Immediately prior to the closing of the Offering, the following transactions, among others, occurred pursuant to the Contribution Agreement:

- QEP Field Services contributed to the General Partner, as a capital contribution, a limited liability company interest in the Operating Company with a value equal to 2.0% of the equity value of OEP Midstream at the closing of the Offering;
- the General Partner contributed to QEP Midstream, as a capital contribution, the limited liability company interest in the Operating Company in exchange for (a) 1,090,000 general partner units representing the continuation of an aggregate 2.0% general partner interest in QEP Midstream and (b) all the incentive distribution rights of QEP Midstream;
- QEP Field Services contributed to QEP Midstream, as a capital contribution, its remaining limited liability company interests in the Operating Company in exchange for (a) 6,701,750 common units representing a 12.3% limited partner interest in QEP Midstream, (b) 26,705,000 subordinated units representing a 49.0% limited partner interest in QEP Midstream and (c) the right to receive a distribution from QEP Midstream; and
- the public, through the underwriters, contributed \$420.0 million in cash (or \$390.7 million, net of the underwriters' discounts and commissions, structuring fees and offering expenses of approximately \$29.3 million) to QEP Midstream in exchange for the issuance of 20,000,000 common units.

Subsequent to the Offering, the underwriters exercised their over-allotment option to purchase an additional 3,000,000 common units in QEP Midstream, which reduced QEP's limited partner common unit interest in QEP Midstream from 12.3% to 6.8% and QEP's total ownership interest from 63.3% to 57.8%.

Omnibus Agreement

In connection with the Offering, QEP entered into an Omnibus Agreement (the Omnibus Agreement) with QEP Midstream on August 14, 2013, that addresses the following matters:

- QEP Midstream's payment of an annual amount to QEP, initially in the amount of approximately \$13.8 million, for the provision of certain general and administrative services by QEP and its affiliates to QEP Midstream, including a fixed annual fee of approximately \$1.4 million for providing certain executive management services by certain officers of the General Partner. The remaining portion of this annual amount reflects an estimate of the costs QEP and its affiliates expect to incur in providing the services;
- QEP Midstream's obligation to reimburse QEP for any out-of-pocket costs and expenses incurred by QEP in providing general and administrative services (which reimbursement is in addition to certain expenses of the General Partner and its affiliates that are reimbursed under QEP Midstream's partnership agreement), as well as any other out-of-pocket expenses incurred by QEP on QEP Midstream's behalf; and
- an indemnity by QEP for certain environmental and other liabilities, and QEP Midstream's obligation to indemnify QEP and its subsidiaries for
 events and conditions associated with the operation of QEP Midstream's assets that occur after the closing of the Offering.

As long as QEP controls the General Partner, the Omnibus Agreement will remain in full force and effect. If QEP ceases to control the General Partner, either party may terminate the Omnibus Agreement, but the indemnification obligations will remain in full force and effect in accordance with their terms.

Fixed Price Condensate Purchase Agreement

On August 14, 2013, QEP entered into a fixed price Condensate Purchase Transaction Agreement (the "Condensate Purchase Agreement") with QEP Midstream. The Condensate Purchase Agreement has a primary term of five years and requires QEP Midstream to sell and QEP to purchase all of the condensate volumes collected on QEP Midstream's gathering systems at a fixed price of \$85.25 per barrel.

Note 4 - Acquisition and Divestitures

Acquisitions

On September 27, 2012, QEP Energy completed an acquisition of oil and gas properties in the Williston Basin for an aggregate purchase price of \$1.4 billion (the 2012 Acquisition). The properties are located in Williams and McKenzie counties of North Dakota, approximately 12 miles west of QEP's then-existing core acreage in the Williston Basin.

The 2012 Acquisition meets the definition of a business combination under ASC 805, *Business Combinations*, as it included proved properties. QEP allocated the cost of the 2012 Acquisition to assets acquired and liabilities assumed based on fair values as of the acquisition date. Revenues of \$76.0 million and \$190.1 million and net income of \$19.2 million and \$42.5 million were generated from the acquired properties during the three and nine months ended September 30, 2013, respectively, and are included in QEP's Condensed Consolidated Statements of Operations.

QEP Energy recorded the 2012 Acquisition on its Condensed Consolidated Balance Sheet. The following table presents a summary of the Company's purchase accounting entries:

	As of So	eptember, 2013
	(ir	millions)
Consideration given:		
Cash consideration	\$	1,392.7
Amounts recognized for fair value of assets acquired and liabilities assumed:		
Proved properties	\$	713.8
Unproved properties		683.4
Asset retirement obligations		(0.9)
Liabilities assumed		(4.4)
Other assets		0.8
Total fair value	\$	1,392.7

The following unaudited, pro forma results of operations are provided for the three and nine months ended September 30, 2012. These supplemental pro forma results of operations are provided for illustrative purposes only and may not be indicative of the actual results that would have been achieved by the acquired properties for the period presented or that may be achieved by such properties in the future. Future results may vary significantly from the results reflected in this pro forma financial information because of future events and transactions, as well as other factors. The pro forma information is based on QEP's consolidated results of operations for the three and nine months ended September 30, 2012, on the acquired properties' historical results of operations and on estimates of the effect of the transaction on the combined results. The pro forma results of operations have been prepared by adjusting the historical results of QEP to include the historical results of the acquired properties based on information provided by the seller and the impact of the purchase price allocation. The pro forma results of operations do not include any cost savings or other synergies that may result from the 2012 Acquisition or any estimated costs that have been or will be incurred by the Company to integrate the acquired properties.

	Three Mo				Nine Mor Septembe			
	 Actual	Pro forma		Actual		Pro forma		
	(in millions, except per share data)							
Revenues	\$ 542.4	\$	598.9	\$	1,644.9	\$	1,780.4	
Net income attributable to QEP	(3.1)		4.8		151.4		164.9	
Earnings per common share attributable to QEP								
Basic	\$ (0.02)	\$	0.03	\$	0.85	\$	0.93	
Diluted	(0.02)		0.03		0.85		0.93	

Divestitures

In June 2013, QEP Energy sold its interests in several non-core oil and gas properties located in QEP's Northern Region for total cash proceeds of \$140.2 million and recorded a pre-tax gain on sale of \$97.4 million. In September 2013, QEP Energy sold its interests in several non-core properties located in QEP's Southern Region for total cash proceeds of \$68.8 million and recorded a pre-tax gain on sale of \$17.5 million. Both the cash proceeds and gains on sales are subject to post-closing adjustments. During the quarter and nine months ended September 30, 2013, QEP Energy recorded these gains on its Condensed Consolidated Statement of Operation in "Net gain from asset sales".

Note 5 – Earnings Per Share

Basic earnings per share (EPS) are computed by dividing net income attributable to QEP by the weighted-average number of common shares outstanding during the reporting period. Diluted EPS includes the potential increase in the number of outstanding shares that could result from the exercise of in-themoney stock options. QEP's unvested restricted shares are included in weighted-average basic common shares outstanding because once the shares are granted, the restricted shares are considered issued and outstanding, the historical forfeiture rate is minimal and the restricted shares receive dividends.

Unvested share-based payment awards that contain non-forfeitable rights to dividends or dividend equivalents are considered participating securities and are included in the computation of earnings per share pursuant to the two-class method. The Company's unvested restricted stock awards contain non-forfeitable dividend rights and participate equally with common stock with respect to dividends issued or declared. However, the Company's unvested restricted stock does not have a contractual obligation to share in losses of the Company. The Company's unexercised stock options do not contain rights to dividends. Under the two-class method, the earnings used to determine basic earnings per common share are reduced by an amount allocated to participating securities. When the Company records a net loss, none of the loss is allocated to the participating securities since the securities are not obligated to share in Company losses. Use of the two-class method has an insignificant impact on the calculation of basic and diluted earnings per common share. There were no anti-dilutive shares during the three and nine months ended September 30, 2012, During the three months ended September 30, 2012, 0.8 million shares were not included in diluted common shares outstanding as they were anti-dilutive due to QEP's net loss.

A reconciliation of the components of basic and diluted shares used in the EPS calculation follows:

	Three Mor	nths Ended	Nine Mon	ths Ended		
	Septem	iber 30,	September 30,			
	2013	2012	2013	2012		
		(in mil	lions)			
Weighted-average basic common shares outstanding	179.3	177.9	179.2	177.6		
Potential number of shares issuable upon exercise of in-the-money stock options under the Long-term Stock Incentive Plan						
	0.2		0.2	1.0		
Average diluted common shares outstanding	179.5	177.9	179.4	178.6		

Note 6 - Asset Retirement Obligations

QEP records asset retirement obligations (ARO) when there are legal obligations associated with the retirement of tangible long-lived assets. The Company's ARO liability applies primarily to abandonment costs associated with oil and gas wells, production facilities, midstream assets and certain other properties. The fair values of such costs are estimated by Company personnel based on abandonment costs of similar assets and depreciated over the life of the related assets. Revisions to ARO estimates result from changes in expected cash flows or material changes in estimated asset retirement costs. The ARO liability is adjusted to present value each period through an accretion calculation using a credit-adjusted risk-free interest rate.

The following is a reconciliation of the changes in the Company's asset retirement obligation from January 1, 2013, to September 30, 2013:

	Asset Retirement Obligations			
		2013		
		(in millions)		
ARO liability at January 1,	\$	193.1		
Accretion		7.2		
Liabilities incurred		8.7		
Revisions		(25.3)		
Liabilities settled		(13.5)		
ARO liability at September 30,	\$	170.2		

Note 7 – Fair Value Measurements

QEP measures and discloses fair values in accordance with the provisions of ASC 820 "Fair Value Measurements and Disclosures". This guidance defines fair value in applying GAAP, establishes a framework for measuring fair value and expands disclosures about fair-value measurements, but does not change existing guidance as to whether or not an instrument is carried at fair value. ASC 820 also establishes a fair-value hierarchy. Level 1 inputs are quoted prices (unadjusted) for identical assets or liabilities in active markets that the Company has the ability to access at the measurement date. Level 2 inputs are inputs other than quoted prices included within Level 1 that are observable for the asset or liability, either directly or indirectly. Level 3 inputs are unobservable inputs for the asset or liability.

QEP has determined its commodity derivative instruments are Level 2. The Level 2 fair value of commodity derivative contracts (see Note 8 - Derivative Contracts) is based on market prices posted on the respective commodity exchanges on the last trading day of the reporting period and industry standard discounted cash flow models. QEP primarily applies the market approach for recurring fair value measurements and maximizes its use of observable inputs and minimizes its use of unobservable inputs. QEP considers bid and ask prices for valuing the majority of its assets and liabilities measured and reported at fair value. In addition to using market data, QEP makes assumptions in valuing its assets and liabilities, including assumptions about risk and the risks inherent in the inputs to the valuation technique. The Company's policy is to recognize significant transfers between Levels at the end of the reporting period.

However, certain of the Company's commodity derivative instruments are valued using industry standard models that consider various inputs, including quoted forward prices for commodities, time value, volatility, and current market and contractual prices for the underlying instruments, as well as other relevant economic measures. Substantially all of these inputs are observable in the marketplace throughout the full term of the instrument and can be derived from observable data or are supported by observable prices at which transactions are executed in the marketplace. The determination of fair value for derivative assets and liabilities also incorporates nonperformance risk for counterparties and for QEP. Derivative contract fair values are reported on a net basis to the extent a legal right of offset with the counterparty exists.

In addition, QEP has interest rate swaps that it has determined are Level 2 financial instruments. The fair values of the interest rate swaps are determined using the market standard methodology of discounting the future expected cash flows that would occur under the contractual terms of the swap. The variable interest rates used in the calculation of projected cash flows are based on an expectation of future interest rates derived from observable market interest rate curves. QEP incorporates credit valuation adjustments to reflect both its nonperformance risk and the respective counterparty's nonperformance risk in the fair value measurements. While the credit valuation adjustments are not observable inputs, they are not significant to the overall valuation and the other inputs used to value the interest rate swaps are observable Level 2 inputs.

The fair value of financial assets and liabilities at September 30, 2013, is shown in the table below:

Fair Value Measurements September 30, 2013

September 30, 2013									
Gross Amou		Gross Amounts of Assets and Liabilities Level 1 Level 2 Level 3				– Netting Adjustments ⁽¹⁾			Net Amounts Presented on the Condensed Consolidated Balance Sheet
(in millions))				
\$	_	\$	58.0	\$	_	\$	(18.6)	\$	39.4
	_		5.8		_				5.8
\$		\$	63.8	\$		\$	(18.6)	\$	45.2
\$	_	\$	26.2	\$	_	\$	(18.6)	\$	7.6
	_		2.5		_		_		2.5
\$	_	\$	28.7	\$		\$	(18.6)	\$	10.1
	\$ \$ \$	Level 1 \$ \$ \$ \$	Level 1	Level 1 Level 2 \$ - \$ - \$ - \$ 63.8 \$ - \$ - \$ - \$ - 26.2 - 2.5	Level 1 Level 2 \$ - - 5.8 \$ - \$ 63.8 \$ - \$ - \$ - 26.2 \$ - 2.5	Gross Amounts of Assets and Liabilities Level 1 Level 2 Level 3 (in mill) \$ - 58.0 \$ - - 5.8 - \$ - \$ 63.8 \$ \$ - \$ 26.2 \$ - - - 2.5 - -	Gross Amounts of Assets and Liabilities Level 1 Level 2 Level 3 (in millions) \$ — \$ — 5.8 — \$ — \$ \$ — \$ \$ — \$	Gross Amounts of Assets and Liabilities Level 1 Level 2 Level 3 Netting Adjustments ⁽¹⁾ (in millions) \$ — \$ 58.0 \$ — \$ (18.6) — 5.8 — — \$ — \$ 63.8 \$ — \$ (18.6) \$ — \$ 26.2 \$ — \$ (18.6) — 2.5 — —	Gross Amounts of Assets and Liabilities Level 1 Level 2 Level 3 Netting Adjustments(1) (in millions) \$ — \$ 58.0 \$ — \$ (18.6) \$ — 5.8 — — — \$ — \$ 63.8 \$ — \$ (18.6) \$ \$ — \$ 26.2 \$ — \$ (18.6) \$ — 2.5 — — —

⁽¹⁾ The Company nets its derivative contract assets and liabilities outstanding with the same counterparty on the Condensed Consolidated Balance Sheet as the contracts contain netting provisions. Refer to Note 8 - Derivative Contracts, for additional information regarding the Company's derivative contracts.

The fair value of financial assets and liabilities at December 31, 2012, is shown in the table below:

Fair Value Measurements

	December 31, 2012									
		Gross Amounts of Assets and Liabilities Level 1 Level 2 Level 3						Netting Adjustments ⁽¹⁾	P	Net Amounts resented on the Condensed solidated Balance Sheet
						(in mill	ions))		
Financial Assets										
Commodity derivative instruments - short-term	\$	_	\$	189.7	\$	_	\$	(1.0)	\$	188.7
Commodity derivative instruments - long-term		_		4.2		_		(0.1)		4.1
Total financial assets	\$		\$	193.9	\$		\$	(1.1)	\$	192.8
Financial Liabilities										
Commodity derivative instruments - short-term	\$	_	\$	1.0	\$	_	\$	(1.0)	\$	_
Interest rate swaps - short-term		_		2.6		_		_		2.6
Commodity derivative instruments - long-term		_		0.1		_		(0.1)		_
Interest rate swaps - long-term				3.6						3.6
Total financial liabilities	\$	_	\$	7.3	\$		\$	(1.1)	\$	6.2

⁽¹⁾ The Company nets its derivative contract assets and liabilities outstanding with the same counterparty on the Condensed Consolidated Balance Sheet as the contracts contain netting provisions. Refer to Note 8 - Derivative Contracts, for additional information regarding the Company's derivative contracts.

The following table discloses the fair value and related carrying amount of certain financial instruments not disclosed in other notes to the condensed consolidated financial statements in this quarterly report on Form 10-Q:

		Carrying Level 1 Amount Fair Value				Carrying Amount]	Level 1 Fair Value
		Septemb	2013	December 31, 2012				
	(in millions)							
Financial assets								
Cash and cash equivalents	\$	123.0	\$	123.0	\$	_	\$	_
Financial liabilities								
Checks outstanding in excess of cash balances	\$	1.6	\$	1.6	\$	39.7	\$	39.7
Long-term debt	\$	2,882.3	\$	2,900.5	\$	3,206.9	\$	3,420.7

The carrying amounts of cash and cash equivalents and checks outstanding in excess of cash balances approximate fair value. The fair value of fixed-rate long-term debt is based on the trading levels and dollar prices for the Company's debt at the end of the quarter. The carrying amount of variable-rate long-term debt approximates fair value because the floating interest rate paid on such debt was set for periods of one month.

The initial measurement of asset retirement obligations at fair value is calculated using discounted cash flow techniques and based on internal estimates of future retirement costs associated with property, plant and equipment. Significant Level 3 inputs used in the calculation of asset retirement obligations include plugging costs and remaining reserve lives. A reconciliation of the Company's asset retirement obligations is presented in Note 6 – Asset Retirement Obligations.

Note 8 - Derivative Contracts

QEP has established policies and procedures for managing commodity price volatility through the use of derivative instruments. In the normal course of business, QEP uses commodity price derivative instruments to reduce the impact of potential downward movements in commodity prices on cash flow, returns on capital investment, and other financial results. However, these instruments typically limit gains from favorable price movements. The volume of production subject to commodity derivative instruments and the mix of the instruments are frequently evaluated and adjusted by management in response to changing market conditions. QEP may enter into commodity derivative contracts for up to 100% of forecasted production from proved reserves. In addition, QEP may enter into commodity derivative contracts on a portion of its extracted NGL volumes in its midstream business and a portion of its natural gas sales and purchases for marketing transactions. QEP does not enter into commodity derivative instruments for speculative purposes.

QEP uses commodity derivative instruments known as fixed-price swaps to realize a known price for a specific volume of production delivered into a regional sales point. QEP's commodity derivative instruments do not require the physical delivery of natural gas, crude oil, or NGL between the parties at settlement. Swap transactions are settled in cash with one party paying the other for the net difference in prices, multiplied by the contract volume, for the settlement periods. Natural gas price derivative instruments are typically structured as fixed-price swaps at regional price indices. Oil price derivative instruments are typically structured as NYMEX fixed-price swaps based at Cushing, Oklahoma. QEP also has oil price derivative fixed-price swaps that use Brent crude oil prices as the reference price. Brent crude oil contracts are traded on the IntercontinentalExchange, Inc. (ICE). NGL price derivative instruments are typically structured as Mont Belvieu, Texas fixed-price swaps.

QEP enters into commodity derivative transactions that do not have margin requirements or collateral provisions that would require payments prior to the scheduled settlement dates. Commodity derivative contract counterparties are normally financial institutions and energy trading firms with investment-grade credit ratings. QEP routinely monitors and manages its exposure to counterparty risk by requiring specific minimum credit standards for all counterparties and avoids concentration of credit exposure by transacting with multiple counterparties.

Effective January 1, 2012, QEP elected to de-designate all of its natural gas, crude oil and NGL derivative contracts that were previously designated as cash flow hedges and discontinue hedge accounting prospectively. As a result of discontinuing hedge accounting, the mark-to-market values at December 31, 2011, were fixed in Accumulated Other Comprehensive Income (AOCI) as of the de-designation date and are being reclassified into the Condensed Consolidated Statement of Operations as the transactions settle and affect earnings. At September 30, 2013, AOCI included \$27.3 million (\$17.2 million after tax) of unrealized gains that will be settled and reclassified from AOCI to the Condensed Consolidated Statements of Operations

during the remainder of 2013. During the nine months ended September 30, 2013 and 2012, \$60.4 million and \$133.8 million, respectively, of unrealized gains, after tax, were reclassified from AOCI into the Condensed Consolidated Statement of Operations in "Realized and unrealized (losses) gains on derivative contracts" as the transactions settled. QEP expects to reclassify into earnings from AOCI the fixed value related to de-designated natural gas, oil and NGL derivatives over the remainder of 2013. Currently, QEP recognizes all gains and losses from changes in the fair value of natural gas, oil and NGL derivative contracts immediately in earnings rather than deferring any such amounts in AOCI. All commodity derivative instruments are recorded on the Condensed Consolidated Balance Sheets as either assets or liabilities measured at their fair values and all realized and unrealized gains and losses from derivative instruments incurred after January 1, 2012, are presented in the Condensed Consolidated Statement of Operations in "Realized and unrealized (losses) gains on derivative contracts" below operating income.

QEP also uses interest rate swaps to mitigate a portion of its exposure to interest rate volatility risk. During the second quarter of 2012, QEP entered into variable-to-fixed interest rate swap agreements having a combined notional principal amount of \$300.0 million to minimize the interest rate volatility risk associated with its \$300.0 million senior, unsecured term loan. QEP locked in a fixed interest rate of 1.07% in exchange for a variable interest rate indexed to the one-month LIBOR rate. The interest rate swaps settle monthly and will mature in March of 2017.

QEP Energy Derivative Contracts

The following table sets forth QEP Energy's quantities and average prices for its commodity derivative contracts as of September 30, 2013:

Insert Title Here

Year	Type of Contract	Type of Contract Index		Average Swap price per unit	
			(in millions)		
Natural gas			(MMBtu)		
2013	Swap	IFNPCR ⁽¹⁾	18.4	\$	5.49
2013	Swap	NYMEX	14.7	\$	3.81
2014	Swap	IFNPCR ⁽¹⁾	32.9	\$	4.00
2014	Swap	NYMEX	25.6	\$	4.19
Crude oil			(Bbls)		
2013	Swap	NYMEX WTI	2.1	\$	98.27
2013	Swap	BRENT ICE	0.1	\$	107.80
2014	Swap	NYMEX WTI	8.8	\$	93.63

⁽¹⁾ Inside FERC monthly settlement index for the Northwest Pipeline Corp. Rocky Mountains.

QEP Marketing Derivative Contracts

QEP Marketing enters into commodity derivative transactions to lock in a margin on natural gas volumes placed into storage and for marketing transactions in which QEP Marketing sells gas volumes at a fixed price. The following table sets forth QEP Marketing's volumes and swap prices for its commodity derivative contracts as of September 30, 2013:

Year	Type of Contract	Type of Contract Index]	age Swap price MMBtu
			(in millions)		
Natural gas sales			(MMBtu)		
2013	Swap	IFNPCR	1.6	\$	3.91
2014	Swap	IFNPCR	3.8	\$	3.71
Natural gas purchases			(MMBtu)		
2013	Swap	IFNPCR	1.7	\$	3.46
2014	Swap	IFNPCR	0.2	\$	3.82

QEP's Derivative Contracts

The following table sets forth QEP's notional amount and interest rate for its interest rate swaps outstanding as of September 30, 2013:

_	Notional amount	Type of Contract	Maturity	Fixed Rate Paid	Variable Rate Received
	(in millions)				
	\$300.0	Swap	March 2017	1.07%	One month LIBOR

QEP Derivative Financial Statement Presentation

The following table identifies the balance sheet location of QEP's outstanding derivative contracts on a gross contract basis as opposed to the net contract basis presentation in the Condensed Consolidated Balance Sheets and the related fair values at the balance sheet dates:

			Gross asse			Gross liability derivative instruments fair value				
	Balance Sheet line item	Sep	September 30, 2013 December 31, 2012		September 30, 2 2013		December 31, 2012			
		(in millions)					(in m	illions)		
Current:										
Commodity	Fair value of derivative contracts	\$	58.0	\$	189.7	\$	26.2	\$	1.0	
Interest rate swaps	Fair value of derivative contracts		_		_		2.5		2.6	
Long-term:										
Commodity	Fair value of derivative contracts		5.8		4.2		_		0.1	
Interest rate swaps	Fair value of derivative contracts		_		_		_		3.6	
Total derivative instruments		\$	63.8	\$	193.9	\$	28.7	\$	7.3	

The effects of the change in fair value and settlement of QEP's derivative contracts recorded in "Realized and unrealized (losses) gains on derivative contracts" on the Condensed Consolidated Statements of Operations are summarized in the following tables:

		Three Mo				Nine Months Ended September 30,				
Derivative instruments not designated as cash flow hedges		2013	2012			2013		2012		
Realized gains (losses) on commodity derivative contracts				(in m	illion	s)				
QEP Energy										
Natural gas derivative contracts	\$	42.5	\$	86.2	\$	112.0	\$	283.8		
Oil derivative contracts		(15.3)		2.7		(3.7)		2.2		
NGL derivative contracts		_		3.4		_		6.5		
QEP Field Services										
NGL derivative contracts		_		1.9		_		6.3		
QEP Marketing										
Natural gas derivative contracts		(0.3)		(0.4)		0.7		3.7		
Total realized gains on commodity derivative contracts		26.9		93.8		109.0		302.5		
Unrealized gains (losses) on commodity derivative contracts										
QEP Energy										
Natural gas derivative contracts		(6.6)		(50.6)		(9.6)		3.3		
Oil derivative contracts		(46.2)		4.1		(49.1)		31.2		
NGL derivative contracts		_		(4.4)		_		3.4		
QEP Field Services										
NGL derivative contracts		_		(2.5)		_		2.0		
QEP Marketing										
Natural gas derivative contracts		0.1		(1.4)		(0.3)		(0.5)		
Total unrealized (losses) gains on commodity derivative contracts		(52.7)		(54.8)		(59.0)		39.4		
Total realized and unrealized (losses) gains on commodity derivative contracts	\$	(25.8)	\$	39.0	\$	50.0	\$	341.9		
Realized gains (losses) on interest rate swaps										
Realized losses on interest rate swaps	\$	(0.6)	\$	(0.6)	\$	(1.9)	\$	(0.6)		
Unrealized gains (losses) on interest rate swaps		· , ,		. ,		, ,				
Unrealized (losses) gains on interest rate swaps		(1.4)		(2.3)		3.5		(6.6)		
Total realized and unrealized (losses) gains on interest rate swaps	\$	(2.0)	\$	(2.9)	\$	1.6	\$	(7.2)		
Total net realized gains on derivative contracts	\$	26.3	\$	93.2	\$	107.1	\$	301.9		
Total net unrealized (losses) gains on derivative contracts	*	(54.1)	-	(57.1)	7	(55.5)	-	32.8		
Grand Total	\$	(27.8)	\$	36.1	\$	51.6	\$	334.7		

Note 9 - Restructuring Costs

During the first quarter of 2012, QEP began incurring costs related to the closure of its Oklahoma City office and the subsequent consolidation of its Southern Region operations into a single regional office located in Tulsa. During the second half of 2012, QEP incurred additional restructuring and reorganization costs related to consolidating various corporate and accounting functions to the Denver corporate headquarters. The creation of one office for QEP's Southern Region as well as the consolidation of corporate and accounting functions was intended to increase efficiency, team-based collaboration and organizational productivity over the long term. As part of the reorganization, QEP incurred costs associated with the severance, retention and relocation of employees, additional pension expenses, exit costs associated with the termination of operating

leases arising from office space that will no longer be utilized by the Company and other expenses. The Company currently estimates that the remaining restructuring costs will be incurred during the remainder of 2013.

The following table summarizes, by line of business, each major type of cost expected to be incurred and the total amounts recorded in "General and administrative" expense on the Condensed Consolidated Statement of Operations for the respective periods indicated:

Total	Restru	cturing	Costs

	_	Recognized in Income										
	ī	Total Expected to	Period from Inception to September 30,		Three Mo Septembe	nth	s Ended	Nine Months September 30				
		e Incurred	2013		2013		2012	2	013	:	2012	
QEP Energy					(in million	ns)						
One-time termination benefits	\$	3.3	\$ 3.3	\$	0.1	\$	0.2		0.4		2.1	
Retention & relocation expense		3.7	3.6		0.1		0.1		0.3		3.2	
Lease termination costs		0.6	0.6		_		_		_		_	
Total restructuring costs	\$	7.6	\$ 7.5	\$	0.2	\$	0.3		0.7		5.3	
QEP Field Services												
One-time termination benefits	\$	_	\$ —	\$	_	\$	_		_		_	
Retention & relocation expense		0.1	0.1		0.1		_		0.1		_	
Lease termination costs		_	_		_		_		_		_	
Total restructuring costs	\$	0.1	\$ 0.1	\$	0.1	\$	_		0.1		_	
QEP Marketing												
One-time termination benefits	\$	0.3	\$ 0.3	\$	_	\$	_		0.1		_	
Retention & relocation expense		_	_		_		_		_		_	
Lease termination costs		_	_		_		_		_		_	
Total restructuring costs	\$	0.3	\$ 0.3	\$	_	\$	_		0.1		_	
Total QEP												
One-time termination benefits	\$	3.6	\$ 3.6	\$	0.1	\$	0.2	\$	0.5	\$	2.1	
Retention & relocation expense	· · · · · · · · · · · · · · · · · · ·	3.8	3.7		0.2		0.1	•	0.4	•	3.2	
Lease termination costs		0.6	0.6		_		_		_		_	
Total restructuring costs	\$	8.0	\$ 7.9	\$	0.3	\$	0.3	\$	0.9	\$	5.3	
				_		-		_				

The following is a reconciliation of the restructuring liability, by line of business, which is included within "Accounts payable and accrued expenses" on the Condensed Consolidated Balance Sheets:

	QEP Energy		QEP Serv		-	QEP Marketing		Total
Balance at December 31, 2012	\$	1.0	\$	_	\$	_	\$	1.0
Costs incurred and charged to expense		0.7		0.1		0.1		0.9
Costs paid or otherwise settled		(1.7)		(0.1)		(0.1)		(1.9)
Balance at September 30, 2013	\$	_		_			\$	

Note 10 - Debt

As of the indicated dates, the principal amount of QEP's debt, including amounts outstanding under its and QEP Midstream's revolving credit facilities, QEP's term loan and QEP's senior notes consisted of the following:

	Se	ptember 30, 2013	De	ecember 31, 2012
		(in mi	llions)	
QEP's revolving credit facility due 2016	\$	365.0	\$	690.0
QEP Midstream's revolving credit facility due 2018		_		_
Term loan due 2017		300.0		300.0
6.05% Senior Notes due 2016		176.8		176.8
6.80% Senior Notes due 2018		134.0		134.0
6.80% Senior Notes due 2020		136.0		136.0
6.875% Senior Notes due 2021		625.0		625.0
5.375% Senior Notes due 2022		500.0		500.0
5.25% Senior Notes due 2023		650.0		650.0
Total principal amount of debt		2,886.8		3,211.8
Less unamortized discount		(4.5)		(4.9)
Total long-term debt outstanding	\$	2,882.3	\$	3,206.9

Of the total debt outstanding on September 30, 2013, amounts outstanding under QEP's revolving credit facility due August 25, 2016, QEP Midstream's revolving credit facility due August 14, 2018, QEP's term loan due April 18, 2017, the 6.05% Senior Notes due September 1, 2016, and the 6.80% Senior Notes due April 1, 2018, will mature within the next five years.

Credit Facilities

QEP's Credit Facility

QEP's revolving credit facility, which matures in August 2016, provides for loan commitments of \$1.5 billion from a group of financial institutions. The credit facility provides for borrowing at short-term interest rates and contains customary covenants and restrictions. The credit facility also contains an accordion provision that would allow for the amount of the facility to be increased to \$2.0 billion and for the maturity to be extended for up to two additional one-year periods, with the agreement of the lenders.

During the nine months ended September 30, 2013 and 2012, QEP's weighted-average interest rate on borrowings from its credit facility was 2.56% and 2.05%, respectively. At September 30, 2013 and December 31, 2012, QEP was in compliance with the covenants under the credit agreement. At September 30, 2013, there was \$365.0 million outstanding and \$3.7 million of letters of credit issued under the credit facility.

QEP Midstream's Credit Facility

On August 14, 2013, QEP Midstream entered into a \$500.0 million senior secured revolving credit facility with a group of financial institutions, which matures on August 14, 2018. QEP Midstream's credit facility contains an accordion provision that allows for the amount of the facility to be increased to \$750.0 million with the agreement of the lenders. QEP Midstream's credit facility is available for QEP Midstream's working capital, capital expenditures, permitted acquisitions and general corporate purposes, including distributions. Substantially all of QEP Midstream's assets, excluding equity in and assets of certain joint ventures and unrestricted subsidiaries, are pledged as collateral under the credit facility. In addition, the credit facility contains restrictions and events of default customary for agreements of this nature.

There have been no borrowings under QEP Midstream's credit facility, and at September 30, 2013, QEP Midstream was in compliance with the covenants under the QEP Midstream credit facility agreement.

QEP is not a borrower or guarantor of QEP Midstream's credit facility. In addition, QEP is not subject to any of the restrictions or covenants contained in QEP Midstream's credit agreement. Outstanding indebtedness under QEP Midstream's credit facility is not included in the definition of indebtedness under QEP's credit facility.

Term Loan

QEP's \$300.0 million term loan facility provides for borrowings at short-term interest rates and contains covenants, restrictions and interest rates that are substantially the same as QEP's revolving credit facility. The term loan matures in April 2017, and the maturity date may be extended one year with the agreement of the lenders. During the nine months ended September 30, 2013 and 2012, QEP's weighted-average interest rate on borrowings from the term loan was 2.23% and 2.02%, respectively. At September 30, 2013 and December 31, 2012, QEP was in compliance with the covenants under the term loan credit agreement.

Senior Notes

At September 30, 2013, the Company had \$2,221.8 million principal amount of senior notes outstanding with maturities ranging from September 2016 to May 2023 and coupons ranging from 5.25% to 6.875%. The senior notes pay interest semi-annually, are unsecured senior obligations and rank equally with all of our other existing and future unsecured and senior obligations. QEP may redeem some or all of its senior notes at any time before their maturity at a redemption price based on a make-whole amount plus accrued and unpaid interest to the date of redemption. The indentures governing QEP's senior notes contain customary events of default and covenants that may limit QEP's ability to, among other things, place liens on its property or assets.

Note 11– Contingencies

QEP is involved in various commercial and regulatory claims, litigation and other legal proceedings that arise in the ordinary course of its business. QEP assesses these claims in an effort to determine the degree of probability and range of possible loss for potential accrual in its consolidated financial statements. In accordance with ASC 450, *Contingencies*, an accrual is recorded for a loss contingency when its occurrence is probable and damages can be reasonably estimated based on the anticipated most likely outcome or the minimum amount within a range of possible outcomes. Because legal proceedings are inherently unpredictable and unfavorable resolutions could occur, assessing contingencies is highly subjective and requires judgments about uncertain future events. When evaluating contingencies, QEP may be unable to provide a meaningful estimate due to a number of factors, including the procedural status of the matter in question, the presence of complex or novel legal theories, and/or the ongoing discovery and development of information important to the matter. QEP's litigation loss contingencies are discussed below. QEP is unable to estimate reasonably possible losses in excess of recorded accruals for these contingencies for the reasons set forth above. QEP believes, however, that the resolution of pending proceedings will not have a material effect on QEP's consolidated financial position, results of operations or cash flows.

Environmental Claims

In October 2009, QEP received a cease and desist order from the U.S. Army Corps of Engineers (COE) to refrain from unpermitted work resulting in the discharge of dredged and/or fill material into waters of the United States at three sites located in Caddo and Red River Parishes, Louisiana. Region 6 of the U.S. Environmental Protection Agency (EPA) has assumed lead responsibility for enforcement of the cease and desist order and any possible future orders for the removal of unauthorized fills and/or civil penalties under the Clean Water Act. On June 28, 2013, the EPA issued to QEP an Administrative Complaint for the alleged violations. QEP and the EPA reached an agreement to settle the alleged violations through an Administrative Order, under the terms of which QEP paid an administrative penalty of \$0.2 million. The Administrative Order is final. In 2012, QEP completed a field audit, which identified 112 additional instances affecting approximately 90 acres where work may have been conducted in violation of the Clean Water Act. QEP has disclosed each of these instances to the EPA under the EPA's Audit Policy (to reduce penalties) and to the COE. QEP is working with the EPA and the COE to resolve these matters, which will require the Company to undertake certain mitigation and permitting activities, and may require QEP to pay a monetary penalty.

In July 2010, QEP received a Notice of Potential Penalty (NOPP) from the Louisiana Department of Environmental Quality (LDEQ) regarding the assumption of ownership and operatorship of a single facility in Louisiana prior to transferring the facility's air quality permit. In 2011, QEP completed an internal audit, which identified 424 facilities in Louisiana for which QEP both failed to submit a complete permit application and to receive approval from the department prior to construction, modification, or operation. QEP has corrected and disclosed all instances of non-compliance to the LDEQ and is working with the department to resolve the NOPP. The LDEQ has assumed lead responsibility for enforcement of the NOPP and may require the Company to pay a monetary penalty.

Litigation

Chieftain Royalty Company v. QEP Energy Company, Case No CIV-11-0212-R, U. S. District Court for the Western District of Oklahoma. This statewide class action was filed in January 2011 on behalf of QEP's Oklahoma royalty owners asserting

various claims for damages related to royalty valuation on all of QEP's Oklahoma wells operated by QEP or from which QEP marketed gas. These claims include breach of contract, breach of fiduciary duty, fraud, unjust enrichment, tortious breach of contract, conspiracy, and conversion, based generally on asserted improper deduction of post-production costs. The Court certified the class as to the breach of contract, breach of fiduciary duty and unjust enrichment claims. The parties successfully mediated the case in January 2013. On February 13, 2013, the parties executed a Stipulation and Agreement of Settlement (the Chieftain Settlement Agreement) providing for a cash payment from QEP to the class in the amount of \$115.0 million. In consideration for the settlement payment, QEP received a full release of all claims regarding the calculation, reporting and payment of royalties from the sale of natural gas and its constituents for all periods prior to February 28, 2013, and all class members are enjoined from asserting claims related to such royalties. As part of the Chieftain Settlement Agreement, the parties also agreed on the methodology for the calculation and payment of future royalties payable by QEP, or its successors and assigns, under all class leases for the life of such leases. On May 31, 2013, the Court issued its order approving the settlement, which is now final.

Questar Gas Company v. QEP Field Services Company, Civil No. 120902969, Third Judicial District Court, State of Utah. At the closing of the Offering, the assets and agreement discussed below were assigned to QEP Midstream. QEP Field Services' former affiliate, Questar Gas Company (QGC), filed this complaint in state court in Utah on May 1, 2012, asserting claims for (i) breach of contract, (ii) breach of implied covenant of good faith and fair dealing, (iii) an accounting and (iv) declaratory judgment related to a 1993 gathering agreement (the 1993 Agreement) executed when the parties were affiliates. Under the 1993 Agreement, certain of QEP Field Services' systems provide gathering services to QGC charging an annual gathering rate which is based on cost of service. QGC is disputing the annual calculation of the gathering rate. The annual gathering rate has been calculated in the same manner under the 1993 Agreement since it was amended in 1998, without any prior objection or challenge by QGC. QGC was netting the disputed amount from its monthly payments of the gathering fees to QEP Field Services and has continued to net such amounts from its monthly payment to QEP Midstream. As of September 30, 2013, QEP Midstream has deferred revenue of \$7.0 million related to the QGC disputed amount. Specific monetary damages are not asserted. QEP Field Services has filed counterclaims seeking damages and a declaratory judgment relating to its gathering services under the 1993 Agreement. QGC may seek to amend its complaint to add QEP Midstream as a defendant in the litigation. QEP Midstream has been indemnified by QEP for costs, expenses and other losses incurred by QEP Midstream in connection with the QGC dispute, subject to certain limitations, as set forth in the Omnibus Agreement (defined below in "Note 3 - QEP Midstream").

Note 12 - Share-Based Compensation

QEP issues stock options and restricted shares under its Long-Term Stock Incentive Plan (LTSIP) and awards performance-based share units under its Cash Incentive Plan (CIP) to certain officers, employees, and non-employee directors. QEP recognizes expense over time as the stock options, restricted shares, and performance-based share units vest. Deferred share-based compensation is included in additional paid-in capital in the Condensed Consolidated Balance Sheets. There were 11.9 million shares available for future grants under the LTSIP at September 30, 2013. Share-based compensation expense is recognized in "General and administrative" on the Condensed Consolidated Statements of Operations. During the three and nine months ended September 30, 2013, QEP recognized \$6.8 million and \$20.0 million, respectively, in total compensation expense related to share-based compensation compared to \$7.2 million and \$19.5 million during the three and nine months ended September 30, 2012, respectively.

QEP Midstream maintains a unit-based compensation plan for officers, directors and employees of the general partner of QEP Midstream and its affiliates and any consultants, affiliates of the general partner, or other individuals who perform services for QEP Midstream. The QEP Midstream 2013 Long-Term Incentive Plan (the QEP Midstream LTIP) permits various types of awards, including awards of restricted units, phantom units, unit options, unit appreciation rights, distribution equivalent rights, profits interest units and other unit-based awards. Awards granted during 2013 under the QEP Midstream LTIP will be settled with QEP Midstream units. During the three and nine months ended September 30, 2013, QEP recognized \$0.1 million in compensation expense related to QEP Midstream LTIP.

Stock Options

QEP uses the Black-Scholes-Merton mathematical model to estimate the fair value of stock option awards at the date of the grant. Fair-value calculations rely upon subjective assumptions used in the mathematical model and may not be representative of future results. The Black-Scholes-Merton model is intended for measuring the value of options traded on an exchange. The Company utilizes the "simplified" method to estimate the expected term of the stock options. QEP uses a historical volatility method to estimate the fair value of stock option awards and the risk-free interest rate is based on the yield on U.S. Treasury strips with maturities similar to those of the expected term of the stock options. The stock options typically vest in equal installments over a three-year period from the grant date and are exercisable immediately upon vesting through the seventh anniversary of the grant date.

The calculated fair value of options granted and major assumptions used in the model at the date of grant are listed below:

	Stock Option Assumptions			
	Nine Months Ended			
	September 30, 2013			
Weighted-average grant-date fair value of awards granted during the period	\$	15.16		
Weighted-average risk-free interest rate		1.00%		
Weighted-average expected price volatility		58.3%		
Expected dividend yield		0.27%		
Expected term in years at the date of grant		5.5		

Stock option transactions under the terms of the LTSIP are summarized below:

	Options Outstanding	Average Exercise		Average Exercise Remaining		Aggregate trinsic Value
			(per share)	(in years)		in millions)
Outstanding at December 31, 2012	1,697,471	\$	25.23			
Granted	330,592		30.06			
Exercised	(209,500)		9.60			
Forfeited	_		_			
Outstanding at September 30, 2013	1,818,563	\$	27.90	3.9	\$	3.6
Options Exercisable at September 30, 2013	1,236,211	\$	26.36	3.0	\$	3.6
Unvested Options at September 30, 2013	582,352	\$	31.18	5.9	\$	_

The total intrinsic value (the difference between the market price at the exercise date and the exercise price) of options exercised was \$4.2 million and \$7.1 million during the nine months ended September 30, 2013 and 2012, respectively. The Company realized \$1.4 million and \$2.2 million of income tax benefit for the nine months ended September 30, 2013 and 2012, respectively, which increased its Additional Paid-in-Capital (APIC) pool by \$1.4 million as of September 30, 2013. As of September 30, 2013, \$4.4 million of unrecognized compensation cost related to stock options granted under the LTSIP is expected to be recognized over a weighted-average period of 2.1 years. During the nine months ended September 30, 2013, QEP received \$0.5 million in cash in relation to the exercise of stock options.

Restricted Shares

Restricted share grants typically vest in equal installments over a three-year period from the grant date. The grant date fair value is determined based on the closing bid price of the Company's common stock on the grant date. The total fair value of restricted stock that vested during the nine months ended September 30, 2013 and 2012, was \$18.4 million and \$16.6 million, respectively. The Company realized \$0.4 million and \$0.2 million of income tax expense for the nine months ended September 30, 2013 and 2012, respectively, with \$0.1 million impact to the Company's APIC pool as of September 30, 2013. The weighted average grant-date fair value of restricted stock was \$30.03 per share and \$30.59 per share for the nine months ended September 30, 2013 and 2012, respectively. As of September 30, 2013, \$23.5 million of unrecognized compensation cost related to restricted shares granted under the LTSIP is expected to be recognized over a weighted-average vesting period of 2.2 years.

Transactions involving restricted shares under the terms of the LTSIP are summarized below:

	Restricted Shares Outstanding	Avera	eighted- nge Grant- Fair Value
		(pe	er share)
Unvested balance at December 31, 2012	1,300,588	\$	31.78
Granted	850,824		30.03
Vested	(620,065)		31.42
Forfeited	(75,442)		30.82
Unvested balance at September 30, 2013	1,455,905	\$	30.96

Performance Share Units

The performance share units' cash payouts are dependent upon the Company's total shareholder return compared to a group of its peers over a three-year period. The awards are denominated in share units but delivered in cash at the end of the performance period. The weighted average grant-date fair value of the performance share units was \$30.12 per share and \$30.90 per share for the nine months ended September 30, 2013 and 2012, respectively. As of September 30, 2013, \$8.5 million of unrecognized compensation cost, representing the fair market value of performance shares granted under the CIP, is expected to be recognized over a weighted-average vesting period of 2.0 years.

Transactions involving performance share units under the terms of the CIP are summarized below:

	Performance Share Units Outstanding	Weigh Average Date Fai	Grant-
Unvested balance at December 31, 2012	283,484	\$	34.01
Granted	223,844		30.12
Vested	_		_
Forfeited	(1,163)		30.12
Unvested balance at September 30, 2013	506,165	\$	32.30

Note 13 - Employee Benefits

The Company maintains closed, defined-benefit pension and postretirement medical plans. QEP's pension plans include a qualified and a nonqualified retirement plan. The Company's postretirement medical plan is unfunded and provides certain health care and life insurance benefits for certain retired employees. During the nine months ended September 30, 2013, the Company made contributions of \$8.1 million to its funded pension plan, and \$2.7 million to its unfunded pension plan. Contributions to funded plans increase plan assets while contributions to unfunded plans are used to fund current benefit payments. During the remainder of 2013, the Company expects to contribute approximately \$0.4 million to its unfunded pension plans and approximately \$0.1 million for retiree health care and life insurance benefits. No additional contributions are planned for the funded pension plan during the remainder of 2013.

The following table sets forth the Company's pension and postretirement benefits net periodic benefit costs:

	Pension								
	 Three Months Ended September 30,			Nine Months September					
	 2013		, 2012		2013		, 2012		
			(in m	illions)					
Service cost	\$ 0.7	\$	1.1	\$	2.6	\$	3.0		
Interest cost	1.2		1.3		3.7		3.7		
Expected return on plan assets	(0.9)		(0.9)		(2.9)		(2.7)		
Amortization of prior service costs	1.2		1.3		3.7		3.9		
Amortization of actuarial loss	0.5		0.6		1.7		1.0		
Periodic expense	\$ 2.7	\$	3.4	\$	8.8	\$	8.9		
	Postretirement Benefits								
	Three Mo	nths Eı	ıded		Nine Moi	ıths En	Ended		
	Septen	nber 30	,	September 30,					
	2013		2012		2013		2012		
			(in m	illions)					
Service cost	\$ 0.1	\$	0.1	\$	0.1	\$	0.1		
Interest cost	_		_		0.2		0.2		
Amortization of prior service costs	0.1		0.1		0.3		0.3		
Recognized net actuarial loss	0.1		_		0.1		_		
Periodic expense	\$ 0.3	\$	0.2	\$	0.7	\$	0.6		

Note 14 - Operations by Line of Business

QEP's lines of business include natural gas and oil exploration and production (QEP Energy), midstream field services, which includes the ownership and operation of QEP Midstream (QEP Field Services), and marketing and corporate (QEP Marketing & Resources). The lines of business are managed separately and therefore the financial information is presented separately due to the distinct differences in the nature of operations of each line of business, among other factors. QEP owns a 57.8% ownership interest in QEP Midstream and it is consolidated under the voting interest model in QEP Field Services' operating results. The outside ownership interest in QEP Midstream is presented separately as a noncontrolling interest.

The following table is a summary of operating results for the three months ended September 30, 2013, by line of business:

	OEI	DEnover	QEP Field Services	QEP Marketing & Resources	Eliminations	C	QEP onsolidated
	QE	P Energy	Services	(in millions)	Elilillations	C	onsondated
Revenues				(III IIIIIIIIIII)			
From unaffiliated customers	\$	538.5	\$ 66.9	\$ 167.4	s —	\$	772.8
From affiliated customers		_	33.2	272.8	(306.0)		_
Total Revenues		538.5	100.1	440.2	(306.0)		772.8
Operating expenses							
Purchased gas, oil and NGL expense		39.2	_	438.2	(272.4)		205.0
Lease operating expense		43.5	_	_	1.5		45.0
Natural gas, oil and NGL transportation and other handling	g	0= 4	2.4		(2.4.0)		24 -
costs		65.1	3.4	_	(34.0)		34.5
Gathering, processing and other			22.0	0.4	_		22.4
General and administrative		33.5	15.8	1.1	(1.1)		49.3
Production and property taxes		40.4	2.0	0.1	_		42.5
Depreciation, depletion and amortization		236.0	16.8	0.3	_		253.1
Other operating expenses		5.6			_		5.6
Total operating expenses		463.3	60.0	440.1	(306.0)		657.4
Net gain from asset sales		12.8					12.8
Operating income (loss)		88.0	 40.1	0.1	_		128.2
Realized and unrealized losses on derivative contracts		(25.6)	_	(2.2)	_		(27.8)
Interest and other income		2.6	8.0	52.3	(51.7)		4.0
Income from unconsolidated affiliates		_	8.0	_	_		0.8
Interest expense		(49.2)	 (3.2)	(41.0)	51.7		(41.7)
Income before income taxes		15.8	38.5	9.2	_		63.5
Income tax provision		(6.2)	(12.7)	(3.5)			(22.4)
Net income		9.6	 25.8	5.7	_		41.1
Net income attributable to noncontrolling interest		_	(3.8)				(3.8)
Net income attributable to QEP	\$	9.6	\$ 22.0	\$ 5.7	s —	\$	37.3

The following table is a summary of operating results for the three months ended September 30, 2012, by line of business:

	QEP Energ	Sy	•	P Field rvices	& Reso	QEP Marketing & Resources		Eliminations		QEP nsolidated
Revenues					(in mi	llions)				
From unaffiliated customers	\$ 37	1 N	\$	77.9	\$	90.5	\$		\$	542.4
From affiliated customers	φ 3/·	+.0	Ф	31.8	•	145.8	Ф	(177.6)	Ф	342.4
	27	4.0						(177.6)		F 40, 4
Total Revenues	3/-	4.0		109.7		236.3		(177.6)		542.4
Operating expenses	4	- 0		4.0		226 7		(1.4.4.0)		1.40 C
Purchased gas, oil and NGL expense		5.9		4.9		236.7		(144.9)		142.6
Lease operating expense		3.1		_		_		(0.9)		42.2
Natural gas, oil and NGL transportation and other handling costs		9.8		6.9		_		(30.4)		36.3
Gathering, processing and other				21.8		0.1		0.2		22.1
General and administrative	3	1.6		10.5		1.2		(1.6)		41.7
Production and property taxes	2:	2.5		1.7		0.1		_		24.3
Depreciation, depletion and amortization	21	3.5		15.9		0.2		_		234.6
Other operating expenses	1	1.2		_		_		_		11.2
Total operating expenses	43.	2.6		61.7		238.3		(177.6)		555.0
Operating (loss) income	(5	3.6)		48.0		(2.0)		_		(12.6)
Realized and unrealized gains (losses) on derivative										
contracts	4	1.4		(0.6)		(4.7)		_		36.1
Interest and other income	(0.2)		_		28.4		(28.4)		(0.2)
Income from unconsolidated affiliates		—		2.3		_		_		2.3
Loss on early extinguishment of debt		—		_		_		_		_
Interest expense	(2	4.1)		(3.5)		(30.8)		28.4		(30.0)
(Loss) income before income taxes	(4	1.5)		46.2		(9.1)				(4.4)
Income tax benefit (provision)	1	5.3		(16.5)		3.5		_		2.3
Net (loss) income	(2	5.2)		29.7		(5.6)		_		(2.1)
Net income attributable to noncontrolling interest		—		(1.0)		_		_		(1.0)
Net (loss) income attributable to QEP	\$ (2	5.2)	\$	28.7	\$	(5.6)	\$	_	\$	(3.1)

The following table is a summary of operating results for the nine months ended September 30, 2013, by line of business:

	QEP Energy	QEP Field Services	QEP Marketing & Resources	Eliminations	QEP Consolidated
Revenues			(in millions)		
From unaffiliated customers	\$ 1,575.2	\$ 203.6	\$ 441.5	s —	\$ 2,220.3
From affiliated customers		91.4	711.7	(803.1)	
Total Revenues	1,575.2	295.0	1,153.2	(803.1)	2,220.3
Operating expenses	1,575.2	255.0	1,155.2	(003.1)	2,220.0
Purchased gas, oil and NGL expense	159.8	8.6	1,150.6	(710.2)	608.8
Lease operating expense	130.2	_	_	(2.8)	127.4
Natural gas, oil and NGL transportation and other handling costs	180.8	11.6	_	(86.6)	105.8
Gathering, processing and other	_	65.3	1.2	— (c)	66.5
General and administrative	100.2	36.2	3.3	(3.5)	136.2
Production and property taxes	112.7	4.8	0.2	` <u>_</u>	117.7
Depreciation, depletion and amortization	712.1	44.3	0.7	_	757.1
Other operating expenses	13.5	_	_	_	13.5
Total operating expenses	1,409.3	170.8	1,156.0	(803.1)	1,933.0
Net gain (loss) from asset sales	113.4	(0.4)	_	_	113.0
Operating income (loss)	279.3	123.8	(2.8)	_	400.3
Realized and unrealized gains on derivative contracts	49.6	_	2.0	_	51.6
Interest and other income	7.5	1.1	158.2	(157.7)	9.1
Income from unconsolidated affiliates	_	3.7	_	_	3.7
Interest expense	(143.4)	(12.5)	(124.3)	157.7	(122.5)
Income before income taxes	193.0	116.1	33.1	_	342.2
Income tax provision	(71.1)	(40.3)	(13.6)		(125.0)
Net income	121.9	75.8	19.5	_	217.2
Net income attributable to noncontrolling interest		(5.8)			(5.8)
Net income attributable to QEP	\$ 121.9	\$ 70.0	\$ 19.5	\$ —	\$ 211.4

The following table is a summary of operating results for the nine months ended September 30, 2012, by line of business:

	QEP Field QEP Marketing QEP Energy Services & Resources (in millions)		Eliminations	Co	QEP onsolidated		
Revenues				(III IIIIIIIII)			
From unaffiliated customers	\$ 1,106.3	\$	5 254.9	\$ 283.7	\$ —	\$	1,644.9
From affiliated customers	_		88.1	396.3	(484.4)		_
Total Revenues	1,106.3		343.0	680.0	(484.4)		1,644.9
Operating expenses							
Purchased gas, oil and NGL expense	159.0		9.0	681.7	(393.8)		455.9
Lease operating expense	125.3		_	_	(2.5)		122.8
Natural gas, oil and NGL transportation and other handling costs	167.4		27.7	_	(83.6)		111.5
Gathering, processing and other	_		65.6	0.8	_		66.4
General and administrative	92.7		23.7	2.6	(4.5)		114.5
Production and property taxes	63.6		4.6	0.2	_		68.4
Depreciation, depletion and amortization	600.2		47.6	0.5	_		648.3
Other operating expenses	77.2		_	_	_		77.2
Total operating expenses	1,285.4		178.2	685.8	(484.4)		1,665.0
Net gain from asset sales	1.5		_	_	_		1.5
Operating (loss) income	(177.6))	164.8	(5.8)	_		(18.6)
Realized and unrealized gains (losses) on derivative contracts	330.4		8.3	(4.0)	_		334.7
Interest and other income	2.2		0.1	81.1	(81.0)		2.4
Income from unconsolidated affiliates	0.1		5.5	_	_		5.6
Loss on early extinguishment of debt	_		_	(0.6)	_		(0.6)
Interest expense	(71.1))	(9.4)	(83.4)	81.0		(82.9)
Income (loss) before income taxes	84.0		169.3	(12.7)	_		240.6
Income taxes	(32.4))	(59.2)	5.1	_		(86.5)
Net income (loss)	51.6		110.1	(7.6)	_		154.1
Net income attributable to noncontrolling interest	_		(2.7)	_	_		(2.7)
Net income (loss) attributable to QEP	\$ 51.6	\$	5 107.4	\$ (7.6)	\$	\$	151.4

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

Management's Discussion and Analysis of Financial Condition and Results of Operations (MD&A) is intended to provide a reader of the financial statements with a narrative from the perspective of management on the financial condition, results of operations, liquidity and certain other factors that may affect the Company's operating results. MD&A should be read in conjunction with the Condensed Consolidated Financial Statements and related notes included in Item 1 of this Quarterly Report on Form 10-Q.

The following information updates the discussion of QEP's financial condition provided in its 2012 Annual Report on Form 10-K filing and analyzes the changes in the results of operations between the three and nine months ended September 30, 2013 and 2012. For definitions of commonly used gas and oil terms found in this Quarterly Report on Form 10-Q, please refer to the "Glossary of Commonly Used Terms" provided in QEP's 2012 Annual Report on Form 10-K.

OVERVIEW

QEP Resources, Inc. (QEP or the Company) is a holding company with three major lines of business: natural gas and crude oil exploration and production; midstream field services; and energy marketing. These businesses are conducted through the Company's three principal subsidiaries:

- QEP Energy Company (QEP Energy) acquires, explores for, develops and produces natural gas, crude oil, and natural gas liquids (NGL);
- QEP Field Services Company (QEP Field Services, which includes the ownership and operations of QEP Midstream Partners, LP (QEP Midstream)),
 provides midstream field services, including natural gas gathering, processing, compression, and treating services, for affiliates and third parties, and;
- QEP Marketing Company (QEP Marketing) markets affiliate and third-party natural gas and oil, and owns and operates an underground gas storage reservoir.

QEP's operations are focused in two major regions: the Northern Region (primarily in North Dakota, Wyoming and Utah) and the Southern Region (primarily in Oklahoma, Louisiana and the Texas Panhandle) of the United States. QEP's corporate headquarters are located in Denver, Colorado.

Strategies

We create value for our shareholders through returns-focused growth, superior execution, and a low cost structure. To achieve these objectives we strive to:

- operate in a safe and environmentally responsible manner;
- allocate capital to those projects that generate the highest returns;
- acquire businesses and assets that complement or expand our current business;
- maintain a sustainable, diverse inventory of low-cost, high-margin resource plays;
- be in the highest-potential areas of the resource plays in which we operate;
- build contiguous acreage positions that drive operating efficiencies;
- be the operator of our assets, whenever possible;
- be the low-cost driller and producer in each area where we operate;
- own a controlling interest in and operate midstream infrastructure in our core producing areas to capture value downstream of the wellhead;
- build gas processing plants to extract liquids from our natural gas streams;
- own or control assets to gather, compress and treat our production to drive down costs;
- support the growth of our midstream business through the growth of a master limited partnership;
- actively market our QEP Energy production to maximize value;
- utilize derivative contracts to mitigate the impact of natural gas, crude oil or NGL price volatility and fluctuating interest rates, while locking in
 acceptable cash flows required to support future capital expenditures;
- attract and retain the best people; and
- maintain a capital structure that allows us the necessary financial flexibility with which to invest in organic growth and potential acquisition opportunities, as they may arise.

Outlook

The Company has substantial acreage positions and operations in some of the most prolific hydrocarbon resource plays in the continental United States, including the Williston Basin, Pinedale Anticline, Uinta Basin, Woodford "Cana" and Haynesville Shale. These resource plays are characterized by unconventional oil or natural gas accumulations in continuous tight sands or shales that underlie broad geographic areas. The lateral continuity of such resource plays means that aside from wells abandoned due to mechanical issues, the Company does not expect to drill many unsuccessful wells as it develops these resource plays. Resource plays allow the Company the opportunity to gain considerable operational efficiencies through high-density, repeatable drilling and completion operations. The Company has a large inventory of lower-risk, predictable development drilling locations across its acreage holdings in the onshore United States that provide a solid base for consistent growth in organic production and reserves. QEP believes that it has one of the lowest cash operating structures among its exploration and production company peers. However, in certain of its resource plays, QEP, along with its peers, has experienced increased drilling and completion costs which could impact near term drilling plans.

While historically a natural gas producer, the Company has increased its focus on growing the relative proportion of crude oil and NGL production in its exploration and production business. As part of the Company's liquids growth strategy, during the

third quarter of 2012, QEP Energy acquired oil and gas properties in the Williston Basin for an aggregate purchase price of \$1.4 billion (the 2012 Acquisition). During the first nine months of 2013, QEP Energy increased its crude oil and NGL (natural gas liquids) production by 34% compared to the first three quarters of 2012. During the first three quarters of 2013, crude oil and NGL revenue accounted for approximately 57% of QEP Energy's field-level production revenues, compared to 50% during the first three quarters of 2012.

While QEP believes that it can grow production and reserves from its extensive inventory of identified drilling locations, the Company continues to evaluate acquisition opportunities that might create significant long-term value. QEP believes that its experience, expertise, and substantial presence in its core operating areas, combined with its low-cost operating model and financial strength, enhance its ability to pursue acquisition opportunities. In addition, the Company will occasionally divest select non-core portfolio assets to redirect capital towards higher-return projects.

QEP owns and operates, directly or through QEP Midstream, gathering and natural gas processing and treatment facilities in the majority of its core producing areas outside of Oklahoma and Texas. These assets enable the Company to promptly connect its wells, better control its costs, and generate a significant, consistent revenue stream by providing gathering and processing services to third parties. In August 2014, QEP Midstream, a master limited partnership formed by QEP to hold certain midstream gathering assets, completed its initial public offering. QEP expects that QEP Midstream will support the growth of QEP's midstream business and allow QEP to raise additional capital through the drop-down of assets to QEP Midstream.

Financial and Operating Results

QEP Energy reported total equivalent production of 78.0 Bcfe during the third quarter of 2013 and 233.9 Bcfe in the first three quarters of 2013, a decrease of 4% and 1%, respectively, over the same periods in 2012. Crude oil and NGL production in the third quarter and first three quarters of 2013 was 3,798.6 Mbbls and 10,546.2 Mbbls, a combined increase of 34% from the same periods in 2012. The Company's 2012 Acquisition contributed 831.2 Mbbls and 2,131.8 Mbbls of crude oil or NGL production in the third quarter and first three quarters of 2013, respectively.

During the third quarter of 2013, QEP Field Services' gathering throughput volumes decreased 13%, and NGL sales volumes decreased 42%, while fee-based processing volumes were 5% higher than the third quarter of 2012. During the first three quarters of 2013, QEP Field Services' gathering throughput volumes decreased 13%, NGL sales volumes decreased 47% and fee-based processing volumes were 1% lower compared to the first three quarters of 2012.

During the third quarter and first three quarters of 2013, QEP Energy benefited from increased average realized prices compared to 2012. QEP Energy's average total net realized equivalent price (including the impact of settled commodity derivatives) increased 31% to \$6.71 per Mcfe for the third quarter of 2013 and increased 24% to \$6.50 per Mcfe for the first three quarters of 2013 compared to the first three quarters of 2012. As a result of low ethane prices relative to natural gas prices, QEP Field Services' processing plants, in regard to its keep-whole processing activities, operated in ethane rejection mode (where the majority of ethane is left in the production stream and sold as natural gas) through the end of the third quarter of 2013. When in ethane rejection mode, NGL volumes are lower and average NGL prices are higher as a result of the remaining components of the NGL stream having a higher price than ethane. During the third quarter and first three quarters of 2013, NGL sales volumes declined, the impact of which was partially offset by an increase in average net realized NGL sales prices of 32% and 12%, respectively. During the third quarter and first three quarters of 2013, QEP Field Services' fee-based processing rates increased 7% and 11%, respectively, while fee-based gathering rates were flat during the third quarter and the first three quarters of 2013.

On August 14, 2013, QEP Midstream completed its initial public offering (the Offering) of 20,000,000 common units, representing limited partner interests in QEP Midstream, at a price to the public of \$21.00 per common unit. QEP Midstream received net proceeds of \$390.7 million from the sale of the common units, after deducting underwriting discounts and commissions, structuring fees and offering expenses of approximately \$29.3 million. Following the Offering, the underwriters exercised their over-allotment option to purchase an additional 3,000,000 common units, at a price of \$21.00 per common unit, providing additional net proceeds of \$58.9 million, after deducting underwriters' discounts and commissions and structuring fees of \$4.1 million, to QEP Midstream. QEP Midstream used the net proceeds to repay its outstanding debt balance with QEP, which was assumed with the assets contributed to the Partnership, pay revolving credit facility origination fees and make a cash distribution to QEP, a portion of which was used to reimburse QEP for certain capital expenditures it incurred with respect to assets contributed to QEP Midstream.

QEP contributed gathering assets which are located in, or within close proximity to, the Green River Basin located in Wyoming and Colorado, the Uinta Basin located in eastern Utah, and the portion of the Williston Basin located in North Dakota. QEP utilized the proceeds of the cash distribution it received from QEP Midstream in connection with the Offering to fund ongoing

operations, to repay debt under the Company's revolving credit facility and for general corporate purposes. Following the Offering, QEP owns a 57.8% ownership interest in QEP Midstream and consolidates QEP Midstream for financial reporting purposes.

Factors Affecting Results of Operations

Oil, Natural Gas, and NGL Prices

Historically, field-level prices received for QEP's natural gas, NGL, and crude oil production have been volatile and unpredictable, and that volatility is expected to continue. In recent years, domestic natural gas supply has grown faster than natural gas demand, driven by advances in drilling and completion technologies, including horizontal drilling and multi-stage hydraulic fracturing. These changes have allowed producers to extract increased quantities of natural gas from shale, tight sand formations, and other unconventional reservoirs. Increased natural gas supplies have resulted in downward pressure on natural gas prices, while concern about the global economy and other factors has created volatility in the price of crude oil. Changes in the market prices for natural gas, crude oil, and NGL directly impact many aspects of QEP's business, including its financial condition, revenues, results of operations, planned drilling activity and related capital expenditures, liquidity, rate of growth, and costs of goods and services required to drill and complete wells, and may impact the carrying value of its oil and natural gas properties.

QEP uses commodity derivatives to reduce the volatility of the prices QEP receives for a portion of its production and to protect cash flow and returns on invested capital from a drop in commodity prices. Generally, QEP intends to enter into commodity derivative contracts for approximately 50% of its forecasted annual production by the end of the first quarter of each fiscal year. At September 30, 2013, assuming 2013 annual production of 312.5 Bcfe, QEP Energy had approximately 55% of its forecasted natural gas equivalent production covered with fixed-price swaps, including 59% of its forecasted natural gas production and 66% of its forecasted crude oil production covered with fixed-price swaps. See Item 3 "Quantitative and Qualitative Disclosures about Market Risk—Commodity Price Risk Management" for further details concerning QEP's commodity derivatives transactions. In addition, as a result of the continued spread between oil and natural gas prices, QEP Energy has allocated approximately 98% of its forecasted 2013 drilling and completion capital expenditure budget to oil and liquids-rich natural gas projects in its portfolio.

Global Geopolitical and Macroeconomic Factors

QEP continues to monitor the outlook of the global economy, including the European debt crisis and its potential impact on global economic growth and the banking and financial sectors, political unrest in the Middle East and Africa, a slowing of growth in Asia, the United States federal budget deficit and debt ceiling crisis, the recent partial shut-down of Federal government offices including the Department of Interior (including the Bureau of Land Management (BLM) and Bureau of Indian Affairs (BIA) which process permits to drill and rights-of-way for construction of gathering lines and other midstream infrastructure on federal (BLM) and Native American (BIA and BLM) minerals and surface), changes in regulatory oversight policy and commodity price volatility. A dramatic decline in regional or global economic conditions, a major recession or depression, regional political instability, economic sanctions, war, or other factors beyond the control of QEP could have a significant impact on natural gas, NGL and crude oil supply, demand and prices, the Company's ability to continue its planned drilling programs on federal and Native American lands, and could materially impact the Company's financial position, results of operations and cash flow from operations.

Supply, Demand and Other Market Risk Factors

U.S. natural gas directed drilling rig count decreased during 2012 and continued to decrease in the first two quarters of 2013 and flattened in the third quarter of 2013 as producers reduced drilling for natural gas in response to lower natural gas prices. A reduction in natural gas production has lagged the downturn in the natural gas rig count, because natural gas producers have a significant inventory of drilled wells waiting on completion, continued efficiency gains allowing more wells to be drilled and completed per operating rig, and higher per-well natural gas production from horizontal wells which now dominate U.S. completions. As a result of the lag, U.S. natural gas production did not decline in 2012 and has not yet declined in 2013. However, strong natural gas demand from electric power generation has resulted in a general firming of natural gas prices during the last half of 2012 and 2013. Despite increased natural gas prices during the third quarter and first nine months of 2013, QEP expects U.S. natural gas prices to remain range-bound over the near term. Relatively low natural gas prices have caused U.S. E&P companies, including QEP, to shift capital investments away from predominantly dry gas areas toward plays that are known to have liquids-rich natural gas and crude oil. This shift in focus has caused domestic NGL production to increase dramatically. Increased NGL production, several warmer-than-average winters, and price dislocations from infrastructure bottlenecks in certain regions, have all contributed to a weakening of domestic NGL prices, particularly ethane. QEP expects NGL prices to remain range-bound for the foreseeable future. QEP anticipates global crude oil prices to remain near current levels, assuming the global economy and socio-political backdrops remain relatively stable. Disruption to the global oil supply system, political and/or economic instability, and/or other factors could trigger additional volatility in crude

oil prices. In addition, transportation, refining, or other infrastructure constraints could introduce significant price differentials between regional markets where QEP sells its crude oil production and national (NYMEX or Cushing) and global (Brent or U.S. Gulf Coast) markets. Because of the global and regional price volatility and the uncertainty around the commodity price environment, QEP continues to manage its capital spending program and financial flexibility accordingly.

Potential for Future Asset Impairments

The carrying value of the Company's properties is sensitive to declines in natural gas, crude oil and NGL prices. These assets are at risk of impairment if future prices for natural gas, crude oil or NGL prices decline and/or drilling and completion costs increase. The cash flow model that the Company uses to assess proved properties for impairment includes numerous assumptions, such as management's estimates of future oil, gas and NGL production, market outlook on forward commodity prices, operating and development costs, and discount rates. All inputs to the cash flow model must be evaluated at each date of estimate. However, a decrease in forward natural gas, crude oil or NGL prices alone could result in an impairment of properties. The Company did not record any proved property impairments during the first three quarters of 2013.

Multi-Well Pad Drilling

To reduce the costs of well location construction and rig mobilization and demobilization and to obtain other efficiencies, QEP utilized multi-well pad drilling where practical. Because wells drilled on a pad are not brought into production until all wells on the pad are drilled and completed and the drilling rig is moved from the location, multi-well pad drilling delays the commencement of production, which may cause volatility in QEP's quarterly operating results.

Critical Accounting Estimates

QEP's significant accounting policies are described in Item 7 of Part II of its 2012 Annual Report on Form 10-K. The Company's Condensed Consolidated Financial Statements are prepared in accordance with United States Generally Accepted Accounting Principles (GAAP). The preparation of the Company's Condensed Consolidated Financial Statements requires management to make assumptions and estimates that affect the reported results of operations and financial position. QEP's accounting policies on gas and oil reserves, successful efforts accounting for gas and oil operations, impairment of gas and oil properties, asset retirement obligations, accounting for derivative contracts, revenue recognition, environmental obligations, litigation and other contingencies, benefit plan obligations, share-based compensation, income taxes, and purchase price allocations, among others, may involve a high degree of complexity and judgment on the part of management.

RESULTS OF OPERATIONS

Net Income (Loss)

QEP's net income for the third quarter of 2013 was \$37.3 million, or \$0.21 per diluted share, compared to a net loss of \$3.1 million, or \$0.02 per diluted share, in the third quarter of 2012. The increase in the third quarter of 2013 was due to a \$35.8 million increase in QEP Energy's net income and an \$11.3 million increase in QEP Marketing and Resources' net income, offset by a \$6.7 million decrease in QEP Field Services' net income when compared to the third quarter of 2012. QEP Energy's net income increase was primarily related to the higher realized equivalent prices and increased oil volumes and the \$12.8 million net gain primarily recorded in connection with sales of certain non-core properties partially offset by lower realized gains on derivative instruments, higher depreciation, depletion and amortization expenses, and higher interest expense when compared to the third quarter of 2012. The increase in QEP Marketing and Resources' net income is related to intercompany interest income from interest expense charges to QEP's subsidiaries. QEP Field Services' decrease in net income during the third quarter of 2013 was driven by an 18% decrease in the keep-whole processing margin and a 13% lower gathering margin.

QEP's net income for the first three quarters of 2013 was \$211.4 million, or \$1.18 per diluted share, compared to net income of \$151.4 million, or \$0.85 per diluted share in the first three quarters of 2012. The increase in the first three quarters of 2013 was primarily due to a \$70.3 million increase in QEP Energy's net income and a \$27.1 million increase in QEP Marketing and Resources' net income, offset by a \$37.4 million decrease in QEP Field Services' net income. QEP Energy's net income increased in the first three quarters of 2013 due to an increase in realized equivalent prices, increased oil volumes, gain from asset sales, and a decrease in property impairments, partially offset by an increase in depreciation, depletion and amortization expense, increase in production and property taxes, and interest expense and a decrease in unrealized gains on commodity derivative contracts. The increase in QEP Marketing and Resources' net income is related to intercompany interest income from interest expense charges to QEP's subsidiaries. QEP Field Services' decrease in net income during the first three quarters of 2013 was driven by a 43% decrease in the keep-whole processing margin and 13% lower gathering margins.

The following table provides a summary of net income (loss) attributable to QEP by line of business:

	Three Months Ended September 30,				er 30,	Nine Months Ended September 30,					
	2013		2012		Change		2013		2012		Change
					(in m	illions	s)				_
QEP Energy	\$ 9.6	\$	(26.2)	\$	35.8	\$	121.9	\$	51.6	\$	70.3
QEP Field Services	22.0		28.7		(6.7)		70.0		107.4		(37.4)
QEP Marketing and Resources	 5.7		(5.6)		11.3		19.5		(7.6)		27.1
Net income (loss) attributable to QEP	\$ 37.3	\$	(3.1)	\$	40.4	\$	211.4	\$	151.4	\$	60.0
Earnings per diluted share	\$ 0.21	\$	(0.02)	\$	0.23	\$	1.18	\$	0.85	\$	0.33
Average diluted shares	179.5		177.9		1.6		179.4		178.6		0.8

Adjusted EBITDA

Management believes Adjusted EBITDA (a non-GAAP measure) is an important measure of the Company's cash flow, liquidity, and ability to incur and service debt, fund capital expenditures and make distributions to shareholders. The use of this measure allows investors to understand how management evaluates financial performance to make operating decisions and allocate resources. It is also an important measure for comparing the Company's financial performance to other gas and oil producing companies. Management defines Adjusted EBITDA as net income attributable to QEP before the following items: certain significant accrued litigation loss contingencies, depreciation, depletion and amortization (DD&A), exploration expense, impairment, gains and losses from asset sales, unrealized gains and losses on derivative contracts, interest and other income, interest expense, loss on early extinguishment of debt and income taxes.

The following table provides a summary of Adjusted EBITDA by line of business:

	Three Months Ended September 30,					Nine Months Ended September 30,					
	 2013	:	2012		Change	_	2013		2012		Change
					(in n	nilli	ons)				
QEP Energy	\$ 344.0		267.7	\$	76.3	\$	999.8		801.7	\$	198.1
QEP Field Services	51.6		66.4		(14.8)		163.1		219.4		(56.3)
QEP Marketing and Resources	(0.5)		(2.8)		2.3		(3.3)		(2.2)		(1.1)
Adjusted EBITDA	\$ 395.1	\$	331.3	\$	63.8	\$	1,159.6	\$	1,018.9	\$	140.7

Adjusted EBITDA increased to \$395.1 million in the third quarter of 2013 from \$331.3 million in the third quarter of 2012, due to a 31% increase in average net realized equivalent prices and an 83% increase in oil production at QEP Energy offset in part by decreases in processing and gathering margins at QEP Field Services.

During the first three quarters of 2013, Adjusted EBITDA increased to \$1,159.6 million from \$1,018.9 million in 2012, due to 5% higher net realized natural gas prices, 7% higher net realized crude oil prices, 19% higher realized NGL prices as a result of the rejection of lower value ethane and an 80% increase in total oil production at QEP Energy that was partially offset by decreases in processing and gathering margins at QEP Field Services and lower natural gas production volumes at QEP Energy.

The following tables are reconciliations of Adjusted EBITDA to net income (loss) attributable to QEP, the most comparable GAAP financial measure:

	QEP Energy	QEP Field Services	QEP Marketing & Resources	QEP
Three Months Ended September 30, 2013		(in mi	illions)	
Net income attributable to QEP	9.6	22.0	5.7	\$ 37.3
Unrealized losses on derivative contracts	52.8	_	1.3	54.1
Net gain from asset sales	(12.8)	_	_	(12.8)
Interest and other income	(2.6)	(0.8)	(0.6)	(4.0)
Income tax provision	6.2	12.7	3.5	22.4
Interest expense ⁽¹⁾	49.2	3.1	(10.7)	41.6
Depreciation, depletion and amortization ⁽²⁾	236.0	14.6	0.3	250.9
Impairment	3.8	_	_	3.8
Exploration expenses	1.8			1.8
Adjusted EBITDA	\$ 344.0	\$ 51.6	\$ (0.5)	\$ 395.1
Three Months Ended September 30, 2012				
Net (loss) income attributable to QEP	(26.2)	28.7	(5.6)	\$ (3.1)
Unrealized losses on derivative contracts	50.9	2.5	3.7	57.1
Interest and other income	0.2	_	_	0.2
Income tax (benefit) provision	(15.3)	16.5	(3.5)	(2.3)
Interest expense	24.1	3.5	2.4	30.0
Accrued litigation loss contingency ⁽³⁾	4.3	_	_	4.3
Loss on early extinguishment of debt	_	_	_	_
Depreciation, depletion and amortization ⁽²⁾	218.5	15.2	0.2	233.9
Impairment	9.0	_	_	9.0
Exploration expenses	2.2			2.2
Adjusted EBITDA	\$ 267.7	\$ 66.4	\$ (2.8)	\$ 331.3
Nine Months Ended September 30, 2013				
Net income attributable to QEP	121.9	70.0	19.5	\$ 211.4
Unrealized losses (gains) on derivative contracts	58.7	_	(3.2)	55.5
Net (gain) loss from asset sales	(113.4)	0.4	_	(113.0)
Interest and other income	(7.5)	(1.1)	(0.5)	(9.1)
Income tax provision	71.1	40.3	13.6	125.0
Interest expense (income) (1)	143.4	12.4	(33.4)	122.4
Depreciation, depletion and amortization ⁽²⁾	712.1	41.1	0.7	753.9
Impairment	4.0	_	_	4.0
Exploration expenses	9.5			9.5
Adjusted EBITDA	\$ 999.8	\$ 163.1	\$ (3.3)	\$ 1,159.6
Nine Months Ended September 30, 2012				
Net income (loss) attributable to QEP	51.6	107.4	(7.6)	
Unrealized (gains) losses on derivative contracts	(37.9)	(2.0)	7.1	(32.8)
Net gain from asset sales	(1.5)	_	_	(1.5)
Interest and other income	(2.2)	(0.1)	(0.1)	(2.4)

	QEP Energy	QEP Field Services	QEP Marketing & Resources	QEP
Income tax provision (benefit)	32.4	59.2	(5.1)	86.5
Interest expense	71.1	9.4	2.4	82.9
Accrued litigation loss contingency ⁽³⁾	10.8	_	_	10.8
Loss on early extinguishment of debt	_	_	0.6	0.6
Depreciation, depletion and amortization ⁽²⁾	600.2	45.5	0.5	646.2
Impairment	70.9	_	_	70.9
Exploration expenses	6.3	_	_	6.3
Adjusted EBITDA	\$ 801.7	\$ 219.4	\$ (2.2)	\$ 1,018.9

⁽¹⁾ Excludes noncontrolling interest's share of \$0.1 million during the three and nine months ended September 30, 2013, of interest expense attributable to OEP Midstream.

⁽²⁾ Excludes noncontrolling interests' share of \$2.2 million and \$0.7 million during the three months ended September 30, 2013 and 2012, respectively, and \$3.2 million and \$2.1 million during the nine months ended September 30, 2013 and 2012, respectively, of depreciation, depletion and amortization attributable to Rendezvous Gas Services, L.L.C and QEP Midstream.

⁽³⁾ Includes certain significant litigation contingency items for the three and nine months ended September 30, 2012.

QEP ENERGY

The following table provides a summary of QEP Energy's financial and operating results:

	Three Months Ended September 30,						Nine Months Ended September 30,					
		2013		2012		Change		2013		2012		Change
Revenues						(in mi	llioi	ns)				
Natural gas sales	\$	194.8	\$	170.3	\$	24.5	\$	610.5	\$	470.4	\$	140.1
Oil sales		253.8		117.7		136.1		656.3		335.7		320.6
NGL sales		47.7		38.6		9.1		144.4		134.3		10.1
Purchased gas, oil and NGL sales		39.3		45.3		(6.0)		156.6		159.1		(2.5)
Other		2.9		2.1		0.8		7.4		6.8		0.6
Total Revenues		538.5		374.0		164.5		1,575.2		1,106.3		468.9
Operating expenses												
Purchased gas, oil and NGL expense		39.2		45.9		(6.7)		159.8		159.0		0.8
Lease operating expense		43.5		43.1		0.4		130.2		125.3		4.9
Natural gas, oil and NGL transportation and other handling costs		65.1		59.8		5.3		180.8		167.4		13.4
General and administrative		33.5		31.6		1.9		100.2		92.7		7.5
Production and property taxes		40.4		22.5		17.9		112.7		63.6		49.1
Depreciation, depletion and amortization		236.0		218.5		17.5		712.1		600.2		111.9
Exploration expenses		1.8		2.2		(0.4)		9.5		6.3		3.2
Impairment		3.8		9.0		(5.2)		4.0		70.9		(66.9)
Total Operating Expenses		463.3		432.6		30.7		1,409.3		1,285.4		123.9
Net gain from asset sales		12.8		_		12.8		113.4		1.5		111.9
Operating Income (Loss)		88.0		(58.6)		146.6	_	279.3		(177.6)		456.9
Realized gains on derivative instruments		27.2		92.3		(65.1)		108.3		292.5		(184.2)
Unrealized (losses) gains on derivative instruments		(52.8)		(50.9)		(1.9)		(58.7)		37.9		(96.6)
Interest and other income (loss)		2.6		(0.2)		2.8		7.5		2.2		5.3
Income from unconsolidated affiliates		_		_		_		_		0.1		(0.1)
Interest expense		(49.2)		(24.1)		(25.1)		(143.4)		(71.1)		(72.3)
Income (Loss) before Income Taxes		15.8		(41.5)		57.3		193.0		84.0		109.0
Income tax (provision) benefit		(6.2)		15.3		(21.5)		(71.1)		(32.4)		(38.7)
Net Income (Loss) Attributable to QEP	\$	9.6	\$	(26.2)	\$	35.8	\$	121.9	\$	51.6	\$	70.3
Production volumes (Bcfe)												
Northern Region												
Pinedale		25.2		28.0		(2.8)		70.1		73.9		(3.8)
Williston Basin		11.8		3.7		8.1		31.9		10.2		21.7
Uinta Basin		7.3		6.4		0.9		20.1		16.9		3.2
Legacy		2.6		3.6		(1.0)		9.6		10.4		(8.0)
Southern Region												
Haynesville/Cotton Valley		16.2		27.9		(11.7)		57.3		86.8		(29.5)
Midcontinent		14.9		11.9		3.0		44.9		37.1		7.8
Total production		78.0		81.5		(3.5)		233.9		235.3		(1.4)
Total equivalent prices (per Mcfe)												
Average equivalent field-level price	\$	6.36	\$	4.01	\$	2.35	\$	6.03	\$	4.00	\$	2.03
Commodity derivative impact		0.35		1.13		(0.78)		0.47		1.24		(0.77)
Net realized equivalent price	\$	6.71	\$	5.14	\$	1.57	\$	6.50	\$	5.24	\$	1.26

Revenue, Volume and Price Variance Analysis

The following table shows volume and price related changes for each of QEP Energy's major revenue categories for the three and nine months ended September 30, 2013 compared to the three and nine months ended September 30, 2012:

	Natural Gas			Oil		NGL	Total
				(in m	illions)	
QEP Energy Production Revenues							
Three months ended September 30, 2012 Revenues	\$	170.3	\$	117.7	\$	38.6	\$ 326.6
Changes associated with volumes (1)		(24.2)		98.1		(6.5)	67.4
Changes associated with prices (2)		48.7		38.0		15.6	102.3
Three months ended September 30, 2013 Revenues	\$ 194.8		\$	253.8	\$ 47.7		\$ 496.3
QEP Energy Production Revenues							
Nine Months ended September 30, 2012 Revenues		470.4		335.7		134.3	\$ 940.4
Changes associated with volumes (1)		(44.2)		270.0		(18.2)	207.6
Changes associated with prices (2)		184.3		50.6		28.3	263.2
Nine Months ended September 30, 2013 Revenues	\$	610.5	\$	656.3	\$	144.4	\$ 1,411.2

⁽¹⁾ The revenue variance attributed to the change in volume is calculated by multiplying the change in volumes from the three and nine months ended September 30, 2013, as compared to the three and nine months ended September 30, 2012, by the average field-level price for the three and nine months ended September 30, 2012.

Natural Gas Volumes and Prices

	Т	hree Mo	nths E	nded Sep	oteml	oer 30,	Niı	ne Mont	ths Ende	d Sept	embe	er 30,
	2	2013	2	2012	(Change	20)13	20	12	С	hange
Natural gas production volumes (Bcf)												
Northern Region												
Pinedale		22.0		21.7		0.3		61.1		56.9		4.2
Williston Basin		0.6		0.1		0.5		2.1		0.3		1.8
Uinta Basin		5.1		4.5		0.6		14.1		11.8		2.3
Legacy		2.2		2.9		(0.7)		8.2		8.7		(0.5)
Southern Region												
Haynesville/Cotton Valley		16.1		27.7		(11.6)		57.0		86.5		(29.5)
Midcontinent		9.2		7.6		1.6		28.1		23.8		4.3
Total production		55.2		64.5		(9.3)		170.6		188.0		(17.4)
Natural gas prices (per Mcf)												
Northern Region	\$	3.48	\$	2.53	\$	0.95	\$	3.59	\$	2.42	\$	1.17
Southern Region		3.57		2.73		0.84		3.56		2.56		1.00
Average field-level price	\$	3.52	\$	2.64	\$	0.88	\$	3.58	\$	2.50	\$	1.08
Commodity derivative impact		0.77		1.34		(0.57)		0.65		1.51		(0.86)
Net realized price	\$	4.29	\$	3.98	\$	0.31	\$	4.23	\$	4.01	\$	0.22

Natural gas revenues increased \$24.5 million, or 14%, in the third quarter of 2013 when compared to the third quarter of 2012 due to higher field-level prices partially offset by lower volumes. The decrease in production volumes was driven by the suspension of QEP's Haynesville/Cotton Valley operated drilling program partially offset by increased production in QEP's

⁽²⁾ The revenue variance attributed to the change in price is calculated by multiplying the change in average field-level prices from the three and nine months ended September 30, 2013, as compared to the three and nine months ended September 30, 2012, by volumes for the three and nine months ended September 30, 2013.

Pinedale, Midcontinent, Uinta Basin and Williston Basin properties. Natural gas field-level prices increased 33% during the third quarter of 2013 as a result of increased near-term demand.

Natural gas revenues increased \$140.1 million, or 30%, during the first three quarters of 2013 when compared to the first three quarters of 2012 due to higher field-level prices partially offset by lower volumes. The decrease in production volumes was driven by the suspension of QEP's Haynesville/Cotton Valley operated drilling program partially offset by increased production from its drilling programs in Pinedale, the Midcontinent, Uinta Basin and Williston Basin. Natural gas field-level prices increased 43% during the first three quarters of 2013 as a result of increased near-term demand.

Oil Volumes and Prices

		Three Mo	nths	Ended Sep	tem	ber 30,	Nine Mont	ths I	Ended Sept	emb	er 30,
		2013		2012		Change	2013		2012	(Change
Crude oil production volumes (Mbbl)											
Northern Region											
Pinedale		180.7		187.4		(6.7)	490.6		493.5		(2.9)
Williston Basin		1,799.1		575.9		1,223.2	4,641.9		1,584.3		3,057.6
Uinta Basin		265.7		210.4		55.3	717.3		630.7		86.6
Legacy		48.5		85.0		(36.5)	200.5		222.6		(22.1)
Southern Region											
Haynesville/Cotton Valley		11.5		12.1		(0.6)	33.9		34.5		(0.6)
Midcontinent		339.2		371.8		(32.6)	1,084.6		1,007.5		77.1
Total production		2,644.7		1,442.6		1,202.1	7,168.8		3,973.1		3,195.7
Crude oil prices (per bbl)	_										
Northern Region	\$	95.05	\$	80.00	\$	15.05	\$ 91.32	\$	82.04	\$	9.28
Southern Region		102.06		86.01		16.05	92.80		91.38		1.42
Average field-level price	\$	95.98	\$	81.60	\$	14.38	\$ 91.55	\$	84.49	\$	7.06
Commodity derivative impact		(5.79)		1.83		(7.62)	(0.52)		0.55		(1.07)
Net realized price	\$	90.19	\$	83.43	\$	6.76	\$ 91.03	\$	85.04	\$	5.99

Oil revenues increased \$136.1 million, or 116%, in the third quarter of 2013 when compared to the third quarter of 2012 due to higher volumes and higher average prices. The increase in production volumes was primarily driven by increases in the Williston Basin due to the properties acquired in the 2012 Acquisition, which contributed 773.4 Mbbls during the third quarter of 2013, and an additional 449.8 Mbbls increase in Williston Basin production related to increased development drilling on QEP's existing pre-acquisition acreage throughout 2013. Field-level oil prices increased 18% in the third quarter of 2013, primarily due to higher average West Texas Intermediate (WTI) and Brent spot crude oil prices in the third quarter of 2013 compared to 2012 and higher pricing in the Williston Basin due to an improvement in price differentials and the sale of crude oil referenced against Brent prices, which were higher than WTI prices in the third quarter of both 2013 and 2012.

Oil revenues increased \$320.6 million, or 96%, in the first three quarters of 2013 when compared to the first three quarters of 2012 due to higher volumes and higher average prices. The increase in production volumes was primarily driven by increases in the Williston Basin due to the properties acquired in the 2012 Acquisition, which contributed 1,911.9 Mbbls during the first three quarters of 2013, and an additional 1,145.7 Mbbls increase in Williston Basin's production related to increased development drilling on QEP's existing pre-acquisition acreage throughout 2013. Additionally, Uinta Basin and Midcontinent oil volumes increased during the first three quarters of 2013 due to new well completions throughout the first nine months of 2013. Field-level oil prices increased 8% in the first three quarters of 2013 due to improved pricing for Williston Basin crude volumes despite a decrease in Brent crude oil prices and only a slight increase in WTI crude prices.

	Three Mo	nths	Ended Sep	tem	ber 30,	Nine Mor	ıths	Ended Sep	temb	er 30,
	2013		2012	(Change	2013		2012	(Change
NGL production volumes (Mbbl)										
Northern Region										
Pinedale	355.8		861.3		(505.5)	1,005.4		2,327.2		(1,321.8)
Williston Basin	66.8		23.8		43.0	321.8		64.4		257.4
Uinta Basin	111.0		116.7		(5.7)	283.1		224.6		58.5
Legacy	12.9		23.7		(10.8)	43.8		72.7		(28.9)
Southern Region										
Haynesville/Cotton Valley	3.6		2.0		1.6	12.9		6.4		6.5
Midcontinent	603.8		359.2		244.6	1,710.4		1,210.9		499.5
Total production	1,153.9		1,386.7		(232.8)	3,377.4		3,906.2		(528.8)
NGL prices (per bbl)										
Northern Region	\$ 54.91	\$	30.75	\$	24.16	\$ 55.22	\$	36.79	\$	18.43
Southern Region	29.18		19.56		9.62	30.79		29.05		1.74
Average field-level price	\$ 41.36	\$	27.83	\$	13.53	\$ 42.75	\$	34.38	\$	8.37
Commodity derivative impact	_		2.46		(2.46)	_		1.66		(1.66)
Net realized price	\$ 41.36	\$	30.29	\$	11.07	\$ 42.75	\$	36.04	\$	6.71

NGL revenues increased \$9.1 million, or 24%, during the third quarter of 2013 when compared to the third quarter of 2012 due to an increased average price per barrel partially offset by decreased production volumes. NGL prices increased 49% during the third quarter of 2013 primarily as a result of not recovering ethane from the wet gas production stream. When ethane is sold as part of the natural gas stream instead of being recovered as an NGL, the average NGL barrel sales price increases as the price of the remaining NGL components are higher than the ethane price. Pinedale NGL volumes decreased due to our processing plants running in ethane rejection mode. The decrease in Pinedale was partially offset by increases in the Midcontinent and Williston Basin. The Midcontinent increase was driven by new well completions in the Woodford "Cana" shale and Granite Wash areas while Williston Basin NGL volumes grew as a result of the 2012 Acquisition, which contributed 57.8 Mbbls in the third quarter of 2013.

During the first three quarters of 2013, NGL revenues increased \$10.1 million, or 8%, when compared to the first three quarters of 2012 due to an increased price per barrel partially offset by decreased production volumes. NGL field-level prices increased 24% as a result of the rejection of lower value ethane. Production volumes decreased in Pinedale due to ethane rejection. The decrease in Pinedale was partially offset by increases in the Midcontinent, Williston Basin and Uinta Basin. The Midcontinent increase was driven by additional Woodford "Cana" shale wells, while the Williston Basin's NGL volume growth was primarily the result of the 2012 Acquisition, which contributed 219.9 Mbbls to the first three quarters of 2013. In addition, the Uinta Basin NGL volumes increased despite processing plants running in ethane rejection mode due to QEP Energy executing a fee-based processing agreement in mid-2012 with QEP Field Services for a portion of the natural gas production from the Red Wash Unit.

QEP Energy Resale Margin

QEP Energy purchases and resells gas, oil and NGL products in order to fulfill firm transportation contract commitments and mitigate potential losses. The difference between the price of the products purchased and sold creates a resale margin that represents a gain or loss for the Company. The following table is a summary of QEP Energy's financial results from its gas, oil and NGL resale activities:

	Tì	ree Mon	ths 1	Ended Se	ptem	ber 30,		Nine Months Ended September 30,						
		2013		2012	С	hange		2013		2012	C.	nange		
Resale Margin	(in mill						lion	s)						
Purchased gas, oil and NGL sales	\$ 39.3 \$ 45.3 \$			\$	(6.0)	\$	156.6	\$	159.1	\$	(2.5)			
Purchased gas, oil and NGL expense		(39.2)		(45.9)		6.7		(159.8)		(159.0)		(8.0)		
Resale margin gain (loss)	\$	0.1	\$	(0.6)	\$	0.7	\$	(3.2)	\$	0.1	\$	(3.3)		

During the third quarter of 2013, QEP Energy recorded a gain on resale margin of \$0.1 million and during the first three quarters of 2013, QEP Energy recorded a loss on resale margin of \$3.2 million as a result of its activities to utilize pipeline transportation commitments in Louisiana. The Company has transportation commitments in excess of its current production as a result of the suspension of its Haynesville drilling program.

QEP Energy Drilling Activity

The following table presents operated and non-operated well completions for the three and nine months ended September 30, 2013:

		Operated Co	ompletions		Non-operated Completions						
	Three Mont September		Nine Montl September		Three Mont September		Nine Month September				
	Gross	Net	Gross	Net	Gross	Net	Gross	Net			
Northern Region						,					
Pinedale	22	17.3	79	59.3	_	_	_	_			
Williston Basin	21	18.4	48	42.2	27	0.9	79	3.6			
Uinta Basin	14	13.6	34	31.0	_	_	93	0.2			
Legacy	_	_	_	_	_	_	6	0.2			
Southern Region											
Haynesville/Cotton Valley	_	_	5	2.4	_	_	2	0.2			
Midcontinent	3	2.2	22	17.6	62	8.3	127	12.1			

The following table presents operated and non-operated wells being drilled or waiting on completion at September 30, 2013:

		Opera	ated		Non-operated							
	Being d	rilled	Waiting on c	ompletion	Being d	rilled	Waiting on c	ompletion				
	Gross	Net	Gross	Net	Gross	Net	Gross	Net				
Northern Region												
Pinedale	12	10.2	31	24.2	_	_	_	_				
Williston Basin	15	13.6	12	10.9	24	1.6	10	0.6				
Uinta Basin ⁽¹⁾	2	2.0	4	4.0	_	_	_	_				
Legacy	_		_	_	_	_	_	_				
Southern Region												
Haynesville/Cotton Valley	_		_	_	9	0.1	6	0.8				
Midcontinent	_	_	1	0.7	13	1.2	47	4.5				

(1) The non-operated well total for the Uinta Basin does not include wells that are being drilled by an outside operator in the Monument Butte unit in which QEP owns a very small working interest.

The term "gross" refers to all wells or acreage in which QEP has at least a partial working interest and the term "net" refers to QEP's ownership represented by that working interest. Each gross well completed in more than one producing zone is counted as a single well. QEP utilizes multi-well pad drilling where practical. Wells drilled are not brought into production until all wells on the pad are drilled and completed and the drilling rig is moved from the location. As a result, QEP had 48 gross operated wells waiting on completion as of September 30, 2013.

Operating expenses

The following table presents certain QEP Energy operating expenses on a per unit of production basis.

	Three Months Ended September 30,							Nine Months Ended September 30,					
		2013		2012		Change		2013		2012		Change	
	(per Mc				(cfe)								
Depreciation, depletion and amortization	\$	3.02	\$	2.68	\$	0.34	\$	3.04	\$	2.55	\$	0.49	
Lease operating expense		0.56		0.53		0.03		0.56		0.53		0.03	
Natural gas, oil and NGL transportation and other													
handling costs		0.84		0.73		0.11		0.77		0.71		0.06	
Production taxes		0.51		0.27		0.24		0.48		0.27		0.21	
Operating Expenses	\$	4.93	\$	4.21	\$	0.72	\$	4.85	\$	4.06	\$	0.79	

Depreciation, depletion and amortization. DD&A expense increased \$17.5 million, or \$0.34 per Mcfe, in the third quarter of 2013 compared to the third quarter of 2012 due to increases in the Williston Basin and Haynesville/Cotton Valley partially offset by a decrease in the Uinta Basin. The increase in the Williston Basin rate is due to the additional proved costs recorded as part of the 2012 Acquisition while the increase in the Haynesville/Cotton Valley rate was due to a year-end 2012 negative revision of proved undeveloped reserves associated with lower prices. These increases were partially offset by a decrease in the Uinta Basin rate due to a 2012 proved property impairment and the addition of proved undeveloped reserves recorded at year-end 2012.

During the first three quarters of 2013, DD&A expense increased \$111.9 million, or \$0.49 per Mcfe, due to increases in the Williston Basin and Haynesville/Cotton Valley partially offset by a decrease in the Uinta Basin. The increase in the Williston Basin rate is due to the additional proved costs recorded as part of the 2012 Acquisition while the increase in the Haynesville/Cotton Valley rate was due to a year-end 2012 negative revision of proved undeveloped reserves as a result of lower natural gas prices. These increases were partially offset by a decrease in the Uinta Basin rate due to a 2012 proved property impairment and the addition of proved undeveloped reserves recorded at year-end 2012.

Lease operating expense. The following table presents lease operating expenses (LOE) for QEP Energy by region on a unit of production basis:

	Three	Months E	nded	September 3	0,		Nine	e Months En			
		2013		2012	(Change		2013	2012		Change
						(per	Mcfe)			
Northern Region	\$	0.51	\$	0.61	\$	(0.10)	\$	0.59	\$ 0.59	\$	_
Southern Region		0.62		0.44		0.18		0.51	0.48		0.03
Average lease operating expense		0.56		0.53		0.03		0.56	0.53		0.03

QEP Energy's LOE increased \$0.4 million, or \$0.03 per Mcfe, during the third quarter of 2013 compared to the third quarter of 2012. The Southern Region's LOE per Mcfe increase during the third quarter of 2013 was driven by a per Mcfe increase in its Haynesville/Cotton Valley properties due to declining production volume but relatively flat labor and pumper costs, fixed operating expenses due to the slight increase in total well count and increased repairs and maintenance costs. The Northern Region decrease was driven primarily by a per Mcfe decrease in the Williston due to cost efficiencies in the current year attributable to the 2012 Acquisition.

QEP Energy's LOE increased \$4.9 million, or \$0.03 per Mcfe, during the first three quarters of 2013 compared to the first three quarters of 2012. The Northern Region's LOE per Mcfe was flat period to period due to a decrease per Mcfe in its Williston Basin properties that was offset by an increase per Mcfe in its Pinedale properties. The decrease in LOE per Mcfe in the Williston Basin was primarily due to cost efficiencies in the current year attributable to the 2012 Acquisition. The increase in LOE per Mcfe in Pinedale was primarily the result of increased workover expense in the first three quarters of 2013. The Southern Region's LOE per Mcfe increase in the first three quarters of 2013 was primarily driven by a 21% per Mcfe increase in its Haynesville properties due to declining production volumes with relatively flat LOE which was partially offset by a per Mcfe decrease at its Woodford properties due to additional cost savings measures in 2013.

Natural gas, oil and NGL transportation and other handling costs. Natural gas, oil and NGL transportation and other handling costs increased \$5.3 million, or \$0.11 per Mcfe, in the third quarter of 2013 when compared to the third quarter of 2012, due to cost increases in the Midcontinent and declining production volumes in the Haynesville/Cotton Valley area.

Natural gas, oil and NGL transportation and other handling costs increased \$13.4 million, or \$0.06 per Mcfe, in the first three quarters of 2013 when compared to the first three quarters of 2012 due to cost increases in the Williston Basin, Haynesville/Cotton Valley field and the Midcontinent. Transportation and other handling costs per Mcfe in the Williston Basin increased 45% due to increased gathering costs associated with the acquired properties from the 2012 Acquisition. Haynesville/Cotton Valley transportation and other handling costs per Mcfe increased 29% due to firm transportation commitments and declining production volumes.

Production and property taxes. In most states in which QEP Energy operates, QEP pays production taxes based on a percentage of field-level revenue, except in Louisiana, where severance taxes are volume-based. Production taxes increased \$17.9 million, or \$0.24 per Mcfe, during the third quarter of 2013 as a result of increased natural gas, oil and NGL revenues due to higher field-level natural gas, oil and NGL prices and higher oil production.

Production taxes increased \$49.1 million, or \$0.21 per Mcfe, during the first three quarters of 2013 when compared to the first three quarters of 2012 as a result of increased revenues due to higher field-level natural gas, oil and NGL prices and higher oil production volumes somewhat offset by lower natural gas production volumes.

Exploration expense. Exploration expenses decreased \$0.4 million during the third quarter of 2013 when compared with the third quarter of 2012. The decrease primarily related to decreases in exploration-related labor.

During the first three quarters of 2013, exploration expenses increased \$3.2 million compared to the first three quarters of 2012. The 2013 increase primarily related to an increase in seismic and geographical studies of \$3.0 million and an increase in exploration consulting services expense.

Impairment expense. During the third quarter and first three quarters of 2013, impairment expense was \$3.8 million and \$4.0 million, respectively, due to the write-off of expiring unproved leaseholds. Impairment expenses of \$9.0 million and \$70.9 million were recorded during the third quarter and first three quarters of 2012 on certain proved and unproved properties due to lower natural gas prices and leasehold expirations.

QEP FIELD SERVICES

The following table provides a summary of QEP Field Services' financial and operating results:

	Three Months Ended September 30,						Nine Months Ended September 30,				
	2013		2012		Change		2013		2012		Change
					(in mill	ion	s)				
Revenues											
NGL sales	\$ 23.6	\$	28.9	\$	(5.3)	\$	70.6	\$	112.7	\$	(42.1)
Processing (fee based)	20.6		18.2		2.4		56.4		51.8		4.6
Other processing fees	8.2		5.4		2.8		13.1		8.4		4.7
Gathering	39.2		43.9		(4.7)		114.6		131.6		(17.0)
Other gathering	8.5		8.0		0.5		31.8		28.6		3.2
Purchased gas, oil and NGL sales			5.3		(5.3)		8.5		9.9		(1.4)
Total Revenues	100.1		109.7		(9.6)		295.0		343.0		(48.0)
Operating expenses	'										
Purchased gas, oil and NGL expense	_		4.9		(4.9)		8.6		9.0		(0.4)
Processing	4.4		4.7		(0.3)		12.6		12.1		0.5
Processing plant fuel and shrinkage	7.2		8.1		(0.9)		22.4		26.6		(4.2)
Gathering	10.4		9.0		1.4		30.3		26.9		3.4
Natural gas, oil and NGL transportation and other											
handling costs	3.4		6.9		(3.5)		11.6		27.7		(16.1)
General and administrative	15.8		10.5		5.3		36.2		23.7		12.5
Taxes other than income taxes	2.0		1.7		0.3		4.8		4.6		0.2
Depreciation, depletion and amortization	16.8		15.9		0.9		44.3		47.6		(3.3)
Total Operating Expenses	60.0		61.7		(1.7)		170.8		178.2		(7.4)
Net (loss) from asset sales	_		_				(0.4)				(0.4)
Operating Income	40.1		48.0		(7.9)		123.8		164.8		(41.0)
Interest and other income	0.8		_		8.0		1.1		0.1		1.0
Income from unconsolidated affiliates	0.8		2.3		(1.5)		3.7		5.5		(1.8)
Realized gains on derivative instruments	_		1.9		(1.9)		_		6.3		(6.3)
Unrealized (losses) gains on derivative instruments	_		(2.5)		2.5		_		2.0		(2.0)
Interest expense	(3.2)		(3.5)		0.3		(12.5)		(9.4)		(3.1)
Income before Income Taxes	38.5		46.2		(7.7)		116.1		169.3		(53.2)
Income taxes	(12.7)		(16.5)		3.8		(40.3)		(59.2)		18.9
Net income	25.8		29.7		(3.9)		75.8		110.1		(34.3)
Net income attributable to noncontrolling interest	(3.8)		(1.0)		(2.8)		(5.8)		(2.7)		(3.1)
Net Income Attributable to QEP	\$ 22.0	\$	28.7	\$	(6.7)	\$	70.0	\$	107.4	\$	(37.4)

Gathering Margin

The following tables present a summary of QEP Field Services' financial and operating results from gathering activities:

		Three M	s Ended Sep	ber 30,	Nine Months Ended September 30,							
		2013		2012	Change			2013		2012		Change
Gathering Margin						(in mil	lions)					
Gathering revenues	\$	39.2	\$	43.9	\$	(4.7)	\$	114.6	\$	131.6	\$	(17.0)
Other gathering revenues		8.5		8.0		0.5		31.8		28.6		3.2
Gathering expense		(10.4)		(9.0)		(1.4)		(30.3)		(26.9)		(3.4)
Gathering margin	\$	37.3	\$	42.9	\$	(5.6)	\$	116.1	\$	133.3	\$	(17.2)
Operating Statistics												
Natural gas gathering volumes (in millions of MM	(Btu)											
For unaffiliated customers		53.7		60.2		(6.5)		162.6		184.4		(21.8)
For affiliated customers		58.3		69.1		(10.8)		172.7		202.5		(29.8)
Total Gas Gathering Volumes		112.0		129.3		(17.3)		335.3		386.9		(51.6)
Average gas gathering revenue (per MMBtu)	\$	0.35	\$	0.34	\$	0.01	\$	0.34	\$	0.34	\$	_

During the third quarter of 2013, gathering margin declined 13% when compared to the third quarter of 2012 due to a 13% decrease in gathering system throughput and increased gathering expense. Gathering system throughput volumes decreased primarily as a result of a 46% decline at QEP Field Services' Northwest Louisiana Hub primarily due to lower QEP Energy production resulting from the suspension of drilling in Haynesville starting in mid-2012. The increase in gathering expense was primarily the results of increased labor and benefits costs due to additional compensation costs from QEP's annual incentive program allocated to QEP Field Services.

During the first three quarters of 2013, gathering margin declined 13% when compared to the first three quarters of 2012 due to a 13% decrease in gathering system throughput volumes and increased gathering expense. Gathering system throughput volumes decreased due to the 42% decline at QEP Field Services' Northwest Louisiana Hub from decreased production in the Haynesville shale play. The increase in gathering expense was primarily the result of increased labor and benefits costs due to additional compensation costs from QEP's annual incentive program allocated to QEP Field Services.

Processing Margin

The following tables present a summary of QEP Field Services' gas processing financial and operating results:

	Three Months Ended September 30,							Nine Months Ended September 30,					
		2013		2012		Change		2013		2012		Change	
Processing Margin						(in n	nillio	ons)					
NGL sales	\$	23.6	\$	28.9	\$	(5.3)	\$	70.6	\$	112.7	\$	(42.1)	
Realized gains from commodity derivative contract	:t												
settlements		_		1.9		(1.9)		_		6.3		(6.3)	
Processing (fee-based) revenues		20.6		18.2		2.4		56.4		51.8		4.6	
Other processing fees		8.2		5.4		2.8		13.1		8.4		4.7	
Processing expense		(4.4)		(4.7)		0.3		(12.6)		(12.1)		(0.5)	
Processing plant fuel and shrink expense		(7.2)		(8.1)		0.9		(22.4)		(26.6)		4.2	
Natural gas, oil and NGL transportation and other handling costs		(3.4)		(6.9)		3.5		(11.6)		(27.7)		16.1	
0	_				_				Φ.		_		
Processing margin	\$	37.4	\$	34.7	\$	2.7	\$	93.5	\$	112.8	\$	(19.3)	
Keep-whole processing margin ⁽¹⁾	\$	13.0	\$	15.8	\$	(2.8)	\$	36.6	\$	64.7	\$	(28.1)	
Operating Statistics													
Natural gas processing volumes													
NGL sales (Mbbl)		482.5		831.7		(349.2)		1,532.4		2,893.7		(1,361.3)	
Average net realized NGL sales price (per bbl) ⁽²⁾	\$	48.86	\$	37.03	\$	11.83	\$	46.09	\$	41.11	\$	4.98	
Fee-based processing volumes (in millions of MM	Btu)												
For unaffiliated customers		26.6		26.0		0.6		75.4		83.7		(8.3)	
For affiliated customers		41.7		39.0		2.7		112.1		105.5		6.6	
Total fee-based processing volumes		68.3		65.0		3.3		187.5		189.2		(1.7)	
Average fee-based processing revenue (per MMBtu)	\$	0.30	\$	0.28	\$	0.02	\$	0.30	\$	0.27	\$	0.03	

⁽¹⁾ Keep-whole processing margin is calculated as NGL sales less processing plant fuel and shrink, natural gas, oil and NGL transportation and other handling costs.

Although a significant portion of QEP Field Services' gas processing services is performed for a volumetric-based fee, QEP Field Services also provides "keep-whole" processing services for certain customers. Under a keep-whole processing contract, QEP Field Services retains and sells NGL extracted at its processing plants and keeps the customer "whole" by delivering a Btu-equivalent amount of natural gas to the customer. Keep-whole processing exposes the Company to the "frac" spread. The frac spread is the difference between the market value of NGL extracted at the processing plant and the market value of an energy-equivalent volume of natural gas.

QEP Field Services' keep-whole processing margin decreased 18% during the third quarter of 2013 compared to the third quarter of 2012 due to a 42% decrease in NGL sales volumes. The decrease in NGL sales volumes is the result of processing plants running in ethane rejection mode in 2013 due to negative ethane processing margins. Partially offsetting this decline was an increase in the average net realized NGL sales price. Including the impact of gains on derivative contract settlements, average NGL realized prices increased 32% in the third quarter of 2013, primarily the result of rejection of ethane, which is normally the lower-value component of the composite NGL barrel. In addition, keep-whole margin was positively impacted in the third quarter of 2013 by decreased natural gas, oil, and NGL transportation and fractionation costs. Transportation costs were lower in the third quarter of 2013 due to the reduction in ethane volumes.

Fee-based processing revenues increased during the third quarter of 2013 compared to the third quarter of 2012 due to a 7% increase in average fee-based processing rate and a 5% increase in fee-based processing volumes. During the third quarter of 2013, the increase in fee-based processing volumes was the result of additional gas processed at the Blacks Fork and Iron Horse

⁽²⁾ Average net realized NGL sales price per gallon is calculated as NGL sales including realized gains from commodity derivative contracts settlements divided by NGL sales volumes.

plants. Approximately 85% and 80% of QEP Field Services' net operating revenue was derived from fee-based gathering and processing agreements in the third quarter of 2013 and 2012, respectively.

QEP Field Services' keep-whole processing margin decreased 43% during the first three quarters of 2013 compared to the first three quarters of 2012 due to a 47% decrease in NGL sales volumes as a result of its processing plants running in ethane rejection mode in 2013 due to negative ethane processing margins. Partially offsetting this decline was an increase in the average net realized NGL sales price. Including the impact of gains on derivative contract settlements, average NGL realized prices increased 12% in the first three quarters of 2013, due to the higher value NGL barrel. Keep-whole margins were positively impacted in the first three quarters of 2013 from decreased natural gas, oil, and NGL transportation and fractionation costs due to the reduction in ethane volumes.

Fee-based processing revenues increased during the first three quarters of 2013 compared to the first three quarters of 2012 due to an 11% increase in average fee-based processing rate, offset by a 1% decrease in fee-based processing volumes. During the first three quarters of 2013, the decrease in fee-based processing volumes was the result of lower unaffiliated volumes from a third-party shipper disruption caused by a fire at one of the shipper's compression stations. Other processing fees increased 56% in the first three quarters of 2013 due to increased deficiency payments from customers who did not meet their contractual annual minimum throughput commitments for gathering or processing volumes. Approximately 83% and 76% of QEP Field Services' net operating revenue was derived from fee-based gathering and processing agreements in the first three quarters of 2013 and 2012, respectively.

QEP MARKETING AND RESOURCES

The following table provides a summary of QEP Marketing and Resources' financial and operating results:

	Three Months Ended September 30,							Nine Months Ended September 30,					
	2013			2012		Change		2013		2012		Change	
						(in mil	llioı	ns)					
Revenues													
Purchased gas, oil and NGL sales	\$ 439	.1	\$	234.9	\$	204.2	\$	1,148.5	\$	674.7	\$	473.8	
Other	1	.1		1.4		(0.3)		4.7		5.3		(0.6)	
Total Revenues	440	.2		236.3		203.9		1,153.2		680.0		473.2	
Operating expenses													
Purchased gas, oil and NGL expense	438	.2		236.7		201.5		1,150.6		681.7		468.9	
Gathering, processing and other	0	.4		0.1		0.3		1.2		8.0		0.4	
General and administrative	1	.1		1.2		(0.1)		3.3		2.6		0.7	
Production and property taxes	0	.1		0.1		_		0.2		0.2		_	
Depreciation, depletion and amortization	0	.3		0.2		0.1		0.7		0.5		0.2	
Total Operating Expenses	440	.1		238.3		201.8		1,156.0		685.8		470.2	
Operating Income (Loss)	0	.1		(2.0)		2.1		(2.8)		(5.8)		3.0	
Realized (losses) gains on derivative instruments	(0	.9)		(1.0)		0.1		(1.2)		3.1		(4.3)	
Unrealized (losses) gains on derivative instruments	(1	.3)		(3.7)		2.4		3.2		(7.1)		10.3	
Interest and other income	52	.3		28.4		23.9		158.2		81.1		77.1	
Loss on extinguishment of debt		_		_		_		_		(0.6)		0.6	
Interest expense	(41	.0)		(30.8)		(10.2)		(124.3)		(83.4)		(40.9)	
Income (Loss) before Income Taxes	9	.2		(9.1)		18.3		33.1		(12.7)		45.8	
Income tax (provision) benefit	(3	.5)		3.5	_	(7.0)		(13.6)		5.1		(18.7)	
Net Income (Loss) Attributable to QEP	\$ 5	.7	\$	(5.6)	\$	11.3	\$	19.5	\$	(7.6)	\$	27.1	

Resale Margin

The following table is a summary of QEP Marketing's financial results from resale activities:

	Three Months Ended September 30,							Nine Months Ended September 30,				
		2013		2012	12 Change			2013	2012		Change	
Resale Margin						(in milli	ons)					
Purchased gas, oil and NGL sales	\$	439.1	\$	234.9	\$	204.2	\$	1,148.5	\$	674.7	\$	473.8
Purchased gas, oil and NGL expense		(438.2)		(236.7)		(201.5)		(1,150.6)		(681.7)		(468.9)
Realized gain (loss) on derivative instruments		(0.9)		(1.0)		0.1		(1.2)		3.1		(4.3)
Resale margin loss	\$		\$	(2.8)	\$	2.8	\$	(3.3)	\$	(3.9)	\$	0.6

QEP Marketing's loss on resale margin was primarily the result of the fulfillment of firm transportation contract commitments, resulting in firm transportation expenses. Purchased gas, oil and NGL sales increased by \$204.2 million, or 87%, during the third quarter of 2013 compared to the third quarter of 2012 due to a \$41.9 million increase in resale gas sales and a \$162.3 million increase in resale oil and NGL sales. Resale natural gas sales increased due to a 31% increase in the resale price and a 4% increase in resale gas volumes. Resale oil and NGL sales increased due to a 27% increase in resale price and a 84% increase in resale volumes.

Purchased gas, oil and NGL expense, which includes transportation expense, increased 85% in the third quarter of 2013 compared to the third quarter of 2012 due to a \$40.6 million increase in resale gas purchases and a \$161.9 million increase in resale oil and NGL purchases. Resale natural gas purchased due to a 35% increase in the resale price whereas resale purchase volumes were consistent period to period. Resale oil and NGL sales increased due to an 84% increase in resale purchase volumes and a 27% increase in resale purchase price.

Purchased gas, oil and NGL sales increased by \$473.8 million, or 70%, during the first three quarters of 2013 compared to the first three quarters of 2012 due to a \$152.8 million increase in resale gas sales and a \$321.0 million increase in resale oil and NGL sales. Resale natural gas sales increased due to a 46% increase in the resale price while resale gas volumes increased 1% period to period. Resale oil and NGL sales increased due to a 69% increase in resale volumes and a 13% increase in resale price.

Purchased gas, oil and NGL expense, which includes transportation expense, increased 69% in the first three quarters of 2013 compared to the first three quarters of 2012 due to a \$148.6 million increase in resale gas purchases and a \$320.6 million increase in resale oil and NGL purchases. Resale natural gas purchases increased due to a 50% increase in the resale price, partially offset by a 1% decrease in resale purchase volumes. Resale oil and NGL sales increased due to a 69% increase in resale purchase volumes and a 13% increase in resale purchase price.

OTHER CONSOLIDATED EXPENSES AND INCOME

General and administrative expense. During the third quarter of 2013, general and administrative (G&A) expense increased \$7.6 million, or 18% compared to the third quarter of 2012. The increase in G&A in 2013 was primarily due to a \$5.4 million increase in labor costs due to the increased number of employees and the Company's annual compensation program, and a \$4.7 million increase in professional and outside services mainly related to the ongoing implementation of a new Enterprise Resource Planning system and feasibility studies. These increases were partially offset by a \$2.2 million decrease in the mark-to-market value of the deferred compensation wrap plan and Cash Incentive Plan (CIP).

During the first three quarters of 2013, G&A expense increased \$21.7 million, or 19%, compared to the first three quarters of 2012. The increase in G&A in 2013 was primarily due to a \$13.9 million increase in labor costs due to the increased number of employees and the Company's annual compensation program, and a \$15.6 million increase in professional and outside services mainly related to the ongoing implementation of a new Enterprise Resource Planning system, QEP Midstream start up costs, feasibility studies, software maintenance costs and other contracted or professional services. These increases were partially offset by a \$4.5 million decrease in restructuring costs and a \$4.4 million decrease in the mark-to-market value of the deferred compensation wrap plan and CIP.

Net gain from asset sales. In the first nine months of 2013, QEP Energy sold its interest in several non-core oil and gas properties for total cash proceeds of \$209.1 million and a pre-tax gain on sale of \$114.9 million. Both the cash proceeds and gain on sale are subject to post-closing adjustments.

Realized and unrealized (losses) gain on derivative contracts. Gains and losses on derivative instruments are comprised of both realized and unrealized gains and losses on QEP's commodity derivative contracts and interest rate swaps, which are marked-to-market each quarter. During the third quarter of 2013, losses on commodity derivative instruments were \$25.8 million, of which \$26.9 million was realized gains and \$52.7 million was unrealized losses. During the third quarter of 2012, gains on commodity derivative instruments were \$39.0 million, of which \$93.8 million was realized gains and \$54.8 million was unrealized losses. Additionally, during the third quarter of 2013, losses from interest rate swaps were \$2.0 million, of which \$0.6 million was realized and \$1.4 million was unrealized.

During the first three quarters of 2013, gains on commodity derivative instruments were \$50.0 million, of which \$109.0 million was realized gains and \$59.0 million was unrealized losses. During the first three quarters of 2012, gains on commodity derivative instruments were \$341.9 million, of which \$302.5 million was realized and \$39.4 million was unrealized. Additionally, during the first three quarters of 2013, gains from interest rate swaps were \$1.6 million, of which \$1.9 million was realized losses and \$3.5 million was unrealized gains.

Interest expense. Interest expense increased \$11.7 million, or 39%, during the third quarter of 2013 compared to the third quarter of 2012. The increase was attributable to average debt levels that were approximately \$619.8 million, or 25% higher than average debt levels in the third quarter of 2012. The increase in average debt levels is primarily related to the issuance of QEP's \$650.0 million 2023 senior notes in the third quarter of 2012, which was used to fund the 2012 Acquisition, partially offset by a lower balance under our revolving credit facility for the period ended September 30, 2013 after repayment of the revolving credit facility with the net proceeds provided by the Offering.

Interest expense increased \$39.6 million, or 48%, during the first three quarters of 2013 compared to the first three quarters of 2012. The increase was attributable to average debt levels that were approximately \$614.0 million, or 25%, higher than average debt levels in the first three quarters of 2012. The increase in average debt levels is primarily related to the issuance of QEP's \$650.0 million 2023 senior notes in the third quarter of 2012, which was used to fund the 2012 Acquisition, partially offset by a lower balance under our revolving credit facility for the period ended September 30, 2013 after repayment of the revolving credit facility with the net proceeds provided by the Offering.

Income taxes. Income tax provision was \$22.4 million during the third quarter of 2013 compared to an income tax benefit of \$2.3 million during the third quarter of 2012. The increased provision was primarily the result of higher income before income taxes for the third quarter of 2013 compared to a loss before income taxes for the third quarter of 2012.

Income tax provision increased \$38.5 million, or 45%, during the first three quarters of 2013 compared to the first three quarters of 2012. The increase was primarily the result of higher income before income taxes and a higher combined effective federal and state income tax rate of 36.5% for the first three quarters of 2013 compared to 36.0% for the first three quarters of 2012.

LIQUIDITY AND CAPITAL RESOURCES

QEP seeks to fund its development projects by employing a capital structure and financing strategy to provide sufficient liquidity to withstand commodity price swings. QEP maintains a commodity price derivative strategy to reduce commodity price volatility and to provide certainty to cash flows. QEP funds its operations, capital expenditures and working capital requirements with cash flow from its operating activities and borrowings under its credit facilities. Periodically, QEP accesses debt and capital markets and sells properties to provide additional liquidity. The Company believes cash flow from operations, cash-on-hand and availability under its credit facility will be sufficient to fund the Company's planned capital expenditures and operating expenses during the next 12 months and the foreseeable future. To the extent actual operating results differ from the Company's estimates, QEP's liquidity could be adversely affected.

The following table provides QEP's available liquidity and debt to equity ratio compared to the previous period:

	Sep	September 30, 2013		cember 31, 2012		
		(in million	s, except %	except %)		
Cash and cash equivalents	\$	123.0	\$	_		
Amount available under the QEP credit facility (1)		1,131.3		805.9		
Total liquidity	\$	1,254.3	\$	805.9		
Total debt	\$	2,882.3	\$	3,206.9		
Total common shareholders' equity		3,426.1		3,266.0		
Ratio of debt to total capital (2)		46%		50%		

⁽¹⁾ See discussion of revolving credit facility below. Availability under the QEP credit facility is reduced by outstanding letters of credit of \$3.7 million as of September 30, 2013, and \$4.1 million as of December 31, 2012 and does not include \$500.0 million available under QEP Midstream's credit facility.

QEP's Credit Facility

QEP's revolving credit facility, which matures in August 2016, provides for loan commitments of \$1.5 billion from a syndicate of financial institutions. The credit facility provides for borrowings at short-term interest rates and contains customary covenants and restrictions. The credit facility also contains provisions which would allow for the amount of the facility to be increased to \$2.0 billion and the maturity to be extended for two additional one-year periods. QEP's weighted-average interest rate on borrowings from its credit facility was 2.56% during the first three quarters of 2013. At September 30, 2013, QEP was in compliance with the debt covenants under the credit agreement. At October 31, 2013, QEP had \$390.5 million of borrowings and \$3.8 million of letters of credit outstanding under its credit facility.

QEP Midstream's Credit Facility

On August 14, 2013, QEP Midstream entered into a \$500.0 million senior secured revolving credit facility with a group of financial institutions, which matures on August 14, 2018. The credit facility contains an accordion provision that allows for the amount of the facility to be increased to \$750.0 million with the agreement of the lenders. QEP Midstream's credit facility is available for QEP Midstream's working capital, capital expenditures, permitted acquisitions and general corporate purposes, including distributions. In addition, QEP Midstream's credit facility includes a sublimit of up to \$50.0 million for letters of credit and a sublimit of up to \$25.0 million for swing line loans. Substantially all of QEP Midstream's assets, excluding equity in and assets of certain joint ventures and unrestricted subsidiaries, are pledged as collateral under the credit facility. In addition, the credit agreement contains restrictions and events of default customary for agreements of this nature.

There have been no borrowings under QEP Midstream's credit facility, and at September 30, 2013, QEP Midstream was in compliance with the covenants under the OEP Midstream credit agreement.

QEP is not a borrower or guarantor of QEP Midstream's credit facility. In addition, QEP is not subject to any of the restrictions or covenants contained in QEP Midstream's credit agreement. Outstanding indebtedness under QEP Midstream's credit facility is not included in the definition of indebtedness under QEP's credit agreement.

Term Loan

QEP's \$300.0 million term loan facility provides for borrowings at short-term interest rates and contains covenants, restrictions and interest rates that are substantially the same as the Company's revolving credit facility. The term loan matures in April 2017, and the maturity date may be extended one year with the agreement of the lenders. During the first three quarters of 2013, QEP's weighted-average interest rate on the term loan was 2.23%. In conjunction with the term loan, QEP entered into interest rate swap contracts with a combined notional principal amount of \$300.0 million which will mature in March 2017. Under the swap contracts, QEP pays 1.07% for the life of the swaps and receives one-month LIBOR. The interest rate at September 30, 2013, under the term loan is one-month LIBOR, plus 2.00% (the Applicable Margin) which, when combined with the fixed interest rate swaps, results in an effective rate of 3.07% for borrowings under the term loan. To the extent that the Applicable Margin under the term loan changes, the effective fixed rate paid for borrowings under the term loan will change.

⁽²⁾ Defined as total debt divided by the sum of total debt plus common shareholders' equity.

Senior Notes

The Company's senior notes outstanding as of September 30, 2013, totaled \$2,221.8 million principal amount and are comprised of six issuances as follows:

- \$176.8 million 6.05% Senior Notes due September 2016
- \$134.0 million 6.80% Senior Notes due April 2018
- \$136.0 million 6.80% Senior Notes due March 2020
- \$625.0 million 6.875% Senior Notes due March 2021
- \$500.0 million 5.375% Senior Notes due October 2022
- \$650.0 million 5.25% Senior Notes due May 2023

Cash Flow from Operating Activities

Cash flows from operations are primarily affected by natural gas, oil and NGL production volumes and commodity prices (including the effects of settlements of the Company's derivative contracts) and by changes in working capital. QEP enters into commodity derivative transactions covering a substantial, but varying, portion of its anticipated future gas, oil and NGL production for the next 18 months.

Net cash provided by operating activities during the first three quarters of 2013 increased \$6.9 million compared to the first three quarters of 2012 due to increased net income and noncash net income adjustments partially offset by a decrease in the use of cash from operating assets and liabilities. Changes in operating assets and liabilities used \$8.1 million of cash in the first three quarters of 2013, mainly due to a decrease in accounts payable and accrued expenses primarily due to the \$115.0 million Chieftain settlement payment in the first quarter of 2013 and an increase in deferred income taxes. Changes in operating assets and liabilities provided \$54.6 million of cash in the first three quarters of 2012 primarily due to decreases in accounts receivable and increases in accounts payable and accrued expenses. Net cash provided from operating activities is presented below:

	Nine Months Ended September 30,				
	2013		2012		Change
		(ir	n millions)		
Net income	\$ 217.2	\$	154.1	\$	63.1
Noncash adjustments to net income	769.8		763.3		6.5
Changes in operating assets and liabilities	(8.1)		54.6		(62.7)
Net cash provided from operating activities	\$ 978.9	\$	972.0	\$	6.9

Cash Flow from Investing Activities

In the first three quarters of 2013, net cash used in investing activities was \$920.6 million, compared to \$2,435.7 million in the first three quarters of 2012. The decrease in cash flows used in investing activities was primarily the result of the \$1.4 billion of capital expenditures incurred in the third quarter of 2012 related to the 2012 Acquisition and the 2013 proceeds of \$208.3 million received from the disposition of assets, mainly attributable to QEP Energy's second and third quarters 2013 oil and gas property divestitures. The capital expenditures for the 2012 Acquisition and proceeds from asset dispositions during the first three quarters of 2013 were partially offset by an increase in cash capital expenditures during the first three quarters of 2013 when compared to the first three quarters of 2012. A comparison of capital expenditures for the first three quarters of 2013 and 2012 and a forecast for calendar year 2013 are presented in the table below:

	Nine Months Ended September 30,						7	Forecast Twelve Months Ended ⁽¹⁾		Prior Forecast Twelve Months Ended ⁽²⁾	
		2013		2012		Change		December 31, 2013		December 31, 2013	
						(in millions)					
QEP Energy	\$	1,108.9	\$	2,390.9	\$	(1,282.0)	\$	1,475.0	\$	1,530.0	
QEP Field Services		55.5		141.2		(85.7)		80.0		120.0	
QEP Marketing		0.8		0.7		0.1		1.0		1.0	
QEP		17.4		5.7		11.7		24.0		24.0	
Total accrued capital expenditures		1,182.6		2,538.5		(1,355.9)		1,580.0		1,675.0	
Change in accruals		(53.7)		(97.5)		43.8		_		_	
Total cash capital expenditures	\$	1,128.9	\$	2,441.0	\$	(1,312.1)	\$	1,580.0	\$	1,675.0	

Current

During the first three quarters of 2013, capital expenditures on a cash basis decreased 54% to \$1,128.9 million, compared to \$2,441.0 million during the first three quarters 2012. The decrease of \$1,312.1 million in cash capital expenditures during the first three quarters of 2013 was primarily the result of QEP Energy's increased capital expenditures related to the 2012 Acquisition.

In the first three quarters of 2013, QEP Energy's capital investment, on an accrual basis, decreased \$1,282.0 million over the first three quarters of 2012 to a total of \$1,108.9 million. QEP Energy's capital expenditures includes \$35.3 million related to property acquisitions in the Williston Basin and \$4.0 million to post-closing adjustments for the 2012 Acquisition incurred in the first three quarters of 2013 compared to \$1,400.3 million of property acquisitions in first three quarters of 2012 related to the 2012 Acquisition. In addition, capital expenditures increased \$301.0 million in the Williston Basin due to additional drilling activity and operations in the area partially offset by a \$59.2 decrease million in the Haynesville/Cotton Valley area due to the suspended drilling program, a \$102.9 million decrease in Pinedale due to the reduction in the number of drilling rigs from seven to four and a \$58.3 million decrease in Midcontinent capital expenditures due to reduced drilling activity.

In the first three quarters of 2013 compared to the first three quarters of 2012, QEP Field Services capital investment decreased \$85.7 million, on an accrual basis, due to the higher capital expenditures during the first three quarters of 2012 for the new 150 MMcfd cryogenic gas processing plant in the Uinta Basin (Iron Horse II), which was completed during the first quarter of 2013, and the expansion of the fractionation facility at the Blacks Fork processing complex. Currently, there are no processing plants under significant construction at QEP Field Services.

At September 30, 2013, forecasted capital investments for 2013 is expected to be \$1,580.0 million, comprised of \$1,475.0 million at QEP Energy, \$80.0 million at QEP Field Services, and \$25.0 million for QEP Marketing and Resources. For the remainder of 2013, QEP intends to fund capital expenditures with cash flow from operating activities, and, if needed, borrowings under its revolving credit facility. As a result of the continued low natural gas prices, QEP plans minimal capital expenditures for the Haynesville Shale and other dry-gas development areas and to increase capital expenditures for higher return projects, including Pinedale, and oil-directed horizontal drilling in the Williston Basin and Midcontinent during the remainder of 2013. QEP Energy has allocated approximately 98% of its forecasted 2013 drilling and completion capital expenditure budget to crude oil and liquids-rich natural gas plays. QEP plans to invest a total of approximately \$80.0 million in capital expenditures during 2013 to maintain and grow its midstream business including the completion of the Iron Horse II cryogenic processing plant in the Uinta Basin in the first quarter of 2013, the expansion of its gathering system in the Uinta Basin and the completion of a 10,000 Bbl/d expansion of the NGL fractionation facility located at the Blacks Fork processing complex in the second quarter of 2013. The remaining QEP Field Services' capital expenditures will be used on compressor projects, new well connections and gathering line expansion. QEP plans to invest approximately \$24.0 million in capital expenditures related to corporate activities, primarily the implementation of a new Enterprise Resource Planning system. The aggregate levels of capital expenditures for 2013 and the allocation of those expenditures are dependent on a variety of factors, including drilling results, natural gas, oil and NGL prices, industry conditions, the extent to which properties or working interests are acquired, the availability of capital resources to fund t

⁽¹⁾ Represents the mid-point of the most recent guidance.

⁽²⁾ Forecast as reported in the 2013 Quarterly Report on Form 10-Q, filed on July 31, 2013.

assessments as to where QEP's capital can be most profitably deployed. Accordingly, the actual levels of capital expenditures and the allocation of those expenditures may vary materially from QEP's estimates.

Cash Flow from Financing Activities

In the first three quarters of 2013, net cash proceeds from financing activities were \$64.7 million compared to \$1,463.7 million in the first three quarters of 2012. During the first three quarters of 2013, QEP had borrowings from the credit facility of \$1,075.0 million and repayments on the credit facility of \$1,400.0 million partially funded by the net proceeds provided from the Offering partially offset by increases to the checks outstanding in excess of cash balances of \$38.1 million during the first three quarters of 2013 and no debt issuances in the first three quarters of 2013. In the nine months ended September 30, 2013, QEP paid \$10.8 million of regular quarterly dividends.

At September 30, 2013, long-term debt consisted of \$365.0 million outstanding under the credit facility, \$300.0 million under the term loan and \$2,221.8 million in senior notes (including \$4.5 million of net original issue discount). The \$325.0 million decrease in borrowings under the credit facility during the first nine months of 2013 was primarily the results of repayment of the credit facility funded from the net proceeds provided by the Offering in the third quarter of 2013.

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

QEP's primary market risk exposures arise from changes in the market price for natural gas, oil and NGL, and to volatility in interest rates. These risks can affect revenues and cash flows from operating, investing and financing activities. Commodity prices have historically been volatile and are subject to wide fluctuations in response to relatively minor changes in supply and demand. If commodity prices fluctuate significantly, revenues and cash flow may significantly decrease or increase. QEP Energy and QEP Marketing also have long-term contracts for pipeline capacity and are obligated to pay for transportation services with no guarantee that QEP will be able to fully utilize the contractual capacity of these transportation commitments. In addition, a non-cash write-down of the Company's oil and gas properties may be required if future oil and natural gas commodity prices experience a sustained, significant decline. Furthermore, the Company's credit facility and term loan agreement have floating interest rates which expose QEP to interest rate risk. To manage the Company's exposure to these risks, QEP enters into commodity derivative contracts in the form of fixed-price swaps to manage commodity price risk and periodically interest rate swaps to manage interest rate risk.

Commodity Price Risk Management

QEP's subsidiaries use commodity price derivative instruments in the normal course of business to reduce the risk of adverse commodity price movements. However, these same arrangements typically limit future gains from favorable price movements. The types of commodity derivative instruments currently utilized by the Company are fixed-price swaps. The volume of commodity derivative instruments utilized by the Company may vary from year-to-year. The derivative instruments currently utilized by the Company do not have margin requirements or collateral provisions that would require payments prior to the scheduled cash settlement dates. As of September 30, 2013, QEP held commodity price derivative contracts totaling 98.9 million MMBtu of natural gas and 11.0 million barrels of oil. At December 31, 2012, the QEP derivative contracts covered 139.4 million MMBtu of natural gas and 6.9 million barrels of oil.

The following table presents 2013 derivative positions as of October 31, 2013:

QEP Energy Commodity Derivative Positions

Year	Type of Contract	Index	Total Volumes	Wei	Weighted-average price per unit	
			(in millions)			
Natural gas			(MMBtu)			
2013	Swap	IFNPCR	18.4	\$	5.49	
2013	Swap	NYMEX	14.7	\$	3.81	
2014	Swap	IFNPCR	32.9	\$	4.00	
2014	Swap	NYMEX	25.6	\$	4.19	
Crude oil			(Bbls)			
2013	Swap	NYMEX WTI	2.1	\$	98.27	
2013	Swap	BRENT ICE	0.1	\$	107.80	
2014	Swap	NYMEX WTI	8.8	\$	93.63	

QEP Marketing Commodity Derivative Positions

Year	Type of Contract	Index	Total Volumes	Wei	ghted-average price per MMBtu
			(in millions)		
Natural gas sales			(MMBtu)		
2013	Swap	IFNPCR	1.6	\$	3.91
2014	Swap	IFNPCR	4.7	\$	3.77
Natural gas purchases			(MMBtu)		
2013	Swap	IFNPCR	2.6	\$	3.56
2014	Swap	IFNPCR	0.2	\$	3.82

Changes in the fair value of derivative contracts from December 31, 2012 to September 30, 2013, are presented below:

		mmodity tive contracts
	(in	millions)
Net fair value of gas and oil derivative contracts outstanding at December 31, 2012	\$	192.8
Contracts settled		(109.0)
Change in gas and oil prices on futures markets		(46.9)
Contracts added		0.7
Net fair value of gas and oil derivative contracts outstanding at September 30, 2013	\$	37.6

The following table shows sensitivity of fair value of gas and oil derivative contracts to changes in the market price of gas, oil and NGL and basis differentials:

	September 30, 2013	
	(in	millions)
Net fair value - asset (liability)	\$	37.6
Fair value if market prices of gas and oil and basis differentials decline by 10%		177.0
Fair value if market prices of gas and oil and basis differentials increase by 10%		(101.9)

Utilizing the actual derivative contractual volumes, a 10% increase in underlying commodity prices would reduce the fair value of these instruments by \$139.5 million, while a 10% decrease in underlying commodity prices would increase the fair value of these instruments by \$139.4 million as of September 30, 2013. However, a gain or loss eventually would be substantially offset by the actual sales value of the physical production covered by the derivative instruments. For additional information regarding the Company's commodity derivative transactions, see Note 8 – Derivative Contracts under Part I, Item 1 of this Form 10-Q.

Interest Rate Risk Management

The Company's ability to borrow and the rates offered by lenders can be adversely affected by illiquid credit markets as described in the risk factors in Part I, Item 1A of the Company's Annual Report on Form 10-K for the year ended December 31, 2012. The Company's credit facility has a floating interest rate which exposes QEP to interest rate risk. At September 30, 2013, the Company had \$365.0 million outstanding under its revolving credit facility. If interest rates were to increase or decrease 10% over the nine months ended September 30, 2013, at our average level of borrowing for those same periods, our interest expense would increase or decrease by \$1.3 million for both the three and nine months ended September 30, 2013 and 2012, or approximately 1% in each period.

The Company's term loan has a floating interest rate which also exposes QEP to interest rate risk. At September 30, 2013, the Company had \$300.0 million outstanding under the term loan. During the second quarter of 2012, QEP entered into interest rate swap contracts, with an aggregate notional amount of \$300.0 million, to minimize the interest rate volatility risk associated with its \$300.0 million term loan. QEP pays a fixed interest rate and receives a floating interest rate indexed to the one-month LIBOR. At September 30, 2013, the fair value of the interest rate swaps was a derivative liability balance of \$2.5 million. A 50 basis point decrease would cause the fair value of the interest rate swaps to decrease by \$4.6 million while a 50 basis point increase would cause the fair value of the interest rate swaps to increase by \$5.0 million.

The remaining \$2,221.8 million of the Company's debt are Senior Notes with fixed interest rates and therefore are not affected by interest rate movements. For additional information regarding the Company's debt instruments, see Note 10 – Debt under Part I, Item 1 of this Form 10-Q.

Forward-Looking Statements

This quarterly report contains information that includes or is based upon "forward-looking statements" within the meaning of Section 27A of the Securities Act of 1933, as amended, and Section 21E of the Securities Exchange Act of 1934, as amended. Forward-looking statements give expectations or forecasts of future events. You can identify these statements by the fact that they do not relate strictly to historical or current facts. They use words such as "anticipate," "estimate," "expect," "project," "intend," "plan," "believe," and other words and terms of similar meaning in connection with a discussion of future operating or financial performance. Forward-looking statements include statements relating to, among other things:

- QEP's growth strategies;
- natural gas, oil and NGL prices and factors affecting the volatility of such prices;
- plans to drill or participate in wells and to defer completion of wells;
- results from planned drilling operations and production operations;
- QEP's low cash operating costs and ability to control costs;
- ability to pursue acquisition opportunities;
- proforma results for acquired properties;
- expected restructuring costs;
- the amount and timing of the reclassification of the fixed-value related to de-designated hedges;
- recognition of compensation costs related to equity compensation grants;
- impact of pension legislation;
- expected gain on sale of assets;
- · amount and allocation of forecasted capital expenditures and plans for funding capital expenditures and operating expenses;
- · estimated accrual for loss contingencies and other items;
- impact of lower or higher commodity prices and interest rates;
- effect of recession;
- plans to enter into derivative contracts for a portion of forecasted production;
- future expenses and operating costs;
- operation of processing plants at assumed capacities;
- the amount and timing of the settlement of derivative contracts;
- incurrence of unrealized derivative gains and losses;
- impact of nonperformance by trade creditors or joint venture partners;

- the outcome of contingencies such as legal proceedings;
- expected contributions to the Company's pension plans;
- impact of recently issued accounting pronouncements;
- the significance of Adjusted EBITDA as a measure of cash flow and liquidity;
- payment of dividends;
- potential for future asset impairments; and
- estimated future purchase accounting adjustments.

Any or all forward-looking statements may turn out to be incorrect. They can be affected by inaccurate assumptions or by known or unknown risks and uncertainties. Many such factors will be important in determining actual future results. These statements are based on current expectations and the current economic environment. They involve a number of risks and uncertainties that are difficult to predict. These statements are not guarantees of future performance. Actual results could differ materially from those expressed or implied in the forward-looking statements. Factors that could cause actual results to differ materially include, but are not limited to the following:

- the risk factors discussed in Part I, Item 1A of the Company's Annual Report on Form 10-K for the year ended December 31, 2012;
- changes in natural gas, oil and NGL prices;
- general economic conditions, including the performance of financial markets and interest rates;
- drilling results;
- shortages of oilfield equipment, services and personnel;
- lack of available pipeline capacity;
- QEP's inability to successfully integrate acquired assets or dispose of non-core assets;
- · the outcome of contingencies such as legal proceedings;
- permitting delays;
- operating risks such as unexpected drilling conditions;
- weather conditions;
- changes in maintenance and construction costs, including possible inflationary pressures;
- the availability and cost of debt and equity financing;
- changes in laws or regulations;
- · legislation regarding climate change and other initiatives related to drilling and completion techniques, including hydraulic fracturing;
- derivative activities;
- substantial liabilities from legal proceedings and environmental claims;
- · failure of internal controls and procedures;
- elimination of federal income tax deductions for oil and gas exploration and development costs;
- future opportunities that QEP's board of directors may determine present greater potential value to stockholders than planned divestiture of assets;
- regulatory approvals and compliance with contractual obligations;
- actions, or inaction, by federal, state, local or tribal governments;
- fluctuations in processing margins;
- unexpected changes in costs for constructing, modifying or operating midstream facilities;
- lack of, or disruptions in, adequate and reliable transportation for QEP's products; and
- other factors, most of which are beyond the Company's control.

QEP undertakes no obligation to publicly correct or update the forward-looking statements in this quarterly report, in other documents, or on the website to reflect future events or circumstances. All such statements are expressly qualified by this cautionary statement.

ITEM 4. CONTROLS AND PROCEDURES

Evaluation of Disclosure Controls and Procedures

The Company's Chief Executive Officer and Chief Financial Officer have evaluated the effectiveness of the Company's disclosure controls and procedures (as such term is defined in Rules 13a-15(e) under the Securities Exchange Act of 1934, as amended, the Exchange Act) as of September 30, 2013. Based on such evaluation, such officers have concluded that, as of September 30, 2013, the Company's disclosure controls and procedures are designed and effective to ensure that information required to be disclosed in the Company's reports filed or submitted under the Exchange Act is recorded, processed, summarized and reported within the time periods specified by the SEC's rules and forms and that information is accumulated and communicated to the Company's management, including its principal executive officer and principal financial officer, or persons performing similar functions, as appropriate, to allow timely decisions regarding required disclosure.

In designing and evaluating the Company's disclosure controls and procedures, management recognizes that any controls and procedures, no matter how well designed and operated, can provide only reasonable, and not absolute, assurance that the objectives of the control system will be met. In addition, the design of any control system is based in part upon certain assumptions about the likelihood of future events and the application of judgment in evaluating the cost-benefit relationship of possible controls and procedures. Because of these and other inherent limitations of control systems, there is only reasonable assurance that the Company's controls will succeed in achieving their goals under all potential future conditions.

Changes in Internal Controls.

There were no changes in the Company's internal controls over financial reporting during the quarter ended September 30, 2013, that have materially affected, or that are reasonably likely to materially affect, the Company's internal control over financial reporting.

PART II. OTHER INFORMATION

ITEM 1. LEGAL PROCEEDINGS

Information regarding legal proceedings is set forth in Note 11 - Contingencies to the Company's consolidated financial statements included in Item 1 of Part I of this Quarterly Report on Form 10-Q and is incorporated herein by reference.

ITEM 1A. RISK FACTORS

Risk factors relating to the Company are set forth in its Annual Report on Form 10-K for the year ended December 31, 2012. Below are material changes to such risk factors have occurred during the three months ended September 30, 2013.

One of our subsidiaries acts as the general partner of a publicly traded master limited partnership, QEP Midstream, which may involve a greater exposure to legal liability than our historic business operations.

One of our subsidiaries acts as the general partner of QEP Midstream, a publicly traded master limited partnership. Our control of the general partner of QEP Midstream may increase the possibility of claims of breach of fiduciary duties including claims of conflicts of interest related to QEP Midstream. Any liability resulting from such claims could have a material adverse effect on our future business, financial condition, results of operations and cash flows.

Multi-well pad drilling may result in volatility in our operating results.

QEP utilized multi-well pad drilling where practical. Because wells drilled on a pad are not brought into production until all wells on the pad are drilled and completed and the drilling rig is moved from the location, multi-well pad drilling delays the commencement of production, which may cause volatility in QEP's quarterly operating results.

ITEM 2. UNREGISTERED SALES OF EQUITY SECURITIES AND USE OF PROCEEDS

None.

ITEM 3. DEFAULTS UPON SENIOR SECURITIES

None.

ITEM 4. MINE SAFETY DISCLOSURES

None.

ITEM 5. OTHER INFORMATION

Indemnification Agreements

On November 4, 2013, the Company entered into an indemnification agreement with each of its directors and officers (the "Indemnification Agreement"). The Indemnification Agreement supplements the indemnification rights provided under the Company's certificate of incorporation, bylaws, and applicable law. The Indemnification Agreement requires the Company to indemnify the officer or director to the fullest extent permitted under Delaware law against liability that may arise by reason of his or her service to the Company and to advance expenses incurred as a result of any proceeding against him or her as to which he or she could be indemnified. The Indemnification Agreement also provides procedures for the determination of an indemnitee's right to receive indemnification and the advancement of expenses. The Company anticipates that it will enter into substantially similar Indemnification Agreements with any new directors and officers. The form of indemnification agreement entered into by each officer and director of the Company's subsidiaries is substantially similar to the Indemnification Agreement.

The foregoing is a summary of the material terms of the Indemnification Agreement. A copy of the form of Indemnification Agreement is filed as Exhibit 10.8 to this Quarterly Report on Form 10-Q and is incorporated herein by reference.

ITEM 6. EXHIBITS

The following exhibits are being filed as part of this report:

Exhibit No.	Exhibits
10.1	Contribution, Conveyance and Assumption Agreement, dated as of August 14, 2013, by and among QEP Midstream Partners, LP, QEP Midstream Partners GP, LLC, QEP Field Services Company and QEP Midstream Partners Operating, LLC, incorporated by reference to Exhibit 10.1 to the QEP Resources, Inc.'s Current Report on Form 8-K, filed with the Securities and Exchange Commission on August 16, 2013.
10.2	First Amendment to Term Loan Agreement, dated as of August 13, 2013, by and among QEP Resources, Inc., the lenders party thereto and Wells Fargo Bank, National Association, in its capacity as administrative agent for the lenders, incorporated by reference to Exhibit 10.2 to QEP Resources, Inc.'s Current Report on Form 8-K, filed with the Securities and Exchange Commission on August 16, 2013.
10.3	Second Amendment to Credit Agreement, dated as of August 13, 2013, by and among QEP Resources, Inc., the lenders party thereto and Wells Fargo Bank, National Association, in its capacity as administrative agent for the lenders, incorporated by reference to Exhibit 10.3 to QEP Resources, Inc.'s Current Report on Form 8-K, filed with the Securities and Exchange Commission on August 16, 2013.
10.4	Credit Agreement, dated as of August 14, 2013, among QEP Midstream Partners Operating, LLC, as the borrower, QEP Midstream Partners, LP, as the parent guarantor, Wells Fargo Bank, National Association, as administrative agent, and the lenders from time to time party thereto.
10.5	QEP Midstream Partners, LP 2013 Long-Term Incentive Plan.
10.6	Form of QEP Midstream Partners, LP 2013 Long-Term Incentive Plan Phantom Unit Award Agreement.
10.7	Omnibus Agreement, dated as of August 14, 2013, by and among QEP Midstream Partners, LP, QEP Midstream Partners GP, LLC, QEP Resources, Inc., QEP Field Services Company and QEP Midstream Partners Operating, LLC.
10.8	Form of Indemnification Agreement for directors and officers.
31.1	Certification signed by Charles B. Stanley, QEP Resources, Inc.'s Chief Executive Officer, pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
31.2	Certification signed by Richard J. Doleshek, QEP Resources, Inc.'s Chief Financial Officer, pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
32.1	Certification signed by Charles B. Stanley and Richard J. Doleshek, QEP Resources, Inc.'s Chief Executive Officer and Chief Financial Officer, respectively, pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
101.INS	XBRL Instance Document
101.SCH	XBRL Schema Document
101.CAL	XBRL Calculation Linkbase Document
101.LAB	XBRL Label Linkbase Document
101.PRE	XBRL Presentation Linkbase Document
101.DEF	XBRL Definition Linkbase Document

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

QEP RESOURCES, INC.

(Registrant)

November 5, 2013 /s/ Charles B. Stanley

Charles B. Stanley,

Chairman, President and Chief Executive Officer

November 5, 2013 /s/ Richard J. Doleshek

Richard J. Doleshek,

Executive Vice President,

Chief Financial Officer and Treasurer

EXECUTION COPY

Published CUSIP Number: 74736BAB7

CREDIT AGREEMENT

Dated as of August 14, 2013

among

QEP MIDSTREAM PARTNERS OPERATING, LLC,

as the Borrower

QEP MIDSTREAM PARTNERS, LP,

as Parent Guarantor

WELLS FARGO BANK, NATIONAL ASSOCIATION,

as Administrative Agent and Swing Line Lender,

and

the Lenders and L/C Issuers Party Hereto

WELLS FARGO SECURITIES, LLC CITIGROUP GLOBAL MARKETS INC. J.P. MORGAN SECURITIES LLC

and

U.S. BANK NATIONAL ASSOCIATION

Joint Bookrunners

WELLS FARGO SECURITIES, LLC CITIGROUP GLOBAL MARKETS INC. DEUTSCHE BANK SECURITIES INC. J.P. MORGAN SECURITIES LLC MORGAN STANLEY SENIOR FUNDING, INC.

and

U.S. BANK NATIONAL ASSOCIATION

Joint Lead Arrangers

CITIBANK, N.A., JPMORGAN CHASE BANK, N.A.

and

U.S. BANK NATIONAL ASSOCIATION

Co-Syndication Agents

DEUTSCHE BANK AG NEW YORK BRANCH

and

MORGAN STANLEY SENIOR FUNDING, INC.

Co-Documentation Agents

TABLE OF CONTENTS

Section	Page
Article I, DEFINITIONS AND ACCOUNTING TERMS	1
1.01 Defined Terms	1
1.02 Other Interpretive Provisions	31
1.03 Accounting Terms	32
1.04 Rounding	32
1.05 References to Agreements and Laws	33
1.06 Times of Day	33
1.07 Letter of Credit Amounts	33
Article II. THE COMMITMENTS AND CREDIT EXTENSIONS	33
2.01 Loans	33
2.02 Borrowings, Conversions and Continuations of Loans	33
2.03 Letters of Credit	34
2.04 Swing Line Loans	42
2.05 Prepayments	45
2.06 Termination or Reduction of Commitments	46
2.07 Repayment of Loans and Swing Line Loans	46
2.08 Interest	46
2.09 Fees	47
2.10 Computation of Interest and Fees	48
2.11 Evidence of Debt	48
2.12 Payments Generally	49
2.13 Sharing of Payments by Lenders	50
2.14 Increase in Commitments	51
2.15 Defaulting Lenders	51

54

Article III. TAXES, YIELD PROTECTION AND ILLEGALITY

3.02 Illegality	57
3.03 Inability to Determine Rates	58
3.04 Increased Cost and Reduced Return; Capital Adequacy; Reserves on Eurodollar Rate Loans	58
3.05 Compensation for Losses	59
3.06 Matters Applicable to all Requests for Compensation.	60
3.07 Survival	60
Article IV. CONDITIONS PRECEDENT TO CREDIT EXTENSIONS	60
4.01 Conditions to Effectiveness of this Agreement (Execution Date)	60
4.02 Conditions of the Closing Date	61
4.03 Conditions to all Credit Extensions	65
Article V. REPRESENTATIONS AND WARRANTIES	65
5.01 No Default	65
5.02 Organization and Good Standing	66
5.03 Authorization	66
5.04 No Conflicts or Consents	66
5.05 Enforceable Obligations	66
5.06 Financial Statements	66
5.07 Full Disclosure	67

54

3.01 Taxes

5.08 Litigation	67
5.09 Labor Disputes and Acts of God	67
5.10 ERISA	67
5.11 Environmental and Other Laws	68
5.12 Borrower's Subsidiaries; Equity Investments	68
5.13 Title to Properties; Licenses	68
5.14 Government Regulation	68
5.15 Solvency	69
5.16 Compliance with Laws and Material Agreements	69
5.17 Taxes	69
5.18 OFAC	69
5.19 Insurance	69
5.20 Collateral Documents	69
Article VI. AFFIRMATIVE COVENANTS	70
6.01 Books, Financial Statements and Reports	70
6.02 Other Information and Financial Records; Inspections	72
6.03 Notice of Material Events; Notice of Changes in Certain Information Pertaining to the Collateral	72
6.04 Maintenance of Properties	73
6.05 Maintenance of Existence and Qualifications	73
6.06 Payment of Obligations	73
6.07 Insurance	73
6.08 Compliance with Law	74
6.09 Use of Proceeds	74
6.10 Subordination of Intercompany Indebtedness	74
6.11 Additional Material Subsidiaries; Collateral	74
6.12 Further Assurances	76
6.13 Keepwell Requirements	76

Article VII. NEGATIVE COVENANTS	76
7.01 Indebtedness	77
7.02 Limitation on Liens	78
7.03 Sale Lease-back Transactions	78
7.04 Line of Businesses	79
7.05 Investments.	79
7.06 Material Subsidiaries and Immaterial Subsidiaries; Unrestricted Subsidiaries	80
7.07 Mergers, Consolidations and Sales of All or Substantially All Assets	81
7.08 Transactions with Affiliates	81
7.09 Restrictive Agreements	81
7.10 ERISA	82
7.11 Dispositions of Property	82
7.12 Swap Contracts	83
7.13 Financial Covenants	83
7.14 Restricted Payments	83
7.15 Negative Pledge on Pipeline Real Property	84
7.16 Modification, Terminations, etc. of Certain Other Indebtedness, Agreements and Organization Documents	84
7.17 Fiscal Year	84
7.18 Certain Covenants Pertaining to Rendezvous Gas	84

Article	VIII. EVENTS OF DEFAULT AND REMEDIES	85
8.01	Events of Default	85
8.02	Remedies upon Event of Default	87
8.03	Application of Funds	87
Article	2 IX. ADMINISTRATIVE AGENT	88
9.01	Appointment and Authority	88
9.02	Rights as a Lender	89
9.03	Exculpatory Provisions	89
9.04	Reliance by Administrative Agent	90
9.05	Delegation of Duties	90
9.06	Resignation of Administrative Agent	91
9.07	Non-Reliance on Administrative Agent and Other Lenders	91
9.08	Administrative Agent May File Proofs of Claim	92
9.09	Other Agents; Arrangers and Managers	92
9.10	Collateral Matters	92
9.11	Secured Hedge Agreements and Secured Cash Management Agreements	94
Article	X. Continuing Guaranty OF THE PARENT GUARANTOR	94
	Guaranty	94
10.02	Rights of Lenders	94
10.03	Certain Waivers	95
10.04	Obligations Independent	95
10.05	Subrogation	96
10.06	Termination; Reinstatement	96
10.07	Subordination	96
10.08	Stay of Acceleration	96
Article	xI. MISCELLANEOUS	97
	Amendments, Etc.	97
	Notices; Electronic Communications	98
	No Waiver; Cumulative Remedies; Enforcement	100
	Expenses; Indemnity; Damage Waiver	101
	Payments Set Aside	103

103

11.06 Successors and Assigns

11.07	Confidentiality	107
11.08	Set-off	108
11.09	Interest Rate Limitation	108
11.10	Counterparts	109
11.11	Integration	109
11.12	Survival of Representations and Warranties	109
11.13	Severability	109
11.14	Replacement of Lenders	109
11.15	Governing Law	110
11.16	Waiver of Right to Trial by Jury	111
11.17	No Advisory or Fiduciary Responsibility	111
11.18	Electronic Execution of Assignments and Certain Other Documents	112
11.19	USA PATRIOT Act Notice	112
11.20	ENTIRE AGREEMENT	112

SCHEDULES

SCHEDULE 2.01 Commitments and Pro Rata Shares
SCHEDULE 2.04 Swing Line Rate Calculation
SCHEDULE 5.09 Labor Disputes and Acts of God
SCHEDULE 5.12 Subsidiaries and Equity Investments

SCHEDULE 5.19 Insurance

SCHEDULE 7.05 Investments on the Execution Date and Closing Date

SCHEDULE 11.02 Administrative Agent's Office, Certain Addresses for Notices

EXHIBITS

EXHIBIT A-1: Form of Loan Notice

EXHIBIT A-2: Form of Swing Line Loan Notice

EXHIBIT B: Form of Note

EXHIBIT C: Form of Compliance Certificate

EXHIBIT D: Form of Assignment and Assumption

EXHIBIT E-1: Form of U.S. Tax Compliance Certificate (For Foreign Lenders That Are Not Partnerships for U.S. Federal Income Tax

Purposes)

EXHIBIT E-2: Form of U.S. Tax Compliance Certificate (For Foreign Participants That Are Not Partnerships for U.S. Federal Income Tax

Purposes)

EXHIBIT E-3: Form of U.S. Tax Compliance Certificate (For Foreign Participants That Are Partnerships For U.S. Federal Income Tax

Purposes)

EXHIBIT E-4: Form of U.S. Tax Compliance Certificate (For Foreign Lenders That Are Partnerships For U.S. Federal Income Tax Purposes)

CREDIT AGREEMENT

This CREDIT AGREEMENT (this "<u>Agreement</u>") is entered into as of August 14, 2013, among QEP MIDSTREAM PARTNERS OPERATING, LLC, a Delaware limited liability company (the "<u>Borrower</u>"), QEP MIDSTREAM PARTNERS, LP, a Delaware limited partnership ("<u>Parent Guarantor</u>"), each lender from time to time party hereto (collectively, the "<u>Lenders</u>" and individually, a "<u>Lender</u>"), WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent, L/C Issuer and Swing Line Lender, and the other L/C Issuers from time to time party hereto.

The Borrower has requested that the Lenders provide a revolving credit facility, and the Lenders are willing to do so on the terms and conditions set forth herein.

In consideration of the mutual covenants and agreements herein contained, the parties hereto covenant and agree as follows:

ARTICLE I. DEFINITIONS AND ACCOUNTING TERMS

- 1.01 **Defined Terms**. As used in this Agreement, the following terms shall have the meanings set forth below:
- "Acceptable Changes" when used with respect to the Registration Statement means amendments, supplements and other changes thereto that are either (a) not material and adverse to the Lenders as determined by the Joint Bookrunners in their reasonable judgment or (b) are agreed to by the Joint Bookrunners, such agreement not to be unreasonably withheld or delayed.
- "Acquired Businesses" means the businesses, and interests in businesses, described in the Registration Statement acquired by the Borrower on or before the first day of the Availability Period.
- "<u>Acquisition</u>" means a Permitted Acquisition, a Permitted Affiliate Acquisition or any other acquisition of any assets by a Loan Party or a Restricted Subsidiary that is not prohibited by this Agreement.
- "Acquisition Period" means the period from and after a Qualifying Acquisition to and including the last day of the second full fiscal quarter following the fiscal quarter in which such Qualifying Acquisition occurred; provided, that during any Acquisition Period, no additional Acquisition Period shall commence, nor shall such Acquisition Period be extended by any subsequent Qualifying Acquisition until the current Acquisition Period shall have terminated and the Parent Guarantor shall be in compliance with Section 7.13.
- "Administrative Agent" means Wells Fargo Bank in its capacity as administrative agent under any of the Loan Documents, or any successor administrative agent.
- "<u>Administrative Agent's Office</u>" means the Administrative Agent's address as set forth on <u>Schedule 11.02</u>, and the account designated by the Administrative Agent or such other address or account as the Administrative Agent may from time to time notify the Borrower and the Lenders.
 - "Administrative Questionnaire" means an Administrative Questionnaire in a form supplied by the Administrative Agent.

- "Affiliate" means, as to any Person, each other Person that directly or indirectly (through one or more intermediaries or otherwise) controls, is controlled by, or is under common control with, such Person. A Person shall be deemed to be "controlled by" any other Person if such other Person possesses, directly or indirectly, the power to direct or cause the direction of the management or policies of such Person, whether through the ability to exercise voting power, by contract or otherwise.
- "Affiliate Non-Wholly Owned Subsidiary" means each Non-Wholly Owned Subsidiary of the Parent Guarantor of which all the Equity Interests are owned by (x) one or more Loan Parties and/or Restricted Subsidiaries and (y) QRI and/or its Subsidiaries.
- "Agent-Related Persons" means the Administrative Agent, together with its Affiliates, the Arrangers, and the officers, directors, employees, agents and attorneys-in-fact of such Persons and Affiliates.
- "Aggregate Commitments" means the Commitments of all the Lenders in an amount not to exceed \$500,000,000, except as such amount may be increased pursuant to Section 2.14.
 - "Agreement" means this Agreement.
- "<u>Applicable Rate</u>" means, from time to time, the following percentages per annum, based upon the Consolidated Leverage Ratio as set forth in the most recent Compliance Certificate received by the Administrative Agent pursuant to <u>Section 6.01</u>:

Applicable Rate

Pricing Level		Consolidated Leverage Ratio	Commitment Fee	Eurodollar Spread for Eurodollar Loans and+ Letter of Credit Fees	Applicable Margin for Base Rate Loans
	1	< 3.00:1.00	0.325%	1.750%	0.750%
	2	³ 3.00 and < 3.50:1.00	0.350%	2.000%	1.000%
	3	³ 3.50 and < 4.00:1.00	0.375%	2.250%	1.250%
	4	³ 4.00	0.500%	2.500%	1.500%

Any increase or decrease in the Applicable Rate resulting from a change in the Consolidated Leverage Ratio shall become effective as of the first Business Day immediately following the date a Compliance Certificate is delivered pursuant to Section 6.01; provided, however, that if a Compliance Certificate is not delivered when due in accordance with such Section, then Pricing Level 4 shall apply as of the first Business Day after the date on which such Compliance Certificate was required to have been delivered and shall remain in effect until the date on which such Compliance Certificate is delivered. The Applicable Rate in effect on the Closing Date shall be determined based upon Pricing Level 1.

Notwithstanding anything to the contrary contained in this definition, the determination of the Applicable Rate for any period shall be subject to the provisions of <u>Section 2.10(b)</u>.

"Approved Fund" has the meaning specified in Section 11.06(h).

"<u>Arrangers</u>" means Wells Fargo Securities LLC, Citigroup Global Markets, Inc., Deutsche Bank Securities Inc., J.P. Morgan Securities LLC, Morgan Stanley Senior Funding, Inc. and U.S. Bank National Association, in their respective capacities as joint lead arrangers.

"Asset Sale" means (a) the sale, transfer or other disposition (by way of merger or otherwise) of any Equity Interests by the Parent Guarantor, the Borrower or any Restricted Subsidiary, including the issuance of Equity Interests by any Restricted Subsidiary, to any Person other than the Parent Guarantor, the Borrower or a Wholly Owned Restricted Subsidiary and (b) the sale, transfer or other disposition (by way of merger or otherwise) of any other property or assets by the Parent Guarantor, the Borrower or any Restricted Subsidiary to any Person other than the Parent Guarantor, the Borrower or any Wholly Owned Restricted Subsidiary (in each case other than any Disposition permitted under clauses (a) through (h) of Section 7.11).

"Assignee Group" means two or more Eligible Assignees that are Affiliates of one another or two or more Approved Funds managed by the same investment advisor.

"Assignment and Assumption" means an Assignment and Assumption substantially in the form of Exhibit D.

"Attorney Costs" means and includes all reasonable fees, expenses and disbursements of any law firm or other external counsel.

"Attributable Indebtedness" means, on any date, (a) in respect of any capital lease of any Person, the capitalized amount thereof that would appear on a balance sheet of such Person prepared as of such date in accordance with GAAP, and (b) in respect of any Synthetic Lease Obligation, the capitalized amount of the remaining lease payments under the relevant lease that would appear on a balance sheet of such Person prepared as of such date in accordance with GAAP if such lease were accounted for as a capital lease.

"Availability Period" means the period from and including the Closing Date to the earliest of (a) the Maturity Date, (b) the date of termination of the Aggregate Commitments pursuant to Section 2.06, and (c) the date of termination of the Commitment of each Lender to make Loans and of the obligation of the L/C Issuer to make L/C Credit Extensions pursuant to Section 8.02.

"Base Rate" means for any day a fluctuating rate per annum equal to the highest of (a) the Federal Funds Rate plus 1/2 of 1%, (b) the rate of interest in effect for such day as publicly announced from time to time by Wells Fargo Bank as its "prime rate", and (c) the Eurodollar Rate for a one month Interest Period for such day (or if such day is not a Business Day, the immediately preceding Business Day) plus 1%. The "prime rate" is a rate set by Wells Fargo Bank based upon various factors including Wells Fargo Bank's costs and desired return, general economic conditions and other factors, and is used as a reference point for pricing some loans, which may be priced at, above, or below such announced rate. Any change in such rate announced by Wells Fargo Bank shall take effect at the opening of business on the day specified in the public announcement of such change.

"Base Rate Loan" means a Loan that bears interest based on the Base Rate.

"Borrower" has the meaning specified in the introductory paragraph hereto.

"Borrower Materials" has the meaning specified in Section 6.01.

"Borrowing" means a borrowing consisting of simultaneous Loans of the same Type and, in the case of Eurodollar Rate Loans, having the same Interest Period made by each of the Lenders pursuant to Section 2.0.

"Business Day" means any day other than a Saturday, Sunday or other day on which commercial banks are authorized to close under the Laws of, or are in fact closed in, the state where the Administrative Agent's Office is located and, if such day relates to any Eurodollar Rate Loan, means any such day on which dealings in Dollar deposits are conducted by and between banks in the London interbank eurodollar market.

"Cash Collateralize" means to pledge and deposit with or deliver to the Administrative Agent, for the benefit of the L/C Issuer or Lenders, as collateral for L/C Obligations or obligations of Lenders to fund participations in respect of L/C Obligations, cash or deposit account balances or, if the Administrative Agent and the L/C Issuer shall agree in their sole discretion, other credit support, in each case pursuant to documentation in form and substance satisfactory to the Administrative Agent and the applicable L/C Issuers. "Cash Collateral" shall have a meaning correlative to the foregoing and shall include the proceeds of such cash collateral and other credit support.

"<u>Cash Management Agreement</u>" means any agreement between or among one or more Loan Parties or Restricted Subsidiaries and a Cash Management Bank, to provide cash management services, including treasury, depository, overdraft, credit or debit card, electronic funds transfer and other cash management arrangements.

"Cash Management Bank" means any Lender, an Affiliate of a Lender, the Administrative Agent and any Affiliate of the Administrative Agent, in its capacity as a party to such Cash Management Agreement.

"Casualty Event" means any loss of or damage to or destruction of, or any condemnation or other taking of, any property of a Loan Party or a Restricted Subsidiary.

"Change in Law" means (a) the adoption of any law, rule or regulation after the date of this Agreement, (b) any change in any law, rule or regulation or in the interpretation or application thereof by any Governmental Authority after the date of this Agreement or (c) compliance by any Lender with any request, guideline or directive (whether or not having the force of law) of any Governmental Authority made or issued after the date of this Agreement; provided, however, that notwithstanding anything herein to the contrary, (i) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, guidelines or directives thereunder or issued in connection therewith and (ii) all requests, rules, guidelines or directives promulgated by any Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States or foreign regulatory authorities, in each case pursuant to Basel III, shall in each case be deemed to be a "Change in Law", regardless of the date enacted, adopted or issued.

"Change of Control" means as of any date, any of the following shall occur: (a) the Parent Guarantor fails to own and control 100% of the Equity Interests of the Borrower, (b) QRI fails to own and Control Equity Interests representing more than 50% of the outstanding Equity Interests of the General Partner, (c) QRI fails to have the right or ability by voting power, contract or otherwise to elect or designate for election more than 50% of the board of directors or equivalent governing body of the General Partner, (d) the General Partner fails to be the sole general partner of and Control the Parent Guarantor, (e) the occurrence of a "Change of Control" as defined in the QRI Credit Agreement as in effect on the date hereof, or (f) the occurrence of a "change of control" or similar event under documentation governing any Qualified Notes Offering.

"Closing Date" means the first date all the conditions precedent in Section 4.02 are satisfied or waived in accordance with Section 11.01.

"Code" means the Internal Revenue Code of 1986.

"Collateral" means all collateral under or as defined in any Collateral Document.

"Collateral Documents" means, collectively, the Guaranty, the Security Agreements, and other similar documents now or hereafter delivered to the Administrative Agent pursuant to Section 6.11, and each other agreement, document or instrument that now or hereafter creates or perfects or purports to create or perfect, or that governs the terms of, a Lien in favor of the Administrative Agent for the ratable benefit of the Secured Parties.

"Commercial Operation Date" means the date on which a Material Project achieves commercial operation.

"Commitment" means, as to each Lender, its obligation to (a) make Loans to the Borrower pursuant to Section 2.01, (b) purchase participations in L/C Obligations, and (c) purchase participations in Swing Line Loans, in an aggregate principal amount at any one time outstanding not to exceed the amount set forth opposite such Lender's name on Schedule 2.01 or in the Assignment and Assumption pursuant to which such Lender becomes a party hereto, as applicable, as such amount may be adjusted from time to time in accordance with this Agreement.

"Commodity Exchange Act" means the Commodity Exchange Act (7 U.S.C. § 1 et seq.), any rule, regulation or order of the Commodity Futures Trading Commission (or the application or official interpretation of any thereof).

"Compliance Certificate" means a certificate substantially in the form of Exhibit C.

"Connection Income Taxes" means Other Connection Taxes that are imposed on or measured by net income (however denominated) or that are franchise Taxes or branch profits Taxes.

"Consolidated" or "consolidated" means (a) when used with reference to a Subsidiary of a Person, a Subsidiary of such Person the financial statements of which are (or should be) consolidated with the financial statement of such Person in accordance with GAAP, and (b) when used with respect to financial statements or financial statement items of any Person, such statements or items on a consolidated basis in accordance with applicable principles of consolidation under GAAP.

"Consolidated EBITDA" means:

"Consolidated EBITDA of the Parent Guarantor" means, for any period, an amount equal to Consolidated Net Income of the Parent Guarantor plus (a) the following to the extent deducted in calculating such Consolidated Net Income of the Parent Guarantor: (i) Consolidated Interest Charges for such period, (ii) Federal, state, local and foreign income taxes payable for such period, (iii) depreciation, depletion and amortization expense, (iv) any extraordinary losses (including losses on sales of assets outside of the ordinary course of business), and (v) unrealized losses under Swap Contracts, plus (b) the amount of cash dividends and other distributions actually received during such period by the Parent Guarantor, the Borrower and Restricted Subsidiaries from (i) Specified Non-Wholly Owned Subsidiaries, (ii) Unrestricted Subsidiaries and (iii) Joint Ventures, minus (c) the following to the extent included in calculating such Consolidated Net Income of the Parent Guarantor: (i) Consolidated Net Income of Specified Non-Wholly Owned Subsidiaries and any amounts added pursuant to clauses (a)(i) through (v) of this definition that are attributable to Specified Non-Wholly Owned Subsidiaries, (ii) Consolidated Net Income of Unrestricted Subsidiaries and any amounts added pursuant to

<u>clauses (a)(i)</u> through <u>(v)</u> of this definition that are attributable to Unrestricted Subsidiaries, (iii) any amount included in Consolidated Net Income that is attributable to Joint Ventures and any amounts added pursuant to clauses (a)(i) through (v) of this definition that are attributable to Joint Ventures, (iv) Federal, state, local and foreign income tax credits for such period, (v) all non-cash items increasing Consolidated Net Income for such period, (vi) gains on sales of assets outside the ordinary course of business, and (vii) unrealized gains under Swap Contracts.

"Consolidated EBITDA" of any Person other than the Parent Guarantor means, for any period, an amount equal to Consolidated Net Income of such Person and its Consolidated Subsidiaries plus (a) the following to the extent deducted in calculating the Consolidated Net Income of such Person:

(i) Consolidated Interest Charges for such period, (ii) Federal, state, local and foreign income taxes payable for such period, (iii) depreciation, depletion and amortization expense, (iv) any extraordinary losses (including losses on sales of assets outside of the ordinary course of business), and (v) unrealized losses under Swap Contracts, plus the following, to the extent not included in Consolidated Net Income (b) the amount of cash dividends and other distributions actually received during such period by such Person and its Consolidated Subsidiaries from Joint Ventures minus (c) the following to the extent included in calculating such Consolidated Net Income: (i) Federal, state, local and foreign income tax credits for such period, (ii) all non-cash items increasing Consolidated Net Income for such period, (iii) gains on sales of assets outside the ordinary course of business, and (vi) unrealized gains under Swap Contracts.

Unless otherwise specifically stated herein, references to "Consolidated EBITDA" shall mean "Consolidated EBITDA of the Parent Guarantor".

In the calculation of Consolidated EBITDA of the Parent Guarantor, Consolidated Net Income of the Parent Guarantor, and the other items in clauses (a), (b) and (c) of the calculation above, shall be adjusted with respect to Non-Wholly Owned Subsidiaries to reflect the Parent Guarantor's pro rata ownership interest therein. In the calculation of Consolidated EBITDA, Consolidated Net Income, and the other items in clauses (a), (b) and (c) of the calculation above, shall be adjusted with respect to Non-Wholly Owned Subsidiaries to reflect such Person's pro rata ownership interest therein.

For purposes of calculating Consolidated EBITDA of the Parent Guarantor or Consolidated EBITDA, as applicable:

- (1) Consolidated EBITDA of the Parent Guarantor, and Consolidated EBITDA, for any relevant period shall be calculated to give pro forma effect to any Acquisition and any Disposition of assets, in each case for a purchase price or sale price, as the case may be, equal to \$10,000,000 or more, consummated at any time after the first day of the relevant period as if each such Permitted Acquisition and each such Disposition had occurred on the first day of such period, provided that any such pro forma adjustments shall be (A) calculated by the Parent Guarantor in good faith, (B) supported by reasonably detailed calculations presented by the Parent Guarantor together with the Compliance Certificate for the applicable period, and (C) subject to the approval of the Administrative Agent (such approval not to be unreasonably withheld or delayed);
- (2) Consolidated EBITDA of the Parent Guarantor, and Consolidated EBITDA, for the first twelve months subsequent to the Closing Date shall be calculated on an annualized 365 day basis for the number of days actually elapsed since the Closing Date until the date of determination; and

(3) If a Loan Party or a Restricted Subsidiary undertakes a Material Project, then Consolidated EBITDA of the Parent Guarantor may include, at the Borrower's option, a Material Project Consolidated EBITDA Adjustment for such Material Project, subject to Administrative Agent's review and approval of each component of such Material Project Consolidated EBITDA Adjustment, provided that: (A) no such additions shall be allowed with respect to any Material Project unless: (y) not later than 15 days (or such lesser period as is acceptable to the Administrative Agent) prior to the delivery of any Compliance Certificate required by Section 6.01(a) or 6.01(b) to the extent Material Project Consolidated EBITDA Adjustments will be made to Consolidated EBITDA, the Borrower shall have delivered to the Administrative Agent written pro forma projections of Consolidated EBITDA attributable to such Material Project, and (z) prior to the date such Compliance Certificate is required to be delivered, the Administrative Agent shall have approved such projections (such approval not to be unreasonably withheld or delayed) and shall have received such other information and documentation as the Administrative Agent may reasonably request, all in form and substance reasonably satisfactory to the Administrative Agent, and (B) the aggregate amount of all Material Project Consolidated EBITDA Adjustments during any period shall be limited to 20% of the total actual Consolidated EBITDA of the Parent Guarantor for such period (which total actual Consolidated EBITDA of the Parent Guarantor shall be determined without including any Material Project Consolidated EBITDA Adjustments).

"Consolidated Funded Debt" means the aggregate of the Indebtedness of the Parent Guarantor, the Borrower and their Restricted Subsidiaries (other than Specified Non-Wholly Owned Subsidiaries) described in clauses (a), (b), (c), (c), (c), (d), (e), (f) and (g) (other than Indebtedness consisting of Guarantees in respect of net obligations under any Swap Contract permitted under this Agreement) of the definition of Indebtedness in Section 1.01, on a consolidated basis after elimination of intercompany items. Notwithstanding the foregoing, Indebtedness of a Restricted Subsidiary (other than Specified Non-Wholly Owned Subsidiaries) that is not Wholly Owned by the Parent Guarantor shall be included in Consolidated Funded Debt only to the extent of the Parent Guarantor's proportional interest therein, unless such Indebtedness is recourse to the Parent Guarantor, the Borrower or any Restricted Subsidiary, in which case the full amount of such Indebtedness that is recourse to the Parent Guarantor, the Borrower or such Restricted Subsidiary shall be included in the calculation of Consolidated Funded Debt.

"Consolidated Interest Charges" for a Person means, for any period, the sum of the following for such Person and its Consolidated Subsidiaries (a) all interest, premium payments, amortization or write-off of debt discount, fees, charges, issuance costs and commissions and related expenses in connection with borrowed money (including capitalized interest) or in connection with the deferred purchase price of assets, in each case to the extent treated as interest in accordance with GAAP, and (b) the portion of rent expense with respect to such period under capital leases that is treated as interest in accordance with GAAP.

"Consolidated Interest Coverage Ratio" means, as of any date of determination, the ratio of (a) Consolidated EBITDA of the Parent Guarantor for the period of four consecutive fiscal quarters ending on such date to (b) Consolidated Interest Charges of the Parent Guarantor and its Restricted Subsidiaries for the period of four consecutive fiscal quarters ending on such date.

"Consolidated Leverage Ratio" means, as of any date of determination, the ratio of (a) Consolidated Funded Debt as of such date to (b) Consolidated EBITDA of the Parent Guarantor for the period of the four consecutive fiscal quarters ending on such date.

"Consolidated Net Income" of any Person means, for any period, consolidated net income of such Person and its Consolidated Subsidiaries (excluding extraordinary gains and extraordinary losses) for that

period; provided that Consolidated Net Income of any Person for any period shall exclude the net income (but not loss) of any Subsidiary thereof to the extent that the declaration or payment of dividends or similar distributions by such Subsidiary to such Person is not at the time permitted by operation of the terms of its Organization Documents, Contractual Obligation (other than pursuant to this Agreement) or applicable law, or is otherwise restricted or prohibited.

"Consolidated Net Tangible Assets of the Parent Guarantor" means at any date of determination, the total amount of Consolidated assets of the Parent Guarantor, the Borrower and the Restricted Subsidiaries (other than Specified Non-Wholly Owned Subsidiaries) after deducting therefrom: (a) all current liabilities (excluding (i) any current liabilities that by their terms are extendable or renewable at the option of the obligor thereon to a time more than 12 months after the time as of which the amount thereof is being computed, and (ii) current maturities of long-term debt) of the Parent Guarantor, the Borrower and the Restricted Subsidiaries (other than Specified Non-Wholly Owned Subsidiaries); and (b) the value of all goodwill, trade names, trademarks, patents and other like intangible assets (other than assets of Specified Non-Wholly Owned Subsidiaries), all as set forth on the Consolidated balance sheet of the Parent Guarantor. For the avoidance of doubt, Consolidated Net Tangible Assets of the Parent Guarantor excludes all amounts attributable to the assets of Specified Non-Wholly Owned Subsidiaries and Unrestricted Subsidiaries, but includes all equity investments in Specified Non-Wholly Owned Subsidiaries and Unrestricted Subsidiaries.

"Consolidated Senior Secured Funded Debt" means, as of any date of determination, all Consolidated Funded Debt that is secured by a Lien on any assets of the Parent Guarantor, the Borrower or any Restricted Subsidiary (other than Specified Non-Wholly Owned Subsidiaries).

"Consolidated Senior Secured Leverage Ratio" means, as of any date of determination, the ratio of (a) Consolidated Senior Secured Funded Debt as of such date to (b) Consolidated EBITDA of the Parent Guarantor for the period of four consecutive fiscal quarters ended on such date.

"Consolidated Total Assets" means:

"Consolidated Total Assets of the Parent Guarantor" means, as of any date of determination, the total assets of Parent Guarantor, the Borrower and the Restricted Subsidiaries (other than Specified Non-Wholly Owned Subsidiaries) determined in accordance with GAAP, as set forth on the Consolidated balance sheet of the Parent Guarantor prepared in accordance with GAAP. For the avoidance of doubt, Consolidated Total Assets of the Parent Guarantor excludes all amounts attributable to the assets of Specified Non-Wholly Owned Subsidiaries and Unrestricted Subsidiaries, but includes all equity investments in Specified Non-Wholly Owned Subsidiaries and Unrestricted Subsidiaries.

"Consolidated Total Assets" of a Person other than the Parent Guarantor means, as of any date of determination, the total assets of such Person and its Consolidated Subsidiaries, determined in accordance with GAAP, as set forth on the Consolidated balance sheet of such Person and its Consolidated Subsidiaries prepared in accordance with GAAP.

"<u>Contractual Obligation</u>" means, as to any Person, any provision of any security issued by such Person or of any agreement, instrument or other undertaking to which such Person is a party or by which it or any of its property is bound.

"Contribution" means the direct or indirect transfer (in one or more transactions) by QRI and certain of its Subsidiaries to the Parent Guarantor and its Subsidiaries of the Acquired Businesses on or before the Closing Date.

"Control" shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through the ownership of voting securities, by contract or otherwise, and the terms "Controlling" and "Controlled" shall have meanings correlative thereto.

"Credit Extension" means each of the following: (a) a Borrowing, (b) an L/C Credit Extension, and (c) a Swing Line Borrowing.

"<u>Debtor Relief Laws</u>" means the Bankruptcy Code of the United States, and all other liquidation, conservatorship, bankruptcy, assignment for the benefit of creditors, moratorium, rearrangement, receivership, insolvency, reorganization, or similar debtor relief Laws of the United States or other applicable jurisdictions from time to time in effect and affecting the rights of creditors generally.

"<u>Default</u>" means any event or condition that constitutes an Event of Default or that, with the giving of any notice, the passage of time, or both, would be an Event of Default.

"<u>Default Rate</u>" means (a) when used with respect to Obligations other than Letter of Credit Fees, an interest rate equal to (i) the Base Rate plus (ii) the Applicable Rate for Base Rate Loans plus (iii) 2% per annum; <u>provided</u>, <u>however</u>, that with respect to a Eurodollar Rate Loan, the Default Rate shall be an interest rate equal to the interest rate (including the Applicable Rate for Eurodollar Loans) otherwise applicable to such Loan plus 2% per annum, and (b) when used with respect to Letter of Credit Fees, a rate equal to the Applicable Rate plus 2% per annum, in all cases to the fullest extent permitted by applicable Laws.

"Defaulting Lender" means, subject to Section 2.15(f), any Lender that (a) has failed to (i) fund all or any portion of its Loans within two Business Days of the date such Loans were required to be funded hereunder unless such Lender notifies the Administrative Agent and the Borrower in writing that such failure is the result of such Lender's determination that one or more conditions precedent to funding (each of which conditions precedent, together with any applicable default, shall be specifically identified in such writing) has not been satisfied, or (ii) pay to the Administrative Agent, the L/C Issuer, the Swing Line Lender or any other Lender any other amount required to be paid by it hereunder (including in respect of its participation in Letters of Credit or Swing Line Loans) within two Business Days of the date when due, (b) has notified the Borrower, the Administrative Agent, the L/C Issuer or the Swing Line Lender in writing that it does not intend to comply with its funding obligations hereunder, or has made a public statement to that effect (unless such writing or public statement relates to such Lender's obligation to fund a Loan hereunder and states that such position is based on such Lender's determination that a condition precedent to funding (which condition precedent, together with any applicable default, shall be specifically identified in such writing or public statement) cannot be satisfied), (c) has failed, within three Business Days after written request by the Administrative Agent or the Borrower, to confirm in writing to the Administrative Agent and the Borrower that it will comply with its prospective funding obligations hereunder (provided that such Lender shall cease to be a Defaulting Lender pursuant to this clause (c) upon receipt of such written confirmation by the Administrative Agent and the Borrower), or (d) has, or has a direct or indirect parent company that has, (i) become the subject of a proceeding under any Debtor Relief Law, or (ii) had appointed for it a receiver, custodian, conservator, trustee, administrator, assignee for the benefit of creditors or similar Person charged with reorganization or liquidation of its business or assets, including the Federal Deposit Insurance Corporation or any other state or federal regulatory authority acting in such a capacity; provided that a Lender shall not be a Defaulting Lender solely by virtue of the ownership or acquisition of any equity interest in that Lender or any direct or indirect parent company thereof by a Governmental Authority so long as such ownership interest does not result in or provide such Lender with immunity from the jurisdiction of courts within the United States or from the enforcement of judgments or writs of attachment on its assets or permit such Lender (or such

Governmental Authority) to reject, repudiate, disavow or disaffirm any contracts or agreements made with such Lender. Any determination by the Administrative Agent that a Lender is a Defaulting Lender under any one or more of clauses (a) through (d) above shall be conclusive and binding absent manifest error, and such Lender shall be deemed to be a Defaulting Lender (subject to Section 2.15(f)) upon delivery of written notice of such determination to the Borrower, the L/C Issuer, the Swing Line Lender and each Lender.

"<u>Disposition</u>" or "<u>Dispose</u>" means the sale, transfer, license, lease or other disposition (including any sale and leaseback transaction) of any property by any Person, including any sale, assignment, transfer or other disposal, with or without recourse, of any notes or accounts receivable or any rights and claims associated therewith.

"<u>Disqualified Equity</u>" means any Equity Interest that, by its terms (or by the terms of any security into which it is convertible or for which it is exchangeable), or upon the happening of any event, (a) matures (excluding any maturity as the result of an optional redemption by the issuer thereof) or is mandatorily redeemable, pursuant to a sinking fund obligation or otherwise, or is redeemable at the option of the holder thereof, in whole or in part, or requires the payment of any cash dividend or any other scheduled payment constituting a return of capital, in each case at any time on or prior to the six months after the Maturity Date, or (b) is convertible into or exchangeable (unless at the sole option of the issuer thereof) for (i) debt securities or (ii) any Equity Interest referred to in clause (a) above, in each case at any time earlier to the six months after the latest Maturity Date.

"Dollar" and "\$" mean lawful money of the United States.

"Eligible Assignee" has the meaning specified in Section 11.06(h).

"Environmental Laws" means any and all Federal, state, local, and foreign statutes, laws, regulations, ordinances, rules, judgments, orders, decrees, permits, concessions, grants, franchises, licenses, agreements or governmental restrictions relating to pollution and the protection of the environment or the release of any materials into the environment, including those related to hazardous substances or wastes, air emissions and discharges to waste or public systems.

"Environmental Liability" means any liability, contingent or otherwise (including any liability for damages, costs of environmental remediation, fines, penalties or indemnities), of any Loan Party or any Subsidiary thereof directly or indirectly resulting from or based upon (a) violation of any Environmental Law, (b) the generation, use, handling, transportation, storage, treatment or disposal of any Hazardous Materials, (c) exposure to any Hazardous Materials, (d) the release or threatened release of any Hazardous Materials into the environment or (e) any contract, agreement or other consensual arrangement pursuant to which liability is assumed or imposed with respect to any of the foregoing.

"Equity Interests", "equity interests" and "equity securities" means, with respect to any Person, all of the shares of capital stock of (or other ownership or profit interests in) such Person, all of the warrants, options or other rights for the purchase or acquisition from such Person of shares of capital stock of (or other ownership or profit interests in) such Person, all of the securities convertible into or exchangeable for shares of capital stock of (or other ownership or profit interests in) such Person or warrants, rights or options for the purchase or acquisition from such Person of such shares (or such other interests), and all of the other ownership or profit interests in such Person (including partnership, member or trust interests therein), whether voting or non-voting, and whether or not such shares, warrants, options, rights or other interests are outstanding on any date of determination. The term Equity Interests shall also include other securities or instruments that have both debt and equity features.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"ERISA Affiliate" means any trade or business (whether or not incorporated) under common control with the Borrower within the meaning of Section 414(b) or (c) of the Code (and Sections 414(m) and (o) of the Code for purposes of provisions relating to Section 412 of the Code).

"ERISA Event" means (a) a Reportable Event with respect to a Pension Plan; (b) a withdrawal by the Borrower or any ERISA Affiliate from a Pension Plan subject to Section 4063 of ERISA during a plan year in which it was a substantial employer (as defined in Section 4001(a)(2) of ERISA) or a cessation of operations that is treated as such a withdrawal under Section 4062(e) of ERISA; (c) a complete or partial withdrawal by the Borrower or any ERISA Affiliate from a Multiemployer Plan or notification that a Multiemployer Plan is in reorganization; (d) the filing of a notice of intent to terminate, the treatment of a Pension Plan amendment as a termination under Sections 4041 or 4041A of ERISA, or the commencement of proceedings by the PBGC to terminate a Pension Plan or Multiemployer Plan; (e) an event or condition which constitutes grounds under Section 4042 of ERISA for the termination of, or the appointment of a trustee to administer, any Pension Plan; or (f) the imposition of any liability under Title IV of ERISA, other than for PBGC premiums due but not delinquent under Section 4007 of ERISA, upon the Borrower or any ERISA Affiliate.

"Eurodollar Rate" means, for any Interest Period with respect to a Eurodollar Rate Loan, the rate per annum equal to the British Bankers Association LIBOR Rate (or the successor thereto if the British Bankers Association is no longer making a LIBOR Rate available) ("LIBOR"), as published by Reuters (or other commercially available source providing quotations of LIBOR as designated by the Administrative Agent from time to time) at approximately 11:00 a.m., London time, two Business Days prior to the commencement of such Interest Period, for Dollar deposits (for delivery on the first day of such Interest Period) with a term equivalent to such Interest Period. If such rate is not available at such time for any reason, then the "Eurodollar Rate" for such Interest Period shall be the rate per annum determined by the Administrative Agent to be the rate at which deposits in Dollars for delivery on the first day of such Interest Period in same day funds in the approximate amount of the Eurodollar Rate Loan being made, continued or converted by Wells Fargo Bank and with a term equivalent to such Interest Period would be offered by Wells Fargo Bank's London Branch to major banks in the London interbank eurodollar market at their request at approximately 11:00 a.m. (London time) two Business Days prior to the commencement of such Interest Period.

"<u>Eurodollar Rate Loan</u>" means a Loan that bears interest at a rate based on the Eurodollar Rate (other than pursuant to <u>clause (c)</u> of the definition of "<u>Base Rate</u>").

"Event of Default" has the meaning specified in Section 8.01.

"Exchange Act" means the United States Securities Exchange Act of 1934, as amended.

"Excluded Property" has the meaning specified in the Security Agreement.

"Excluded Swap Obligation" means, with respect to any Guarantor, any Swap Obligation if, and to the extent that, all or a portion of the Guarantee of such Guarantor of, or the grant by such Guarantor of a security interest to secure, such Swap Obligation (or any Guarantee thereof) is or becomes illegal under the Commodity Exchange Act by virtue of such Guarantor's failure for any reason to constitute an "eligible contract participant" as defined in the Commodity Exchange Act thereunder at the time the Guarantee of such Guarantor or the grant of such security interest becomes effective with respect to such Swap Obligation. If a Swap Obligation arises under a master agreement governing more than one swap, such exclusion shall apply only to the portion of such Swap Obligation that is attributable to swaps for which such Guarantee or security interest is or becomes illegal.

"Excluded Taxes" means any of the following Taxes imposed on or with respect to a Recipient or required to be withheld or deducted from a payment to a Recipient, (a) Taxes imposed on or measured by net income (however denominated), franchise Taxes, and branch profits Taxes, in each case, (i) imposed as a result of such Recipient being organized under the laws of, or having its principal office or, in the case of any Lender, its applicable Lending Office located in, the jurisdiction imposing such Tax (or any political subdivision thereof) or (ii) that are Other Connection Taxes (b) U.S. federal withholding Taxes imposed on amounts payable to or for the account of such Lender with respect to an applicable interest in a Loan or Commitment pursuant to a law in effect on the date on which (i) such Lender acquires such interest in the Loan or Commitment (other than pursuant to an assignment request by the Borrower under Section 11.14) or (ii) such Lender changes its Lending Office, except in each case to the extent that, pursuant to Section 3.01, amounts with respect to such Taxes were payable either to such Lender's assignor immediately before such Lender became a party hereto or to such Lender immediately before it changed its Lending Office, (c) Taxes attributable to such Recipient's failure to comply with Section 3.01(g) and (d) any U.S. federal withholding Taxes imposed under FATCA.

"Execution Date" means the first date all conditions precedent in Section 4.01 are satisfied or waived in accordance with Section 11.01.

"<u>FATCA</u>" means Sections 1471 through 1474 of the Code, as in effect on the date of this Agreement (or any amended or successor version that is substantively comparable and not materially more onerous to comply with), any current or future regulations thereunder or official interpretations thereof and any agreements entered into pursuant to Section 1471(b)(1) of the Code or otherwise pursuant to any of the foregoing.

"Federal Funds Rate" means, for any day, the rate per annum equal to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers on such day, as published by the Federal Reserve Bank of New York on the Business Day next succeeding such day; provided that (a) if such day is not a Business Day, the Federal Funds Rate for such day shall be such rate on such transactions on the next preceding Business Day as so published on the next succeeding Business Day, and (b) if no such rate is so published on such next succeeding Business Day, the Federal Funds Rate for such day shall be the average rate (rounded, if necessary, to the nearest 1/100 of 1%) charged to Wells Fargo Bank on such day on such transactions as determined by the Administrative Agent.

"Fee Letters" means (a) the Administrative Agency Fee Letter, dated July 19, 2013, between the Borrower and the Administrative Agent and (b) and the Fee Letter, dated July 19, 2013, among the Borrower, the Administrative Agent and the Arrangers.

"<u>Finance Subsidiary</u>" means a direct Wholly Owned Subsidiary of the Parent Guarantor that (a) (i) is formed solely to be a co-issuer, jointly and severally, with the Parent Guarantor, of any senior unsecured notes of the Parent Guarantor and (ii) has nominal assets and conducts no business operations or (b) (i) engages in no material business other than (x) lending funds to a Loan Party or a direct or indirect Wholly Owned Subsidiary of a Loan Party, and (y) activities incidental to the foregoing, and (ii) has no Indebtedness other than to a Loan Party.

"Financial Covenants" means the covenants contained in Section 7.13.

"<u>First Priority</u>" means, with respect to any Lien purported to be created and granted in any Collateral pursuant to any Collateral Document, that such Lien is the most senior Lien to which such Collateral is subject, except to the extent otherwise permitted by <u>Section 7.02</u>.

"<u>Foreign Lender</u>" means (a) if the Borrower is a U.S. Person, a Lender that is not a U.S. Person, and (b) if the Borrower is not a U.S. Person, a Lender that is resident or organized under the Laws of a jurisdiction other than that in which the Borrower is resident for tax purposes.

"FRB" means the Board of Governors of the Federal Reserve System of the United States.

"Fronting Exposure" means, at any time there is a Defaulting Lender, (a) with respect to a L/C Issuer, such Defaulting Lender's Pro Rata Share of the outstanding L/C Obligations with respect to Letters of Credit issued by such L/C Issuer other than L/C Obligations as to which such Defaulting Lender's participation obligation has been reallocated to other Lenders or Cash Collateralized in accordance with the terms hereof, and (b) with respect to the Swing Line Lender, such Defaulting Lender's Pro Rata Share of outstanding Swing Line Loans made by the Swing Line Lender other than Swing Line Loans as to which such Defaulting Lender's participation obligation has been reallocated to other Lenders in accordance with the terms hereof.

"Fund" has the meaning set forth in Section 11.06(h).

"GAAP" means generally accepted accounting principles in the United States set forth in the opinions and pronouncements of the Accounting Principles Board and the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board or such other principles as may be approved by a significant segment of the accounting profession in the United States, that are applicable to the circumstances as of the date of determination, consistently applied.

"General Partner" means QEP Midstream Partners GP, LLC, a Delaware limited liability company, general partner of the Parent Guarantor.

"Governmental Authority" means any nation or government, any state or other political subdivision thereof, any agency, authority, instrumentality, regulatory body, court, administrative tribunal, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to government.

"Guarantee" means, as to any Person, (a) any obligation, contingent or otherwise, of such Person guaranteeing or having the economic effect of guaranteeing any Indebtedness or other obligation payable or performable by another Person (the "primary obligor") in any manner, whether directly or indirectly, and including any obligation of such Person, direct or indirect, (i) to purchase or pay (or advance or supply funds for the purchase or payment of) such Indebtedness or other obligation, (ii) to purchase or lease property, securities or services for the purpose of assuring the obligee in respect of such Indebtedness or other obligation of the payment or performance of such Indebtedness or other obligation, (iii) to maintain working capital, equity capital or any other financial statement condition or liquidity or level of income or cash flow of the primary obligor so as to enable the primary obligor to pay such Indebtedness or other obligation, or (iv) entered into for the purpose of assuring in any other manner the obligee in respect of such Indebtedness or other obligation of the payment or performance thereof or to protect such obligee against loss in respect thereof (in whole or in part), or (b) any Lien on any assets of such Person securing any Indebtedness or other obligation of any other Person, whether or not such Indebtedness or other obligation is assumed by such Person. The amount of any Guarantee shall be deemed to be an amount equal to the stated or determinable amount of the related primary obligation, or

portion thereof, in respect of which such Guarantee is made or, if not stated or determinable, the maximum reasonably anticipated liability in respect thereof as determined by the guaranteeing Person in good faith. The term "Guarantee" as a verb has a corresponding meaning.

"Guaranteed Obligations" has the meaning set forth in Section 10.01.

"Guarantors" means, collectively, the Parent Guarantor and each Subsidiary Guarantor.

"Guaranty" means, collectively, the guaranty made by the Parent Guarantor under <u>Article X</u>, the Subsidiary Guaranty Agreement and any other guaranty agreements delivered pursuant to this Agreement.

"<u>Hazardous Materials</u>" means all explosive or radioactive substances or wastes and all hazardous or toxic substances, wastes or other pollutants, including petroleum or petroleum distillates, asbestos or asbestos-containing materials, polychlorinated biphenyls, radon gas, infectious or medical wastes and all other substances or wastes of any nature regulated pursuant to any Environmental Law.

"Hedge Bank" has the meaning set forth in the definition of "Secured Hedge Agreement."

"Immaterial Subsidiary" means any Subsidiary designated by the Borrower pursuant to Section 7.06 as an Immaterial Subsidiary, if and for so long as (a) such Subsidiary does not have (i) Consolidated assets in excess of 5% of Consolidated Total Assets of the Parent Guarantor or (ii) Consolidated EBITDA for the four fiscal quarters ending on the most recent Quarter-End Date that accounts for more than 5% of Consolidated EBITDA of the Parent Guarantor for such period, and (b) such Immaterial Subsidiary, together with all other Subsidiaries designated by the Borrower as Immaterial Subsidiaries, (i) does not have Consolidated assets exceeding 10% of the Consolidated Total Assets of the Parent Guarantor and (ii) does not have Consolidated EBITDA for the four fiscal quarters ending on the most recent Quarter-End Date that accounts for more than 10% of Consolidated EBITDA of the Parent Guarantor for such period.

A Subsidiary will not be an Immaterial Subsidiary to the extent that any of the required terms set forth in clauses (a) or (b) of this paragraph is not satisfied, and if a Subsidiary that has been designated by the Borrower as an Immaterial Subsidiary ceases to satisfy any of such terms, from and after the date that it ceases to satisfy any such terms, such Subsidiary shall cease to be an Immaterial Subsidiary and shall be a Material Subsidiary.

"Indebtedness" means, as to any Person at a particular time, without duplication, all of the following, whether or not included as indebtedness or liabilities in accordance with GAAP:

- (a) all obligations of such Person for borrowed money and all obligations of such Person evidenced by bonds, debentures, notes, loan agreements or other similar instruments;
- (b) all direct or contingent obligations of such Person arising under letters of credit (including standby and commercial), bankers' acceptances, bank guaranties, surety bonds and similar instruments;.
 - (c) net obligations of such Person under any Swap Contract;
- (d) all obligations of such Person to pay the deferred purchase price of property or services (other than trade accounts payable in the ordinary course of business);

- (e) indebtedness (excluding prepaid interest thereon) secured by (or for which the holder of such indebtedness has an existing right, contingent or otherwise, to be secured by) a Lien on property owned or being acquired by such Person (including indebtedness arising under conditional sales or other title retention agreements), whether or not such indebtedness shall have been assumed by such Person or is limited in recourse;
 - (f) capital leases and Synthetic Lease Obligations; and
 - (g) all Guarantees of such Person in respect of any of the foregoing.

For all purposes hereof, the Indebtedness of any Person shall include the Indebtedness of any partnership or joint venture (other than a joint venture that is itself a corporation or limited liability company) in which such Person is a general partner or a joint venturer, unless such Indebtedness is expressly made non-recourse to such Person. The amount of any net obligation under any Swap Contract on any date shall be deemed to be the Swap Termination Value thereof as of such date. The amount of any capital lease or Synthetic Lease Obligation as of any date shall be deemed to be the amount of Attributable Indebtedness in respect thereof as of such date.

"Indemnified Taxes" means (a) Taxes, other than Excluded Taxes, imposed on or with respect to any payment made by or on account of any obligation of any Loan Party under any Loan Document and (b) to the extent not otherwise described in (a), Other Taxes.

"Indemnitees" has the meaning specified in Section 11.04(b).

"Information" has the meaning specified in Section 11.07.

"<u>Initial Financial Statements</u>" means the unqualified audited combined financial statements of the predecessor of the Acquired Businesses for the fiscal year ended December 31, 2012 and the unqualified unaudited combined financial statements of the Acquired Businesses for the period from January 1, 2013 to March 31, 2013.

"Initial L/C Issuer" means each of Wells Fargo Bank, Citibank, N.A., Deutsche Bank AG New York Branch, JPMorgan Chase Bank, N.A., Morgan Stanley Bank, N.A. and U.S. Bank National Association. Any Initial L/C Issuer may, in its discretion, arrange for one or more Letters of Credit to be issued by Affiliates of such Initial L/C Issuer, in which case the term "Initial L/C Issuer" shall include any such Affiliate with respect to Letters of Credit issued by such Affiliate.

"Interest Payment Date" means, (a) as to any Loan other than a Base Rate Loan, the last day of each Interest Period applicable to such Loan and the Maturity Date; provided, however, that if any Interest Period for a Eurodollar Rate Loan exceeds three months, the respective dates that fall every three months after the beginning, of such Interest Period shall also be Interest Payment Dates; (b) as to any Base Rate Loan, the last Business Day of each March, June, September and December and the Maturity Date; and (c) as to any Swing Line Loan, the Maturity Date thereof.

"Interest Period" means, as to each Eurodollar Rate Loan, the period commencing on the date such Eurodollar Rate Loan is disbursed or converted to or continued as a Eurodollar Rate Loan and ending on the date (a) one week, or one, two or three months thereafter, or (b) upon consent of all Lenders, twelve months thereafter, in any case as selected by the Borrower in its Loan Notice; <u>provided</u> that:

- (i) any Interest Period that would otherwise end on a day that is not a Business Day shall be extended to the next succeeding Business Day unless such Business Day falls in another calendar month, in which case such Interest Period shall end on the next preceding Business Day;
- (ii) any Interest Period that begins on the last Business Day of a calendar month (or on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period) shall end on the last Business Day of the calendar month at the end of such Interest Period; and
 - (iii) no Interest Period shall extend beyond the Maturity Date.

"Investment" means, as to any Person, any direct or indirect acquisition or investment by such Person, whether by means of (a) the purchase or other acquisition of capital stock or other securities of another Person, (b) a loan, advance or capital contribution to, Guarantee or assumption of debt of, or purchase or other acquisition of any other debt or equity participation or interest in, another Person, including any partnership or joint venture interest in such other Person, or (c) the purchase or other acquisition (in one transaction or a series of transactions) of assets of another Person that constitute a line of business. For purposes of covenant compliance, the amount of any Investment shall be the amount actually invested, without adjustment for subsequent increases or decreases in the value of such Investment.

"IPO" means the initial underwritten public offering of Equity Interests in the Parent Guarantor pursuant to the Registration Statement

"IRS" means the United States Internal Revenue Service.

"ISP" means, with respect to any Letter of Credit, the "International Standby Practices 1998" published by the Institute of International Banking Law & Practice (or such later version thereof as may be in effect at the time of issuance).

"Issuer Documents" means with respect to any Letter of Credit, the Letter of Credit Application, and any other document, agreement and instrument entered into by the L/C Issuer that is the issuer of such Letter of Credit and the Borrower (or any Subsidiary) or in favor of such L/C Issuer and relating to any such Letter of Credit.

"Joint Bookrunners" means Wells Fargo Securities LLC, Citigroup Global Markets, Inc., and U.S. Bank National Association, in their respective capacities as joint bookrunners.

"<u>Joint Venture</u>" means an entity, other than a Subsidiary, in which any Loan Party owns directly or indirectly five percent (5%) or more of any class of ownership interests. As of the Execution Date and as of the Closing Date, the only Joint Venture is Three Rivers Gathering, L.L.C.

"Laws" means, collectively, all international, foreign, Federal, state and local statutes, treaties, rules, guidelines, regulations, ordinances, codes and administrative or judicial precedents or authorities, including the interpretation or administration thereof by any Governmental Authority charged with the enforcement, interpretation or administration thereof, and all applicable administrative orders, directed duties, requests, licenses, authorizations and permits of, and agreements with, any Governmental Authority, in each case whether or not having the force of law.

"L/C Advance" means, with respect to each Lender, such Lender's funding of its participation in any L/C Borrowing in accordance with its Pro Rata Share.

"L/C Borrowing" means an extension of credit resulting from a drawing under any Letter of Credit which has not been reimbursed on the date when made or refinanced as a Borrowing.

"L/C Commitment" has the meaning set forth in the definition of L/C Issuer.

"L/C Credit Extension" means, with respect to any Letter of Credit, the issuance thereof or extension of the expiry date thereof, or the increase of the amount thereof.

"L/C Issuer" means with respect to each Letter of Credit issued hereunder, each Initial L/C Issuer or such other Lender that has issued or agreed to issue such Letter of Credit at the request of the Borrower and that is reasonably acceptable to the Administrative Agent (such acceptance not to be unreasonably withheld, conditioned or delayed), in its capacity as the issuer of such Letter of Credit, or any successor issuer of Letters of Credit hereunder. The dollar amount of the commitment of each L/C Issuer to issue Letters of Credit hereunder (the "L/C Commitment") shall be as agreed in writing between the Borrower, such L/C Issuer and the Administrative Agent. As of the Closing Date, the L/C Commitment of each of Wells Fargo Bank, Citibank, N.A., JPMorgan Chase Bank, N.A., and U.S. Bank National Association is \$8,500,000.00, and the L/C Commitment of each of Deutsche Bank AG New York Branch and Morgan Stanley Bank, N.A. is \$8,000,000.00. Any L/C Issuer may, in its discretion, arrange for one or more Letters of Credit to be issued by Affiliates of such L/C Issuer, in which case the term "L/C Issuer" shall include any such Affiliate with respect to Letters of Credit issued by such Affiliate. As used herein, the term "the L/C Issuer" shall mean "each L/C Issuer" or "the applicable L/C Issuer," as the context may require.

"L/C Obligations" means, as at any date of determination, the aggregate undrawn amount of all outstanding Letters of Credit plus the aggregate of all Unreimbursed Amounts, including all L/C Borrowings. For all purposes of this Agreement, if on any date of determination a Letter of Credit has expired by its terms but any amount may still be drawn thereunder by reason of the operation of Rule 3.14 of the ISP, such Letter of Credit shall be deemed to be "outstanding" in the amount so remaining available to be drawn.

"Lender" has the meaning specified in the introductory paragraph hereto and, unless the context otherwise requires, includes each L/C Issuer and the Swing Line Lender.

"<u>Lending Office</u>" means, as to any Lender, the office or offices of such Lender described as such in such Lender's Administrative Questionnaire, or such other office or offices as a Lender may from time to time notify the Borrower and the Administrative Agent.

"Letter of Credit" means any standby letter of credit issued hereunder.

"<u>Letter of Credit Application</u>" means an application and agreement for the issuance or amendment of a Letter of Credit in the form from time to time in use by the applicable L/C Issuer.

"<u>Letter of Credit Expiration Date</u>" means the day that is seven days prior to the Maturity Date (or, if such day is not a Business Day, the next preceding Business Day).

"Letter of Credit Fee" has the meaning specified in Section 2.03(i).

"<u>Letter of Credit Sublimit</u>" means an amount equal to the lesser of (a) \$50,000,000 and (b) the Aggregate Commitments. The Letter of Credit Sublimit is part of, and not in addition to, the Aggregate Commitments.

"<u>Lien</u>" means any mortgage, pledge, hypothecation, assignment, deposit arrangement, encumbrance, lien (statutory or other), charge, or preference, priority or other security interest or preferential arrangement in the nature of a security interest of any kind or nature whatsoever (including any conditional sale or other title retention agreement, and any financing lease having substantially the same economic effect as any of the foregoing).

"Loan" has the meaning specified in Section 2.01.

"<u>Loan Documents</u>" means this Agreement, each Note, each Issuer Document, the Fee Letters, the Letters of Credit, the Collateral Documents and each other document executed and delivered by a Loan Party or a Subsidiary in connection therewith that is designated as a Loan Document therein. "Loan Documents" do not include any Secured Hedge Agreement or any Secured Cash Management Agreement.

"<u>Loan Notice</u>" means a written notice of (a) a Borrowing, (b) a conversion of Loans from one Type to the other, or (c) a continuation of Eurodollar Rate Loans, pursuant to <u>Section 2.02(a)</u>, which shall be substantially in the form of <u>Exhibit A-1</u>.

"Loan Parties" means the Borrower, the Parent Guarantor and each Subsidiary Guarantor.

"Material Adverse Effect" means (a) a material adverse effect on the property, business, operations, liabilities, condition (financial or otherwise) or operating results of (x) the Borrower and the Restricted Subsidiaries taken as a whole or (y) the Parent Guarantor, the Borrower and its Restricted Subsidiaries taken as a whole, (b) a material impairment of the ability of the Borrower or any other Loan Party to perform any of its material obligations under any Loan Document, or (c) a material impairment of any right or remedy of or benefit available to the Lenders or the Administrative Agent under any Loan Document.

"<u>Material Agreement</u>" means any agreement or contract to which a Loan Party or any Restricted Subsidiary is a party, which, if terminated or cancelled (other than an expiration in accordance with its terms), would reasonably be expected to result in a Material Adverse Effect.

"<u>Material Project</u>" means any capital project of the Borrower or any Restricted Subsidiary the cost of which is reasonably expected by the Borrower to exceed, or exceeds, \$10,000,000.

"Material Project Consolidated EBITDA Adjustment" means:

(i) prior to the Commercial Operation Date of such Material Project (but including the fiscal quarter in which such Commercial Operation Date occurs), a percentage (equal to the then-current completion percentage of such Material Project) of an amount to be approved by the Administrative Agent (such approval not to be unreasonably withheld or delayed) as the projected Consolidated EBITDA attributable to such Material Project for the first 12-month period following the scheduled Commercial Operation Date of such Material Project (such amount to be determined based on projected revenues from customer contracts relating to such Material Project, the creditworthiness and applicable projected production of the prospective customers, capital and other costs, operating and administrative expenses, scheduled Commercial Operation Date, commodity price assumptions and other factors reasonably deemed appropriate by the Administrative Agent), which may, at the Borrower's option (subject to Administrative Agent's review and approval (such approval not to be unreasonably withheld or delayed) of each component of such Material Project Consolidated EBITDA Adjustment), be added to actual Consolidated EBITDA for the fiscal quarter in which construction of such Material Project

commences and for each fiscal quarter thereafter until the Commercial Operation Date of such Material Project (including the fiscal quarter in which such Commercial Operation Date occurs, but net of any actual Consolidated EBITDA attributable to such Material Project following such Commercial Operation Date); provided that if the actual Commercial Operation Date does not occur by the scheduled Commercial Operation Date, then the foregoing amount shall be reduced, for quarters ending after the scheduled Commercial Operation Date to (but excluding) the first full quarter after its actual Commercial Operation Date, by the following percentage amounts depending on the period of delay (based on the period of actual delay or then-estimated delay, whichever is longer): (i) 90 days or less, 0%, (ii) longer than 90 days, but not more than 180 days, 25%, (iii) longer than 180 days but not more than 270 days, but not more than 365 days, 75% and (v) longer than 365 days, 100%; and

(ii) beginning with the first full fiscal quarter following the Commercial Operation Date of a Material Project and for the two immediately succeeding fiscal quarters, an amount to be approved by the Administrative Agent as the projected Consolidated EBITDA attributable to such Material Project (determined in the same manner as set forth in clause (i) above) for the balance of the four full fiscal quarter period following such Commercial Operation Date, which may, at Borrower's option, be added to actual Consolidated EBITDA for such fiscal quarters (but net of any actual Consolidated EBITDA attributable to such Material Project following such Commercial Operation Date).

"Material Real Property" has the meaning set forth in Section 6.11(c).

"Material Subsidiary" means, at any time, each of the following: (a) QEP Gathering I, LLC, (b) Rendezvous Gas, and (b) each other Subsidiary that is not an Immaterial Subsidiary at such time.

"Maturity Date" means August 14, 2018; provided that the Maturity Date for any Swing Line Loan shall be the earlier of such date and five Business Days after such Swing Line Loan is made.

"Minimum Collateral Amount" means, at any time, (a) with respect to Cash Collateral consisting of cash or deposit account balances, an amount equal to the Fronting Exposure of the L/C Issuers with respect to Letters of Credit issued and outstanding at such time and (b) otherwise, an amount determined by the Administrative Agent and the applicable L/C Issuers in their sole discretion.

"Moody's" means Moody's Investors Service, Inc. and any successor thereto.

"<u>Multiemployer Plan</u>" means any employee benefit plan of the type described in Section 4001(a)(3) of ERISA, to which the Borrower or any ERISA Affiliate makes or is obligated to make contributions, or during the preceding five plan years, has made or been obligated to make contributions.

"Net Cash Proceeds" means

- (a) with respect to any Asset Sale or Casualty Event, the cash proceeds (including cash proceeds subsequently received (as and when received) in respect of noncash consideration initially received), net of:
- (i) selling expenses (including reasonable broker's fees or commissions, legal fees, transfer and similar Taxes and the Borrower's good faith estimate of income Taxes paid or payable by the Borrower in connection with such sale),

- (ii) amounts provided as a reserve, in accordance with GAAP, against (x) any liabilities under any indemnification obligations associated with such Asset Sale or Casualty Event or (y) any other liabilities retained by Borrower or any of its Restricted Subsidiaries associated with the properties sold or transferred in such Asset Sale or Casualty Event (provided that, to the extent and at the time any such amounts are released from such reserve, such amounts shall constitute Net Cash Proceeds);
- (iii) Borrower's good faith estimate of payments required to be made with respect to unassumed liabilities relating to the properties sold or transferred within 90 days of such Asset Sale or Casualty Event (provided that, to the extent such cash proceeds are not used to make payments in respect of such unassumed liabilities within 90 days of such Asset Sale or Casualty Event, such cash proceeds shall constitute Net Cash Proceeds);
- (iv) the principal amount, premium or penalty, if any, interest and other amounts on any Indebtedness for borrowed money which is secured by a Permitted Lien on the properties sold or transferred in such Asset Sale or Casualty Event and which is repaid with such proceeds (other than any such Indebtedness assumed by the purchaser of such properties); and
- (v) in the case of a Casualty Event, all reasonable costs and expenses incurred in connection with the collection of proceeds, awards or other compensation in respect of a Casualty Event;

provided, however, that, if the Borrower shall deliver a certificate of a Responsible Officer to the Administrative Agent not later than the seventh Business Day following the receipt by the Borrower or any Restricted Subsidiary of such cash proceeds setting forth the Borrower's intent to reinvest (or to cause its Restricted Subsidiaries to reinvest) such proceeds in productive assets of a kind then used or usable in the business of the Borrower or any Restricted Subsidiary (or 100% of the Equity Interests of any entity that shall become a Restricted Subsidiary hereunder that owns such productive assets) within 360 days of receipt of such proceeds and in each case such proceeds are used for such reinvestment within such 360 day period (or, if committed to be so used within such period, are so reinvested within a further 180 days thereafter) and (y) no Event of Default shall have occurred and shall be continuing at the time of such certificate or at the proposed time of the application of such proceeds, such proceeds shall not constitute Net Cash Proceeds except to the extent not so used at the end of such period, at which time such proceeds shall be deemed to be Net Cash Proceeds; and

(b) with respect to any issuance or incurrence of Indebtedness, the cash proceeds thereof, net of all Taxes and reasonable and customary fees, commissions, costs and other expenses incurred in connection therewith.

"Non-Wholly Owned" when used with respect to a Subsidiary, means a Subsidiary that is not Wholly Owned.

"Note" means a promissory note made by the Borrower in favor of a Lender evidencing Loans made by such Lender, substantially in the form of Exhibit B.

"Obligations" means all advances to, and debts, liabilities, obligations, covenants and duties of the Borrower or any other Loan Party arising under any Loan Document or otherwise with respect to any Loan, Swing Line Loan or Letter of Credit and including all L/C Obligations, whether such Obligations are direct or indirect (including those acquired by assumption), absolute or contingent, due or to become

"OFAC" means the U.S. Department of the Treasury's Office of Foreign Assets Control.

"Organization Documents" means, (a) with respect to any corporation, the certificate or articles of incorporation and the bylaws (or equivalent or comparable constitutive documents with respect to any non-U.S. jurisdiction); (b) with respect to any limited liability company, the certificate or articles of formation or organization and operating agreement; and (c) with respect to any partnership, joint venture, trust or other form of business entity, the partnership, joint venture or other applicable agreement of formation or organization and any agreement, instrument, filing or notice with respect thereto filed in connection with its formation or organization with the applicable Governmental Authority in the jurisdiction of its formation or organization and, if applicable, any certificate or articles of formation or organization of such entity.

"Other Connection Taxes" means, with respect to any Recipient, Taxes imposed solely as a result of a present or former connection between such Recipient and the jurisdiction imposing such Tax (other than connections arising from such Recipient having executed, delivered, become a party to, performed its obligations under, received payments under, received or perfected a security interest under, engaged in any other transaction pursuant to or enforced any Loan Document, or sold or assigned an interest in any Loan Document).

"Other Taxes" means all present or future stamp, court or documentary, intangible, recording, filing or similar Taxes that arise from any payment made under, from the execution, delivery, performance, enforcement or registration of, from the receipt or perfection of a security interest under, or otherwise with respect to, any Loan Document, except any such Taxes that are Other Connection Taxes imposed with respect to an assignment (other than an assignment made pursuant to Section 11.14).

"Outstanding Amount" means (i) with respect to Loans on any date, the aggregate outstanding principal amount thereof after giving effect to any borrowings and prepayments or repayments of Loans, as the case may be, occurring on such date; (ii) with respect to any L/C Obligations on any date, the amount of such L/C Obligations on such date after giving effect to any L/C Credit Extension occurring on such date and any other changes in the aggregate amount of the L/C Obligations as of such date, including as a result of any reimbursements of outstanding unpaid drawings under any Letters of Credit or any reductions in the maximum amount available for drawing under Letters of Credit taking effect on such date; and (iii) with respect to Swing Line Loans, the aggregate outstanding principal amount thereof after giving effect to any borrowing or repayment of Swing Line Loans, as the case may be, occurring on such date.

"Parent Guarantor" has the meaning specified in the introductory paragraph hereto.

"Participant" has the meaning specified in Section 11.06(d).

"Participant Register" has the meaning specified in Section 11.06(f).

"Payment In Full" means payment in full in cash of all Secured Obligations (other than contingent obligations for which no claim has been made), termination of the Commitments of all Lenders and L/C Issuers, and termination of all Letters of Credit (or Cash Collateralization thereof as acceptable to the applicable L/C Issuers).

"PBGC" means the Pension Benefit Guaranty Corporation.

"Pension Funding Rules" means the rules of the Code and ERISA regarding minimum required contributions (including any installment payment thereof) to Pension Plans and set forth in, with respect to plan years ending prior to the effective date of the Pension Protection Act of 2006, as amended ("Pension Act"), Section 412 of the Code and Section 302 of ERISA, each as in effect prior to the Pension Act and, thereafter, Section 412, 430, 431, 432, and 436 of the Code and Sections 302, 303, 304 and 305 of ERISA.

"Pension Plan" means any "employee pension benefit plan" (as such term is defined in Section 3(2) of ERISA), other than a Multiemployer Plan, that is subject to Title IV of ERISA and is sponsored or maintained by the Borrower or any ERISA Affiliate or to which the Borrower or any ERISA Affiliate contributes or has an obligation to contribute, or in the case of a multiple employer or other plan described in Section 4064(a) of ERISA, has made contributions at any time during the immediately preceding five plan years.

"<u>Permitted Acquisition</u>" means the acquisition by a Loan Party or a Restricted Subsidiary of assets of a Person constituting a line of business of such Person, or not less than 100% of the Equity Interests (other than directors' qualifying shares) of a Person (each such Person being referred to herein as the "<u>Acquired Entity</u>"), whether from QRI or a Subsidiary thereof or from any other Person; <u>provided</u> that (a) at the time of such acquisition (i) both immediately before and after giving effect thereto, no Default shall have occurred and be continuing; (ii) immediately before and after consummating such acquisition, the Parent Guarantor shall be in Pro Forma Compliance with the Financial Covenants; (iii) if the Acquired Entity becomes a Material Subsidiary as a result of the acquisition, the Borrower and any applicable Restricted Subsidiary shall comply, and shall cause the Acquired Entity to comply, with <u>Section 6.11</u>.

"Permitted Affiliate Acquisition" means the acquisition by a Loan Party or a Restricted Subsidiary from QRI or a Subsidiary thereof of less than 100% of the Equity Interests or assets of a Person (each such Person being referred to herein as the "Acquired Entity"); provided that (a) at the time of such acquisition (i) both immediately before and after giving effect thereto, no Default shall have occurred and be continuing; (ii) immediately before and after consummating such acquisition, the Parent Guarantor shall be in Pro Forma Compliance with the Financial Covenants; (iii) all Equity Interests in the Acquired Entity that are not being acquired as part of such acquisition shall be owned by QRI or a Subsidiary thereof; (b) such Loan Party or Restricted Subsidiary shall pledge such Equity Interests to the Administrative Agent pursuant to a Collateral Document; and (c) the aggregate consideration for all Permitted Affiliate Acquisitions by all Loan Parties and Restricted Subsidiaries shall not exceed \$250,000,000.

"Permitted Investments" means:

(a) direct obligations of, or obligations the principal of and interest on which are unconditionally guaranteed by, the United States of America (or by any agency thereof to the extent such obligations are backed by the full faith and credit of the United States of America), in each case maturing within twelve months from the date of acquisition thereof;

- (b) Investments in commercial paper maturing within twelve months from the date of acquisition thereof and having, at such date of acquisition, the highest credit rating obtainable from S&P or from Moody's;
- (c) investments in certificates of deposit, banker's acceptances and time deposits maturing within twelve months from the date of acquisition thereof issued or guaranteed by or placed with, and demand, savings and money market deposit accounts issued or offered by, the Administrative Agent or any Affiliate of the Administrative Agent, any Arranger or any Affiliate of any Arranger or any domestic office of any commercial bank organized under the laws of the United States of America or any State thereof that has a combined capital and surplus and undivided profits of not less than \$500,000,000 and that issues (or the parent of which issues) commercial paper rated at least "Prime-2" (or the then equivalent grade) by Moody's or "A-2" (or the then equivalent grade) by S&P;
- (d) fully collateralized repurchase agreements with a term of not more than 30 days for securities described in clause (a) above and entered into with a financial institution satisfying the criteria of clause (c) above; and
- (e) investments in "money market funds" within the meaning of Rule 2a-7 of the Investment Company Act of 1940, as amended, at least 95% of whose assets are invested in investments of the type described in clauses (a) through (d) above.

"Permitted Liens" means:

- (a) statutory Liens for taxes, assessments and governmental charges, statutory mechanics', materialmen's, carriers', workman's and warehousemen's Liens, and other similar statutory Liens, <u>provided</u> that in each case under this subparagraph (a), such Liens secure only indebtedness, liabilities and obligations which are not delinquent for a period of more than 60 days or which are being contested by appropriate proceedings and for which adequate reserves are provided on the books of the applicable Loan Party or Restricted Subsidiary;
 - (b) Liens on the office facilities of the Loan Parties and Restricted Subsidiaries;
 - (c) Liens to secure the Obligations;
- (d) Liens on any property or asset existing prior to the acquisition thereof pursuant to an Acquisition by a Loan Party and Liens on any property or asset of any Person that becomes a Restricted Subsidiary after the Closing Date if such Liens exist prior to the time such Person becomes a Restricted Subsidiary, provided that (i) such Liens secure only Indebtedness permitted by Section 7.01(c), (ii) such Liens are not created in contemplation of or in connection with such acquisition or such Person becoming a Restricted Subsidiary, (iii) such Liens shall not apply to any other property or assets of any Loan Party or any Restricted Subsidiary, (iv) such Liens shall secure only those obligations which it secured on the date of such acquisition or the date such Person becomes a Restricted Subsidiary, and any Permitted Refinancing Indebtedness in respect of the foregoing, and (v) such Liens may not cover any Pipeline Real Property other than Pipeline Real Property acquired after the Closing Date pursuant to an Acquisition;
- (e) (i) pledges of cash and cash equivalents incurred or deposits made to secure obligations (other than Indebtedness) under workers' compensation laws or similar legislation or to secure public or statutory obligations, in each case in the ordinary course of business, or deductibles, self-insurance, insurance premiums, co-payment, coinsurance, retention and similar obligations to providers of insurance in respect of such obligations in the ordinary course of business; and

- (ii) pledges of cash and cash equivalents to secure the obligations of any one or more Loan Parties or Restricted Subsidiaries with respect to Swap Contracts permitted by this Agreement, provided that the aggregate amount of cash and cash equivalents encumbered by Liens permitted by this clause (e)(ii) shall not exceed at any time \$25,000,000;
- (f) encumbrances consisting of easements, restrictions, servitudes, permits, conditions, covenants, exceptions or reservations in any Real Property of a Loan Party or any Restricted Subsidiary for the purpose of roads, pipelines, transmission lines, transportation lines and distribution lines, or otherwise imposed by law or arising in the ordinary course of business that do not secure Indebtedness or other monetary obligations and, in the aggregate, are not substantial in amount and do not materially impair the use of such property by such Loan Party or Restricted Subsidiary in the operation of its business and which do not in any case materially detract from the value of the property subject thereto;
- (g) deposits made in the ordinary course of business to secure the performance of bids, trade contracts (other than for debt for borrowed money), leases (other than Indebtedness) and surety bonds;
- (h) Liens securing Indebtedness of one or more Loan Parties or Restricted Subsidiaries permitted by Section 7.01(b) to finance the acquisition, construction or improvement of fixed or capital assets, provided that (i) such Liens and the Indebtedness secured thereby are incurred prior to or within 180 days after such acquisition or the completion of construction or improvement of such fixed or capital assets, (ii) such Liens do not at any time encumber any property other than the property financed by such Indebtedness, (iii) the amount of Indebtedness initially secured thereby is not more than 100% of the purchase price or cost of construction or improvement of such fixed or capital asset and (iv) such Liens may not cover any Pipeline Real Property other than Pipeline Real Property acquired after the Closing Date pursuant to an Acquisition;
- (i) the interest or title of a lessor under any lease entered into by a Loan Party or any Restricted Subsidiary in the ordinary course of its business and covering only the assets so leased;
 - (j) Liens with respect to judgments which do not result in an Event of Default under Section 8.01(h);
- (k) Liens not securing Indebtedness arising solely by virtue of any statutory or common law provision relating to banker's liens, rights of set-off or similar rights and remedies and burdening only deposit accounts or other funds maintained with a creditor depository institution, <u>provided</u> that no such deposit account is a dedicated cash collateral account or is subject to restrictions against access by the depositor in excess of those set forth by regulations promulgated by the FRB and no such deposit account is intended by a Loan Party or any Restricted Subsidiary to provide collateral to the depository institution;
- (l) (i) Liens on Equity Interests in an Unrestricted Subsidiary securing Indebtedness of such Unrestricted Subsidiary (or its parent company that is an Unrestricted Subsidiary) that is non-recourse to the Loan Parties and the Restricted Subsidiaries and (ii) Liens on Equity Interests in a Joint Venture, owned by a Loan Party or Restricted Subsidiary, that are in favor of the other holders of Equity Interests in such Joint Venture to secure the performance of such Loan Party's or Restricted Subsidiary's obligations under the joint venture agreement governing such joint venture;

- (m) Liens constituting earnest money deposits made by a Loan Party or any Restricted Subsidiary in connection with any letter of intent or purchase agreement;
- (n) Liens created or evidenced by or resulting from precautionary financing statements filed by lessors of property (but only related to the leased property), other than in connection with capital leases and sale and leaseback transactions; and
- (o) Liens securing Indebtedness of one or more Loan Parties or Restricted Subsidiaries, which Liens are not otherwise permitted by <u>clauses</u> (a) through (m) above, <u>provided</u> that (x) the aggregate outstanding principal amount of the Indebtedness of all Loan Parties and Restricted Subsidiaries secured thereby, when added to the aggregate principal amount of Indebtedness permitted by <u>Section 7.01(j)</u> (without duplication), does not exceed at any time an amount equal to the greater of \$20,000,000 and 3% of Consolidated Net Tangible Assets of the Parent Guarantor, and (y) no such Liens shall encumber Pipeline Real Property;

<u>provided</u> that nothing in this definition shall in and of itself constitute or be deemed to constitute an agreement or acknowledgment by the Administrative Agent or any Lender that the Indebtedness subject to or secured by any such Permitted Lien ranks (apart from the effect of any Lien included in or inherent in any such Permitted Liens) in priority to the Obligations.

"Permitted Refinancing Indebtedness" means Indebtedness issued in exchange for, or proceeds of which are used to extend, refinance, renew, replace, defease or refund (collectively, to "Refinance"), the Indebtedness being Refinanced (or previous refinancings thereof constituting Permitted Refinancing Indebtedness), provided, that (i) the principal amount of the Permitted Refinancing Indebtedness does not exceed the principal amount of the Indebtedness being Refinanced except by an amount equal to unpaid accrued interest and premium thereon plus other reasonable amounts paid, and fees and expenses reasonably incurred, in connection with such modification, refinancing, refunding, renewal or extension and by an amount equal to any existing commitments unutilized thereunder; (ii) the average life to maturity of such Permitted Refinancing Indebtedness is greater than or equal to that of the Indebtedness being Refinanced; (iii) if the Indebtedness being Refinanced is subordinated in right of payment to the Obligations, such Permitted Refinancing Indebtedness is subordinated in right of payment to the Obligations on terms at least as favorable to the Lenders as those contained in the documentation governing the Indebtedness being Refinanced; (iv) no Permitted Refinancing Indebtedness shall have different obligors, or greater guarantees or security, than the Indebtedness being Refinanced; and (v) at the time of incurrence of such Permitted Refinancing Indebtedness, no Event of Default shall have occurred and be continuing.

"Permitted Unsecured Indebtedness" means Indebtedness permitted by Section 7.01(e).

"Person" means any natural person, corporation, limited liability company, trust, joint venture, association, company, partnership, Governmental Authority or other entity.

"<u>Pipeline Real Property</u>" shall mean all easements, rights of way, and other Real Property or other interests therein comprising any part of the pipelines, including transmission pipelines and gathering pipelines, of any Loan Party or any Subsidiary, or upon which any part of such transmission and/or gathering pipelines has been built, passes over or through or which is used in or reasonably necessary for the operation thereof.

"Plan" means any "employee benefit plan" (as such term is defined in Section 3(3) of ERISA) established by the Borrower or, with respect to any such plan that is subject to Section 412 of the Code or Title IV of ERISA, any ERISA Affiliate.

"Platform" has the meaning specified in Section 6.01.

"Pro Forma Compliance" means, on any date (the "Determination Date") when used in connection with the Financial Covenants, that the Parent Guarantor is in compliance with the covenant levels set forth in Section 7.13 as of the most recent Quarter-End Date, in each case calculated on a pro forma basis to give effect to (i) Indebtedness as of such date of determination and any incurrence of Indebtedness on the Determination Date and (ii) any Acquisition occurring after such Quarter-End Date, in each case, as if such events had occurred on the first day of the four fiscal quarter period ended on such Quarter-End Date.

"Pro Rata Share" means, with respect to each Lender at any time, a fraction (expressed as a percentage, carried out to the ninth decimal place), the numerator of which is the amount of the Commitment of such Lender at such time and the denominator of which is the amount of the Aggregate Commitments at such time; provided that if the commitment of each Lender to make Credit Extensions and the obligation of the L/C Issuer to make L/C Credit Extensions have been terminated pursuant to Section 8.02, then the Pro Rata Share of each Lender shall be determined based on the Pro Rata Share of such Lender immediately prior to such termination and after giving effect to any subsequent assignments made pursuant to the terms hereof. The initial Pro Rata Share of each Lender is set forth opposite the name of such Lender on Schedule 2.01 or in the Assignment and Assumption pursuant to which such Lender becomes a party hereto, as applicable.

"Public Lender" has the meaning specified in Section 6.01.

"QEPM Gathering" means QEPM Gathering I, LLC, a Delaware limited liability company.

"QRI" means QEP Resources, Inc., a Delaware corporation.

"QRI Credit Agreement" means the Credit Agreement referenced in Section 4.02(i).

"Qualified ECP Guarantor" means, in respect of any Swap Obligation, each Loan Party that has total assets exceeding \$10,000,000 at the time the relevant Guarantee or grant of the relevant Lien becomes effective with respect to such Swap Obligation or such other Person as constitutes an "eligible contract participant" under the Commodity Exchange Act and can cause another person to qualify as an "eligible contract participant" at such time by entering into a keepwell under Section 1a(18)(A)(v)(II) of the Commodity Exchange Act.

"Qualified Notes Offering" means issuance by the Parent Guarantor or the Borrower of unsecured Indebtedness that satisfies the requirements of Section 7.01(e) and the gross proceeds of which equal \$200,000,000 or more.

"Qualifying Acquisition" shall mean any Acquisition by the Parent, the Borrower or a Restricted Subsidiary for a purchase price in excess of \$50,000,000.

"Quarter-End Date" means the last day of the fiscal quarter of the Parent Guarantor most recently ended for which financial statements are available to the Parent Guarantor or for which financial statements are required to be delivered pursuant to Section 6.01(a) or 6.01(b).

"<u>Real Property</u>" means, collectively, all right, title and interest in and to any and all parcels of or interests in real property owned or leased by any Person, together with, in each case, all easements, hereditaments and appurtenances relating thereto, all improvements and appurtenant fixtures and equipment, all general intangibles and contract rights and other property and rights incidental to the ownership, lease or operation thereof.

"Recipient" means (a) the Administrative Agent, (b) any Lender and (c) any L/C Issuer, as applicable.

"Register" has the meaning specified in Section 11.06(c).

"Registration Statement" means the Registration Statement on Form S-1, under the Exchange Act, of the Parent Guarantor filed with the SEC on May 9, 2013 as amended by all changes thereto filed prior to the Execution Date, including the exhibits filed therewith, as amended and supplemented by Acceptable Changes.

"Related Parties" means, with respect to any Person, such Person's Affiliates and the partners, directors, officers, employees, agents, trustees and advisors of such Person and of such Person's Affiliates.

"Remaining Present Value" shall mean, as of any date with respect to any lease, the present value as of such date of the scheduled future lease payments with respect to such lease, determined with a discount rate equal to a market rate of interest for such lease reasonably determined at the time such lease was entered into.

"Rendezvous Gas" means Rendezvous Gas Services, L.L.C., a Delaware limited liability company.

"Reportable Event" means any of the events set forth in Section 4043(c) of ERISA, other than events for which the 30 day notice period has been waived.

"Request for Credit Extension" means (a) with respect to a Borrowing, conversion or continuation of Loans, a Loan Notice, and (b) with respect to an L/C Credit Extension, a Letter of Credit Application, and (c) with respect to a Swing Line Borrowing, a Swing Line Loan Notice.

"Required Lenders" means, as of any date of determination, Lenders having more than 50% of the Aggregate Commitments or, if the commitment of each Lender to make Credit Extensions and the obligation of each L/C Issuer to make L/C Credit Extensions have been terminated pursuant to Section 8.02, Lenders holding in the aggregate more than 50% of the Total Outstandings (with the aggregate amount of each Lender's risk participation and funded participation in L/C Obligations and Swing Line Loans being deemed "held" by such Lender for purposes of this definition); provided that the Commitment of, and the portion of the Total Outstandings held or deemed held by, any Defaulting Lender shall be excluded for purposes of making a determination of Required Lenders.

"Responsible Officer" of a Person means the chairman of the board, chief executive officer, president or chief financial officer of such Person (or, in the case of a Partnership, of its general partner or in the case of the Borrower, to the extent the Borrower is managed by the Parent Guarantor, of the Parent Guarantor). Any document delivered hereunder that is signed by a Responsible Officer of a Loan Party shall be conclusively presumed to have been authorized by all necessary corporate, partnership and/or other action on the part of such Loan Party and such Responsible Officer shall be conclusively presumed to have acted on behalf of such Loan Party. Unless otherwise specifically stated herein, references to "Responsible Officer" shall mean "Responsible Officer of the Borrower".

"Restricted Payment" means any dividend or other distribution (whether in cash, securities or other property) with respect to any capital stock or other equity interest of the Borrower or any Subsidiary, or any payment (whether in cash, securities or other property), including any sinking fund or similar deposit, on account of the purchase, redemption, retirement, acquisition, cancellation or termination of any such capital stock or other equity interest or of any option, warrant or other right to acquire any such capital stock or other equity interest.

"Restricted Subsidiary" means any Subsidiary of the Parent Guarantor that is not an Unrestricted Subsidiary.

"S&P" means Standard & Poor's Ratings Services, a division of The McGraw-Hill Companies, Inc. and any successor thereto.

"<u>Sanctioned Country</u>" means a country subject to a sanctions program identified on the list maintained by OFAC and available at http://www.treasury.gov/resource-center/sanctions/Programs/ Pages/Programs.aspx, or as otherwise published from time to time.

"Sanctioned Person" means (a) a Person named on the list of "Specially Designated Nationals and Blocked Persons" maintained by OFAC available at http://www.treasury.gov/resource-center/sanctions/SDN-List/Pages/default.aspx, or as otherwise published from time to time, or (b) (i) an agency of the government of a Sanctioned Country, (ii) an organization controlled by a Sanctioned Country, or (iii) a person resident in a Sanctioned Country, to the extent subject to a sanctions program administered by the U.S. Department of the Treasury's Office of Foreign Assets Control.

"SEC" means the Securities and Exchange Commission, or any Governmental Authority succeeding to any of its principal functions.

"Secured Cash Management Agreement" means any Cash Management Agreement that is entered into by and between any Loan Party and any Cash Management Bank.

"Secured Hedge Agreement" means each Swap Contract entered into by and between any Loan Party or any Restricted Subsidiary and any Person (a "Hedge Bank") that is, at the time such Swap Contract is entered into, a Lender or an Affiliate of a Lender.

"Secured Obligations" means, collectively, (a) the Obligations and (b) all other advances to, and debts, liabilities, obligations, covenants and duties of the Borrower or any other Loan Party arising under any Secured Hedge Agreement or under any Secured Cash Management Agreement, in each case including interest and fees that accrue after the commencement by or against any Loan Party or any Affiliate thereof of any proceeding under any Debtor Relief Laws naming such Person as the debtor in such proceeding, regardless of whether such interest and fees are allowed claims in such proceeding; provided that solely with respect to any Loan Party that is not an "eligible contract participant" under the Commodity Exchange Act, Excluded Swap Obligations of such Loan Party shall in any event be excluded from "Secured Obligations" owing by such Loan Party.

"Secured Parties" means, collectively, the Administrative Agent, the L/C Issuers, the Lenders, the Hedge Banks and the Cash Management Banks.

"Security Agreements" means the Security Agreement dated as of even date with this Agreement, executed by the Loan Parties in favor of the Administrative Agent for the ratable benefit of the Secured Parties, together with any other security agreements that may from time to time be executed in favor of the Administrative Agent for the ratable benefit of the Secured Parties, by any Person to secure the Secured Obligations, in each case which shall be in form and substance acceptable to the Administrative Agent.

"Solvent" and "Solvency" mean, with respect to any Person on any date of determination, that on such date (a) the fair value of the property of such Person is greater than the total amount of liabilities, including contingent liabilities, of such Person, (b) the present fair salable value of the assets of such Person is not less than the amount that will be required to pay the probable liability of such Person on its debts as they become absolute and matured, (c) such Person does not intend to, and does not believe that it will, incur debts or liabilities beyond such Person's ability to pay such debts and liabilities as they mature, (d) such Person is not engaged in business or a transaction, and is not about to engage in business or a transaction, for which such Person's property would constitute an unreasonably small capital, and (e) such Person is able to pay its debts and liabilities, contingent obligations and other commitments as they mature in the ordinary course of business. The amount of contingent liabilities at any time shall be computed as the amount that, in the light of all the facts and circumstances existing at such time, represents the amount that can reasonably be expected to become an actual or matured liability.

"Specified Non-Wholly Owned Subsidiary" means each of (i) Rendezvous Gas, for so long as Rendezvous Gas is a Non-Wholly Owned Subsidiary of the Parent Guarantor and (ii) each Affiliate Non-Wholly Owned Subsidiary, so long as such Subsidiary is a Non-Wholly Owned Subsidiary of the Parent Guarantor.

"Subsidiary" of a Person means a corporation, partnership, joint venture, limited liability company or other business entity of which a majority of the shares of securities or other interests having ordinary voting power for the election of directors or other governing body (other than securities or interests having such power only by reason of the happening of a contingency) are at the time beneficially owned, or the management of which is otherwise controlled, directly, or indirectly through one or more intermediaries, or both, by such Person. Unless otherwise specified, all references herein to a "Subsidiary" or to "Subsidiaries" shall refer to a direct or indirect Subsidiary or Subsidiaries of the Parent Guarantor.

"Subsidiary Guarantor" means, at any time, each Subsidiary that is party to a Subsidiary Guaranty Agreement as a Guarantor.

"Subsidiary Guaranty Agreement" means the guaranty agreement dated as of even date with this Agreement executed by the Subsidiary Guarantors for the benefit of the Administrative Agent and the Secured Parties, and any other guaranty agreements that may from time to time be executed by any Subsidiary.

"Swap Contract" means (a) any and all rate swap transactions, basis swaps, credit derivative transactions, forward rate transactions, commodity swaps, commodity options, forward commodity contracts, equity or equity index swaps or options, bond or bond price or bond index swaps or options or forward bond or forward bond price or forward bond index transactions, interest rate options, forward foreign exchange transactions, cap transactions, floor transactions, culrency swap transactions, cross-currency rate swap transactions, currency options, spot contracts, or any other similar transactions or any combination of any of the foregoing (including any options to enter into any of the foregoing), whether or not any such transaction is governed by or subject to any master agreement, and (b) any and all transactions of any kind, and the related confirmations, which are subject to the terms and

conditions of, or governed by, any form of master agreement published by the International Swaps and Derivatives Association, Inc., any International Foreign Exchange Master Agreement, or any other master agreement (any such master agreement, together with any related schedules, a "Master Agreement"), including any such obligations or liabilities under any Master Agreement.

"Swap Obligation" means, with respect to any Guarantor, any obligation to pay or perform under any agreement, contract or transaction that constitutes a "swap" within the meaning of Section 1a(47) of the Commodity Exchange Act.

"Swap Termination Value" means, in respect of any one or more Swap Contracts, after taking into account the effect of any legally enforceable netting agreement relating to such Swap Contracts, (a) for any date on or after the date such Swap Contracts have been closed out and termination value(s) determined in accordance therewith, such termination value(s), and (b) for any date prior to the date referenced in clause (a), the amount(s) determined as the mark-to-market value(s) for such Swap Contracts, as determined by the counterparties to such Swap Contracts.

"Swing Line Borrowing" means a borrowing of a Swing Line Loan pursuant to Section 2.04.

"Swing Line Lender" means Wells Fargo Bank in its capacity as provider of Swing Line Loans, or any successor swing line lender hereunder.

"Swing Line Loan" has the meaning specified in Section 2.04(a).

"Swing Line Loan Notice" means a notice of a Swing Line Borrowing pursuant to Section 2.04(b), which, if in writing, shall be substantially in the form of Exhibit A-2.

"Swing Line Rate" means either (a) the Base Rate plus the Applicable Rate for Base Rate Loans or (b) a rate to be established as provided on Schedule 2.04, as selected by the Borrower and specified in the Swing Line Loan Notice.

"Swing Line Sublimit" means an amount equal to the lesser of (a) \$25,000,000 and (b) the Aggregate Commitments. The Swing Line Sublimit is part of, and not in addition to, the Aggregate Commitments.

"Synthetic Lease Obligation" means the monetary obligation of a Person under (a) a so-called synthetic, off-balance sheet or tax retention lease, or (b) an agreement for the use or possession of property creating obligations that do not appear on the balance sheet of such Person but which, upon the insolvency or bankruptcy of such Person, would be characterized as the indebtedness of such Person (without regard to accounting treatment).

"<u>Taxes</u>" means all present or future taxes, levies, imposts, duties, deductions, withholdings (including backup withholding), assessments, fees or other charges imposed by any Governmental Authority, including any interest, additions to tax or penalties applicable thereto.

"Threshold Amount" means \$25,000,000.

"Total Outstandings" means the aggregate Outstanding Amount of all Loans, all L/C Obligations, and all Swing Line Loans.

"Transactions" means the Contribution and the IPO.

- "Type" means, with respect to a Loan, its character as a Base Rate Loan or a Eurodollar Rate Loan.
- "United States" and "U.S." mean the United States of America.
- "Unreimbursed Amount" has the meaning specified in Section 2.03(c)(i).
- "<u>Unrestricted Subsidiary</u>" means any Immaterial Subsidiary of the Parent Guarantor that is designated as an Unrestricted Subsidiary by the Borrower pursuant to a notice given to the Administrative Agent pursuant to <u>Section 7.06</u>. If at any time a Subsidiary that has been designated as an Unrestricted Subsidiary becomes a Material Subsidiary, from and after such time, such Subsidiary will cease to be an Unrestricted Subsidiary and shall be a Restricted Subsidiary as provided in <u>Section 7.06(c)(iv)</u>. As of the Execution Date and as of the Closing Date, there are no Unrestricted Subsidiaries.
 - "U.S. Person" means any Person that is a "United States Person" as defined in Section 7701(a)(30) of the Code.
 - "U.S. Tax Compliance Certificate" has the meaning assigned to such term in paragraph (ii)(B)(iii) of Section 3.01(g).
 - "Wells Fargo Bank" means Wells Fargo Bank, National Association and its successors.
- "Wholly Owned", when used with respect to a Subsidiary of a Person, means that all of the Equity Interests of such Subsidiary are directly or indirectly (through one or more Wholly Owned Subsidiaries) owned by such Person, excluding directors' qualifying shares that are required to be held by directors under applicable law.

"Withholding Agent" means any Loan Party and the Administrative Agent.

1.02 Other Interpretive Provisions.

With reference to this Agreement and each other Loan Document, unless otherwise specified herein or in such other Loan Document:

- (a) The meanings of defined terms are equally applicable to the singular and plural forms of the defined terms.
- (b) (i) The words "hereto," "hereof" and "hereunder" and words of similar import when used in any Loan Document shall refer to such Loan Document as a whole and not to any particular provision thereof.
 - (ii) Article, Section, Exhibit and Schedule references are to the Loan Document in which such reference appears.
 - (iii) The terms "include", "includes" "including" is by way of example and not limitation.
- (iv) The term "<u>documents</u>" includes any and all instruments, documents, agreements, certificates, notices, reports, financial statements and other writings, however evidenced, whether in physical or electronic form.

- (v) The words "asset" and "property" (and plural forms thereof) shall be construed to have the same meaning and effect and to refer to any and all tangible and intangible assets and properties, including cash, securities, accounts, contract rights, and Equity Interests.
- (c) In the computation of periods of time from a specified date to a later specified date, the word "from" means "from and including;" the words "to" and "until" each mean "to but excluding;" and the word "through" means "to and including."
 - (d) Any reference herein to any Person shall be construed to include such Person's successors and assigns.
- (e) Section headings herein and in the other Loan Documents are included for convenience of reference only and shall not affect the interpretation of this Agreement or any other Loan Document.

1.03 Accounting Terms.

- (a) All accounting terms not specifically or completely defined herein shall be construed in conformity with, and all financial data (including financial ratios and other financial calculations) required to be submitted pursuant to this Agreement shall be prepared in conformity with, GAAP applied on a consistent basis, as in effect from time to time, applied in a manner consistent with that used in preparing the Initial Financial Statements, <u>except</u> as otherwise specifically prescribed herein.
- (b) If at any time any change in GAAP would affect the computation of any financial ratio, covenant in respect of Indebtedness or other requirement set forth in any Loan Document, and either the Borrower or the Required Lenders shall so request, the Administrative Agent, the Lenders and the Borrower shall negotiate in good faith to amend such ratio or requirement to preserve the original intent thereof in light of such change in GAAP (subject to the approval of the Required Lenders); provided that, until so amended, (i) such ratio or requirement shall continue to be computed in accordance with GAAP prior to such change therein and (ii) the Parent Guarantor shall provide to the Administrative Agent and the Lenders financial statements and other documents required under this Agreement or as reasonably requested hereunder setting forth a reconciliation between calculations of such ratio or requirement made before and after giving effect to such change in GAAP.
- (c) Notwithstanding any other provision contained herein, all terms of an accounting or financial nature used herein shall be construed, and all computations of amounts and ratios referred to herein shall be made (i) without giving effect to any election under Accounting Standards Codification 825-10-25 (or any other Accounting Standards Codification or Financial Accounting Standard having a similar result or effect) to value any Indebtedness or other liabilities of the Parent Guarantor, the Borrower or any Subsidiary at "fair value", as defined therein, and (ii) without giving effect to any treatment of Indebtedness in respect of convertible debt instruments under Accounting Standards Codification 470-20 (or any other Accounting Standards Codification or Financial Accounting Standard having a similar result or effect) to value any such Indebtedness in a reduced or bifurcated manner as described therein, and such Indebtedness shall at all times be valued at the full stated principal amount thereof.

1.04 Rounding.

Any financial ratios required to be maintained by the Borrower pursuant to this Agreement shall be calculated by dividing the appropriate component by the other component, carrying the result to one place more than the number of places by which such ratio is expressed herein and rounding the result up or down to the nearest number (with a rounding-up if there is no nearest number).

1.05 References to Agreements and Laws.

Unless otherwise expressly provided herein, (a) references to Organization Documents, agreements (including the Loan Documents) and other contractual instruments shall be deemed to include all subsequent amendments, restatements, extensions, supplements and other modifications thereto, but only to the extent that such amendments, restatements, extensions, supplements and other modifications are not prohibited by any Loan Document; and (b) references to any Law shall include all statutory and regulatory provisions consolidating, amending, replacing, supplementing or interpreting such Law.

1.06 Times of Day.

Unless otherwise specified, all references herein to times of day shall be references to Central time (daylight or standard, as applicable).

1.07 Letter of Credit Amounts.

Unless otherwise specified, all references herein to the amount of a Letter of Credit at any time shall be deemed to mean the maximum face amount of such Letter of Credit after giving effect to all increases thereof contemplated by such Letter of Credit or the Issuer Documents related thereto, whether or not such maximum face amount is in effect at such time.

ARTICLE II. THE COMMITMENTS AND CREDIT EXTENSIONS

2.01 Loans.

Subject to the terms and conditions set forth herein, each Lender severally agrees to make loans (each such loan, a "Loan") in Dollars to the Borrower from time to time, on any Business Day during the Availability Period, in an aggregate amount not to exceed at any time outstanding the amount of such Lender's Commitment; provided, however, that after giving effect to any Borrowing, (i) the Total Outstandings shall not exceed the Aggregate Commitments, and (ii) the aggregate Outstanding Amount of the Loans of any Lender, plus such Lender's Pro Rata Share of the Outstanding Amount of all L/C Obligations and Swing Line Loans shall not exceed such Lender's Commitment. Within the limits of each Lender's Commitment, and subject to the other terms and conditions hereof, the Borrower may borrow under this Section 2.01, prepay under Section 2.05, and reborrow under this Section 2.01. Loans may be Base Rate Loans or Eurodollar Rate Loans, as further provided herein.

2.02 Borrowings, Conversions and Continuations of Loans.

(a) Each Borrowing, each conversion of Loans from one Type to the other, and each continuation of Eurodollar Rate Loans shall be made upon the Borrower's irrevocable written notice to the Administrative Agent. Each such notice must be received by the Administrative Agent not later than 11:00 a.m. (i) three Business Days prior to the requested date of any Borrowing of, conversion to or continuation of Eurodollar Rate Loans or of any conversion of Eurodollar Rate Loans to Base Rate Loans, and (ii) on the requested date of any Borrowing of Base Rate Loans; provided, however, that if the Borrower wishes to request Eurodollar Rate Loans having an Interest Period other than one week or one, two or three months in duration as provided in the definition of "Interest Period," the Administrative Agent shall give prompt notice to the Lenders of such request and determine whether the requested Interest Period is acceptable to all of them. Each Borrowing of, conversion to or continuation of Eurodollar Rate Loans shall be in a principal amount of \$2,000,000 or a whole multiple of \$1,000,000 in excess thereof. Except as provided in Section 2.03(c), each Borrowing of or conversion to Base Rate Loans shall be in a principal

amount of \$1,000,000 or a whole multiple of \$500,000 in excess thereof. Each Loan Notice (whether telephonic or written) shall specify (i) whether the Borrower is requesting a Borrowing, a conversion of Loans from one Type to the other, or a continuation of Eurodollar Rate Loans, (ii) the requested date of the Borrowing, conversion or continuation, as the case may be (which shall be a Business Day), (iii) the principal amount of Loans to be borrowed, converted or continued, (iv) the Type of Loans to be borrowed or to which existing Loans are to be converted, and (v) if applicable, the duration of the Interest Period with respect thereto. If the Borrower fails to specify a Type of Loan in a Loan Notice or if the Borrower fails to give a timely notice requesting a conversion or continuation, then the applicable Loans shall be made as, or converted to, Base Rate Loans. Any such automatic conversion to Base Rate Loans shall be effective as of the last day of the Interest Period then in effect with respect to the applicable Eurodollar Rate Loans. If the Borrower requests a Borrowing of, conversion to, or continuation of Eurodollar Rate Loans in any such Loan Notice, but fails to specify an Interest Period, it will be deemed to have specified an Interest Period of one month.

- (b) Following receipt of a Loan Notice, the Administrative Agent shall promptly notify each Lender of the amount of its Pro Rata Share of the applicable Loans, and if no timely notice of a conversion or continuation is provided by the Borrower, the Administrative Agent shall notify each Lender of the details of any automatic conversion to Base Rate Loans described in the preceding subsection. In the case of a Borrowing, each Lender shall make the amount of its Loan available to the Administrative Agent in immediately available funds at the Administrative Agent's Office not later than 1:00 p.m. on the Business Day specified in the applicable Loan Notice. Upon satisfaction of the applicable conditions set forth in Section 4.03 (and, if such Borrowing is the initial Credit Extension, Section 4.02), the Administrative Agent shall make all funds so received available to the Borrower in like funds as received by the Administrative Agent either by (i) crediting the account of the Borrower on the books of Wells Fargo Bank with the amount of such funds or (ii) wire transfer of such funds, in each case in accordance with instructions provided to (and reasonably acceptable to) the Administrative Agent by the Borrower; provided, however, that if, on the date the Loan Notice with respect to such Borrowing is given by the Borrower, there are L/C Borrowings or Swing Line Loans outstanding, then the proceeds of such Borrowing, first shall be applied to the payment in full of any such L/C Borrowings or Swing Line Loans, and second, shall be made available to the Borrower as provided above.
- (c) Except as otherwise provided herein, a Eurodollar Rate Loan may be continued or converted only on the last day of an Interest Period for such Eurodollar Rate Loan. During the existence of a Default, no Loans may be requested as, converted to or continued as Eurodollar Rate Loans without the consent of the Required Lenders.
- (d) After giving effect to all Borrowings, all conversions of Loans from one Type to the other, and all continuations of Loans as the same Type, there shall not be more than ten Interest Periods in effect with respect to Loans.

2.03 Letters of Credit.

(a) The Letter of Credit Commitments.

(i) Subject to the terms and conditions set forth herein, (A) each L/C Issuer agrees, in reliance upon the agreements of the other Lenders set forth in this <u>Section 2.03</u>, (1) from time to time on any Business Day during the period from the Closing Date until the Letter of Credit Expiration Date, to issue Letters of Credit in Dollars for the account of any Loan Party, Subsidiary or Joint Venture, and to amend or extend Letters of Credit previously issued by it, in accordance with <u>subsection (b)</u> below, and (2) to honor drawings under the Letters of Credit; and (B) the Lenders severally agree to participate in Letters of Credit issued for the account of any

Loan Party, Subsidiary or Joint Venture and any drawings thereunder; <u>provided</u> that after giving effect to any L/C Credit Extension with respect to any Letter of Credit, (x) the Total Outstandings shall not exceed the Aggregate Commitments, (y) the aggregate Outstanding Amount of the Loans of any Lender, plus such Lender's Pro Rata Share of the Outstanding Amount of all Swing Line Loans, plus such Lender's Pro Rata Share of the Outstanding Amount of all L/C Obligations shall not exceed such Lender's Commitment, and (z) the Outstanding Amount of the L/C Obligations shall not exceed the Letter of Credit Sublimit. Each request by the Borrower for the issuance or amendment of a Letter of Credit shall be deemed to be a representation by the Borrower that the L/C Credit Extension so requested complies with the conditions set forth in the proviso to the preceding sentence. Within the foregoing limits, and subject to the terms and conditions hereof, the Borrower's ability to obtain Letters of Credit shall be fully revolving, and accordingly the Borrower may, during the foregoing period, obtain Letters of Credit to replace Letters of Credit that have expired or that have been drawn upon and reimbursed. The obligations of the L/C Issuers under this Agreement are several and not joint. No L/C Issuer shall be responsible for any actions or any failure to act by or on the part of any other L/C Issuer.

- (ii) A L/C Issuer shall not issue any Letter of Credit, if:
- (A) subject to <u>Section 2.03(b)(iii)</u>, the expiry date of such requested Letter of Credit would occur more than twelve months after the date of issuance or last extension, unless the Required Lenders have approved such expiry date; or
- (B) the expiry date of such requested Letter of Credit would occur after the Letter of Credit Expiration Date, unless all the Lenders have approved such expiry date.
- (iii) A L/C Issuer shall not be under any obligation to issue any Letter of Credit if the L/C Obligations with respect to Letters of Credit issued by it would exceed, after giving effect to the requested Letter of Credit, such L/C Issuer's L/C Commitment, or if:
- (A) any order, judgment or decree of any Governmental Authority or arbitrator shall by its terms purport to enjoin or restrain such L/C Issuer from issuing such Letter of Credit, or any Law applicable to such L/C Issuer or any request or directive (whether or not having the force of law) from any Governmental Authority with jurisdiction over such L/C Issuer shall prohibit, or request that such L/C Issuer refrain from, the issuance of letters of credit generally or such Letter of Credit in particular or shall impose upon such L/C Issuer with respect to such Letter of Credit any restriction, reserve or capital requirement (for which such L/C Issuer is not otherwise compensated hereunder) not in effect on the Closing Date, or shall impose upon such L/C Issuer any unreimbursed loss, cost or expense which was not applicable on the Closing Date and which such L/C Issuer in good faith deems material to it;
 - (B) the issuance of such Letter of Credit would violate any Laws or one or more policies of such L/C Issuer;
- (C) except as otherwise agreed by the Administrative Agent and such L/C Issuer, such Letter of Credit is in an initial face amount less than \$500,000;
 - (D) such Letter of Credit is to be denominated in a currency other than Dollars;

- (E) such Letter of Credit contains any provisions for automatic reinstatement of the stated amount after any drawing thereunder; or
- (F) any Lender is a Defaulting Lender, unless the Fronting Exposure of such L/C Issuer with respect to such Defaulting Lender (after giving effect to any reallocation pursuant to Section 2.15(d)) has been Cash Collateralized in an amount not less than the Minimum Collateral Amount.
- (iv) A L/C Issuer shall not amend any Letter of Credit if such L/C Issuer would not be permitted at such time to issue such Letter of Credit in its amended form under the terms hereof.
- (v) Such L/C Issuer shall be under no obligation to amend any Letter of Credit if (A) such L/C Issuer would have no obligation at such time to issue such Letter of Credit in its amended form under the terms hereof, or (B) the beneficiary of such Letter of Credit does not accept the proposed amendment to such Letter of Credit.

(b) Procedures for Issuance and Amendment of Letters of Credit; Auto-Extension Letters of Credit.

- (i) Each Letter of Credit shall be issued or amended, as the case may be, upon the request of the Borrower delivered to the applicable L/C Issuer (with a copy to the Administrative Agent) in the form of a Letter of Credit Application, appropriately completed and signed by a Responsible Officer. Such Letter of Credit Application must be received by such L/C Issuer and the Administrative Agent not later than 11:00 a.m. at least two Business Days (or, in the case of Morgan Stanley Bank, N.A., at least five Business Days) (or such later date and time as the Administrative Agent and such L/C Issuer may agree in a particular instance in their sole discretion) prior to the proposed issuance date or date of amendment, as the case may be. In the case of a request for an initial issuance of a Letter of Credit, such Letter of Credit Application shall specify in form and detail satisfactory to such L/C Issuer: (A) the proposed issuance date of the requested Letter of Credit (which shall be a Business Day); (B) the amount thereof; (C) the expiry date thereof; (D) the name and address of the beneficiary thereof; (E) the documents to be presented by such beneficiary in case of any drawing thereunder; (G) the purpose and nature of such requested Letter of Credit; and (H) such other matters as such L/C Issuer may require. In the case of a request for an amendment of any outstanding Letter of Credit, such Letter of Credit Application shall specify in form and detail satisfactory to such L/C Issuer (A) the Letter of Credit to be amended; (B) the proposed date of amendment thereof (which shall be a Business Day); (C) the nature of the proposed amendment; and (D) such other matters as such L/C Issuer may require. Additionally, the Borrower shall furnish to such L/C Issuer and the Administrative Agent such other documents and information pertaining to such requested Letter of Credit issuance or amendment, including any Issuer Documents, as such L/C Issuer or the Administrative Agent may require.
- (ii) Unless such L/C Issuer has received written notice from any Lender, the Administrative Agent or the Borrower, at least one Business Day prior to the requested date of issuance or amendment of the applicable Letter of Credit, that one or more applicable conditions contained in <u>Article IV</u> shall not then be satisfied, then, subject to the terms and conditions hereof, such L/C Issuer shall, on the requested date, issue a Letter of Credit for the account of the Borrower or the applicable Subsidiary or enter into the applicable amendment, as the case may be, in each case in accordance with such L/C Issuer's usual and customary business practices.

Immediately upon the issuance of each Letter of Credit by a L/C Issuer, each Lender shall be deemed to, and hereby irrevocably and unconditionally agrees to, purchase from such L/C Issuer a risk participation in such Letter of Credit in an amount equal to the product of such Lender's Pro Rata Share times the amount of such Letter of Credit.

- (iii) If the Borrower so requests in any applicable Letter of Credit Application, the applicable L/C Issuer may, in its sole and absolute discretion, agree to issue a Letter of Credit that has automatic extension provisions (each, an "Auto-Extension Letter of Credit"); provided that any such Auto-Extension Letter of Credit must permit the L/C Issuer to prevent any such extension at least once in each twelve-month period (commencing with the date of issuance of such Letter of Credit) by giving prior written notice to the beneficiary thereof not later than a specified day (the "Non-Extension Notice Date") in each such twelve-month period to be agreed upon at the time such Letter of Credit is issued. Unless otherwise directed by the L/C Issuer, the Borrower shall not be required to make a specific request to the L/C Issuer for any such extension. Once an Auto-Extension Letter of Credit has been issued, the Lenders shall be deemed to have authorized (but may not require) the L/C Issuer to permit the extension of such Letter of Credit at any time to an expiry date not later than the Letter of Credit Expiration Date; provided, however, that the L/C Issuer shall not permit any such extension if (A) the L/C Issuer has determined that it would not be permitted, or would have no obligation, at such time to issue such Letter of Credit in its revised form (as extended) under the terms hereof (by reason of the provisions of clause (ii) or (iii) of Section 2.03(a) or otherwise), or (B) it has received written notice on or before the day that is seven Business Days before the Non-Extension Notice Date (1) from the Administrative Agent that the Required Lenders have elected not to permit such extension or (2) from the Administrative Agent, any Lender or the Borrower that one or more of the applicable conditions specified in Section 4.03 is not then satisfied, and in each such case directing the L/C Issuer not to permit such extension.
- (iv) If the Borrower so requests in any applicable Letter of Credit Application, the applicable L/C Issuer may, in its sole and absolute discretion, agree to issue a Letter of Credit that permits the automatic reinstatement of all or a portion of the stated amount thereof after any drawing thereunder (each, an "Auto-Reinstatement Letter of Credit"). Unless otherwise directed by the L/C Issuer, the Borrower shall not be required to make a specific request to the L/C Issuer to permit such reinstatement. Once an Auto-Reinstatement Letter of Credit has been issued, except as provided in the following sentence, the Lenders shall be deemed to have authorized (but may not require) the L/C Issuer to reinstate all or a portion of the stated amount thereof in accordance with the provisions of such Letter of Credit. Notwithstanding the foregoing, if such Auto-Reinstatement Letter of Credit permits the L/C Issuer to decline to reinstate all or any portion of the stated amount thereof after a drawing thereunder by giving notice of such non-reinstatement within a specified number of days after such drawing (the "Non-Reinstatement Deadline"), the L/C Issuer shall not permit such reinstatement if it has received a notice (which may be by telephone or in writing) on or before the day that is seven Business Days before the Non-Reinstatement Deadline (A) from the Administrative Agent that the Required Lenders have elected not to permit such reinstatement or (B) from the Administrative Agent, any Lender or the Borrower that one or more of the applicable conditions specified in Section 4.03 is not then satisfied (treating such reinstatement as an L/C Credit Extension for purposes of this clause) and, in each case, directing the L/C Issuer not to permit such reinstatement.
- (v) Promptly after its delivery of any Letter of Credit or any amendment to a Letter of Credit to an advising bank with respect thereto or to the beneficiary thereof, the L/C Issuer will also deliver to the Borrower and the Administrative Agent a true and complete copy of such Letter of Credit or amendment.

(c) <u>Drawings and Reimbursements</u>; <u>Funding of Participations</u>.

- (i) Upon receipt from the beneficiary of any Letter of Credit of any notice of a drawing under such Letter of Credit, the applicable L/C Issuer shall notify the Borrower and the Administrative Agent thereof. Not later than 11:00 a.m. on the date of any payment by a L/C Issuer under a Letter of Credit (each such date, an "Honor Date"), the Borrower shall reimburse the L/C Issuer through the Administrative Agent in an amount equal to the amount of such drawing. If the Borrower fails to so reimburse the L/C Issuer by such time, the Administrative Agent shall promptly notify each Lender of the Honor Date, the amount of the unreimbursed drawing (the "Unreimbursed Amount"), and the amount of such Lender's Pro Rata Share thereof. In such event, the Borrower shall be deemed to have requested a Borrowing of Base Rate Loans to be disbursed on the Honor Date in an amount equal to the Unreimbursed Amount, without regard to the minimum and multiples specified in Section 2.02 for the principal amount of Base Rate Loans, but subject to the amount of the unutilized portion of the Aggregate Commitments and the conditions set forth in Section 4.03 (other than the delivery of a Loan Notice). Any notice given by a L/C Issuer or the Administrative Agent pursuant to this Section 2.03(c)(i) may be given by telephone if immediately confirmed in writing; provided that the lack of such an immediate confirmation shall not affect the conclusiveness or binding effect of such notice.
- (ii) Each Lender (including the Lender acting as L/C Issuer) shall upon any written notice pursuant to <u>Section 2.03(c)(i)</u> make funds available to the Administrative Agent for the account of the L/C Issuer at the Administrative Agent's Office in an amount equal to its Pro Rata Share of the Unreimbursed Amount not later than 1:00 p.m. on the Business Day specified in such written notice by the Administrative Agent, whereupon, subject to the provisions of <u>Section 2.03(c)(iii)</u>, each Lender that so makes funds available shall be deemed to have made a Base Rate Loan to the Borrower in such amount. The Administrative Agent shall remit the funds so received to the L/C Issuer.
- (iii) With respect to any Unreimbursed Amount that is not fully refinanced by a Borrowing of Base Rate Loans because the conditions set forth in Section 4.03 cannot be satisfied or for any other reason, the Borrower shall be deemed to have incurred from the L/C Issuer an L/C Borrowing in the amount of the Unreimbursed Amount that is not so refinanced, which L/C Borrowing shall be due and payable on demand (together with interest) and shall bear interest at the Default Rate. In such event, each Lender's payment to the Administrative Agent for the account of the L/C Issuer pursuant to Section 2.03(c)(ii) shall be deemed payment in respect of its participation in such L/C Borrowing and shall constitute an L/C Advance from such Lender in satisfaction of its participation obligation under this Section 2.03.
- (iv) Until each Lender funds its Loan or L/C Advance pursuant to this <u>Section 2.03(c)</u> to reimburse the L/C Issuer for any amount drawn under any Letter of Credit, interest in respect of such Lender's Pro Rata Share of such amount shall be solely for the account of the L/C Issuer.
- (v) Each Lender's obligation to make Loans or L/C Advances to reimburse the L/C Issuer for amounts drawn under Letters of Credit, as contemplated by this Section 2.03(c), shall be absolute and unconditional and shall not be affected by any circumstance, including (A) any set-off, counterclaim, recoupment, defense or other right which such Lender may have against the L/C Issuer, the Borrower or any other Person for any reason whatsoever; (B) the occurrence or continuance of a Default, or (C) any other occurrence, event or condition, whether or not similar to any of the foregoing; provided, however, that each Lender's obligation to make Loans pursuant to this Section 2.03(c) is subject to the conditions set forth in Section 4.03 (other than delivery by the Borrower of a Loan Notice). No such making of an L/C Advance shall relieve or otherwise impair the obligation of the Borrower to reimburse the L/C Issuer for the amount of any payment made by the L/C Issuer under any Letter of Credit, together with interest as provided herein.

(vi) If any Lender fails to make available to the Administrative Agent for the account of the L/C Issuer any amount required to be paid by such Lender pursuant to the foregoing provisions of this Section 2.03(c) by the time specified in Section 2.03(c)(ii), the L/C Issuer shall be entitled to recover from such Lender (acting through the Administrative Agent), on demand, such amount with interest thereon for the period from the date such payment is required to the date on which such payment is immediately available to the L/C Issuer at a rate per annum equal to the Federal Funds Rate from time to time in effect. A certificate of the L/C Issuer submitted to any Lender (through the Administrative Agent) with respect to any amounts owing under this clause (vi) shall be conclusive absent manifest error.

(d) Repayment of Participations.

- (i) At any time after the L/C Issuer has made a payment under any Letter of Credit and has received from any Lender such Lender's L/C Advance in respect of such payment in accordance with Section 2.03(c), if the Administrative Agent receives for the account of such L/C Issuer any payment in respect of the related Unreimbursed Amount or interest thereon (whether directly from the Borrower or otherwise, including proceeds of Cash Collateral applied thereto by the Administrative Agent), the Administrative Agent will distribute to such Lender its Pro Rata Share thereof (appropriately adjusted, in the case of interest payments, to reflect the period of time during which such Lender's L/C Advance was outstanding) in the same funds as those received by the Administrative Agent.
- (ii) If any payment received by the Administrative Agent for the account of a L/C Issuer pursuant to Section 2.03(c)(i) is required to be returned under any of the circumstances described in Section 11.05 (including pursuant to any settlement entered into by such L/C Issuer in its discretion), each Lender shall pay to the Administrative Agent for the account of such L/C Issuer its Pro Rata Share thereof on demand of the Administrative Agent, plus interest thereon from the date of such demand to the date such amount is returned by such Lender, at a rate per annum equal to the Federal Funds Rate from time to time in effect.
- (e) <u>Obligations Absolute</u>. The obligation of the Borrower to reimburse each L/C Issuer for each drawing under each Letter of Credit issued by it and to repay each L/C Borrowing shall be absolute, unconditional and irrevocable, and shall be paid strictly in accordance with the terms of this Agreement under all circumstances, including the following:
 - (i) any lack of validity or enforceability of such Letter of Credit, this Agreement, or any other Loan Document;
- (ii) the existence of any claim, counterclaim, set-off, defense or other right that the Borrower or any Subsidiary may have at any time against any beneficiary or any transferee of such Letter of Credit (or any Person for whom any such beneficiary or any such transferee may be acting), any L/C Issuer or any other Person, whether in connection with this Agreement, the transactions contemplated hereby or by such Letter of Credit or any agreement or instrument relating thereto, or any unrelated transaction;
- (iii) any draft, demand, certificate or other document presented under such Letter of Credit proving to be forged, fraudulent, invalid or insufficient in any respect or any statement therein being untrue or inaccurate in any respect; or any loss or delay in the transmission or otherwise of any document required in order to make a drawing under such Letter of Credit;

(iv) any payment by the L/C Issuer under such Letter of Credit against presentation of a draft or certificate that does not strictly comply with the terms of such Letter of Credit; or any payment made by the L/C Issuer under such Letter of Credit to any Person purporting to be a trustee in bankruptcy, debtor-in-possession, assignee for the benefit of creditors, liquidator, receiver or other representative of or successor to any beneficiary or any transferee of such Letter of Credit, including any arising in connection with any proceeding under any Debtor Relief Law; or

(v) any other circumstance or happening whatsoever, whether or not similar to any of the foregoing, including any other circumstance that might otherwise constitute a defense available to, or a discharge of, the Borrower or any Subsidiary.

The Borrower shall promptly examine a copy of each Letter of Credit and each amendment thereto that is delivered to it and, in the event of any claim of noncompliance with the Borrower's instructions or other irregularity, the Borrower will immediately notify the L/C Issuer. The Borrower shall be conclusively deemed to have waived any such claim against the L/C Issuer and its correspondents unless such written notice is given as aforesaid.

(f) Role of L/C Issuer. Each Lender and the Borrower agree that, in paying any drawing under a Letter of Credit, the L/C Issuer shall not have any responsibility to obtain any document (other than any sight draft, certificates and documents expressly required by the Letter of Credit) or to ascertain or inquire as to the validity or accuracy of any such document or the authority of the Person executing or delivering any such document. No L/C Issuer, no Agent-Related Person and none of the respective correspondents, participants or assignees of any L/C Issuer shall be liable to any Lender for (i) any action taken or omitted in connection herewith at the request or with the approval of all Lenders or the Required Lenders, as applicable; (ii) any action taken or omitted in the absence of gross negligence or willful misconduct; or (iii) the due execution, effectiveness, validity or enforceability of any document or instrument related to any Letter of Credit or Letter of Credit Application. The Borrower hereby assumes all risks of the acts or omissions of any beneficiary or transferee with respect to its use of any Letter of Credit; provided, however, that this assumption is not intended to, and shall not, preclude the Borrower's pursuing such rights and remedies as it may have against the beneficiary or transferee at law or under any other agreement. No L/C Issuer, no Agent-Related Person, and none of the respective correspondents, participants or assignees of any L/C Issuer, shall be liable or responsible for any of the matters described in clauses (i) through (v) of Section 2.03(e); provided, however, that anything in such clauses to the contrary notwithstanding, the Borrower may have a claim against a L/C Issuer, and a L/C Issuer may be liable to the Borrower, to the extent, but only to the extent, of any direct, as opposed to consequential or exemplary, damages suffered by the Borrower which the Borrower proves were caused by such L/C Issuer's willful misconduct or gross negligence or such L/C Issuer's willful failure to pay under any Letter of Credit after the presentation to it by the beneficiary of a sight draft and certificate(s) strictly complying with the terms and conditions of a Letter of Credit. In furtherance and not in limitation of the foregoing, the L/C Issuers may accept documents that appear on their face to be in order, without responsibility for further investigation, regardless of any notice or information to the contrary, and the L/C Issuers shall not be responsible for the validity or sufficiency of any instrument transferring or assigning or purporting to transfer or assign a Letter of Credit or the rights or benefits thereunder or proceeds thereof, in whole or in part, which may prove to be invalid or ineffective for any reason.

- (g) <u>Cash Collateral</u>. Upon the request of the Administrative Agent, (i) if any L/C Issuer has honored any full or partial drawing request under any Letter of Credit and such drawing has resulted in an L/C Borrowing, or (ii) if, as of the Letter of Credit Expiration Date, any Letter of Credit for any reason remains outstanding and partially or wholly undrawn, the Borrower shall immediately Cash Collateralize the then Outstanding Amount of all L/C Obligations (in an amount equal to such Outstanding Amount determined as of the date of such L/C Borrowing or the Letter of Credit Expiration Date, as the case may be). At any time that there shall exist a Defaulting Lender, within one Business Day following the written request of the Administrative Agent or any L/C Issuer (with a copy to the Administrative Agent) the Borrower shall Cash Collateralize such L/C Issuer's Fronting Exposure with respect to such Defaulting Lender (determined after giving effect to Section 2.15(d) and any Cash Collateral provided by such Defaulting Lender) in an amount not less than the Minimum Collateral Amount.
- (i) <u>Grant of Security Interest</u>. The Borrower, and to the extent Cash Collateral is provided by any Defaulting Lender, such Defaulting Lender, hereby grants to the Administrative Agent, for the benefit of the L/C Issuers, and agrees to maintain, a first priority security interest in all such Cash Collateral as security for the Defaulting Lenders' obligation to fund participations in respect of L/C Obligations, to be applied pursuant to <u>clause (ii)</u> below. If at any time the Administrative Agent determines that Cash Collateral is subject to any right or claim of any Person other than the Administrative Agent and the L/C Issuers as herein provided (other than any Permitted Liens), or that the total amount of such Cash Collateral is less than the Minimum Collateral Amount, the Borrower will, promptly upon demand by the Administrative Agent, pay or provide to the Administrative Agent additional Cash Collateral in an amount sufficient to eliminate such deficiency (after giving effect to any Cash Collateral provided by the Defaulting Lender).
- (ii) <u>Application</u>. Notwithstanding anything to the contrary contained in this Agreement, Cash Collateral provided under this <u>Section 2.03(g)</u> or <u>Section 2.15</u> in respect of Letters of Credit shall be applied to the satisfaction of the Defaulting Lender's obligation to fund participations in respect of L/C Obligations (including, as to Cash Collateral provided by a Defaulting Lender, any interest accrued on such obligation) for which the Cash Collateral was so provided, prior to any other application of such property as may otherwise be provided for herein.
- (iii) <u>Termination of Requirement</u>. Cash Collateral (or the appropriate portion thereof) provided to reduce one or more L/C Issuer's Fronting Exposure shall no longer be required to be held as Cash Collateral pursuant to this <u>Section 2.03(g)</u> following (A) the elimination of the applicable Fronting Exposure (including by the termination of Defaulting Lender status of the applicable Lender), or (B) the determination by the Administrative Agent and the applicable L/C Issuer(s) that there exists excess Cash Collateral; <u>provided</u> that, subject to <u>Section 2.15</u> the Person providing Cash Collateral and the L/C Issuers may agree that Cash Collateral shall be held to support future anticipated Fronting Exposure or other obligations.
- (h) <u>Applicability of ISP98</u>. Unless otherwise expressly agreed by the applicable L/C Issuer and the Borrower when a Letter of Credit is issued, the rules of the ISP shall apply to each Letter of Credit.
- (i) <u>Letter of Credit Fees</u>. The Borrower shall pay to the Administrative Agent for the account of each Lender in accordance with its Pro Rata Share a Letter of Credit fee (the "<u>Letter of Credit Fee</u>") for each Letter of Credit equal to the Applicable Rate times the daily maximum amount available to be drawn under such Letter of Credit (whether or not such maximum amount is then in effect under such Letter of Credit). Letter of Credit Fees shall be (i) computed on a quarterly basis in arrears and (ii) due and payable on the first Business Day after the end of each March, June, September and December, commencing with the first such date to occur after the issuance of such Letter of Credit, on the Letter of Credit Expiration Date and thereafter on demand. If there is any change in the Applicable Rate during any quarter, the daily maximum amount of each Letter of Credit shall be computed and multiplied by the Applicable Rate

separately for each period during such quarter that such Applicable Rate was in effect. Notwithstanding anything to the contrary contained herein, upon the request of the Required Lenders, while any Event of Default exists, all Letter of Credit Fees shall accrue at the Default Rate.

- (j) <u>Fronting Fee and Documentary and Processing Charges Payable to L/C Issuer</u>. The Borrower shall pay directly to the applicable L/C Issuer for its own account a fronting fee with respect to each Letter of Credit issued by such L/C Issuer in the amount specified in the Fee Letters, payable on the actual daily maximum amount available to be drawn under such Letter of Credit (whether or not such maximum amount is then in effect under such Letter of Credit). Such fronting fee shall be computed on a quarterly basis in arrears. Such fronting fee shall be due and payable on the first Business Day after the end of each March, June, September and December, commencing with the first such date to occur after the issuance of such Letter of Credit, on the Letter of Credit Expiration Date and thereafter on demand. In addition, the Borrower shall pay directly to the applicable L/C Issuer for its own account the customary issuance, presentation, amendment and other processing fees, and other standard costs and charges, of the L/C Issuer relating to letters of credit as from time to time in effect. Such customary fees and standard costs and charges are due and payable on demand and are nonrefundable.
- (k) <u>Conflict with Issuer Documents</u>. In the event of any conflict between the terms hereof and the terms of any Issuer Document, the terms hereof shall control.
- (l) <u>Letters of Credit Issued for Loan Parties</u>, <u>Subsidiaries or Joint Ventures</u>. Notwithstanding that a Letter of Credit issued or outstanding hereunder is in support of any obligations of, or is for the account of, any Loan Party or its Subsidiaries or Joint Ventures, the Borrower shall be obligated to reimburse the applicable L/C Issuer hereunder for any and all drawings under such Letter of Credit. The Borrower hereby acknowledges that the issuance of Letters of Credit for the account of any Loan Party or its Subsidiaries or Joint Ventures inures to the benefit of the Borrower, and that the Borrower's business derives substantial benefits from the businesses of such Loan Parties, Subsidiaries and Joint Ventures.

2.04 Swing Line Loans.

(a) The Swing Line. Subject to the terms and conditions set forth herein, the Swing Line Lender agrees, in reliance upon the agreements of the other Lenders set forth in this Section 2.04, to make loans in Dollars (each such loan, a "Swing Line Loan") to the Borrower from time to time on any Business Day during the Availability Period in an aggregate amount not to exceed at any time outstanding the amount of the Swing Line Sublimit, notwithstanding the fact that such Swing Line Loans, when aggregated with the Pro Rata Share of the Outstanding Amount of Loans and L/C Obligations of the Lender acting as Swing Line Lender, may exceed the amount of such Lender's Commitment; provided, however, that after giving effect to any Swing Line Loan, (i) the Total Outstandings shall not exceed the Aggregate Commitments, and (ii) the aggregate Outstanding Amount of the Loans of any Lender, plus such Lender's Pro Rata Share of the Outstanding Amount of all Swing Line Loans shall not exceed such Lender's Commitment, and provided, further, that the Borrower shall not use the proceeds of any Swing Line Loan to refinance any outstanding Swing Line Loan. Within the foregoing limits, and subject to the other terms and conditions hereof, the Borrower may borrow under this Section 2.04, prepay under Section 2.05, and reborrow under this Section 2.04. Immediately upon the making of a Swing Line Loan, each Lender shall be deemed to, and hereby irrevocably and unconditionally agrees to, purchase from the Swing Line Lender a risk participation in such Swing Line Loan in an amount equal to the product of such Lender's Pro Rata Share times the amount of such Swing Line Loan.

(b) <u>Borrowing Procedures</u>. Each Swing Line Borrowing shall be made upon the Borrower's irrevocable notice to the Swing Line Lender and the Administrative Agent, which may be given by telephone. Each such notice must be received by the Swing Line Lender and the Administrative Agent not later than 1:00 p.m. on the requested borrowing date, and shall specify (i) the amount to be borrowed, which shall be a minimum of \$2,000,000, and (ii) the requested borrowing date, which shall be a Business Day. Each such telephonic notice must be confirmed promptly by delivery to the Swing Line Lender and the Administrative Agent of a written Swing Line Loan Notice, appropriately completed and signed by a Responsible Officer. Promptly after receipt by the Swing Line Lender of any telephonic Swing Line Loan Notice, the Swing Line Lender will confirm with the Administrative Agent (by telephone or in writing) that the Administrative Agent has also received such Swing Line Loan Notice and, if not, the Swing Line Lender will notify the Administrative Agent (by telephone or in writing) of the contents thereof. Unless the Swing Line Lender has received notice (by telephone or in writing) from the Administrative Agent (including at the request of any Lender) prior to 2:00 p.m. on the date of the proposed Swing Line Borrowing (A) directing the Swing Line Lender not to make such Swing Line Loan as a result of the limitations set forth in the first proviso to the first sentence of Section 2.04(a), or (B) that one or more of the applicable conditions specified in Article IV is not then satisfied, then, subject to the terms and conditions hereof, the Swing Line Lender will, not later than 3:00 p.m. on the borrowing date specified in such Swing Line Loan Notice, make the amount of its Swing Line Loan available to the Borrower at its office by crediting the account of the Borrower on the books of the Swing Line Lender in immediately available funds.

(c) Refinancing of Swing Line Loans.

- (i) The Swing Line Lender at any time in its sole and absolute discretion may request, on behalf of the Borrower (which hereby irrevocably authorizes the Swing Line Lender to so request on its behalf), that each Lender make a Base Rate Loan in an amount equal to such Lender's Pro Rata Share of the amount of Swing Line Loans then outstanding. Such request shall be made in writing (which written request shall be deemed to be a Loan Notice for purposes hereof) and in accordance with the requirements of Section 2.02, without regard to the minimum and multiples specified therein for the principal amount of Base Rate Loans, but subject to the unutilized portion of the Aggregate Commitments and the conditions set forth in Section 4.03. The Swing Line Lender shall furnish the Borrower with a copy of the applicable Loan Notice promptly after delivering such notice to the Administrative Agent. Each Lender shall make an amount equal to its Pro Rata Share of the amount specified in such Loan Notice available to the Administrative Agent in immediately available funds for the account of the Swing Line Lender at the Administrative Agent's Office not later than 2:00 p.m. on the day specified in such Loan Notice, whereupon, subject to Section 2.04(c)(ii), each Lender that so makes funds available shall be deemed to have made a Base Rate Loan to the Borrower in such amount. The Administrative Agent shall remit the funds so received to the Swing Line Lender.
- (ii) If for any reason any Swing Line Loan cannot be refinanced by such a Borrowing in accordance with Section 2.04(c)(i), the request for Base Rate Loans submitted by the Swing Line Lender as set forth herein shall be deemed to be a request by the Swing Line Lender that each of the Lenders fund its risk participation in the relevant Swing Line Loan and each Lender's payment to the Administrative Agent for the account of the Swing Line Lender pursuant to Section 2.04(c)(i) shall be deemed payment in respect of such participation.
- (iii) If any Lender fails to make available to the Administrative Agent for the account of the Swing Line Lender any amount required to be paid by such Lender pursuant to the foregoing provisions of this Section 2.04(c) by the time specified in Section 2.04(c)(i), the Swing Line Lender shall be entitled to recover from such Lender (acting through the Administrative

Agent), on demand, such amount with interest thereon for the period from the date such payment is required to the date on which such payment is immediately available to the Swing Line Lender at a rate per annum equal to the greater of the Federal Funds Rate and a rate determined by the Swing Line Lender in accordance with banking industry rules on interbank compensation, plus any administrative, processing or similar fees customarily charged by the Swing Line Lender in connection with the foregoing. If such Lender pays such amount (with interest and fees as aforesaid), the amount so paid shall constitute such Lender's Loan included in the relevant Borrowing or funded participation in the relevant Swing Line Loan, as the case may be. A certificate of the Swing Line Lender submitted to any Lender (through the Administrative Agent) with respect to any amounts owing under this clause shall be conclusive absent manifest error.

(iv) Each Lender's obligation to make Loans or to purchase and fund risk participations in Swing Line Loans pursuant to this Section 2.04(c) shall be absolute and unconditional and shall not be affected by any circumstance, including (A) any setoff, counterclaim, recoupment, defense or other right which such Lender may have against the Swing Line Lender, the Borrower or any other Person for any reason whatsoever, (B) the occurrence or continuance of a Default, or (C) any other occurrence, event or condition, whether or not similar to any of the foregoing; provided, however, that each Lender's obligation to make Loans pursuant to this Section 2.04(c) is subject to the conditions set forth in Section 4.03. No such funding of risk participations shall relieve or otherwise impair the obligation of the Borrower to repay Swing Line Loans, together with interest as provided herein.

(d) Repayment of Participations.

- (i) At any time after any Lender has purchased and funded a risk participation in a Swing Line Loan, if the Swing Line Lender receives any payment on account of such Swing Line Loan, the Swing Line Lender will distribute to such Lender its Pro Rata Share thereof in the same funds as those received by the Swing Line Lender.
- (ii) If any payment received by the Swing Line Lender in respect of principal or interest on any Swing Line Loan is required to be returned by the Swing Line Lender under any of the circumstances described in Section 11.05 (including pursuant to any settlement entered into by the Swing Line Lender in its discretion), each Lender shall pay to the Swing Line Lender its Pro Rata Share thereof on demand of the Administrative Agent, plus interest thereon from the date of such demand to the date such amount is returned, at a rate per annum equal to the Federal Funds Rate. The Administrative Agent will make such demand upon the request of the Swing Line Lender. The obligations of the Lenders under this clause shall survive the payment in full of the Obligations and the termination of this Agreement.
- (e) <u>Interest for Account of Swing Line Lender</u>. The Swing Line Lender shall be responsible for invoicing the Borrower for interest on the Swing Line Loans. Until each Lender funds its Base Rate Loan or risk participation pursuant to this <u>Section 2.04</u> to refinance such Lender's Pro Rata Share of any Swing Line Loan, interest in respect of such Pro Rata Share shall be solely for the account of the Swing Line Lender.
- (f) <u>Payments Directly to Swing Line Lender</u>. The Borrower shall make all payments of principal and interest in respect of the Swing Line Loans directly to the Swing Line Lender.

2.05 Prepayments.

(a) Optional.

- (i) The Borrower may, upon notice to the Administrative Agent, at any time or from time to time voluntarily prepay Loans in whole or in part without premium or penalty; provided that (i) such notice must be received by the Administrative Agent not later than 11:00 a.m. (A) three Business Days prior to any date of prepayment of Eurodollar Rate Loans and (B) on the date of prepayment of Base Rate Loans; (ii) any prepayment of Eurodollar Rate Loans shall be in a principal amount of \$5,000,000 or a whole multiple of \$1,000,000 in excess thereof; and (iii) any prepayment of Base Rate Loans shall be in a principal amount of \$2,000,000 or a whole multiple of \$500,000 in excess thereof; or, in each case, if less, the entire principal amount thereof then outstanding. Each such notice shall specify the date and amount of such prepayment and the Type(s) of Loans to be prepaid. The Administrative Agent will promptly notify each Lender of its receipt of each such notice, and of the amount of such Lender's Pro Rata Share of such prepayment. If such notice is given by the Borrower, the Borrower shall make such prepayment and the payment amount specified in such notice shall be due and payable on the date specified therein. Any prepayment of a Eurodollar Rate Loan shall be accompanied by all accrued interest thereon, together with any additional amounts required pursuant to Section 3.05. Each such prepayment shall be applied to the Loans of the Lenders in accordance with their respective Pro Rata Shares.
- (ii) The Borrower may, upon notice to the Swing Line Lender (with a copy to the Administrative Agent), at any time or from time to time, voluntarily prepay Swing Line Loans in whole or in part without premium or penalty; <u>provided</u> that (A) such notice must be received by the Swing Line Lender and the Administrative Agent not later than 1:00 p.m. on the date of the prepayment, and (B) any such prepayment shall be in a minimum principal amount of \$2,000,000 or a whole multiple of \$500,000 in excess thereof or if less, the entire principal amount thereof then outstanding. Each such notice shall specify the date and amount of such prepayment. If such notice is given by the Borrower, the Borrower shall make such prepayment and the payment amount specified in such notice shall be due and payable on the date specified therein.

(b) Mandatory.

- (i) If for any reason the Total Outstandings at any time exceed the Aggregate Commitments then in effect, the Borrower shall immediately prepay or repay all outstanding Loans and Swing Line Loans and/or Cash Collateralize the L/C Obligations in an aggregate amount equal to such excess; <u>provided</u>, <u>however</u>, that the Borrower shall not be required to Cash Collateralize the L/C Obligations pursuant to this <u>Section 2.05(b)</u> unless after the prepayment in full of the Loans and the Swing Line Loans, the Total Outstandings exceed the Aggregate Commitments then in effect.
- (ii) To the extent that the aggregate amount of Net Cash Proceeds of Asset Sales and Net Cash Proceeds of Casualty Events received by all Loan Parties and Restricted Subsidiaries in a fiscal year exceeds five percent (5%) of Consolidated Net Tangible Assets of the Parent Guarantor as of the end of the most recently ended fiscal year, the Borrower shall apply an amount equal to 100% of such excess Net Cash Proceeds to prepay outstanding Loans and/or Cash Collateralize outstanding Letters of Credit in accordance with Section 2.05(b)(iv).

- (iii) In the event that the Parent Guarantor, the Borrower or any Restricted Subsidiary shall receive Net Cash Proceeds from the issuance or incurrence of Indebtedness for money borrowed (other than any cash proceeds from the issuance of Indebtedness permitted pursuant to Section 7.01) the Borrower shall within three Business Day following the receipt of such Net Cash Proceeds by the Parent Guarantor, the Borrower or such Restricted Subsidiary, apply an amount equal to 100% of such Net Cash Proceeds to prepay outstanding Loans and/or Cash Collateralize outstanding Letters of Credit in accordance with Section 2.05(b)(iv).
- (iv) Mandatory prepayments under this <u>Section 2.05(b)</u> shall be applied first, to outstanding Loans and, second, to Cash Collateralize outstanding Letters of Credit on a pro rata basis, in each case, with no corresponding permanent reduction of the Commitments.
- (v) Prepayments of Loans under this <u>Section 2.05</u> shall be subject to <u>Section 3.05</u>, and, to the extent interest is required to be paid pursuant to <u>Section 2.08(c)</u>, shall be accompanied by accrued and unpaid interest on the principal amount to be prepaid to but excluding the date of payment.
- 2.06 **Termination or Reduction of Commitments.** The Borrower may, upon notice to the Administrative Agent, terminate the Aggregate Commitments, the Letter of Credit Sublimit or the Swing Line Sublimit, or from time to time permanently reduce the Aggregate Commitments, the Letter of Credit Sublimit or the Swing Line Sublimit; provided that (i) any such notice shall be received by the Administrative Agent not later than 11:00 a.m. three Business Days prior to the date of termination or reduction, (ii) any such partial reduction shall be in an aggregate amount of \$5,000,000 or any whole multiple of \$1,000,000 in excess thereof, (iii) the Borrower shall not terminate or reduce the (A) Aggregate Commitments if, after giving effect thereto and to any concurrent prepayments hereunder, the Total Outstandings would exceed the Aggregate Commitments, (B) the Letter of Credit Sublimit if, after giving effect thereto, the Outstanding Amount of L/C Obligations not fully Cash Collateralized hereunder would exceed the Letter of Credit Sublimit, or (C) the Swing Line Sublimit if, after giving effect thereto and to any concurrent prepayments hereunder, the Outstanding Amount of Swing Line Loans would exceed the Swing Line Sublimit, and (iv) if, after giving effect to any reduction of the Aggregate Commitments, the Letter of Credit Sublimit or the Swing Line Sublimit, as applicable, exceeds the amount of the Aggregate Commitments, the Letter of Credit Sublimit or the Sublimit, as applicable, shall be automatically reduced by the amount of such excess. The Administrative Agent will promptly notify the Lenders of any such notice of termination or reduction of the Aggregate Commitments shall be applied to the Commitment of each Lender according to its Pro Rata Share. All fees accrued until the effective date of any termination of the Aggregate Commitments of the L/C Issuers on a pro rata basis among them, unless otherwise agreed by the Borrower and all L/C Issuers.
- 2.07 **Repayment of Loans and Swing Line Loans.** The Borrower shall repay to the Lenders on the Maturity Date the aggregate principal amount of Loans outstanding on such date. The Borrower shall repay to the Swing Line Lender the aggregate principal amount of each Swing Line Loan on the fifth Business Day after such Swing Line Loan is made.

2.08 Interest.

(a) Subject to the provisions of <u>subsection (b)</u> below, (i) each Eurodollar Rate Loan shall bear interest on the outstanding principal amount thereof for each Interest Period at a rate per annum equal to the Eurodollar Rate for such Interest Period plus the Applicable Rate; (ii) each Base Rate Loan shall

bear interest on the outstanding principal amount thereof from the applicable borrowing date at a rate per annum equal to the Base Rate plus the Applicable Rate; and (iii) each Swing Line Loan shall bear interest on the Outstanding Amount thereof from the applicable borrowing date at a rate per annum equal to the Swing Line Rate.

- (b) (i) If any amount of principal of any Loan or Swing Line Loan is not paid when due (without regard to any applicable grace periods), whether at stated maturity, by acceleration or otherwise, such amount shall thereafter bear interest at a fluctuating interest rate per annum at all times equal to the Default Rate to the fullest extent permitted by applicable Laws.
- (ii) If any amount (other than principal of any Loan or Swing Line Loan) payable by the Borrower under any Loan Document is not paid when due (without regard to any applicable grace periods), whether at stated maturity, by acceleration or otherwise, then upon the request of the Required Lenders, such amount shall thereafter bear interest at a fluctuating interest rate per annum at all times equal to the Default Rate to the fullest extent permitted by applicable Laws.
- (iii) Upon the request of the Required Lenders, while any Event of Default exists, the Borrower shall pay interest on the amount of all outstanding Obligations hereunder at a fluctuating interest rate per annum at all times equal to the Default Rate to the fullest extent permitted by applicable Laws.
 - (iv) Accrued and unpaid interest on past due amounts (including interest on past due interest) shall be due and payable upon demand.
- (c) Interest on each Loan and Swing Line Loan shall be due and payable in arrears on each Interest Payment Date applicable thereto and at such other times as may be specified herein. In the event of any repayment or prepayment of any Loan other than a Base Rate Loan or a Swing Line Loan, accrued interest on the principal amount repaid or prepaid shall be payable on the date of such repayment or prepayment. Interest hereunder shall be due and payable in accordance with the terms hereof before and after judgment, and before and after the commencement of any proceeding under any Debtor Relief Law.
 - 2.09 **Fees.** In addition to certain fees described in <u>subsections (i)</u> and (j) of <u>Section 2.03</u>:
- (a) <u>Commitment Fee</u>. The Borrower shall pay to the Administrative Agent for the account of each Lender in accordance with its Pro Rata Share, a commitment fee equal to the Applicable Rate times the actual daily amount by which the Aggregate Commitments exceed the sum of (i) the Outstanding Amount of Loans (for the avoidance of doubt, other than the Swing Line Loans) and (ii) the Outstanding Amount of L/C Obligations. The commitment fee shall accrue at all times during the Availability Period, including at any time during which one or more of the conditions in <u>Article IV</u> is not met, and shall be due and payable quarterly in arrears (i) promptly after Borrower receives notice of the amount of such commitment fee for such quarter, but not earlier than the last Business Day of each March, June, September and December, commencing with the first such date to occur after the Closing Date, and (ii) on the Maturity Date. The commitment fee shall be calculated quarterly in arrears, and if there is any change in the Applicable Rate during any quarter, the actual daily amount shall be computed and multiplied by the Applicable Rate separately for each period during such quarter that such Applicable Rate was in effect.

(b) Other Fees.

- (i) The Borrower shall pay to the Arrangers and the Administrative Agent for their own respective accounts fees in the amounts and at the times specified in the Fee Letters. Such fees shall be fully earned when paid and shall not be refundable for any reason whatsoever.
- (ii) The Borrower shall pay to the Lenders such fees as shall have been separately agreed upon in writing in the amounts and at the times so specified. Such fees shall be fully earned when paid and shall not be refundable for any reason whatsoever.

2.10 Computation of Interest and Fees.

- (a) All computations of interest for Base Rate Loans when the Base Rate is determined by Wells Fargo Bank's "prime rate" shall be made on the basis of a year of 365 or 366 days, as the case may be, and actual days elapsed. All other computations of fees and interest shall be made on the basis of a 360-day year and actual days elapsed (which results in more fees or interest, as applicable, being paid than if computed on the basis of a 365-day year). Interest shall accrue on each Loan for the day on which the Loan is made, and shall not accrue on a Loan, or any portion thereof, for the day on which the Loan or such portion is paid, provided that any Loan that is repaid on the same day on which it is made shall, subject to Section 2.12(a), bear interest for one day.
- (b) If, as a result of any restatement of or other adjustment to the financial statements of the Borrower or for any other reason, the Borrower or the Lenders determine that (i) the Consolidated Leverage Ratio as calculated by the Borrower as of any applicable date was inaccurate and (ii) a proper calculation of the Consolidated Leverage Ratio would have resulted in higher pricing for such period, the Borrower shall immediately and retroactively be obligated to pay to the Administrative Agent for the account of the applicable Lenders, the L/C Issuers or the Swing Line Lender, as the case may be, promptly on demand by the Administrative Agent (or, after the occurrence of an actual or deemed entry of an order for relief with respect to the Borrower under the Bankruptcy Code of the United States, automatically and without further action by the Administrative Agent, any Lender, any L/C Issuer or the Swing Line Lender), an amount equal to the excess of the amount of interest and fees that should have been paid for such period over the amount of interest and fees actually paid for such period. This paragraph shall not limit the rights of the Administrative Agent, any Lender, any L/C Issuer or the Swing Line Lender, as the case may be, under Section 2.03(c)(iii), 2.03(i) or 2.08(b) or under Article VIII.

2.11 Evidence of Debt.

(a) The Credit Extensions made by each Lender shall be evidenced by one or more accounts or records maintained by such Lender and by the Administrative Agent in the ordinary course of business. The accounts or records maintained by the Administrative Agent and each Lender shall be conclusive absent manifest error of the amount of the Credit Extensions made by the Lenders to the Borrower and the interest and payments thereon. Any failure to so record or any error in doing so shall not, however, limit or otherwise affect the obligation of the Borrower hereunder to pay any amount owing with respect to the Obligations. In the event of any conflict between the accounts and records maintained by any Lender and the accounts and records of the Administrative Agent in respect of such matters, the accounts and records of the Administrative Agent shall control in the absence of manifest error. Upon the request of any Lender made through the Administrative Agent, the Borrower shall execute and deliver to such Lender (through the Administrative Agent) a Note, which shall evidence such Lender's Loans in addition to such accounts or records. Each Lender may attach schedules to its Note and endorse thereon the date, Type (if applicable), amount and maturity of its Loans and payments with respect thereto.

(b) In addition to the accounts and records referred to in <u>subsection (a)</u>, each Lender and the Administrative Agent shall maintain in accordance with its usual practice accounts or records evidencing the purchases and sales by such Lender of participations in Letters of Credit. In the event of any conflict between the accounts and records maintained by the Administrative Agent and the accounts and records of any Lender in respect of such matters, the accounts and records of the Administrative Agent shall control in the absence of manifest error.

2.12 Payments Generally.

- (a) All payments to be made by any Loan Party hereunder or under any other Loan Document shall be made without condition or deduction for any counterclaim, defense, recoupment or setoff. Except as otherwise expressly provided herein, all payments by the Borrower hereunder shall be made to the Administrative Agent, for the account of the respective Lenders to which such payment is owed, at the Administrative Agent's Office in Dollars and in immediately available funds not later than 2:00 p.m. on the date specified herein. The Administrative Agent will promptly distribute to each Lender its Pro Rata Share (or other applicable share as provided herein) of such payment in like funds as received by wire transfer to such Lender's Lending Office. All payments received by the Administrative Agent after 2:00 p.m. shall be deemed received on the next succeeding Business Day and any applicable interest or fee shall continue to accrue.
- (b) If any payment to be made by any Loan Party hereunder or under any other Loan Document shall come due on a day other than a Business Day, payment shall be made on the next following Business Day, and such extension of time shall be reflected in computing interest or fees, as the case may be.
- (c) (i) Unless the Administrative Agent shall have received notice from a Lender prior to the proposed date of any Borrowing of Eurodollar Rate Loans (or, in the case of any Borrowing of Base Rate Loans, prior to 12:00 noon on the date of such Borrowing) that such Lender will not make available to the Administrative Agent such Lender's share of such Borrowing, the Administrative Agent may assume that such Lender has made such share available on such date in accordance with Section 2.02 (or, in the case of a Borrowing of Base Rate Loans, that such Lender has made such share available in accordance with and at the time required by Section 2.02) and may, in reliance upon such assumption, make available to the Borrower a corresponding amount. In such event, if a Lender has not in fact made its share of the applicable Borrowing available to the Administrative Agent, then the applicable Lender and the Borrower severally agree to pay to the Administrative Agent forthwith on demand such corresponding amount with interest thereon, for each day from and including the date such amount is made available to the Borrower to but excluding the date of payment to the Administrative Agent, at (A) in the case of a payment to be made by such Lender, the greater of the Federal Funds Rate and a rate determined by the Administrative Agent in accordance with banking industry rules on interbank compensation, and (B) in the case of a payment to be made by the Borrower, the interest rate applicable to Base Rate Loans. If the Borrower and such Lender shall pay such interest to the Administrative Agent for the same or an overlapping period, the Administrative Agent shall promptly remit to the Borrower the amount of such interest paid by the Borrower for such period. If such Lender pays its share of the applicable Borrowing to the Administrative Agent, then the amount so paid shall constitute such Lender's Loan included in such Borrowing. Any payment by the Borrower shall be without prejudice to any claim the Borrower may have against a Len
- (ii) Unless the Administrative Agent shall have received notice from the Borrower prior to the date on which any payment is due to the Administrative Agent for the account of the Lenders or the L/C Issuers hereunder that the Borrower will not make such payment, the Administrative Agent may assume that the Borrower has made such payment on such date in accordance herewith and may, in

reliance upon such assumption, distribute to the Lenders or the L/C Issuers, as the case may be, the amount due. In such event, if the Borrower has not in fact made such payment, then each Lender or L/C Issuer, as the case may be, severally agrees to repay to the Administrative Agent forthwith on demand the amount so distributed to such Lender or such L/C Issuer, with interest thereon, for each day from and including the date such amount is distributed to it to but excluding the date of payment to the Administrative Agent, at the greater of the Federal Funds Rate and a rate determined by the Administrative Agent in accordance with banking industry rules on interbank compensation.

A notice of the Administrative Agent to any Lender or the Borrower with respect to any amount owing under this <u>subsection (c)</u> shall be conclusive, absent manifest error.

- (d) If any Lender makes available to the Administrative Agent funds for any Loan to be made by such Lender as provided in the foregoing provisions of this <u>Article II</u>, and such funds are not made available to the Borrower by the Administrative Agent because the conditions to the applicable Credit Extension set forth in <u>Article IV</u> are not satisfied or waived in accordance with the terms hereof, the Administrative Agent shall return such funds (in like funds as received from such Lender) to such Lender, without interest.
- (e) The obligations of the Lenders hereunder to make Loans, to fund participations in Letters of Credit and to make payments pursuant to $\underline{\text{Section}}$ $\underline{11.04(c)}$ are several and not joint. The failure of any Lender to make any Loan, to fund any such participation or to make any payment under $\underline{\text{Section }}$ $\underline{11.04(c)}$ on any date required hereunder shall not relieve any other Lender of its corresponding obligation to do so on such date, and no Lender shall be responsible for the failure of any other Lender to so make its Loan, to purchase its participation or to make its payment under $\underline{\text{Section }}$ $\underline{11.04(c)}$.
- (f) Nothing herein shall be deemed to obligate any Lender to obtain the funds for any Loan in any particular place or manner or to constitute a representation by any Lender that it has obtained or it will obtain the funds for any Loan in any particular place or manner.
- 2.13 **Sharing of Payments by Lenders.** If any Lender shall, by exercising any right of setoff or counterclaim or otherwise, obtain payment in respect of any principal of or interest on any of its Loans made by it, or the participations in L/C Obligations or in Swing Line Loans held by it resulting in such Lender receiving payment of a proportion of the aggregate amount of such Loans or participations and accrued interest thereon greater than its <u>pro rata</u> share thereof as provided herein, then the Lender receiving such greater proportion shall (a) notify the Administrative Agent of such fact, and (b) purchase (for cash at face value) participations in the Loans and subparticipations in L/C Obligations and Swing Line Loans of the other Lenders, or make such other adjustments as shall be equitable, so that the benefit of all such payments shall be shared by the Lenders ratably in accordance with the aggregate amount of principal of and accrued interest on their respective Loans and other amounts owing them, <u>provided</u> that:
- (i) if any such participations are purchased and all or any portion of the payment giving rise thereto is recovered, such participations shall be rescinded and the purchase price restored to the extent of such recovery, without interest; and
- (ii) the provisions of this Section shall not be construed to apply to (x) any payment made by the Borrower pursuant to and in accordance with the express terms of this Agreement (including the application of funds arising from the existence of a Defaulting Lender), or (y) the application of Cash Collateral in accordance with this Agreement, or (z) any payment obtained by a Lender as consideration for the assignment of or sale of a participation in any of its Loans or participations in L/C Obligations or Swing Line Loans to any assignee or participant, other than to the Borrower or any Subsidiary thereof (as to which the provisions of this Section shall apply).

The Borrower consents to the foregoing and agrees, to the extent it may effectively do so under applicable law, that any Lender acquiring a participation pursuant to the foregoing arrangements may exercise against the Borrower rights of setoff and counterclaim with respect to such participation as fully as if such Lender were a direct creditor of the Borrower in the amount of such participation.

2.14 Increase in Commitments.

- (a) Provided there exists no Default, upon notice to the Administrative Agent (which shall promptly notify the Lenders), the Borrower may from time to time, (with the approval of the Administrative Agent, which approval shall not be unreasonably withheld, conditioned or delayed) request an increase in the Aggregate Commitments by an amount (for all such requests) not exceeding \$250,000,000; provided that any such request for an increase shall be in a minimum amount of \$25,000,000. To achieve the requested increase, the Borrower may ask one or more Lenders to increase their existing Commitments and/or invite additional Eligible Assignees to become Lenders pursuant to a joinder agreement in form and substance reasonably satisfactory to the Administrative Agent. For the avoidance of doubt, no Lender shall be obligated to increase its Commitment pursuant to this Section 2.14.
- (b) If the Aggregate Commitments are increased in accordance with this Section, the Administrative Agent and the Borrower shall determine the effective date (the "Increase Effective Date") and the final allocation of such increase. The Administrative Agent shall promptly notify the Borrower and the Lenders of the final allocation of such increase and the Increase Effective Date. As a condition precedent to such increase, the Borrower shall deliver to the Administrative Agent a certificate dated as of the Increase Effective Date signed by a Responsible Officer (i) certifying and attaching the resolutions approving or consenting to such increase, and (ii) certifying that, before and after giving effect to such increase (A) the representations and warranties contained in Article V are true and correct in all material respects (except that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof) on and as of the Increase Effective Date, except to the extent that such representations and warranties specifically refer to an earlier date, in which case they are true and correct as of such earlier date, and except that for purposes of this Section 2.14, the representations and warranties contained in Sections 5.06(a) and 5.06(b) shall be deemed to refer to the most recent statements furnished pursuant to subsections (a) and (b), respectively, of Section 6.01, and (B) no Default exists. The Borrower shall prepay any Loans outstanding on the Increase Effective Date (and pay any additional amounts required pursuant to Section 3.05) if necessary to keep the outstanding Loans ratable with any revised Pro Rata Shares arising from any nonratable increase in the Commitments under this Section.
 - (c) This Section shall supersede any provisions in <u>Section 2.13</u> or <u>11.01</u> to the contrary.
- 2.15 **Defaulting Lenders.** Notwithstanding anything to the contrary contained in this Agreement, if any Lender becomes a Defaulting Lender, then, until such time as such Lender is no longer a Defaulting Lender, to the extent permitted by applicable Laws:
- (a) <u>Waivers and Amendments</u>. Such Defaulting Lender's right to approve or disapprove any amendment, waiver or consent with respect to this Agreement shall be restricted as set forth in the definition of Required Lenders, subject to the penultimate sentence of <u>Section 11.01</u>.

(b) <u>Defaulting Lender Waterfall</u>. Any payment of principal, interest, fees or other amounts received by the Administrative Agent for the account of such Defaulting Lender (whether voluntary or mandatory, at maturity, pursuant to Article VIII or otherwise) or received by the Administrative Agent from a Defaulting Lender pursuant to Section 11.08 shall be applied at such time or times as may be determined by the Administrative Agent as follows: first, to the payment of any amounts owing by such Defaulting Lender to the Administrative Agent hereunder; second, to the payment on a pro rata basis of any amounts owing by such Defaulting Lender to the L/C Issuers or Swing Line Lender hereunder; third, to Cash Collateralize each L/C Issuer's Fronting Exposure with respect to such Defaulting Lender in accordance with Section 2.03(g); fourth, as the Borrower may request (so long as no Default exists), to the funding of any Loan in respect of which such Defaulting Lender has failed to fund its portion thereof as required by this Agreement, as determined by the Administrative Agent; fifth, if so determined by the Administrative Agent and the Borrower, to be held in a deposit account and released pro rata in order to (x) satisfy such Defaulting Lender's potential future funding obligations with respect to Loans under this Agreement and (y) Cash Collateralize the L/C Issuers' future Fronting Exposure with respect to such Defaulting Lender with respect to future Letters of Credit issued under this Agreement, in accordance with Section 2.03(g); sixth, to the payment of any amounts owing to the Lenders, the L/C Issuers or Swing Line Lender as a result of any judgment of a court of competent jurisdiction obtained by any Lender, any L/C Issuer or Swing Line Lender against such Defaulting Lender as a result of such Defaulting Lender's breach of its obligations under this Agreement; seventh, so long as no Default or Event of Default exists, to the payment of any amounts owing to the Borrower as a result of any judgment of a court of competent jurisdiction obtained by the Borrower against such Defaulting Lender as a result of such Defaulting Lender's breach of its obligations under this Agreement; and eighth, to such Defaulting Lender or as otherwise directed by a court of competent jurisdiction; provided that if (x) such payment is a payment of the principal amount of any Loans or L/C Borrowings in respect of which such Defaulting Lender has not fully funded its appropriate share, and (y) such Loans were made or the related Letters of Credit were issued at a time when the conditions set forth in Section 4.03 were satisfied or waived, such payment shall be applied solely to pay the Loans of, and L/C Borrowings owed to, all non-Defaulting Lenders on a pro rata basis prior to being applied to the payment of any Loans of, or L/C Borrowings owed to, such Defaulting Lender until such time as all Loans and funded and unfunded participations in L/C Obligations and Swing Line Loans are held by the Lenders pro rata in accordance with the Commitments hereunder without giving effect to Section 2.15(d). Any payments, prepayments or other amounts paid or payable to a Defaulting Lender that are applied (or held) to pay amounts owed by a Defaulting Lender or to post Cash Collateral pursuant to this Section 2.15(b) shall be deemed paid to and redirected by such Defaulting Lender, and each Lender irrevocably consents hereto.

(c) Certain Fees.

- (i) No Defaulting Lender shall be entitled to receive any Commitment Fee for any period during which that Lender is a Defaulting Lender (and the Borrower shall not be required to pay any such fee that otherwise would have been required to have been paid to that Defaulting Lender).
- (ii) Each Defaulting Lender shall be entitled to receive Letter of Credit Fees for any period during which that Lender is a Defaulting Lender only to the extent allocable to its Pro Rata Share of the stated amount of Letters of Credit for which it has provided Cash Collateral pursuant to <u>Section 2.03(g)</u>.
- (iii) With respect to any Commitment Fee or Letter of Credit Fee not required to be paid to any Defaulting Lender pursuant to <u>clause (i)</u> or <u>(ii)</u> above, the Borrower shall (A) pay to each non-Defaulting Lender that portion of any such fee otherwise payable to such Defaulting Lender with respect to such Defaulting Lender's participation in L/C Obligations or Swing Line

Loans that has been reallocated to such non-Defaulting Lender pursuant to <u>Section 2.15(d)</u>, (B) pay to the L/C Issuers and the Swing Line Lender, as applicable, the amount of any such fee otherwise payable to such Defaulting Lender to the extent allocable to the L/C Issuers' or Swing Line Lender's Fronting Exposure to such Defaulting Lender, and (C) not be required to pay the remaining amount of any such fee.

- (d) Reallocation of Participations to Reduce Fronting Exposure. All or any part of such Defaulting Lender's participation in L/C Obligations and Swing Line Loans shall be reallocated among the non-Defaulting Lenders in accordance with their respective Pro Rata Shares (calculated without regard to such Defaulting Lender's Commitment) but only to the extent that (i) the conditions set forth in Section 4.03 are satisfied at the time of such reallocation (and, unless the Borrower shall have otherwise notified the Administrative Agent at such time, the Borrower shall be deemed to have represented and warranted that such conditions are satisfied at such time), and (ii) such reallocation does not cause the aggregate Outstanding Amount of any non-Defaulting Lender to exceed such non-Defaulting Lender's Commitment. No reallocation hereunder shall constitute a waiver or release of any claim of any party hereunder against a Defaulting Lender arising from that Lender having become a Defaulting Lender, including any claim of a non-Defaulting Lender as a result of such non-Defaulting Lender's increased exposure following such reallocation.
- (e) <u>Cash Collateral, Repayment of Swing Line Loans</u>. If the reallocation described in <u>Section 2.15(d)</u> cannot, or can only partially, be effected, the Borrower shall, without prejudice to any right or remedy available to it hereunder or under law, (i) <u>first</u>, prepay Swing Line Loans in an amount equal to the Swing Line Lender's Fronting Exposure and (ii) <u>second</u>, Cash Collateralize the L/C Issuers' Fronting Exposure in accordance with the procedures set forth in <u>Section 2.03(g)</u>.
- (f) <u>Defaulting Lender Cure</u>. If the Borrower, the Administrative Agent, the Swing Line Lender and the L/C Issuers agree in writing that a Lender is no longer a Defaulting Lender, the Administrative Agent will so notify the parties hereto, whereupon as of the effective date specified in such notice and subject to any conditions set forth therein (which may include arrangements with respect to any Cash Collateral), that Lender will, to the extent applicable, purchase at par that portion of outstanding Loans of the other Lenders or take such other actions as the Administrative Agent may determine to be necessary to cause the Loans and funded and unfunded participations in Letters of Credit and Swing Line Loans to be held pro rata by the Lenders in accordance with the Commitments under this Agreement (without giving effect to <u>Section 2.15(d)</u>), whereupon such Lender will cease to be a Defaulting Lender; <u>provided</u> that no adjustments will be made retroactively with respect to fees accrued or payments made by or on behalf of the Borrower while that Lender was a Defaulting Lender; and <u>provided</u>, <u>further</u>, that except to the extent otherwise expressly agreed by the affected parties, no change hereunder from Defaulting Lender to Lender will constitute a waiver or release of any claim of any party hereunder arising from that Lender's having been a Defaulting Lender.
- (g) New Swing Line Loans/Letters of Credit. So long as any Lender is a Defaulting Lender, (i) the Swing Line Lender shall not be required to fund any Swing Line Loans unless it is satisfied that it will have no Fronting Exposure after giving effect to such Swing Line Loan and (ii) no L/C Issuer shall be required to issue, extend, renew or increase any Letter of Credit unless it is satisfied that it will have no Fronting Exposure after giving effect thereto.
- (h) <u>Replacement of Defaulting Lender</u>. So long as any Lender is a Defaulting Lender, such Defaulting Lender may be replaced in accordance with <u>Section 11.14</u>.

ARTICLE III. TAXES, YIELD PROTECTION AND ILLEGALITY

3.01 Taxes.

(a) <u>Defined Terms</u>. For purposes of this <u>Section 3.01</u>, the term "Lender" includes each L/C Issuer and the Swing Line Lender and the term "applicable law" includes FATCA.

(b) Payments Free of Taxes.

Any and all payments by or on account of any obligation of any Loan Party under any Loan Document shall to the extent permitted by applicable Laws be made without deduction or withholding for any Taxes, except as required by applicable law. If any applicable law (as determined in the good faith discretion of an applicable Withholding Agent) requires the deduction or withholding of any Tax from any such payment by a Withholding Agent, then the applicable Withholding Agent shall be entitled to make such deduction or withholding and shall timely pay the full amount deducted or withheld to the relevant Governmental Authority in accordance with applicable law and, if such Tax is an Indemnified Tax, then the sum payable by the applicable Loan Party shall be increased as necessary so that after such deduction or withholding has been made (including such deductions and withholdings applicable to additional sums payable under this Section) the applicable Recipient receives an amount equal to the sum it would have received had no such deduction or withholding been made.

- (c) <u>Payment of Other Taxes by the Borrower</u>. The Loan Parties shall timely pay to the relevant Governmental Authority in accordance with applicable law, or at the option of the Administrative Agent timely reimburse it for the payment of, any Other Taxes.
- (d) <u>Indemnification by the Borrower</u>. The Loan Parties shall jointly and severally indemnify each Recipient, within 10 days after demand therefor, for the full amount of any Indemnified Taxes (including Indemnified Taxes imposed or asserted on or attributable to amounts payable under this Section) payable or paid by such Recipient or required to be withheld or deducted from a payment to such Recipient and any reasonable expenses arising therefrom or with respect thereto, whether or not such Indemnified Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. A certificate as to the amount of such payment or liability delivered to the Borrower by a Lender (with a copy to the Administrative Agent), or by the Administrative Agent on its own behalf or on behalf of a Lender, shall be conclusive absent manifest error.
- (e) <u>Indemnification by the Lenders</u>. Each Lender shall severally indemnify the Administrative Agent, within 10 days after demand therefor, for (i) any Indemnified Taxes attributable to such Lender (but only to the extent that a Loan Party has not already indemnified the Administrative Agent for such Indemnified Taxes and without limiting the obligation of the Loan Parties to do so), (ii) any Taxes attributable to such Lender's failure to comply with the provisions of <u>Section 11.06(f)</u> relating to the maintenance of a Participant Register and (iii) any Excluded Taxes attributable to such Lender, in each case, that are payable or paid by the Administrative Agent in connection with any Loan Document, and any reasonable expenses arising therefrom or with respect thereto, whether or not such Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. A certificate as to the amount of such payment or liability delivered to any Lender by the Administrative Agent shall be conclusive absent manifest error. Each Lender hereby authorizes the Administrative Agent to set off and apply any and all amounts at any time owing to such Lender under any Loan Document or otherwise payable by the Administrative Agent to the Lender from any other source against any amount due to the Administrative Agent under this paragraph (e).

(f) <u>Evidence of Payments</u>. As soon as practicable after any payment of Taxes by a Loan Party to a Governmental Authority pursuant to this <u>Section 3.01</u>, such Loan Party shall deliver to the Administrative Agent the original or a certified copy of a receipt issued by such Governmental Authority evidencing such payment, a copy of the return reporting such payment or other evidence of such payment reasonably satisfactory to the Administrative Agent.

(g) Status of Lenders; Tax Documentation.

- (i) Any Lender that is entitled to an exemption from or reduction of withholding Tax with respect to payments made under any Loan Document shall deliver to the Borrower and the Administrative Agent, at the time or times reasonably requested by the Borrower or the Administrative Agent, such properly completed and executed documentation reasonably requested by the Borrower or the Administrative Agent as will permit such payments to be made without withholding or at a reduced rate of withholding. In addition, any Lender, if reasonably requested by the Borrower or the Administrative Agent, shall deliver such other documentation prescribed by applicable law or reasonably requested by the Borrower or the Administrative Agent as will enable the Borrower or the Administrative Agent to determine whether or not such Lender is subject to backup withholding or information reporting requirements. Notwithstanding anything to the contrary in the preceding two sentences, the completion, execution and submission of such documentation (other than such documentation set forth in Section 3.01(g)(ii)(A), (ii)(B) and (ii)(D) below) shall not be required if in the Lender's reasonable judgment such completion, execution or submission would subject such Lender to any material unreimbursed cost or expense or would materially prejudice the legal or commercial position of such Lender.
 - (ii) Without limiting the generality of the foregoing:
- (A) any Lender that is a U.S. Person shall deliver to the Borrower and the Administrative Agent on or prior to the date on which such Lender becomes a Lender under this Agreement (and from time to time thereafter upon the reasonable request of the Borrower or the Administrative Agent), executed originals of IRS Form W-9 certifying that such Lender is exempt from U.S. federal backup withholding tax;
- (B) any Foreign Lender shall, to the extent it is legally entitled to do so, deliver to the Borrower and the Administrative Agent (in such number of copies as shall be requested by the recipient) on or prior to the date on which such Foreign Lender becomes a Lender under this Agreement (and from time to time thereafter upon the reasonable request of the Borrower or the Administrative Agent), whichever of the following is applicable:
- (i) in the case of a Foreign Lender claiming the benefits of an income tax treaty to which the United States is a party (x) with respect to payments of interest under any Loan Document, executed originals of IRS Form W-8BEN establishing an exemption from, or reduction of, U.S. federal withholding Tax pursuant to the "interest" article of such tax treaty and (y) with respect to any other applicable payments under any Loan Document, IRS Form W-8BEN establishing an exemption from, or reduction of, U.S. federal withholding Tax pursuant to the "business profits" or "other income" article of such tax treaty;

(ii) executed originals of IRS Form W-8ECI;

(iii) in the case of a Foreign Lender claiming the benefits of the exemption for portfolio interest under Section 881(c) of the Code, (x) a certificate substantially in the form of Exhibit E-1 to the effect that such Foreign Lender is not a "bank" within the meaning of Section 881(c)(3)(A) of the Code, a "10 percent shareholder" of the Borrower within the meaning of Section 881(c)(3)(B) of the Code, or a "controlled foreign corporation" described in Section 881(c)(3)(C) of the Code (a "U.S. Tax Compliance Certificate") and (y) executed originals of IRS Form W-8BEN; or

(iv) to the extent a Foreign Lender is not the beneficial owner, executed originals of IRS Form W-8IMY, accompanied by IRS Form W-8ECI, IRS Form W-8BEN, a U.S. Tax Compliance Certificate substantially in the form of Exhibit E-3, IRS Form W-9, and/or other certification documents from each beneficial owner, as applicable; provided that if the Foreign Lender is a partnership and one or more direct or indirect partners of such Foreign Lender are claiming the portfolio interest exemption, such Foreign Lender may provide a U.S. Tax Compliance Certificate substantially in the form of Exhibit E-4 on behalf of each such direct and indirect partner.

(C) any Foreign Lender shall, to the extent it is legally entitled to do so, deliver to the Borrower and the Administrative Agent (in such number of copies as shall be requested by the recipient) on or prior to the date on which such Foreign Lender becomes a Lender under this Agreement (and from time to time thereafter upon the reasonable request of the Borrower or the Administrative Agent), executed originals of any other form prescribed by applicable law as a basis for claiming exemption from or a reduction in U.S. federal withholding Tax, duly completed, together with such supplementary documentation as may be prescribed by applicable law to permit the Borrower or the Administrative Agent to determine the withholding or deduction required to be made; and

(D) if a payment made to a Lender under any Loan Document would be subject to U.S. federal withholding Tax imposed by FATCA if such Lender were to fail to comply with the applicable reporting requirements of FATCA (including those contained in Section 1471(b) or 1472(b) of the Code, as applicable), such Lender shall deliver to the Borrower and the Administrative Agent at the time or times prescribed by law and at such time or times reasonably requested by the Borrower or the Administrative Agent such documentation prescribed by applicable law (including as prescribed by Section 1471(b)(3)(C)(i) of the Code) and such additional documentation reasonably requested by the Borrower or the Administrative Agent as may be necessary for the Borrower and the Administrative Agent to comply with their obligations under FATCA and to determine that such Lender has complied with such Lender's obligations under FATCA or to determine the amount to deduct and withhold from such payment. Solely for purposes of this clause (D), "FATCA" shall include any amendments made to FATCA after the date of this Agreement.

Each Lender agrees that if any form or certification it previously delivered expires or becomes obsolete or inaccurate in any respect, it shall update such form or certification or promptly notify the Borrower and the Administrative Agent in writing of its legal inability to do so. For purposes of this Section 3.01(g), references to a Lender shall include the Administrative Agent.

- (h) Treatment of Certain Refunds. If any party determines, in its sole discretion exercised in good faith, that it has received a refund of any Taxes as to which it has been indemnified pursuant to this Section 3.01 (including by the payment of additional amounts pursuant to this Section 3.01), it shall pay to the indemnifying party an amount equal to such refund (but only to the extent of indemnity payments made under this Section with respect to the Taxes giving rise to such refund), net of all out-of-pocket expenses (including Taxes) of such indemnified party and without interest (other than any interest paid by the relevant Governmental Authority with respect to such refund). Such indemnifying party, upon the request of such indemnified party, shall repay to such indemnified party the amount paid over pursuant to this paragraph (h) (plus any penalties, interest or other charges imposed by the relevant Governmental Authority) in the event that such indemnified party is required to repay such refund to such Governmental Authority. Notwithstanding anything to the contrary in this paragraph (h), in no event will the indemnified party be required to pay any amount to an indemnifying party pursuant to this paragraph (h) the payment of which would place the indemnified party in a less favorable net after-Tax position than the indemnified party would have been in if the Tax subject to indemnification and giving rise to such refund had not been deducted, withheld or otherwise imposed and the indemnification payments or additional amounts with respect to such Tax had never been paid. This paragraph shall not be construed to require any indemnified party to make available its Tax returns (or any other information relating to its Taxes that it deems confidential) to the indemnifying party or any other Person.
- (i) <u>Survival</u>. Each party's obligations under this <u>Section 3.01</u> shall survive the resignation or replacement of the Administrative Agent or any assignment of rights by, or the replacement of, a Lender, the termination of the Commitments and the repayment, satisfaction or discharge of all obligations under any Loan Document.
- (j) <u>Designation of a Different Lending Office</u>. If any Lender requests compensation under <u>Section 3.01(b)</u>, or requires the Borrower to make any payments pursuant to <u>Section 3.01(c)</u>, then such Lender shall (at the request of the Borrower) use reasonable efforts to designate a different Lending Office for funding or booking its Loans and, if applicable, Letters of Credit hereunder or to assign its rights and obligations hereunder to another of its offices, branches or affiliates, if, in the judgment of such Lender, such designation or assignment (i) would eliminate or reduce amounts payable pursuant to <u>Section 3.01(b)</u> or <u>Section 3.01(c)</u> in the future, and (ii) would not subject such Lender to any unreimbursed cost or expense and would not otherwise be disadvantageous to such Lender. If the Borrower requests a Lender to designate a different Lending Office or assign its rights and obligations to another of its offices, branches or affiliates, the Borrower hereby agrees to pay all reasonable costs and expenses incurred by such Lender in connection with any such designation or assignment. Subject to the foregoing, Lenders agree to use reasonable efforts to select lending offices which will minimize taxes and other costs and expenses for the Borrower.

3.02 Illegality.

(a) If any Lender determines that any Law has made it unlawful, or that any Governmental Authority has asserted that it is unlawful, for any Lender or its applicable Lending Office to make, maintain or fund Eurodollar Rate Loans, or to determine or charge interest rates based upon the Eurodollar Rate, then, on notice thereof by such Lender to the Borrower through the Administrative Agent, any obligation of such Lender to make or continue Eurodollar Rate Loans or to convert Base Rate Loans to Eurodollar Rate Loans shall be suspended until such Lender notifies the Administrative Agent and the Borrower that the circumstances giving rise to such determination no longer exist. Upon receipt of such notice, the Borrower shall, upon demand from such Lender (with a copy to the Administrative Agent), prepay or, if applicable, convert all Eurodollar Rate Loans of such Lender to Base Rate Loans, either on

the last day of the Interest Period therefor, if such Lender may lawfully continue to maintain such Eurodollar Rate Loans to such day, or immediately, if such Lender may not lawfully continue to maintain such Eurodollar Rate Loans. Upon any such prepayment or conversion, the Borrower shall also pay accrued interest on the amount so prepaid or converted. Each Lender agrees to designate a different Lending Office if such designation will avoid the need for such notice and will not, in the good faith judgment of such Lender, otherwise be materially disadvantageous to such Lender.

(b) If Export Development Corporation as a Lender ("EDC") determines, acting reasonably, that any applicable law of the United States or any State thereof has made it unlawful, or that any Governmental Authority asserts that it is unlawful, for EDC as a Lender to hold or benefit from a Lien on Real Property pursuant to any law of the United States or any State thereof, EDC may notify the Administrative Agent and may disclaim any benefit of such security interest to the extent of such illegality, but such determination or disclaimer shall not invalidate, render unenforceable or otherwise adversely affect in any manner such Lien for the benefit of any other Lender or other holder of Secured Obligations.

3.03 **Inability to Determine Rates.** If the Required Lenders determine that for any reason adequate and reasonable means do not exist for determining the Eurodollar Rate for any requested Interest Period with respect to a proposed Eurodollar Rate Loan, or that the Eurodollar Rate for any requested Interest Period with respect to a proposed Eurodollar Rate Loan does not adequately and fairly reflect the cost to such Lenders of funding such Loan, the Administrative Agent will promptly so notify the Borrower and each Lender. Thereafter, the obligation of the Lenders to make or maintain Eurodollar Rate Loans shall be suspended until the Administrative Agent (upon the instruction of the Required Lenders) revokes such notice. Upon receipt of such notice, the Borrower may revoke any pending request for a Borrowing of, conversion to or continuation of Eurodollar Rate Loans or, failing that, will be deemed to have converted such request into a request for a Borrowing of Base Rate Loans in the amount specified therein.

3.04 Increased Cost and Reduced Return; Capital Adequacy; Reserves on Eurodollar Rate Loans.

- (a) If any Change in Law shall:
- (i) impose, modify or deem applicable any reserve, special deposit, compulsory loan, insurance charge or similar requirement against assets of, deposits with or for the account of, or credit extended or participated in by, any Lender (except any reserve requirement contemplated by <u>Section 3.04(e)</u>) or any L/C Issuer;
- (ii) subject any Recipient to any Taxes (other than (A) Indemnified Taxes, (B) Taxes described in clauses (b) through (d) of the definition of Excluded Taxes and (C) Connection Income Taxes) on its loans, loan principal, letters of credit, commitments, or other obligations, or its deposits, reserves, other liabilities or capital attributable thereto; or
- (iii) impose on any Lender or L/C Issuer or the London interbank market any other condition, cost or expense (other than Taxes) affecting this Agreement or Loans made by such Lender or any Letter of Credit or participation therein;

and the result of any of the foregoing shall be to increase the cost to such Lender of making, continuing, converting to or maintaining any Loan the interest on which is determined by reference to the Eurodollar Rate (or of maintaining its obligation to make any such Loan), or to increase the cost to such Lender or the L/C Issuer of participating in, issuing or maintaining any Letter of Credit (or of maintaining its

obligation to participate in or to issue any Letter of Credit), or to reduce the amount of any sum received or receivable by such Lender or the L/C Issuer hereunder (whether of principal, interest or any other amount) then, upon request of such Lender or the L/C Issuer, the Borrower will pay to such Lender or the L/C Issuer, as the case may be, such additional amount or amounts as will compensate such Lender or the L/C Issuer, as the case may be, for such additional costs incurred or reduction suffered.

- (b) If any Lender or L/C Issuer determines that any Change in Law affecting such Lender or L/C Issuer or any Lending Office of such Lender or such Lender's or L/C Issuer's holding company, if any, regarding capital or liquidity requirements has or would have the effect of reducing the rate of return on such Lender's or L/C Issuer's capital or on the capital of such Lender's or L/C Issuer's holding company, if any, as a consequence of this Agreement, the Commitments of such Lender or the Loans made by, or participations in Letters of Credit held by, such Lender, or the Letters of Credit issued by such L/C Issuer, to a level below that which such Lender or such L/C Issuer or such Lender's or such L/C Issuer's holding company could have achieved but for such Change in Law (taking into consideration such Lender's or such L/C Issuer's policies and the policies of such Lender's or such L/C Issuer's holding company with respect to capital adequacy), then from time to time the Borrower will pay to such Lender or such L/C Issuer, as the case may be, such additional amount or amounts as will compensate such Lender or such L/C Issuer or such L/C Issuer's holding company for any such reduction suffered.
- (c) A certificate of a Lender or a L/C Issuer setting forth the amount or amounts necessary to compensate such Lender or such L/C Issuer or its holding company, as the case may be, as specified in subsection (a) or (b) of this Section and delivered to the Borrower shall be conclusive absent manifest error. The Borrower shall pay such Lender or such L/C Issuer, as the case may be, the amount shown as due on any such certificate within 15 days after receipt thereof.
- (d) Failure or delay on the part of any Lender or L/C Issuer to demand compensation pursuant to the foregoing provisions of this Section shall not constitute a waiver of such Lender's or L/C Issuer's right to demand such compensation, <u>provided</u> that the Borrower shall not be required to compensate a Lender or L/C Issuer pursuant to the foregoing provisions of this Section for any increased costs incurred or reductions suffered more than 270 days prior to the date that such Lender or L/C Issuer, as the case may be, notifies the Borrower of the Change in Law giving rise to such increased costs or reductions and of such Lender's or L/C Issuer's intention to claim compensation therefor (except that, if the Change in Law giving rise to such increased costs or reductions is retroactive, then the 270-day period referred to above shall be extended to include the period of retroactive effect thereof).
- (e) The Borrower shall pay to each Lender, as long as such Lender shall be required to maintain reserves with respect to liabilities or assets consisting of or including Eurocurrency funds or deposits (currently known as "Eurocurrency liabilities"), additional interest on the unpaid principal amount of each Eurodollar Rate Loan, and of each Base Rate Loan bearing interest based on the Eurodollar Rate, equal to the actual costs of such reserves allocated to such Loan by such Lender (as determined by such Lender in good faith, which determination shall be conclusive), which shall be due and payable on each date on which interest is payable on such Loan, provided the Borrower shall have received at least 15 days' prior notice (with a copy to the Administrative Agent) of such additional interest from such Lender. If a Lender fails to give notice 15 days prior to the relevant Interest Payment Date, such additional interest shall be due and payable 15 days from receipt of such notice.
- 3.05 **Compensation for Losses.** Upon demand of any Lender (with a copy to the Administrative Agent) from time to time, the Borrower shall promptly compensate such Lender for and hold such Lender harmless from any loss, cost or expense incurred by it as a result of:

- (a) any continuation, conversion, payment or prepayment of any Loan other than a Base Rate Loan on a day other than the last day of the Interest Period for such Loan (whether voluntary, mandatory, automatic, by reason of acceleration, or otherwise); or
- (b) any failure by the Borrower (for a reason other than the failure of such Lender to make a Loan) to prepay, borrow, continue or convert any Loan other than a Base Rate Loan on the date or in the amount notified by the Borrower; or
- (c) any assignment of a Eurodollar Rate Loan on a day other than the last day of the Interest Period therefor as a result of a request by the Borrower pursuant to Section 11.14;

including any loss of anticipated profits and any loss or expense arising from the liquidation or reemployment of funds obtained by it to maintain such Loan or from fees payable to terminate the deposits from which such funds were obtained. The Borrower shall also pay any customary administrative fees charged by such Lender in connection with the foregoing.

For purposes of calculating amounts payable by the Borrower to the Lenders under this <u>Section 3.05</u>, each Lender shall be deemed to have funded each Eurodollar Rate Loan made by it at the Eurodollar Rate for such Loan by a matching deposit or other borrowing in the London interbank eurodollar market for a comparable amount and for a comparable period, whether or not such Eurodollar Rate Loan was in fact so funded.

3.06 Matters Applicable to all Requests for Compensation.

- (a) A certificate of the Administrative Agent or any Lender claiming compensation under this <u>Article III</u> and setting forth the additional amount or amounts to be paid to it hereunder shall be conclusive in the absence of manifest error. In determining such amount, the Administrative Agent or such Lender may use any reasonable averaging and attribution methods.
- (b) Upon any Lender's making a claim for compensation under $\underline{\text{Section 3.01}}$ or $\underline{3.04}$ or if the Borrower is required to pay any amount for the account of any Lender or L/C Issuer pursuant to $\underline{\text{Section 3.01}}$, the Borrower may replace such Lender in accordance with $\underline{\text{Section 11.14}}$.
- 3.07 **Survival.** All of the Borrower's obligations under this <u>Article III</u> shall survive termination of the Aggregate Commitments and repayment of all other Obligations hereunder.

ARTICLE IV. CONDITIONS PRECEDENT TO CREDIT EXTENSIONS

- 4.01 **Conditions to Effectiveness of this Agreement (Execution Date).** This Agreement shall be effective upon satisfaction (or waiver in accordance with <u>Section 11.01</u>) of the conditions precedent set forth in this <u>Section 4.01</u>; <u>provided</u> that the obligations of the Lenders to make Credit Extensions hereunder are subject to satisfaction (or waiver in accordance with <u>Section 11.01</u>) of the conditions precedent set forth in <u>Section 4.02</u> and <u>Section 4.03</u>:
- (a) The Administrative Agent's receipt of the following, each of which shall be originals or facsimiles (followed promptly by originals) unless otherwise specified, each dated the Execution Date (or, in the case of certificates of governmental officials, a recent date before the Execution Date):

- (i) executed counterparts of this Agreement, in the number requested by the Administrative Agent;
- (ii) the following, each in form and substance satisfactory to the Administrative Agent: a certificate of the secretary or an assistant secretary of the General Partner on behalf of the Parent Guarantor and the Borrower certifying as to the incumbency and genuineness of the signature of each officer of the General Partner executing the Loan Documents and certifying that attached thereto is a true, correct and complete copy of (A) the certificate of limited partnership or formation of such party and all amendments thereto, certified as of a recent date by the appropriate Governmental Authority in its jurisdiction of incorporation or formation, (B) the limited partnership agreement, limited liability company agreement or other governing document of such party as in effect on the Execution Date, and (C) resolutions duly adopted by the board of directors (or other governing body) of such party authorizing and approving the transactions contemplated hereunder and the execution, delivery and performance of this Agreement and the other Loan Documents to which it is a party;
- (iii) a certificate of a Responsible Officer of the General Partner on behalf of the Parent Guarantor and the Borrower stating that all governmental and regulatory approvals necessary in connection with execution and delivery of this Agreement by the Parent Guarantor and the Borrower shall have been obtained and be in full force and effect or stating that no such approvals are required; and
- (iv) certificates evidencing existence and good standing of each Loan Party, issued by the applicable Governmental Authority of the state where each is organized.
- (b) The Lenders shall have received such documentation and other information as may be required by them in order to enable compliance with applicable "know your customer" and anti-money laundering rules and regulations, including the information required by the USA PATRIOT Act and information described in <u>Section 11.19</u>.

Without limiting the generality of the provisions of the last paragraph of <u>Section 9.03</u>, for purposes of determining compliance with the conditions specified in this <u>Section 4.01</u>, each Lender that has signed this Agreement shall be deemed to have consented to, approved or accepted or to be satisfied with, each document or other matter required thereunder to be consented to or approved by or acceptable or satisfactory to a Lender unless the Administrative Agent shall have received notice from such Lender prior to the proposed Execution Date specifying its objection thereto.

The Administrative Agent shall notify the Lenders and the Borrower of the Execution Date, and such notice shall be conclusive and binding.

- 4.02 **Conditions of the Closing Date.** The obligation of each Lender to make its initial Credit Extension hereunder is subject to the occurrence of the Execution Date and satisfaction (or waiver in accordance with <u>Section 11.01</u>) of the following conditions precedent:
- (a) The Administrative Agent's receipt of the following, each of which shall be originals or facsimiles (followed promptly by originals) unless otherwise specified, each properly executed by a Responsible Officer to the extent required to be executed, each dated the Closing Date (or, in the case of certificates of governmental officials, a recent date before the Closing Date) and each in form and substance reasonably satisfactory to the Administrative Agent and the Required Lenders:

- (i) a Note executed by the Borrower in favor of each Lender requesting a Note;
- (ii) executed counterparts of the Subsidiary Guaranty Agreement;
- (iii) executed counterparts of the Security Agreement and such other security agreements, if any, as may be required to be delivered pursuant to Section 6.11, together with
- (A) original stock certificates or other certificates evidencing the Equity Interests pledged pursuant thereto, an undated stock power for each such certificate duly executed in blank by the registered owner thereof and, in the case of a pledge of Equity Interests issued by a Person who is not a party to the Security Agreement, an acknowledgment of pledge executed by such Person;
- (B) copies of the Organization Documents of each issuer of Equity Interests being pledged to the extent not previously delivered pursuant to Section 4.01;
- (C) each original promissory note, if any, pledged pursuant to the Security Agreements together with an undated endorsement for each such promissory note duly executed in blank by the holder thereof;
- (D) financing statements in form appropriate for filing under the Uniform Commercial Code of all jurisdictions in which the Administrative Agent may deem necessary or desirable in order to perfect the Liens created under the Security Agreements, covering the Collateral, provided however that no mortgages or fixture filings shall be required to be filed in county recording offices;
- (E) such UCC-3 termination statements as may be required in order to release Liens on the Collateral, if any, other than Liens permitted by this Agreement;
 - (F) such lien searches as the Administrative Agent shall reasonably request; and
- (G) evidence of the completion of such other actions that the Administrative Agent may deem reasonably necessary or desirable in order to perfect the Liens created thereby;
- (iv) a favorable opinion of (i) Latham & Watkins, LLP, counsel to the Loan Parties, covering such matters concerning the Loan Parties, the Loan Documents and the Collateral as the Arrangers may reasonably request, (ii) Colorado counsel to the Loan Parties, covering such matters concerning any Loan Party organized in Colorado as the Arrangers may reasonably request and (iii) Utah and Wyoming counsel to the Loan Parties, covering the transmitting utility financing statements to be filed in each such state, in the case of each opinion, in form and substance reasonably satisfactory to the Arrangers;
- (v) financial projections of the Borrower and the Acquired Businesses through December 31, 2015, prepared on a basis consistent with the financial projections of the Borrower and the Acquired Businesses delivered to the Arrangers prior to the Closing Date; and
- (vi) certificates of insurance evidencing the insurance required to be maintained pursuant to this Agreement, naming the Administrative Agent, on behalf of the Secured Parties as an additional insured or loss payee, as the case may be, delivered together with a certificate of a

Responsible Officer of the General Partner on behalf of the Parent Guarantor and the Borrower (which may be combined with the certificate delivered pursuant to Section 4.02(b)(iii)) certifying that all insurance required to be maintained pursuant to this Agreement has been obtained and is in effect.

- (b) The Administrative Agent's receipt of the following, each of which shall be originals or facsimiles (followed promptly by originals) unless otherwise specified, each properly executed by a Responsible Officer to the extent required to be executed, each dated the Closing Date (or, in the case of certificates of government officials, a recent date before the Closing Date) and each in form and substance satisfactory to the Administrative Agent:
- (i) a certificate of a secretary or assistant secretary of the General Partner on behalf of each Loan Party certifying as to the incumbency and genuineness of the signature of each officer of the General Partner executing the Loan Documents and certifying that attached thereto is a true, correct and complete copy of each of the following, or certifying that such documents were delivered on the Execution Date and certifying that since such date there have been no changes thereto: (A) the articles or certificate of limited partnership or formation of such Loan Party and all amendments thereto, certified as of a recent date by the appropriate Governmental Authority in its jurisdiction of incorporation or formation, (B) the limited partnership agreement, limited liability company agreement or other governing document of such Loan Party as in effect on the Closing Date, (C) resolutions duly adopted by the board of directors (or other governing body) of the General Partner authorizing and approving the transactions contemplated hereunder and the execution, delivery and performance of this Agreement and the other Loan Documents to which such Loan Party is a party; and
- (ii) certificates evidencing valid existence and good standing of each Loan Party, issued by the applicable governmental authority of the state where each is organized;
- (iii) a certificate, dated the Closing Date and signed by a Responsible Officer of the General Partner on behalf of the Parent Guarantor and the Borrower certifying
- (A) that each of the Borrower and the Parent Guarantor (on a Consolidated basis together with its Consolidated Subsidiaries) is Solvent on such date after giving effect to the Transactions;
 - (B) that, both before and immediately after giving effect to the Transactions, no Default exists on such date;
- (C) that, both before and immediately after giving effect to the Transactions and the incurrence of Indebtedness on such date, if any, the representations and warranties contained in <u>Article V</u> are true and correct on and as of such date;
- (D) that a true and correct copy of each Material Agreement either (i) has been filed with the SEC by the Parent Guarantor as exhibits to its Registration Statement as in effect on the date of the IPO (with certain terms thereof having been redacted as permitted by applicable SEC regulations) or (ii) is attached to such certificate; and
- (E) as to the matters set forth in <u>paragraphs (e)</u> through (i) of this <u>Section 4.02</u>, in form and substance reasonably satisfactory to the Administrative Agent.

- (c) The Parent Guarantor shall have delivered the Initial Financial Statements to the Administrative Agent, which delivery may be accomplished by the filing of the Registration Statement containing such financial statements.
- (d) The Joint Bookrunners shall be reasonably satisfied with the material terms of the agreements (other than the Underwriting Agreement) filed by the Parent Guarantor with the SEC as exhibits to its Registration Statement as in effect on the date of the IPO (it being understood and agreed that the material terms of any such documents described in detail in the Registration Statement as so described or otherwise provided to the Lenders and the Arrangers prior to the Execution Date shall be deemed to be reasonably satisfactory).
- (e) The Contribution and the IPO shall have been, or contemporaneously with the satisfaction (or waiver in accordance with <u>Section 11.01</u>) of the other conditions precedent set forth in this <u>Section 4.02</u> are being, consummated in all material respects as described in the Registration Statement in compliance in all material respects with applicable law and regulatory approvals.
- (f) All partnership and company, governmental and applicable material third-party consents and approvals necessary in connection with the Transactions shall have been obtained.
- (g) There shall not have occurred since December 31, 2012 any event or condition that has had or would reasonably expected to have, either individually or in the aggregate, a material adverse effect on the business, assets (including the assets or business to be contributed to the Borrower and its Subsidiaries pursuant to the Contribution and IPO), financial condition, or operations of the Borrower and its Subsidiaries, taken as a whole.
- (h) Except as disclosed in the Registration Statement, there is no litigation, arbitration or governmental investigation, proceeding or inquiry as to which there is a reasonable possibility of an adverse determination which (i) would reasonably be expected to have a Material Adverse Effect or (ii) which seeks to prevent, enjoin or delay the Contribution or IPO or the making of the initial Credit Extensions.
- (i) The Credit Agreement dated as of August 25, 2011 among QRI, Wells Fargo Bank, as administrative agent, and the lenders and letter of credit issuers parties thereto and the Term Loan Agreement dated as of April 18, 2012 among QRI, Wells Fargo Bank, as administrative agent, and the lenders parties thereto, shall have been amended, or are being amended effective on the Closing Date, to the extent necessary to permit the Contribution and IPO.
- (j) To the extent not previously delivered, the Lenders shall have received such documentation and other information as may be required by them in order to enable compliance with applicable "know your customer" and anti-money laundering rules and regulations, including the information required by the USA PATRIOT Act including information required by the Act and information described in <u>Section 11.19</u>.
- (k) The Administrative Agent and the Arrangers shall have received all fees and expenses that are due and payable by the Borrower on or before the Closing Date, including fees payable to the Lenders, and including, to the extent invoiced, reimbursement or payment of all expenses (including, without limitation, Attorney Costs of one firm as counsel to the Administrative Agent) required to be reimbursed or paid by the Borrower hereunder.

Without limiting the generality of the provisions of the last paragraph of <u>Section 9.03</u>, for purposes of determining compliance with the conditions specified in this <u>Section 4.02</u>, each Lender that has signed this Agreement shall be deemed to have consented to, approved or accepted or to be satisfied with, each document or other matter required thereunder to be consented to or approved by or acceptable or satisfactory to a Lender unless the Administrative Agent shall have received notice from such Lender prior to the proposed Closing Date specifying its objection thereto.

The Administrative Agent shall notify the Borrower and the Lenders of the Closing Date, and such notice shall be conclusive and binding.

Notwithstanding the foregoing, the obligations of the Lenders to make Loans and of the L/C Issuers to issue Letters of Credit hereunder shall not become effective unless each of the foregoing conditions is satisfied (or waived pursuant to Section 11.01) at or prior to 5 p.m., New York City time, on the ninetieth (90th) day after the Execution Date (and, in the event such conditions are not so satisfied or waived, the Commitments shall terminate at such time).

- 4.03 **Conditions to all Credit Extensions.** The obligation of each Lender to honor any Request for Credit Extension (other than a Loan Notice requesting only the conversion of Loans from one Type to another or the continuation of Eurodollar Rate Loans) is subject to the following conditions precedent:
- (a) The representations and warranties of the Borrower, the Parent Guarantor and the other Loan Parties contained in Article V or any other Loan Document, or which are contained in any document furnished by the Borrower to the Administrative Agent or the Lenders under or in connection with this Agreement, shall be true and correct in all material respects (except that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof) on and as of the date of such Credit Extension, except to the extent that such representations and warranties specifically refer to an earlier date, in which case they shall be true and correct as of such earlier date, and except that for purposes of this Section 4.03(a) the representations and warranties contained in Section 5.06(b) shall be deemed to refer to the most recent statements furnished pursuant to clauses (a) and (b), respectively, of Section 6.01.
 - (b) No Default shall exist or would result from such proposed Credit Extension.
- (c) The Administrative Agent and, if applicable, the applicable L/C Issuer or the Swing Line Lender shall have received a Request for Credit Extension in accordance with the requirements hereof.

Each Request for Credit Extension (other than a Loan Notice requesting only a conversion of Loans to the other Type or a continuation of Eurodollar Rate Loans) submitted by the Borrower shall be deemed to be a representation and warranty that the conditions specified in <u>Sections 4.03(a)</u> and (b) have been satisfied on and as of the date of the applicable Credit Extension.

ARTICLE V. REPRESENTATIONS AND WARRANTIES

Each Loan Party represents and warrants to the Administrative Agent and each Lender that:

5.01 **No Default.** No event has occurred and is continuing which constitutes a Default.

- 5.02 **Organization and Good Standing.** Each Loan Party and each of its Material Subsidiaries is duly organized, validly existing and in good standing under the Laws of its jurisdiction of organization, having all powers required to carry on its business and enter into and carry out the transactions contemplated hereby. Each Loan Party and each of its Material Subsidiaries is duly qualified, in good standing, and authorized to do business in all other jurisdictions wherein the character of the properties owned or held by it or the nature of the business transacted by it makes such qualification necessary, except as would not, individually or in the aggregate, have a Material Adverse Effect.
- 5.03 **Authorization.** Each Loan Party has duly taken all necessary corporate action to authorize the execution and delivery by it of the Loan Documents to which it is a party and to authorize the consummation of the transactions contemplated thereby and the performance of its obligations thereunder. The Borrower is duly authorized to borrow funds hereunder.
- 5.04 **No Conflicts or Consents.** The execution and delivery by each Loan Party of the Loan Documents to which it is a party, the performance of its obligations under such Loan Documents, and the consummation of the transactions contemplated by the Loan Documents, do not and will not (a) conflict with any provision of (i) any Law, (ii) the Organization Documents of any Loan Party, or (iii) any material agreement, judgment, license, order or permit applicable to or binding upon any Loan Party, (b) result in the acceleration of any material Indebtedness owed by any Loan Party, or (c) result in or require the creation of any Lien upon any assets or properties of any Loan Party, except as expressly contemplated in the Loan Documents. Except as expressly contemplated in the Loan Documents and, with respect to the Contribution, except as described in the Registration Statement, no material consent, approval, authorization or order of, and no notice to or filing with, any Governmental Authority or third party is required in connection with the execution, delivery or performance by any Loan Party of any Loan Document to which it is a party or to consummate any transactions contemplated by the Loan Documents.
- 5.05 **Enforceable Obligations.** This Agreement is, and the other Loan Documents when duly executed and delivered will be, legal, valid and binding obligations of each Loan Party that is a party thereto, enforceable in accordance with their terms except as such enforcement may be limited by bankruptcy, insolvency or similar Laws of general application relating to the enforcement of creditors' rights or by general principles of equity, regardless of whether considered in a proceeding in equity or at law.

5.06 Financial Statements.

(a) The Initial Financial Statements (i) fairly present, in all material respects, the financial position of the Acquired Businesses at the respective dates thereof and the results of operations and consolidated cash flows of the Acquired Businesses for the respective periods thereof, and (ii) show all material indebtedness and other liabilities, direct or contingent, of the Parent Guarantor and its Subsidiaries as of the dates thereof that are required to be disclosed under GAAP. The Initial Financial Statements were prepared in accordance with GAAP.

- (b) Beginning with the initial delivery of the financial information required under Section 6.01(a) and Section 6.01(b), the financial information delivered by the Parent Guarantor pursuant to such Sections (i) was prepared in accordance with GAAP consistently applied throughout the period covered thereby, except as otherwise expressly noted therein, and (ii) fairly presents, in all material respects, the Parent Guarantor's Consolidated financial condition as of the date thereof and the results of operations for the period covered thereby, subject, in the case of interim statements, to the absence of footnotes and to normal year-end adjustments.
- (c) Since December 31, 2012, there has been no event or circumstance, either individually or in the aggregate, that has had or could reasonably be expected to have, a Material Adverse Effect.
- 5.07 **Full Disclosure.** No certificate, statement or other information delivered herewith or heretofore by any Loan Party to the Administrative Agent or to any Lender in connection with the negotiation of this Agreement or in connection with any transaction contemplated hereby, taken as a whole, contains any untrue statement of a material fact or omits to state any material fact known to any Loan Party (other than industry-wide risks normally associated with the types of businesses conducted by the Loan Parties) necessary to make the statements contained herein or therein, in light of the circumstances under which they were made, not materially misleading as of the date made or deemed made, <u>provided</u> that, with respect to any projected financial information, the Borrower represents only that such information was prepared in good faith based upon assumptions believed by the Borrower to be reasonable at the time (it being recognized, however, that projections as to future events are not to be viewed as facts and that the actual results during the period or periods covered by any projections may materially differ from the projected results). There is no fact known to any Loan Party (other than industry-wide risks normally associated with the types of businesses conducted by the Loan Parties) that has not been disclosed by the Loan Parties to each Lender in writing which would reasonably be expected to have a Material Adverse Effect.
- 5.08 **Litigation.** Except as disclosed in the Registration Statement, there are no actions, suits or legal, equitable, arbitrative or administrative proceedings pending, or to the knowledge of the Borrower threatened, against any one or more Loan Parties or Subsidiaries before any Governmental Authority which, either individually or in the aggregate, would reasonably be expected to have a Material Adverse Effect, and there are no outstanding judgments, injunctions, writs, rulings or orders by any such Governmental Authority against any Loan Party or any Subsidiary which, either individually or in the aggregate, would reasonably be expected to have a Material Adverse Effect.
- 5.09 **Labor Disputes and Acts of God.** Except as disclosed on <u>Schedule 5.09</u>, neither the business nor the properties of any Loan Party or any Material Subsidiary has been affected by any fire, explosion, accident, strike, lockout or other labor dispute, drought, storm, hail, earthquake, embargo, act of God or of the public enemy or other casualty (whether or not covered by insurance), which, either individually or in the aggregate, would reasonably be expected to have a Material Adverse Effect.
- 5.10 **ERISA**. No ERISA Event has occurred or is reasonably expected to occur that, when taken together with all other such ERISA Events for which liability is reasonably expected to occur, would reasonably be expected to result in a Material Adverse Effect. Each Loan Party and each ERISA Affiliate has fulfilled its obligations under the Pension Funding Rules with respect to each Pension Plan, except to the

extent that such noncompliance would not reasonably be expected to result in a Material Adverse Effect. Except as would not reasonably be expected to result in a Material Adverse Effect, no Loan Party and no ERISA Affiliate has (a) sought a waiver of the minimum funding standard under the Pension Funding Rules in respect of any Plan, (b) failed to make any contribution or payment to any Plan or Multiemployer Plan, or made any amendment to any Plan that has resulted or could result in the imposition of a Lien or the posting of a bond or other security under ERISA or the Code, or (c) incurred any liability under Title IV of ERISA other than a liability to the PBGC for premiums under Section 4007 of ERISA that are not past due.

- 5.11 **Environmental and Other Laws.** Except for matters that, individually or in the aggregate, would not reasonably be expected to result in a Material Adverse Effect, no Loan Party or any Subsidiary thereof (i) has failed to comply with any Environmental Law or to obtain, maintain or comply with any permit, license or other approval required under any Environmental Law or (ii) has become subject to any Environmental Liability.
- 5.12 **Borrower's Subsidiaries; Equity Investments.** As of the Execution Date and as of the Closing Date, neither the Parent Guarantor nor the Borrower has (a) any Subsidiaries other than those disclosed on <u>Schedule 5.12</u>, and all of the outstanding Equity Interests in such Subsidiaries have been validly issued, are fully paid and non-assessable (to the extent applicable) and are owned by the Loan Party indicated on <u>Schedule 5.12</u>, or (b) any equity investment in any other entity other than those disclosed on <u>Schedule 5.12</u>.
- 5.13 **Title to Properties; Licenses.** Each Loan Party and Material Subsidiary has good title to, or valid leasehold interests in, its properties and assets material to its business, except for any failure, defect or other matter that would not, in the aggregate, be reasonably expected to have a Material Adverse Effect. The property of each Loan Party and Material Subsidiary material to the conduct of its business is free and clear of all Liens other than Permitted Liens. Each Loan Party and Material Subsidiary possesses all licenses, permits, franchises, patents, copyrights, trademarks and trade names, and other intellectual property (or otherwise possesses the right to use such intellectual property without violation of the rights of any other Person) which are reasonably necessary to carry out its business as presently conducted and as presently proposed to be conducted hereafter, and no Loan Party or Material Subsidiary is in violation in any material respect of the terms under which it possesses such intellectual property or the right to use such intellectual property, except as would not, individually or in the aggregate, have a Material Adverse Effect.

5.14 Government Regulation.

- (a) No Loan Party or Subsidiary is engaged, or will engage, principally or as one of its important activities, in the business of purchasing or carrying margin stock (within the meaning of Regulation U issued by the FRB), or extending credit for the purpose of purchasing or carrying margin stock. No part of the proceeds of any Loan or Letter of Credit will be used for purchasing or carrying margin stock or for any purpose which violates the provision of Regulation T, U or X of the FRB.
- (b) No Loan Party or Subsidiary is subject to regulation under any Law which regulates the incurring by such Person of Indebtedness. Neither the Borrower nor any of its Subsidiaries is an "investment company" as defined in, or regulated by, The Investment Company Act of 1940.

- 5.15 **Solvency.** The Borrower and its Subsidiaries on a Consolidated basis, and the Parent Guarantor and its Subsidiaries on a Consolidated basis, are Solvent.
- 5.16 **Compliance with Laws and Material Agreements.** Each Loan Party and Restricted Subsidiary is in compliance with (a) all Laws applicable to it or its property or assets and (b) all Material Agreements, except where the failure to do so, individually or in the aggregate, would not reasonably be expected to result in a Material Adverse Effect.
- 5.17 **Taxes.** Each Loan Party and its Restricted Subsidiaries has timely filed or caused to be filed all Tax returns and reports required to have been filed and has paid or caused to be paid all Taxes required to have been paid by it, except (a) Taxes that are being contested in good faith by appropriate proceedings, are disclosed on the financial statements of the Parent Guarantor and, if required by GAAP, for which adequate reserves have been provided in accordance with GAAP or (b) to the extent that the failure to do so would not reasonably be expected to result in a Material Adverse Effect.
- 5.18 **OFAC.** No Loan Party or any Subsidiary (i) is an "enemy" or an "ally of the enemy" within the meaning of Section 2 of the Trading with the Enemy Act of the United States (50 U.S.C. App. §§ 1 et seq.), (ii) is in violation of (A) the Trading with the Enemy Act, (B) any of the foreign assets control regulations of the United States Treasury Department (31 CFR, Subtitle B, Chapter V) or any enabling legislation or executive order relating thereto or (C) the PATRIOT Act, (iii) is a Sanctioned Person or (iv) has more than 10% of its assets in Sanctioned Countries. No part of the proceeds of any Loan or drawings under any Letter of Credit will be used directly or indirectly to fund any operations in, finance any investments or activities in or make any payments to, a Sanctioned Person or a Sanctioned Country.
- 5.19 **Insurance.** Schedule 5.19 sets forth an accurate description of all insurance maintained by the Loan Parties and Restricted Subsidiaries as of the Closing Date. As of the Closing Date, such insurance is in full force and effect and all premiums have been duly paid. The Loan Parties and Restricted Subsidiaries have insurance in such amounts and covering such risks and liabilities as are in accordance with normal industry practice.
- 5.20 **Collateral Documents.** Each Collateral Document creates in favor of the Administrative Agent, for the benefit of the Secured Parties, legal, valid and enforceable Liens on, and security interests in, each Loan Party's right, title and interest in and to the Collateral therein described, and when all appropriate filings or recordings are made in the appropriate offices as may be required under applicable Laws, such Collateral Document will constitute First Priority fully perfected Liens on, and security interests in, all right, title and interest of it in such Collateral, in each case with no other Liens except for Liens that are permitted pursuant to this Agreement.

ARTICLE VI. AFFIRMATIVE COVENANTS

Each Loan Party covenants and agrees that until the full and final payment of the Obligations (other than contingent Obligations for which no claim has been made), termination of the Commitments of all Lenders and L/C Issuers, and termination of all Letters of Credit (or Cash Collateralization thereof as acceptable to the applicable L/C Issuer):

- 6.01 **Books, Financial Statements and Reports.** The Parent Guarantor will, and will cause each of its Subsidiaries to, maintain proper books of record and account in which full and correct entries in conformity with GAAP consistently applied shall be made of financial transactions and matters involving the assets and business of the Borrower and its Subsidiaries. The Parent Guarantor will furnish the following statements and reports to the Administrative Agent (which shall make such information available to the Lenders in accordance with its customary practices) and at the Parent Guarantor's expense:
- (a) Within five (5) Business Days after the date required to be delivered to the SEC, but no later than ninety-five (95) days after the end of each fiscal year, complete consolidated financial statements of the Parent Guarantor together with all notes thereto, which shall be prepared in reasonable detail in accordance with GAAP and shall not be subject to any "going concern" or like qualification or exception or any qualification or exception as to the scope of such audit, together with an unqualified opinion based on an audit using generally accepted auditing standards, by an independent certified public accountant of nationally recognized standing, stating that such consolidated financial statements have been so prepared. These financial statements shall contain a consolidated balance sheet as of the end of such fiscal year and consolidated statements of earnings, of cash flows, and of changes in shareholders' equity for such fiscal year, each setting forth in comparative form the corresponding figures for the preceding fiscal year. On the date of delivery of such financial statements to Administrative Agent, the Parent Guarantor will furnish to Administrative Agent a Compliance Certificate signed by a Responsible Officer of the Parent Guarantor and the Borrower, stating that such financial statements fairly present, in all material respects, the financial condition of the Parent Guarantor, stating that such Person has reviewed the Loan Documents, containing all calculations required to be made to show compliance or non-compliance with the Financial Covenants, containing the information required by Section 6.01(d) and further stating that there is no condition or event at the end of such fiscal year or at the time of such certificate which constitutes a Default or specifying the nature and period of existence of any such condition or event.
- (b) Within five (5) Business Days after the date required to be delivered to the SEC, but no later than fifty (50) days after the end of each of the first three fiscal quarters of the Parent Guarantor, starting with the fiscal quarter ending September 30, 2013, the Parent Guarantor's consolidated balance sheet and income statement as of the end of such fiscal quarter and a consolidated statement of cash flows for the period from the beginning of the then current fiscal year to the end of such fiscal quarter, all in reasonable detail and prepared in accordance with GAAP, subject to changes resulting from normal year-end adjustments. In addition, the Parent Guarantor will, together with each such set of financial statements, furnish a Compliance Certificate signed by a Responsible Officer of the Parent Guarantor and the Borrower stating that such financial statements are accurate and complete (subject to normal year-end adjustments), stating that such Person has reviewed the Loan Documents, containing the calculations required to be made to show compliance or noncompliance with the provisions of the Financial Covenants, containing the information required by Section 6.01(d) and further stating that there is no condition or event at the end of such fiscal quarter or at the time of such certificate which constitutes a Default or specifying the nature and period of existence of any such condition or event.

- (c) Promptly upon their becoming available, the Parent Guarantor shall provide copies of all registration statements, periodic reports and other statements and schedules filed by the Parent Guarantor with any securities exchange, the SEC or any similar Governmental Authority.
 - (d) Each Compliance Certificate shall contain
- (i) with respect to any financial statements relating to a period during which any Specified Non-Wholly Owned Subsidiary or any Unrestricted Subsidiary is a Consolidated Subsidiary of the Parent Guarantor, a reasonably detailed reconciliation of each of the components reflected in the calculations of compliance with Financial Covenants to the corresponding amounts set forth in such financial statements;
- (ii) a statement by a Responsible Officer certifying that the Schedules attached to the Security Agreements are accurate and complete in all material respects, and attaching any changes thereto that may be needed to make them accurate and complete in all material respects; and

Documents required to be delivered pursuant to Section 6.01(a), (b), or (c) (to the extent any such documents are included in materials otherwise filed with the SEC) may be delivered electronically and if so delivered, shall be deemed to have been delivered on the earlier of (i) the date on which the Parent Guarantor posts such documents, or provides a link thereto on the Parent Guarantor's website on the Internet at the website address listed on Schedule 11.02 or at http://www.sec.gov or (ii) the date on which such documents are posted on the Parent Guarantor's behalf on an Internet or intranet website, if any, to which each Lender and the Administrative Agent have access (whether a commercial, third-party website or whether sponsored by the Administrative Agent); provided that: (i) the Parent Guarantor shall deliver electronic or paper copies of such documents to the Administrative Agent if requested and (ii) the Parent Guarantor shall notify (which may be by facsimile or electronic mail) the Administrative Agent of the posting of any such documents and provide to the Administrative Agent electronic versions (i.e., soft copies) of such documents. Notwithstanding anything contained herein, in every instance the Parent Guarantor shall be required to provide electronic copies of the Compliance Certificates required by Sections 6.01(a) and (b) to the Administrative Agent. The Administrative Agent shall have no obligation to request the delivery or to maintain copies of the documents referred to above, and in any event shall have no responsibility to monitor compliance by the Parent Guarantor with any such request for delivery, and each Lender shall be solely responsible for requesting delivery to it or maintaining its copies of such documents.

Each Loan Party hereby acknowledges that (a) the Administrative Agent and/or the Arrangers will make available to the Lenders materials and/or information provided by or on behalf of the Borrower or other Loan Parties hereunder (collectively, "Borrower Materials") by posting the Borrower Materials on IntraLinks, Debt Domain, Syndtrak or another similar electronic system (the "Platform") and (b) certain of the Lenders (each, a "Public Lender") may have personnel who do not wish to receive material non-public information with respect to the Parent Guarantor or its Affiliates, or the respective securities of any of the foregoing, and who may be engaged in investment and other market-related activities with respect to such Persons' securities. The Borrower hereby agrees that (w) all Borrower Materials that are to be made available to Public Lenders shall be clearly and conspicuously marked "PUBLIC" which, at a minimum, shall mean that the word "PUBLIC" shall appear prominently on the first page thereof; (x) by marking Borrower Materials "PUBLIC," each Loan Party shall be deemed to have authorized the Administrative Agent, the Arrangers and the Lenders to treat such Borrower Materials as not containing any material non-public information with respect to any Loan Party or its securities for purposes of United States Federal and state securities laws (provided, however, that to the extent such Borrower Materials constitute Information, they shall be treated as set forth in Section 11.07);

- (y) all Borrower Materials marked "PUBLIC" are permitted to be made available through a portion of the Platform designated "Public Side Information;" and (z) the Administrative Agent and the Arrangers shall be entitled to treat any Borrower Materials that are not marked "PUBLIC" as being suitable only for posting on a portion of the Platform not designated "Public Side Information."
- 6.02 **Other Information and Financial Records; Inspections.** Each Loan Party agrees to (a) keep proper books of record and accounts as may be required or necessary to permit the preparation of financial statements in accordance with GAAP; (b) furnish to each Lender any information which Administrative Agent or any Lender may from time to time reasonably request concerning any covenant, provision or condition of the Loan Documents or any matter in connection with the businesses and operations of any Loan Party and its Subsidiaries; and (c) permit, and cause its Restricted Subsidiaries to permit, representatives appointed by Administrative Agent or any Lender (including independent accountants, auditors, agents, attorneys, appraisers and any other representatives), upon reasonable prior notice, to visit and inspect during normal business hours any of the properties, to examine its books of account and other books and records, to make copies thereof, to write down and record any information such representatives obtain, and to discuss its affairs, finance and accounts with its officers, employees and independent accountants. In the absence of an Event of Default, and notwithstanding anything to the contrary in Section 11.04, the Borrower shall not be required to pay for more than one such visit in any year.
- 6.03 **Notice of Material Events; Notice of Changes in Certain Information Pertaining to the Collateral.** The Borrower will promptly notify the Administrative Agent in writing (and Administrative Agent shall make such information available to the Lenders in accordance with its customary practices), stating that such notice is being given pursuant to this Agreement, of:
- (a) the occurrence of any event (including, without limitation, (i) any default by a Loan Party or any Material Subsidiary under a Contractual Obligation, (ii) the filing or commencement of any action, suit or proceeding by or before any arbitrator or Governmental Authority affecting a Loan Party or any Material Subsidiary, or (iii) the occurrence of any ERISA Event) which has had or would reasonably be expected to have a Material Adverse Effect;
 - (b) the occurrence of any Event of Default; and
- (c) any change in any Loan Party's (i) legal name, (ii) jurisdiction of organization or formation, (iii) form of entity, or (iv) Federal Taxpayer Identification Number. The Borrower agrees to make all filings under the Uniform Commercial Code or otherwise that are reasonably requested by the Administrative Agent in order for the Administrative Agent to continue at all times following such change to have a valid, legal and perfected security interest in all the Collateral to the same extent as before such change.

Each notice delivered under <u>Section 6.03(a)</u> shall be accompanied by a statement of a Responsible Officer setting forth the details of the event or development requiring such notice and any action taken or proposed to be taken with respect thereto.

- 6.04 **Maintenance of Properties.** Each Loan Party will, and will cause its Material Subsidiaries to, keep and maintain all property material to the conduct of its business in good, working condition, ordinary wear and tear excepted, except where failure to do so would not reasonably be expected to have a Material Adverse Effect.
- 6.05 **Maintenance of Existence and Qualifications.** Each Loan Party will maintain its legal existence and good standing under the laws of its state of organization. Each Loan Party (a) will cause its Material Subsidiaries to maintain their existence and good standing under the laws of the state of their organization, (b) will take, and will cause each of its Material Subsidiaries to take, all reasonable steps to maintain the rights, licenses, permits, privileges and franchises material to the conduct of its business, except, in the case of <u>clauses (a)</u> and (b), any failure to maintain, preserve or qualify that would not reasonably be expected to have a Material Adverse Effect; <u>provided</u> that the foregoing shall not prohibit (i) any merger, consolidation, liquidation or dissolution permitted under <u>Section 7.07</u> or (ii) a termination of such existence, good standing, rights, licenses, permits, privileges and franchises of any Material Subsidiary if the Borrower determines in good faith that such termination is in the best interest of the Borrower and would not reasonably be expected to have a Material Adverse Effect.
- 6.06 **Payment of Obligations.** Each Loan Party will, and will cause each of its Restricted Subsidiaries to, pay its obligations, including Tax liabilities, that, if not paid, would reasonably be expected to result in a Material Adverse Effect before the same shall become delinquent or in default, except where (a) the validity or amount thereof is being contested in good faith by appropriate proceedings, and (b) such Loan Party or such Restricted Subsidiary has set aside on its books adequate reserves with respect thereto in accordance with GAAP.
 - 6.07 Insurance. Each Loan Party will, and will cause each of its Restricted Subsidiaries to,
- (a) Maintain, with financially sound and reputable insurance companies, insurance in such amounts and against such risks, including property and casualty insurance and comprehensive general liability insurance, in each case as are customarily maintained by companies engaged in the same or similar businesses operating in the same or similar locations;
- (b) (i) Cause such property insurance covering Collateral to name the Administrative Agent as loss payee; (ii) use commercially reasonable efforts to cause the policy or policies evidencing such insurance to be endorsed or otherwise amended to include a customary lender's loss payable endorsement, in form and substance satisfactory to the Administrative Agent; and (iii) use commercially reasonable efforts to provide that such policy or policies shall not be canceled or not renewed (A) by reason of nonpayment of premium upon not less than ten (10) days' prior written notice thereof by the insurer to the Administrative Agent or (B) for any other reason upon not less than 30 days' prior written notice thereof by the insurer to the Administrative Agent;
 - (c) Cause such comprehensive general liability insurance to name the Administrative Agent as an additional insured;
- (d) To the extent Collateral under a mortgage consists of buildings or mobile homes located on Real Property covered by such mortgage, provide the Administrative Agent with a standard flood hazard determination form for such property and if any such property is located in an area designated a

"flood hazard area" in any Flood Insurance Rate Map published by the Federal Emergency Management Agency (or any successor agency), obtain flood insurance in such amount as the Administrative Agent may from time to time reasonably require to ensure compliance with the National Flood Insurance Program as set forth in the Flood Disaster Protection Act of 1973; and

(e) Upon request by the Administrative Agent, deliver a certificate or certificates of insurance evidencing insurance required to be maintained pursuant to this Section 6.07.

6.08 Compliance with Law.

- (a) Each Loan Party will, and will cause each of its Restricted Subsidiaries to, comply with all laws, rules, regulations and orders of any Governmental Authority applicable to it or its property, except in such circumstances in which (a) a requirement of Law or the necessity to comply therewith is being contested in good faith by appropriate proceedings diligently conducted, or (b) the failure to comply therewith would not be reasonably expected to have a Material Adverse Effect.
- (b) Without limiting the obligations of the Loan Parties under Section 6.08(a), each Loan Party will, and will cause its Restricted Subsidiaries to, comply with all Environmental Laws applicable to its or their business, operations and properties; obtain and maintain in full force and effect all material authorizations, registrations, licenses and permits required pursuant to Environmental Law for its business, operations and properties; and perform any investigation, remedial action or cleanup as required pursuant to Environmental Laws, except, in each case with respect to this Section 6.08(b), to the extent the failure to do so would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.
- 6.09 **Use of Proceeds.** The Borrower will use the proceeds of the Credit Extensions for working capital, capital expenditures, Acquisitions and other general corporate purposes, including distributions, not in contravention of any Law or of any Loan Document. No part of the proceeds of the Credit Extensions will be used, whether directly or indirectly, for any purpose that violates any of Regulations T, U or X.
- 6.10 **Subordination of Intercompany Indebtedness.** All Indebtedness of any Loan Party permitted by <u>Section 7.01(f)</u> shall be subject to subordination provisions in form and substance reasonably satisfactory to Administrative Agent.
 - 6.11 Additional Material Subsidiaries; Collateral. Each Loan Party agrees to, and will cause its Material Subsidiaries to:
- (a) <u>Collateral</u>. Cause (x) all present and future Equity Interests now or hereafter owned by the Parent Guarantor, the Borrower and each present and future Material Subsidiary and (y) all other material tangible and intangible property, other than Excluded Property, now or hereafter owned by the Parent, the Borrower and each present and future respective Material Subsidiary, to be subject at all times to perfected First Priority Liens in favor of the Administrative Agent to secure the Secured Obligations pursuant to the terms and conditions of Collateral Documents, <u>provided</u>, however, that the foregoing requirements in this paragraph shall not apply to any Specified Non-Wholly Owned Subsidiary unless and until it is a Wholly Owned Subsidiary, and <u>provided further</u> that

- (i) No mortgages or fixture filings in county records will be required to be executed or recorded on or with respect to pipelines, Pipeline Real Property or other owned or leased Real Property except as provided in <u>Section 6.11(c)</u> with respect to Material Real Property as therein described, provided that transmitting utility Uniform Commercial Code financing statements may be filed in state central filing offices;
 - (ii) Control agreements will not be required with respect to deposit accounts, securities accounts and commodities accounts;
- (iii) A Lien on Equity Interests evidencing ownership of a Joint Venture will not be required for so long as and to the extent that the Joint Venture Organization Documents prohibit such pledge, and a Lien on Equity Interests evidencing ownership of an Unrestricted Subsidiary will not be required if and for so long as (i) such Unrestricted Subsidiary is not a Wholly Owned Subsidiary and the Organization Documents of such Unrestricted Subsidiary prohibit such pledge or (ii) such Equity Interests are required to be pledged to secure debt of such Unrestricted Subsidiary (or its parent company that is an Unrestricted Subsidiary) and the terms of such other pledge prohibit a lien to secure the Secured Obligations;
- (iv) with respect to owned vehicles, rail cars and similar collateral for which perfection of Liens would require taking possession of, or noting Liens on, a certificate of title, Liens on such assets need not be perfected;
- (v) Liens on assets will not be required, and/or perfection of Liens will not be required, in circumstances where the Administrative Agent and the Borrower agree that the cost of obtaining (or perfecting, as applicable) a Lien on such assets is materially disproportionate in relation to the benefit to the Secured Parties afforded thereby.
- (b) Additional Material Subsidiaries. Notify the Administrative Agent if any of the following occurs (each, a "New Material Subsidiary Event") (i) any entity (other than Rendezvous Gas or an Affiliate Non-Wholly Owned Subsidiary) becomes a Material Subsidiary, (ii) any Material Subsidiary (other than Rendezvous Gas or an Affiliate Non-Wholly Owned Subsidiary) is created or acquired or otherwise becomes owned directly or indirectly by the Parent Guarantor or (iii) Rendezvous Gas becomes, or any Affiliate Non-Wholly Owned Subsidiary becomes, a Wholly Owned Subsidiary. Promptly (and in any event within thirty (30) days (or such greater number of days as may be agreed by the Administrative Agent)) after such New Material Subsidiary Event, cause such Material Subsidiary (in the case of clause (i) or (ii)), or cause Rendezvous Gas or such Affiliate Non-Wholly Owned Subsidiary if it is a Material Subsidiary at such time (in the case of clause (iii)), to (i) become a Subsidiary Guarantor by delivering to the Administrative Agent a duly executed supplement to the Subsidiary Guaranty Agreement or such other document as the Administrative Agent shall reasonably deem appropriate for such purpose, (ii) grant a Lien on assets and property owned by such Subsidiary in accordance with Section 6.11(a) by delivering to the Administrative Agent a duly executed supplement to each Security Agreement or such other document as the Administrative Agent shall reasonably deem appropriate for such purpose and comply with the terms of each such Security Agreement, (iii) deliver to the Administrative Agent such documents, certificates and opinion of counsel, each of the type referred to in Section 4.02(b)(i) and (ii), as may be reasonably requested by the Administrative Agent, (iv) to the extent not previously delivered, deliver to the Administrative Agent such original stock or other certificates and stock or other transfer powers evidencing the Equity Interests issued by such Material Subsidiary and evidencing Equity Interests owned by such Material Subsidiary, (v) deliver to the Administrative Agent such updated Schedules to the Loan Documents as reasonably requested by the Administrative Agent with respect to such Material Subsidiary, and (vi) deliver to the Administrative Agent such other documents as may be reasonably requested by the Administrative Agent in connection with the foregoing, in form, content and scope reasonably satisfactory to the Administrative Agent.

- (c) <u>Material Real Property</u>. To the extent that any Material Real Property is acquired (which may be by completion of construction), in any transaction or series of related transactions, by a Loan Party or a Material Subsidiary during any fiscal quarter of the Parent Guarantor, within thirty (30) days (or such longer period of time agreed by the Administrative Agent) after a Compliance Certificate is required to be delivered for such quarter under <u>Section 6.01(a)</u> or (b), deliver (i) a mortgage or deed of trust, (ii) any existing environmental reports, (iii) insurance certificates or certificates of the type referred to in <u>Section 4.02(b)(i)</u> to the extent requested by the Administrative Agent, and (iv) other documents reasonably requested by the Administrative Agent in connection with granting and perfecting a First Priority Lien on such Material Real Property in favor of the Administrative Agent, for the ratable benefit of the Secured Parties, in each case in form and substance reasonably acceptable to the Administrative Agent. As used herein, "<u>Material Real Property</u>" means any processing plant, terminal, and any other Real Property (other than Pipeline Real Property) owned in fee by any Loan Party or a Restricted Subsidiary, or group of related tracts of Real Property (other than Pipeline Real Property), acquired (whether acquired in a single transaction or in a series of transactions) or owned by a Loan Party or any Restricted Subsidiary having a fair market value (including the fair market value of improvements owned by any Loan Party or Restricted Subsidiary and located thereon) on such date of determination equal to \$25.0 million or more.
- 6.12 **Further Assurances.** Execute any and all further documents, agreements and instruments, deliver any financing statements and take such further action (including filing financing statements) that may be required under applicable law, or that the Administrative Agent or the Required Lenders (acting through the Administrative Agent) may reasonably request, for the purposes of implementing or effectuating the provisions of this Agreement and the other Loan Documents, and in order to grant, preserve, protect and perfect the security interests required to be created pursuant to the terms of this Agreement.
- 6.13 **Keepwell Requirements.** Each Qualified ECP Guarantor hereby jointly and severally absolutely, unconditionally and irrevocably undertakes to provide such funds or other support as may be needed from time to time by each other Loan Party to honor all of its obligations under this Agreement or any other Loan Document in respect of Swap Obligations (provided, however, that each Qualified ECP Guarantor shall only be liable under this Section 6.13 for the maximum amount of such liability that can be hereby incurred without rendering its obligations under this Section 6.13, or otherwise under this Agreement or any other Loan Document, voidable under applicable law relating to fraudulent conveyance or fraudulent transfer, and not for any greater amount). The obligations of each Qualified ECP Guarantor under this Section shall remain in full force and effect until the full and final payment of the Obligations (other than contingent Obligations for which no claim has been made), termination of the Commitments of all Lenders and L/C Issuers, and termination of all Letters of Credit (or Cash Collateralization thereof as acceptable to the applicable L/C Issuer). Each Qualified ECP Guarantor intends that this Section 6.13 constitute, and this Section 6.13 shall be deemed to constitute, a "keepwell, support, or other agreement" for the benefit of each other Loan Party for all purposes of Section 1a(18)(A)(v)(II) of the Commodity Exchange Act.

ARTICLE VII. NEGATIVE COVENANTS

Each Loan Party covenants and agrees that until the full and final payment of the Obligations (other than contingent Obligations for which no claim has been made), termination of the Commitments of all Lenders and L/C Issuers, and termination of all Letters of Credit (or Cash Collateralization thereof as acceptable to the applicable L/C Issuer):

- 7.01 **Indebtedness.** Each Loan Party agrees that it will not, and will not permit any of its Restricted Subsidiaries to, in any manner owe or be liable for Indebtedness except:
 - (a) Indebtedness created hereunder and under the other Loan Documents;
- (b) Indebtedness (including capital lease obligations) of any one or more of the Loan Parties or Restricted Subsidiaries incurred to finance the acquisition, construction or improvement of any fixed or capital assets, and Permitted Refinancing Indebtedness in respect thereof; <u>provided</u> that (i) such Indebtedness is incurred prior to or within 180 days after such acquisition or the completion of such construction or improvement and (ii) the aggregate outstanding principal amount of Indebtedness permitted by this <u>Section 7.01(b)</u>, when added to the aggregate outstanding principal amount of Indebtedness permitted by <u>Section 7.01(c)</u> and the Remaining Present Value of all existing leases permitted by <u>Section 7.03</u>, shall not at any time exceed the greater of \$70,000,000 and an amount equal to 10% of Consolidated Net Tangible Assets of the Parent Guarantor;
- (c) Indebtedness of any one or more of the Loan Parties or Restricted Subsidiaries assumed in connection with any Acquisition after the Closing Date and Indebtedness of a Person that becomes a Restricted Subsidiary after the Closing Date pursuant to an Acquisition and any Permitted Refinancing Indebtedness in respect of any of the foregoing; <u>provided</u> that such Indebtedness exists prior to such Acquisition and is not created in contemplation thereof, and <u>provided</u> <u>further</u> that the aggregate principal amount of Indebtedness permitted by this clause (c), when added to the aggregate principal amount of Indebtedness permitted by <u>Section 7.01(b)</u> and the Remaining Present Value of all existing leases permitted by <u>Section 7.03</u> shall not at any time exceed the greater of \$70,000,000 and an amount equal to 10% of Consolidated Net Tangible Assets of the Parent Guarantor;
- (d) the Guarantee by any one or more of the Loan Parties or Restricted Subsidiaries of Indebtedness of any one or more of the other Loan Parties or Restricted Subsidiaries if such Indebtedness of such other Loan Parties or Restricted Subsidiaries was permitted to be incurred by another provision of this Section 7.01; provided that if the Indebtedness being guaranteed is subordinated to the Obligations, then the guarantee shall be subordinated to the same extent as the Indebtedness guaranteed;
- (e) unsecured Indebtedness of one or both of the Parent Guarantor and a Finance Subsidiary and unsecured Indebtedness of the Borrower, provided that (A) immediately prior to and after giving effect to the issuance of such Indebtedness, no Default exists, (B) such Indebtedness does not mature or require any scheduled repayment, defeasance or redemption (or sinking fund therefor) of any principal amount thereof sooner than six (6) months after the Maturity Date, (C) the indenture or other agreement governing such Indebtedness does not contain (1) financial maintenance covenants that are more restrictive than as set forth in this Agreement, or (2) any other covenants or events of default that, taken as a whole, are more restrictive than those set forth in this Agreement, or (3) any mandatory prepayment or redemption provisions or other prepayments required as a result of a "change of control" or similar event or as a result of an asset sale, (D) such Indebtedness is not guaranteed by any Subsidiary, other than a Subsidiary Guarantor, (E) immediately before and after giving incurring such Indebtedness, the Parent Guarantor demonstrates Pro Forma Compliance with the Financial Covenants, and (F) at the time of incurrence of such Indebtedness, a Responsible Officer certifies to the Administrative Agent as to compliance with this Section 7.01(e);

- (f) Indebtedness owed to any Loan Party <u>provided</u> that (i) any such Indebtedness shall be unsecured, (ii) any promissory note evidencing such Indebtedness shall be delivered to the Administrative Agent pursuant to the Security Agreements, and (iii) such Indebtedness shall be subordinated to the Obligations pursuant to subordination terms reasonably satisfactory to the Administrative Agent;
 - (g) Indebtedness under Swap Contracts permitted under Section 7.12;
- (h) Indebtedness consisting of unpaid insurance premiums owing to insurance companies and insurance brokers incurred in connection with the financing of insurance premiums in the ordinary course of business;
- (i) Indebtedness in respect of performance bonds, warranty bonds, bid bonds, appeal bonds, surety bonds, labor bonds and completion or performance guarantees and similar obligations, in each case provided in the ordinary course of business, including those incurred to secure health, safety and environmental obligations in the ordinary course of business and Indebtedness arising out of advances on trade receivables, customer prepayments and similar transactions in the ordinary course of business; and
- (j) other Indebtedness of any one or more of the Loan Parties and Restricted Subsidiaries not described in <u>subsections (a)</u> through (<u>i)</u> above; <u>provided</u> that the aggregate outstanding principal balance of Indebtedness permitted for all Loan Parties and Restricted Subsidiaries under this <u>subsection (j)</u>, when added to the aggregate outstanding principal balance of all outstanding Indebtedness of all Loan Parties and Restricted Subsidiaries secured by Liens permitted by <u>clause (o)</u> of the definition of "<u>Permitted Liens</u>" (without duplication), shall not at any time exceed an amount equal to the greater of \$20,000,000 and 3% of Consolidated Net Tangible Assets of the Parent Guarantor.
- 7.02 **Limitation on Liens.** Each Loan Party agrees that it will not, and will not permit any of its Restricted Subsidiaries to (a) create, assume or permit to exist any Lien upon any of the properties or assets which a Loan Party or such Restricted Subsidiary now owns or hereafter acquires other than Permitted Liens (subject to Section 7.15 in the case of Pipeline Real Property); or (b) allow the filing or continued existence of any financing statement describing as collateral any of its or their respective assets or property, other than financing statements which describe only collateral subject to a Lien permitted under this Section and which name as secured party or lessor only the holder of such Lien.
- 7.03 **Sale Lease-back Transactions.** Each Loan Party agrees that it shall not, and shall not permit any of its Restricted Subsidiaries to, enter into any arrangement pursuant to which it shall sell or transfer any property, real or personal, used or useful in its business, whether now owned or hereafter acquired, and thereafter rent or lease such property or other property that it intends to use for substantially the same purpose or purposes as the property being sold or transferred, unless (a) such transaction is entered into by a Loan Party or a Restricted Subsidiary and (b) after giving effect to such sale or transfer and the execution and delivery of such lease, the Remaining Present Value of such lease and all other existing leases previously entered into as permitted by this Section 7.03, together with the outstanding principal amount of Indebtedness permitted pursuant to Sections 7.01(b) and 7.01(c), shall not exceed the greater of \$70,000,000 and an amount equal to 10% of Consolidated Net Tangible Assets of the Parent Guarantor.

7.04 **Line of Businesses.** Each Loan Party agrees that it will not, and will not permit any Restricted Subsidiaries to, engage directly or indirectly in any material line of business other than the midstream oil and gas business, and any business substantially related or incidental thereto, in the United States.

7.05 Investments.

- (a) Each Loan Party agrees that it will not, and will not permit any of its Restricted Subsidiaries to, make, purchase, acquire or hold any Investments except:
- (i) Investments existing on the Closing Date in Restricted Subsidiaries, Unrestricted Subsidiaries and in Joint Ventures, in each case, as described in <u>Schedule 7.05</u>;
 - (ii) Investments made after the Closing Date in Equity Interests in Wholly Owned Restricted Subsidiaries and in Rendezvous Gas;
 - (iii) Permitted Investments;
 - (iv) Investments constituting loans or advances permitted by Section 7.01(f);
 - (v) Guarantees constituting Indebtedness permitted by Section 7.01;
 - (vi) Investments in Swap Contracts permitted by Section 7.12;
 - (vii) (A) Permitted Acquisitions and (B) Permitted Affiliate Acquisitions;
- (viii) Investments consisting of (A) extensions of credit in the nature of accounts receivable or notes receivable arising from the granting of trade credit in the ordinary course of business, (B) Investments received in satisfaction or partial satisfaction thereof from financially troubled account debtors in order to prevent or limit loss, and (C) Investments received in connection with the bankruptcy or reorganization of, or settlement of delinquent accounts and disputes with, customers and suppliers, in each case in the ordinary course of business;
- (ix) Investments made after the Closing Date in (A) Unrestricted Subsidiaries, (B) Non-Wholly Owned Subsidiaries and (C) Joint Ventures, provided that the aggregate outstanding amount of all Investments permitted for all Loan Parties and Restricted Subsidiaries pursuant to this clause (a)(ix) shall not at any time exceed in the aggregate an amount equal to the greater of \$50,000,000 and 7.5% of Consolidated Net Tangible Assets of the Parent Guarantor; and
- (x) other Investments not permitted by the foregoing clauses in this <u>Section 7.05</u>, <u>provided</u> that the aggregate outstanding amount of all Investments permitted for all Loan Parties and Restricted Subsidiaries pursuant to this clause (a)(x) shall not at any time exceed in the aggregate an amount equal to the greater of \$30,000,000 and 4.0% of Consolidated Net Tangible Assets of the Parent Guarantor.
 - (b) The Parent Guarantor shall not at any time directly own Equity Interests in any Person other than the Borrower and any Finance Subsidiaries.

7.06 Material Subsidiaries and Immaterial Subsidiaries; Unrestricted Subsidiaries.

(a) <u>Restricted Subsidiaries</u>. Each Subsidiary shall be a Restricted Subsidiary unless it is designated by the Borrower as an Unrestricted Subsidiary pursuant to, and meets the requirements set forth in, this <u>Section 7.06</u>.

(b) Material Subsidiaries; Immaterial Subsidiaries.

- (i) Each Subsidiary shall be a Material Subsidiary unless it is designated by the Borrower pursuant to this <u>Section 7.06(b)</u> as an Immaterial Subsidiary. The Borrower may designate a Subsidiary as an Immaterial Subsidiary by written notice to the Administrative Agent, subject to compliance with the requirements set forth in the definition of "Immaterial Subsidiary".
- (ii) If at any time any Subsidiary designated as an Immaterial Subsidiary fails to meet any of the requirements applicable to Immaterial Subsidiaries set forth in the definition thereof or set forth in this <u>Section 7.06(b)</u>, it will thereafter cease to be an Immaterial Subsidiary.
- (iii) The Borrower will not permit the aggregate Consolidated Total Assets of all Immaterial Subsidiaries, or the aggregate Consolidated EBITDA of all Immaterial Subsidiaries, to exceed at any time the limits set forth in the definition of "Immaterial Subsidiary".

(c) Unrestricted Subsidiaries.

- (i) The Borrower may designate any Immaterial Subsidiary as an Unrestricted Subsidiary after the Closing Date by delivering to the Administrative Agent a certificate of a Responsible Officer of the Parent Guarantor or of the Borrower, certifying that such designation complies with the conditions set forth in this <u>Section 7.06(c)</u>. No Material Subsidiary may be designated as an Unrestricted Subsidiary, and at no time may any Material Subsidiary be an Unrestricted Subsidiary.
- (ii) The Borrower may designate any Immaterial Subsidiary as an Unrestricted Subsidiary if immediately before and after such designation, no Default shall have occurred and be continuing.
- (iii) The designation of an Immaterial Subsidiary as an Unrestricted Subsidiary shall constitute an Investment in the Unrestricted Subsidiary on the date of designation in an amount equal to the fair market value of the Parent Guarantor's outstanding investment therein.
- (iv) If at any time any Unrestricted Subsidiary becomes a Material Subsidiary, it will thereafter cease to be an Unrestricted Subsidiary and will be a Restricted Subsidiary. Any Indebtedness of such Subsidiary will be deemed to be incurred by a Restricted Subsidiary as of such date and the Parent Guarantor and Borrower will, and will cause such Subsidiary to, deliver the agreements, documents and other items required by <u>Section 6.11.</u>
- (d) No Loan Party or any Restricted Subsidiary may guaranty or otherwise become liable in respect of any Indebtedness or other obligations of, grant any Lien on any of its property to secure any Indebtedness of or other obligation of, or provide any other form of credit support to, any Unrestricted Subsidiary.
 - (e) No Unrestricted Subsidiary may, directly or indirectly, make any Investment in any Loan Party or any Restricted Subsidiary

7.07 **Mergers, Consolidations and Sales of All or Substantially All Assets.** Each Loan Party agrees that it shall not, and shall not permit any of its Restricted Subsidiaries to, merge into or consolidate with any other Person, or permit any other Person to merge into or consolidate with it, or sell, transfer, lease or otherwise Dispose of (in one transaction or in a series of transactions) all or substantially all of its assets (whether now owned or hereafter acquired), except for transactions permitted by Section 7.11 and except that if at the time thereof and immediately after giving effect hereto no Event of Default shall have occurred and be continuing then any Restricted Subsidiary may merge with (i) the Borrower in a transaction in which the Borrower is the surviving Person, or (ii) the Parent Guarantor in a transaction in which the Parent Guarantor is the surviving Person, (iii) any other Restricted Subsidiary or (iv) any other Person in connection with a Permitted Acquisition or a Permitted Affiliate Acquisition if the surviving Person is a Restricted Subsidiary; provided that (x) if a Loan Party is a party to such merger or consolidation then a Loan Party must be the surviving Person and (y) unless such merger or consolidation is permitted pursuant to Section 7.11, if a Wholly Owned Restricted Subsidiary is a party to such merger then a Wholly Owned Restricted Subsidiary must be the surviving Person.

7.08 **Transactions with Affiliates.** Each Loan Party agrees that it will not, and will not permit any of its Material Subsidiaries to, sell, lease or otherwise transfer any property or assets to, or purchase, lease or otherwise acquire any property or assets from, or otherwise engage in any other transactions with, any of its Affiliates, except (a) transactions on fair and reasonable terms, (b) transactions between or among two or more of the Loan Parties and the Wholly Owned Restricted Subsidiaries not involving any other Affiliate, (c) investments in Unrestricted Subsidiaries permitted pursuant to Section 7.05(a) (ix) or (x), (d) the Contribution and the IPO, (e) transactions entered into with QRI and its Subsidiaries on terms and conditions, taken as a whole, that are fair and reasonable to the Loan Parties and Restricted Subsidiaries, taking into account the totality of the relationship between the Loan Parties and Restricted Subsidiaries, on the one hand, and QRI and its Subsidiaries, on the other, (f) transactions approved by the Conflicts Committee of the board of directors or equivalent governing body of the Parent Guarantor (or the equivalent successor body to such Conflicts Committee) and (g) any Restricted Payment permitted by Section 7.14.

7.09 **Restrictive Agreements.** Each Loan Party agrees that it will not, and will not permit any Material Subsidiary to, enter into any Contractual Obligation (other than this Agreement and other Loan Documents) that limits the ability of any Restricted Subsidiary (i) to make Restricted Payments to the Parent Guarantor, the Borrower or any Restricted Subsidiary, as applicable, (ii) to otherwise transfer property to the Parent Guarantor, the Borrower or any Restricted Subsidiary, or (iii) to Guarantee the Obligations, provided, however, that:

- (a) clause (ii) of this <u>Section 7.09</u> shall not prohibit restrictions or conditions imposed by an agreement related to secured Indebtedness permitted by this Agreement if such restrictions or conditions apply only to the property securing such Indebtedness;
- (b) this <u>Section 7.09</u> shall not apply to customary restrictions and conditions contained in agreements relating to the sale of a Subsidiary pending such sale, provided such restrictions and conditions apply only to the Subsidiary that is to be sold and such sale is permitted hereunder;
 - (c) clause (ii) of this Section 7.09 shall not prohibit customary provisions in leases and other contracts restricting the assignment or transfer thereof;

- (d) clause (ii) of this <u>Section 7.09</u> shall not prohibit customary provisions in Joint Venture agreements and other similar agreements applicable to Joint Ventures restricting the ability to pledge or otherwise transfer Equity Interests in such Joint Ventures; and
- (e) this <u>Section 7.09</u> shall not apply to prohibitions, restrictions or conditions contained in, or existing by reason of, any agreement or instrument relating to any Indebtedness of any Restricted Subsidiary (and any Permitted Refinancing Indebtedness thereof) at the time such Restricted Subsidiary was merged or consolidated with or into, or acquired by, a Loan Party or a Restricted Subsidiary or became a Restricted Subsidiary and not created in contemplation thereof.
- 7.10 **ERISA.** Except as would not reasonably be expected to result in a Material Adverse Effect, no ERISA Affiliate will incur any obligation to contribute to any "multiemployer plan" as defined in Section 4001 of ERISA.
 - 7.11 Dispositions of Property. Each Loan Party agrees that it will not, and will not permit any Restricted Subsidiary to, make any Disposition except:
- (a) (i) equipment which is worthless or obsolete or which is replaced by equipment of equal suitability and value and (ii) inventory which is sold in the ordinary course of business;
 - (b) Dispositions of property to a Loan Party or to a Wholly Owned Restricted Subsidiary;
 - (c) Investments permitted pursuant to Section 7.05;
 - (d) Dispositions of accounts receivable in connection with the collection or compromise thereof;
- (e) Dispositions in the form of licenses, sublicenses, leases or subleases granted to others which do not interfere in any material respect with the business of the Loan Parties and their Subsidiaries;
 - (f) Dispositions of cash equivalents and Permitted Investments for fair market value;
- (g) Dispositions resulting from Casualty Events, provided that the Net Cash Proceeds of such Dispositions are applied in accordance with the requirements of Section 2.05(b) to the extent required by such Section;
- (h) Dispositions of assets acquired in a Permitted Acquisition to the extent determined by a Responsible Officer of the Parent Guarantor to be non-core, immaterial, or ancillary to the Borrower's and its Restricted Subsidiaries' midstream oil and gas business, provided that the Net Cash Proceeds of such Dispositions are applied in accordance with the requirements of <u>Section 2.05(b)</u> to the extent required by such Section; and
- (i) any Disposition not otherwise permitted under this Section 7.11, provided that (i) at the time of such Disposition no Default exists or would result from such Disposition, (ii) the Net Cash Proceeds of such Disposition are applied in accordance with the requirements of Section 2.05(b) to the extent required by such Section, (iii) no less than 75% of the consideration received for such Disposition shall be cash, and (iv) the aggregate sales price of all Dispositions made by all Loan Parties and Restricted Subsidiaries pursuant to this clause (i) during any fiscal year of the Parent Guarantor does not exceed ten percent (10%) of Consolidated Net Tangible Assets of the Parent Guarantor as of the last day of the previous fiscal year.

7.12 **Swap Contracts.** Each Loan Party agrees that it will not, and will not permit any of its Restricted Subsidiaries to, enter into any Swap Contract, except (i) Swap Contracts entered into to hedge or mitigate risks to which any Loan Party or any Restricted Subsidiary has actual or projected exposure (other than those in respect of Equity Interests of any Loan Party or any of its Subsidiaries), and (ii) other Swap Contracts permitted under the risk management policies approved by the Board of Directors (or equivalent governing body) of the General Partner from time to time and not subjecting any Loan Party or its Restricted Subsidiaries to material speculative risks.

7.13 Financial Covenants.

- (a) <u>Interest Coverage Ratio</u>. The Parent Guarantor shall not permit the Consolidated Interest Coverage Ratio to be less than 2.5 to 1.0 as of any Quarter-End Date (commencing with the first Quarter-End Date after the Closing Date).
- (b) <u>Total Leverage Ratio</u>. Prior to consummation of a Qualified Notes Offering, the Parent Guarantor shall not permit the Consolidated Leverage Ratio to exceed 5.0 to 1.0 as of any Quarter-End Date (commencing with the first Quarter-End Date after the Closing Date), <u>provided</u> that if such Quarter-End Date is during an Acquisition Period, the Consolidated Leverage Ratio may not exceed 5.5 to 1.0 as of such Quarter-End Date. From and after consummation of a Qualified Notes Offering, the Parent Guarantor shall not permit the Consolidated Leverage Ratio to exceed 5.50 to 1.0 as of any Quarter-End Date.
- (c) <u>Senior Secured Leverage Ratio</u>. From and after consummation of a Qualified Notes Offering, the Parent Guarantor shall not permit the Consolidated Senior Secured Leverage Ratio to exceed 3.5 to 1.0 as of any Quarter-End Date (commencing with the first Quarter-End Date after the Closing Date).
- 7.14 **Restricted Payments.** The Parent Guarantor will not declare or make, or agree to pay or make, directly or indirectly, any Restricted Payment, except:
- (a) Restricted Payments to the extent of the amount of "Available Cash" as defined in the Parent Guarantor's Partnership Agreement in accordance with the terms of such Partnership Agreement;
- (b) Restricted Payments in exchange for, or out of the net cash proceeds from the substantially concurrent sale (other than to a Subsidiary) of, Equity Interests of the Parent Guarantor (other than Disqualified Equity); and
- (c) Any purchase, redemption or other acquisition (on a pro rata basis with respect to all of its Equity Interests) of Equity Interests issued by it with the proceeds received from the substantially concurrent issuance (other than to a Subsidiary) by it of new Equity Interests (other than Disqualified Equity);

provided that no Default has occurred and is continuing at the time of such declaration or at the time of such payment or would result therefrom.

- 7.15 **Negative Pledge on Pipeline Real Property.** Each Loan Party agrees that it will not, and will not permit any of its Restricted Subsidiaries to create, incur, assume or permit to exist any Lien on any portion of any Pipeline Real Property, other than Liens permitted by clauses (a), (d), (f), (h), (i) and (j) of the definition of Permitted Liens.
- 7.16 **Modification, Terminations, etc. of Certain Other Indebtedness, Agreements and Organization Documents.** Each Loan Party agrees that it will not, and will not permit any Restricted Subsidiary to:
- (a) permit any waiver, supplement, modification, amendment, release, cancellation or termination (other than a termination at the end of a term) of any Material Agreement if the effect of such waiver, supplement, modification, amendment, release, cancellation or termination, individually or in the aggregate, would reasonably be expected to result in a Material Adverse Effect;
- (b) permit any supplement, modification or amendment to any Organization Documents of any Loan Party or any Restricted Subsidiary if such supplement, modification or amendment, individually or in the aggregate, would reasonably be expected to materially and adversely affect the Lenders; or
- (c) make, or agree or offer to pay or make, directly or indirectly, any payment or other distribution (whether in cash, securities or other property) of or in respect of principal of or interest on Permitted Unsecured Indebtedness or any payment or other distribution (whether in cash, securities or other property), including any sinking fund or similar deposit, on account of the purchase, redemption, retirement, acquisition, cancellation or termination of any Permitted Unsecured Indebtedness, except for
 - (i) payments of regularly scheduled interest accrued thereon;
- (ii) so long as no Default is continuing or would result therefrom, prepayments with the proceeds of any Permitted Refinancing Indebtedness in respect thereof; or
- (iii) payments in an amount not to exceed the amount of net cash proceeds received by the Parent Guarantor from the issuance, after the Closing Date, of common Equity Interests, provided that each such payment is made within one hundred twenty (120) days of receipt of such proceeds; provided that, no such prepayments shall be made with the proceeds of Loans.
- 7.17 **Fiscal Year.** Each Loan Party agrees that it will not, and will not permit any of its Restricted Subsidiaries to, change its fiscal year-end from December 31.
- 7.18 **Certain Covenants Pertaining to Rendezvous Gas.** Notwithstanding the provisions of <u>Section 7.01</u>, the Parent Guarantor and the Borrower each agree that they will not permit Rendezvous Gas to incur, assume or permit to exist any Indebtedness for borrowed money; <u>provided</u> that the restrictions in this <u>Section 7.18</u> shall not apply at any time when Rendezvous Gas is a Wholly Owned Subsidiary.

ARTICLE VIII. EVENTS OF DEFAULT AND REMEDIES

- 8.01 **Events of Default.** Each of the following events constitutes an Event of Default under this Agreement:
- (a) The Borrower fails to pay (i) any amount of principal of any Loan, any L/C Obligation, or any Swing Line Loan when and as required to be paid herein, or (ii) any other amount due and payable hereunder or under any other Loan Document on the third Business Day after the date such amount becomes due;
- (b) The Borrower or any other Loan Party fails to duly observe, perform or comply with any term, covenant or agreement contained in <u>Sections 6.02(b)</u>, <u>6.02(c)</u>, <u>6.03(b)</u>, <u>Section 6.05</u> (with respect to maintenance of existence of the Borrower, the Parent Guarantor and QEPM Gathering) or <u>Article VII</u>;
- (c) The Borrower or any other Loan Party fails (other than as referred to in <u>subsections (a)</u> or (b) above) to duly observe, perform or comply with any covenant, agreement, condition or provision of any Loan Document, and such failure remains unremedied for a period of thirty (30) days;
- (d) Any representation or warranty made or deemed made by the Borrower or any other Loan Party herein or in any other Loan Document or in any document delivered in connection herewith shall prove to have been false or incorrect in any material respect on any date on or as of which made;
- (e) This Agreement or any other Loan Document is asserted to be or at any time ceases to be valid, binding and enforceable in any material respect as warranted in Section 5.05 for any reason other than its release or subordination by Administrative Agent;
- (f) (i) Any one or more of the Loan Parties or Restricted Subsidiaries (A) fails to make any payment when due (whether by scheduled maturity, required prepayment, acceleration, demand, or otherwise) in respect of any Indebtedness or Guarantee (other than Indebtedness hereunder and Indebtedness under Swap Contracts) having an aggregate principal amount (including undrawn committed or available amounts and including amounts owing to all creditors under any combined or syndicated credit arrangement) of more than the Threshold Amount, or (B) fails to observe or perform any other agreement or condition relating to any such Indebtedness or Guarantee or contained in any instrument or agreement evidencing, securing or relating thereto, or any other event occurs, the effect of which default or other event is to cause, or to permit the holder or holders of such Indebtedness or the beneficiary or beneficiaries of such Guarantee (or a trustee or agent on behalf of such holder or holders or beneficiary or beneficiaries) to cause, with the giving of notice if required, such Indebtedness to be demanded or to become due or to be repurchased, prepaid, defeased or redeemed (automatically or otherwise), or an offer to repurchase, prepay, defease or redeem such Indebtedness to be made, prior to its stated maturity, or such Guarantee to become payable or cash collateral in respect thereof to be demanded; or (ii) there occurs under any Swap Contract an Early Termination Date (as defined in such Swap Contract) resulting from (A) any event of default under such Swap Contract as to which the Borrower or any Restricted Subsidiary is the Defaulting Party (as defined in such Swap Contract) or (B) any Termination Event (as so defined) under such Swap Contract as to which the Borrower or such Restricted Subsidiary as a result thereof is greater than the Threshold Amount;

- (g) A Loan Party or any of its Material Subsidiaries or the General Partner:
- (i) institutes or consents to the institution of any proceeding under any Debtor Relief Law, or files an answer admitting the material allegations of a petition filed against it in any such proceeding; or makes an assignment for the benefit of creditors; or applies for or consents to the appointment of any receiver, trustee, custodian, conservator, liquidator, rehabilitator or similar officer for it or for all or any material part of its property; or any receiver, trustee, custodian, conservator, liquidator, rehabilitator or similar officer is appointed without the application or consent of such Person and the appointment continues undischarged or unstayed for 60 calendar days; or any proceeding under any Debtor Relief Law relating to any such Person or to all or any material part of its property is instituted without the consent of such Person and continues undismissed or unstayed for 60 calendar days, or an order for relief is entered in any such proceeding; or
 - (ii) becomes unable or admits in writing its inability or fails generally to pay its debts as they become due; or
- (iii) there is issued or levied any writ or warrant of attachment or execution or similar process against all or any material part of the property of any such Person and is not released, vacated or fully bonded within 30 days after its issue or levy;
- (h) There is entered against any one or more of the Loan Parties or Restricted Subsidiaries one or more final judgments or orders for the payment of money in an aggregate amount exceeding the Threshold Amount (to the extent not covered by independent third-party insurance as to which the insurer does not dispute coverage), or one or more non-monetary final judgments that have, or would reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect and, in either case, (A) enforcement proceedings are commenced by any creditor upon such judgment or order, or (B) there is a period of 10 consecutive days during which a stay of enforcement of such judgment, by reason of a pending appeal or otherwise, is not in effect;
- (i) Either (i) any failure to satisfy the minimum funding requirements of the Pension Funding Rules resulting in a liability in excess of the Threshold Amount exists with respect to any Pension Plan, whether or not waived by the Secretary of the Treasury or his delegate, or (ii) any ERISA Event occurs with respect to any Pension Plan or any Multiemployer Plan and (A) with respect to a Pension Plan, the then current value of the accumulated benefit obligation of such Pension Plan exceeds the then current value of the assets of such Plan available for the payment of such benefit liabilities by more than the Threshold Amount, or (B) with respect to a Multiemployer Plan, the combined liability to such Multiemployer Plan is expected to be in excess of the Threshold Amount, or (C) in the case of an ERISA Event involving the withdrawal of a substantial employer from a Multiemployer Plan, the withdrawing employer's proportionate share of such liability exceeds the Threshold Amount;
 - (j) A Change of Control occurs;
- (k) Any material provision of any Loan Document, at any time after execution and delivery thereof, shall for any reason (other than as expressly permitted hereunder or thereunder or the full and final payment of the Obligations (other than contingent Obligations for which no claim has been made), termination of the Commitments of all Lenders and L/C Issuers, and termination of all Letters of Credit (or Cash Collateralization thereof as acceptable to the applicable L/C Issuer)) cease to be valid and binding on any Loan Party or any such Person shall so state in writing; or

- (l) any one or more Collateral Documents after delivery thereof ceases for any reason (other than pursuant to the terms thereof or as permitted hereunder) to create a valid and perfected First Priority Lien (subject to the Liens permitted by <u>Section 7.02</u>) on Collateral purported to be covered thereby having in the aggregate a fair market value equal to or greater than five percent (5%) of Consolidated Net Tangible Assets of the Parent Guarantor.
- 8.02 **Remedies upon Event of Default.** If any Event of Default occurs and is continuing, the Administrative Agent shall, at the request of, or may, with the consent of, the Required Lenders, take any or all of the following actions:
- (a) declare the commitment of each Lender to make Loans and Swing Line Loans and any obligation of each L/C Issuer to make L/C Credit Extensions to be terminated, whereupon such commitments and obligation shall be terminated;
- (b) declare the unpaid principal amount of all outstanding Loans and Swing Line Loans, all interest accrued and unpaid thereon, and all other amounts owing or payable hereunder or under any other Loan Document to be immediately due and payable, without presentment, demand, protest or other notice of any kind, all of which are hereby expressly waived by the Borrower;
 - (c) require that the Borrower Cash Collateralize the L/C Obligations (in an amount equal to 100% of the then Outstanding Amount thereof);
 - (d) enforce any and all security interests, Liens and other remedies pursuant to the Collateral Documents; and
 - (e) exercise on behalf of itself and the Lenders all rights and remedies available to it and the Lenders under the Loan Documents or applicable law;
- provided, however, that upon the occurrence of an Event of Default specified in subsections (g)(i) or (g)(ii) of Section 8.01 with respect to the Borrower or the Parent Guarantor, the obligation of each Lender to make Loans and Swing Line Loans and all obligations of the L/C Issuers to make L/C Credit Extensions shall automatically terminate, the unpaid principal amount of all outstanding Loans and Swing Line Loans and all interest and other amounts as aforesaid shall automatically become due and payable, and the obligation of the Borrower to Cash Collateralize the L/C Obligations as aforesaid shall automatically become effective, in each case without further act of the Administrative Agent, any Lender or any L/C Issuer.
- 8.03 **Application of Funds.** After the exercise of remedies provided for in <u>Section 8.02</u> (or after the Loans have automatically become immediately due and payable and the L/C Obligations have automatically been required to be Cash Collateralized as set forth in <u>Section 8.02</u>), any amounts received on account of Secured Obligations and all net proceeds from the enforcement of Secured Obligations shall be applied by the Administrative Agent in the following order:
- (a) <u>First</u>, to payment of that portion of the Secured Obligations constituting fees, indemnities, expenses and other amounts (including Attorney Costs and amounts payable under <u>Article III</u>) payable to the Administrative Agent in its capacity as such, the L/C Issuers in their capacity as such and the Swing Line Lender in its capacity as such, ratably among the Administrative Agent, the L/C Issuers and Swing Line Lender in proportion to the respective amounts described in this clause (a) payable to them;

- (b) <u>Second</u>, to payment of that portion of the Secured Obligations constituting fees, indemnities and other amounts (other than principal and interest) payable to the Lenders and the L/C Issuers under the Loan Documents (including Attorney Costs and amounts payable under <u>Article III</u>), ratably among them in proportion to the amounts described in this <u>clause (b)</u> payable to them;
- (c) <u>Third</u>, to payment of that portion of the Secured Obligations constituting accrued and unpaid interest on the Loans, Unreimbursed Amounts, L/C Borrowings, Swing Line Loans and other Obligations, ratably among the Lenders and the L/C Issuers in proportion to the respective amounts described in this <u>clause (c)</u> payable to them;
- (d) <u>Fourth</u>, to payment of that portion of the Secured Obligations constituting unpaid principal of the Loans, Unreimbursed Amounts, L/C Borrowings and Swing Line Loans, and payment obligations then owing under Secured Hedge Agreements and Secured Cash Management Agreements, ratably among the Lenders, the L/C Issuers, the Hedge Banks and the Cash Management Banks in proportion to the respective amounts described in this <u>clause (d)</u> held by them:
- (e) <u>Fifth</u>, to the Administrative Agent for the account of the applicable L/C Issuers, to Cash Collateralize that portion of L/C Obligations comprised of the aggregate undrawn amount of Letters of Credit; and
 - (f) Last, the balance, if any, after all of the Obligations have been indefeasibly paid in full, to the Borrower or as otherwise required by Law.

Notwithstanding the foregoing, Secured Obligations arising under Secured Cash Management Agreements and Secured Hedge Agreements shall be excluded from the application described above if the Administrative Agent has not received written notice thereof, together with such supporting documentation as the Administrative Agent may request, from the applicable Cash Management Bank or Hedge Bank, as the case may be. Each Cash Management Bank or Hedge Bank not a party to this Agreement that has given the notice contemplated by the preceding sentence shall, by such notice, be deemed to have acknowledged and accepted the appointment of the Administrative Agent pursuant to the terms of Article IX for itself and its Affiliates as if a "Lender" party hereto.

Subject to Section 2.03(c), amounts used to Cash Collateralize the aggregate undrawn amount of Letters of Credit pursuant to clause (e) above shall be applied to satisfy drawings under such Letters of Credit as they occur. If any amount remains on deposit as Cash Collateral after all Letters of Credit have either been fully drawn or expired, such remaining amount shall be applied to the other Obligations, if any, in the order set forth above.

ARTICLE IX. ADMINISTRATIVE AGENT

9.01 Appointment and Authority.

(a) Each Lender and L/C Issuer hereby irrevocably appoints Wells Fargo Bank to act on its behalf as the Administrative Agent hereunder and under the other Loan Documents and authorizes the Administrative Agent to take such actions on its behalf and to exercise such powers as are delegated to the Administrative Agent by the terms hereof or thereof, together with such actions and powers as are reasonably incidental thereto. The provisions of this Article are solely for the benefit of the Administrative Agent, the Lenders and the L/C Issuers, and the Borrower shall not have rights as a third party beneficiary of such provisions. It is understood and agreed that the use of the term "agent" herein or in any other Loan Documents (or any other similar term) with reference to the Administrative Agent is not intended to

connote any fiduciary or other implied (or express) obligations arising under agency doctrine of any applicable law. Instead such term is used as a matter of market custom, and is intended to create or reflect only an administrative relationship between contracting parties.

- (b) The Administrative Agent shall also act as the "collateral agent" under the Loan Documents, and each Lender (including in its capacity as a Hedge Bank or Cash Management Bank, and on behalf of any Affiliate that is now or hereafter a Hedge Bank or Cash Management Bank) and L/C Issuer hereby irrevocably appoints and authorizes the Administrative Agent to act as the agent of such Lender and L/C Issuer for purposes of acquiring, holding and enforcing any and all Liens on Collateral granted by any of the Loan Parties to secure any of the Secured Obligations, together with such powers and discretion as are reasonably incidental thereto (including, without limitation, to enter into additional Loan Documents or supplements to existing Loan Documents on behalf of the Secured Parties). In this connection, the Administrative Agent, as "collateral agent" and any co-agents, sub-agents and attorneys-in-fact appointed by the Administrative Agent pursuant to this <u>Article IX</u> for purposes of holding or enforcing any Lien on the Collateral (or any portion thereof) granted under the Collateral Documents, or for exercising any rights and remedies thereunder at the direction of the Administrative Agent, shall be entitled to the benefits of all provisions of this <u>Article IX</u> and <u>Article XI</u> (including <u>Section 11.04</u>, as though such co-agents, sub-agents and attorneys-in-fact were the "collateral agent" under the Loan Documents) as if set forth in full herein with respect thereto.
- 9.02 **Rights as a Lender.** The Person serving as the Administrative Agent hereunder shall have the same rights and powers in its capacity as a Lender as any other Lender and may exercise the same as though it were not the Administrative Agent and the term "Lender" or "Lenders" shall, unless otherwise expressly indicated or unless the context otherwise requires, include the Person serving as the Administrative Agent hereunder in its individual capacity. Such Person and its Affiliates may accept deposits from, lend money to, act as the financial advisor or in any other advisory capacity for and generally engage in any kind of business with the Borrower or any Subsidiary or other Affiliate thereof as if such Person were not the Administrative Agent hereunder and without any duty to account therefor to the Lenders.
- 9.03 **Exculpatory Provisions.** The Administrative Agent shall not have any duties or obligations except those expressly set forth herein and in the other Loan Documents. Without limiting the generality of the foregoing, the Administrative Agent:
 - (a) shall not be subject to any fiduciary or other implied duties, regardless of whether a Default has occurred and is continuing;
- (b) shall not have any duty to take any discretionary action or exercise any discretionary powers, except discretionary rights and powers expressly contemplated hereby or by the other Loan Documents that the Administrative Agent is required to exercise as directed in writing by the Required Lenders (or such other number or percentage of the Lenders as shall be expressly provided for herein or in the other Loan Documents), provided that the Administrative Agent shall not be required to take any action that, in its opinion or the opinion of its counsel, may expose the Administrative Agent to liability or that is contrary to any Loan Document or applicable law; and
- (c) shall not, except as expressly set forth herein and in the other Loan Documents, have any duty to disclose, and shall not be liable for the failure to disclose, any information relating to the Borrower or any of its Affiliates that is communicated to or obtained by the Person serving as the Administrative Agent or any of its Affiliates in any capacity.

The Administrative Agent shall not be liable for any action taken or not taken by it (i) with the consent or at the request of the Required Lenders (or such other number or percentage of the Lenders as shall be necessary, or as the Administrative Agent shall believe in good faith shall be necessary, under the circumstances as provided in Sections 11.01 and 8.02) or (ii) in the absence of its own gross negligence or willful misconduct. The Administrative Agent shall be deemed not to have knowledge of any Default unless and until notice describing such Default is given to the Administrative Agent by the Borrower, a Lender or a L/C Issuer.

The Administrative Agent shall not be responsible for or have any duty to ascertain or inquire into (i) any statement, warranty or representation made in or in connection with this Agreement or any other Loan Document, (ii) the contents of any certificate, report or other document delivered hereunder or thereunder or in connection herewith or therewith, (iii) the performance or observance of any of the covenants, agreements or other terms or conditions set forth herein or therein or the occurrence of any Default, (iv) the validity, enforceability, effectiveness or genuineness of this Agreement, any other Loan Document or any other agreement, instrument or document or (v) the satisfaction of any condition set forth in Article IV or elsewhere herein, other than to confirm receipt of items expressly required to be delivered to the Administrative Agent.

9.04 **Reliance by Administrative Agent.** The Administrative Agent shall be entitled to rely upon, and shall not incur any liability for relying upon, any notice, request, certificate, consent, statement, instrument, document or other writing (including any electronic message, Internet or intranet website posting or other distribution) believed by it to be genuine and to have been signed, sent or otherwise authenticated by the proper Person. The Administrative Agent also may rely upon any statement made to it orally or by telephone and believed by it to have been made by the proper Person, and shall not incur any liability for relying thereon. In determining compliance with any condition hereunder to the making of a Loan, or the issuance of a Letter of Credit, that by its terms must be fulfilled to the satisfaction of a Lender or L/C Issuer, the Administrative Agent may presume that such condition is satisfactory to such Lender or L/C Issuer unless the Administrative Agent shall have received notice to the contrary from such Lender or L/C Issuer prior to the making of such Loan or the issuance of such Letter of Credit. The Administrative Agent may consult with legal counsel (who may be counsel for the Borrower), independent accountants and other experts selected by it, and shall not be liable for any action taken or not taken by it in accordance with the advice of any such counsel, accountants or experts.

9.05 **Delegation of Duties.** The Administrative Agent may perform any and all of its duties and exercise its rights and powers hereunder or under any other Loan Document by or through any one or more sub agents appointed by the Administrative Agent. The Administrative Agent and any such sub agent may perform any and all of its duties and exercise its rights and powers by or through their respective Related Parties. The exculpatory provisions of this Article shall apply to any such sub agent and to the Related Parties of the Administrative Agent and any such sub agent, and shall apply to their respective activities in connection with the syndication of the credit facilities provided for herein as well as activities as Administrative Agent.

9.06 **Resignation of Administrative Agent.** The Administrative Agent may at any time give notice of its resignation to the Lenders, the L/C Issuers and the Borrower. Upon receipt of any such notice of resignation, the Required Lenders shall have the right, in consultation with the Borrower, to appoint a successor, which shall be a bank with an office in the United States, or an Affiliate of any such bank with an office in the United States and shall be a United States person within the meaning of 7701(a)(30) of the Code. If no such successor shall have been so appointed by the Required Lenders and shall have accepted such appointment within 30 days after the retiring Administrative Agent gives notice of its resignation, then the retiring Administrative Agent may on behalf of the Lenders and the L/C Issuers, appoint a successor Administrative Agent meeting the qualifications set forth above; provided that if the Administrative Agent shall notify the Borrower and the Lenders that no qualifying Person has accepted such appointment, then such resignation shall nonetheless become effective in accordance with such notice and (1) the retiring Administrative Agent shall be discharged from its duties and obligations hereunder and under the other Loan Documents and (2) except for any indemnity payments owed to the retiring Administrative Agent, all payments, communications and determinations provided to be made by, to or through the Administrative Agent shall instead be made by or to each Lender and L/C Issuer directly, until such time as the Required Lenders appoint a successor Administrative Agent as provided for above in this Section. Upon the acceptance of a successor's appointment as Administrative Agent hereunder, such successor shall succeed to and become vested with all of the rights, powers, privileges and duties of the retiring (or retired) Administrative Agent (other than any rights to indemnity payments owed to the retiring Administrative Agent), and the retiring Administrative Agent shall be discharged from all of its duties and obligations hereunder or under the other Loan Documents (if not already discharged therefrom as provided above in this Section). The fees payable by the Borrower to a successor Administrative Agent shall be the same as those payable to its predecessor unless otherwise agreed between the Borrower and such successor. After the retiring Administrative Agent's resignation hereunder and under the other Loan Documents, the provisions of this Article and Section 11.04 shall continue in effect for the benefit of such retiring Administrative Agent, its sub agents and their respective Related Parties in respect of any actions taken or omitted to be taken by any of them while the retiring Administrative Agent was acting as Administrative Agent.

Any resignation by Wells Fargo Bank as Administrative Agent pursuant to this Section shall also constitute its resignation as L/C Issuer and Swing Line Lender. Upon the acceptance of a successor's appointment as Administrative Agent hereunder, (a) such successor shall succeed to and become vested with all of the rights, powers, privileges and duties of the retiring L/C Issuer and Swing Line Lender, (b) the retiring L/C Issuer and Swing Line Lender shall be discharged from all of their respective duties and obligations hereunder or under the other Loan Documents, and (c) the successor L/C Issuer and Swing Line Lender shall issue letters of credit in substitution for the Letters of Credit, if any, outstanding at the time of such succession or make other arrangements satisfactory to the retiring L/C Issuer to effectively assume the obligations of the retiring L/C Issuer with respect to such Letters of Credit.

9.07 **Non-Reliance on Administrative Agent and Other Lenders.** Each Lender and L/C Issuer acknowledges that it has, independently and without reliance upon the Administrative Agent or any other Lender or any of their Related Parties and based on such documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into this Agreement. Each Lender and L/C Issuer also acknowledges that it will, independently and without reliance upon the Administrative Agent or any other Lender or any of their Related Parties and based on such documents and information as it shall from time to time deem appropriate, continue to make its own decisions in taking or not taking action under or based upon this Agreement, any other Loan Document or any related agreement or any document furnished hereunder or thereunder.

- 9.08 **Administrative Agent May File Proofs of Claim.** In case of the pendency of any proceeding under any Debtor Relief Law or any other judicial proceeding relative to the Borrower or any Subsidiary, the Administrative Agent (irrespective of whether the principal of any Loan or L/C Obligation shall then be due and payable as herein expressed or by declaration or otherwise and irrespective of whether the Administrative Agent shall have made any demand on the Borrower) shall be entitled and empowered, by intervention in such proceeding or otherwise
- (a) to file and prove a claim for the whole amount of the principal and interest owing and unpaid in respect of the Loans, L/C Obligations, Swing Line Loans and all other Obligations that are owing and unpaid and to file such other documents as may be necessary or advisable in order to have the claims of the Lenders, the L/C Issuers, the Swing Line Lender and the Administrative Agent (including any claim for the reasonable compensation, expenses, disbursements and advances of the Lenders, the L/C Issuers, the Swing Line Lender and the Administrative Agent and their respective agents and counsel and all other amounts due the Lenders, the L/C Issuers, the Swing Line Lender and the Administrative Agent under Sections 2.03(i) and (j), 2.09 and 11.04) allowed in such judicial proceeding; and
 - (b) to collect and receive any monies or other property payable or deliverable on any such claims and to distribute the same;

and any custodian, receiver, assignee, trustee, liquidator, sequestrator or other similar official in any such judicial proceeding is hereby authorized by each Lender, L/C Issuer and the Swing Line Lender to make such payments to the Administrative Agent and, in the event that the Administrative Agent shall consent to the making of such payments directly to the Lenders, the L/C Issuers and the Swing Line Lender, to pay to the Administrative Agent any amount due for the reasonable compensation, expenses, disbursements and advances of the Administrative Agent and its agents and counsel, and any other amounts due the Administrative Agent under Sections 2.09 and 11.04.

Nothing contained herein shall be deemed to authorize the Administrative Agent to authorize or consent to or accept or adopt on behalf of any Lender, any L/C Issuer or the Swing Line Lender any plan of reorganization, arrangement, adjustment or composition affecting the Obligations or the rights of any Lender, any L/C Issuer or the Swing Line Lender to authorize the Administrative Agent to vote in respect of the claim of any Lender, any L/C Issuer or the Swing Line Lender in any such proceeding.

9.09 **Other Agents; Arrangers and Managers.** None of Arrangers, Joint Bookrunners, or Lenders or other Persons identified on the facing page of this Agreement as a "co-syndication agent," or "co-documentation agent," shall have any right, power, obligation, liability, responsibility or duty under this Agreement other than, in the case of such Lenders, those applicable to all Lenders as such. Without limiting the foregoing, none of the Lenders or other Persons so identified shall have or be deemed to have any fiduciary relationship with any Lender. Each Lender acknowledges that it has not relied, and will not rely, on any of the Lenders or other Persons so identified in deciding to enter into this Agreement or in taking or not taking action hereunder.

9.10 Collateral Matters.

(a) Each Lender and L/C Issuer authorizes and directs the Administrative Agent to enter into the Collateral Documents for the benefit of the Lenders and other Secured Parties. Each Lender hereby agrees, and each holder of any Loan (or any note evidencing a Loan) by the acceptance thereof will be deemed to agree, that, except as otherwise set forth herein, any action taken by the Required Lenders in accordance with the provisions of this Agreement or the Collateral Documents and the exercise by the Required Lenders of the powers set forth herein or therein, together with such other powers as are reasonably incidental thereto, shall be authorized and binding upon all of the Lenders. The Administrative

Agent is hereby authorized on behalf of all of the Lenders, without the necessity of any notice to or further consent from any Lender, from time to time prior to an Event of Default, to take any action with respect to any Collateral or the Collateral Documents which may be necessary to perfect and maintain perfected the security interest in and Liens upon the Collateral granted pursuant to the Collateral Documents.

- (b) The Lenders hereby authorize the Administrative Agent, at its option and in its discretion,
- (i) to release any Lien granted to or held by the Administrative Agent upon any Collateral and to release any Guaranty (x) upon the full and final payment of the Obligations (other than contingent Obligations for which no claim has been made), termination of the Commitments of all Lenders and L/C Issuers, and termination of all Letters of Credit (or Cash Collateralization thereof as acceptable to the applicable L/C Issuer) or (y) if approved, authorized or ratified in writing by the Required Lenders (or all of the Lenders hereunder, to the extent required by Section 11.01),
- (ii) to release any Lien on property being sold or otherwise disposed of (to Persons other than the Parent Guarantor, the Borrower or a Subsidiary) upon the sale or other disposition thereof in compliance with <u>Section 7.11</u>, or as otherwise may be expressly provided in the relevant Collateral Documents, and
- (iii) release the Guaranty of a Subsidiary upon request made by the Borrower if (x) the Borrower has designated such Subsidiary as an Immaterial Subsidiary in compliance with the terms of this Agreement, such Subsidiary is an Immaterial Subsidiary at the time of such release and no Default exists before or would exist immediately after such release, or (y) all of the Equity Interests in such Subsidiary are being Disposed of in a Disposition permitted by Section 7.11.

Upon request by the Administrative Agent at any time, the Lenders will confirm in writing the Administrative Agent's authority to release particular types or items of Collateral or particular Guaranties pursuant to this Section 9.10. In each case as specified in this Section 9.10(b), the Administrative Agent will, at the Borrower's expense and upon presentation by the Borrower of a certificate of a Responsible Officer dated the date of the requested release and certifying as to the matters in the foregoing clause (i), (ii) or (iii), as applicable, execute and deliver to the applicable Loan Party such documents as such Loan Party may reasonably request to evidence the release such item of Collateral or such Guaranty, as applicable, in each case in accordance with the terms of this Section 9.10(b).

(c) The Administrative Agent shall have no obligation whatsoever to the Lenders or to any other Person to require evidence that or otherwise assure that the Collateral exists or is owned by any Loan Party or any other grantor of a Lien under the Collateral Documents or is cared for, protected or insured or that the Liens granted to the Administrative Agent herein or pursuant hereto have been properly or sufficiently or lawfully created, perfected, protected or enforced or are entitled to any particular priority, or to exercise or to continue exercising at all or in any manner or under any duty of care, disclosure or fidelity any of the rights, authorities and powers granted or available to the Administrative Agent in this Section 9.10 or in any of the Collateral Documents, it being understood and agreed that in respect of the Collateral, or any act, omission or event related thereto, the Administrative Agent may act in any manner it may deem appropriate, in its sole discretion, given the Administrative Agent's own interest in the Collateral as one of the Lenders, and that the Administrative Agent shall have no duty or liability whatsoever to the Lenders, except for its gross negligence or willful misconduct (as determined by a court of competent jurisdiction in a final and non-appealable decision).

9.11 Secured Hedge Agreements and Secured Cash Management Agreements.

- (a) Each Hedge Bank and each Cash Management Bank shall be deemed to appoint the Administrative Agent its nominee and agent, and to act for and on behalf of such Affiliate in connection with the Collateral Documents and to be bound by this <u>Article IX</u>.
- (b) No Cash Management Bank or Hedge Bank that obtains the benefits of Section 8.02 or any Collateral by virtue of the provisions hereof or of any Collateral Document shall have any right to notice of any action or to consent to, direct or object to any action hereunder or under any other Loan Document or otherwise in respect of the Collateral (including the release or impairment of any Collateral) other than in its capacity as a Lender and, in such case, only to the extent expressly provided in the Loan Documents. Notwithstanding any other provision of this Article IX to the contrary, the Administrative Agent shall not be required to verify the payment of, or that other satisfactory arrangements have been made with respect to, Secured Cash Management Agreements and Secured Hedge Agreements unless the Administrative Agent has received written notice of such Secured Cash Management Agreements and Secured Hedge Agreements, together with such supporting documentation as the Administrative Agent may request, from the applicable Cash Management Bank or Hedge Bank, as the case may be.

ARTICLE X. CONTINUING GUARANTY OF THE PARENT GUARANTOR

10.01 **Guaranty.** The Parent Guarantor hereby unconditionally and irrevocably guarantees, jointly with any other guarantors from time to time and severally, as a primary obligor and not merely as a surety, and as a guaranty of payment and performance and not merely as a guaranty of collection, full and prompt payment when due, whether at stated maturity, by required prepayment, upon acceleration, demand or otherwise, and at all times thereafter, of any and all of the Guaranteed Obligations and the punctual performance of all of the terms of the Loan Documents. "Guaranteed Obligations" means (a) the Obligations, and (b) all other advances to, and debts, liabilities, obligations, covenants and duties of the Borrower or any other Loan Party arising under any Secured Hedge Agreement or under any Secured Cash Management Agreement. Without limiting the generality of the foregoing, the Guaranteed Obligations shall include such indebtedness, obligations, and liabilities which may be or hereafter become unenforceable or shall be an allowed or disallowed claim under any Debtor Relief Laws, and shall include interest and fees that accrue after the commencement by or against any Loan Party or any Affiliate thereof of any proceeding under any Debtor Relief Laws.

The Administrative Agent's books and records showing the amount of the Guaranteed Obligations shall be admissible in evidence in any action or proceeding, and shall be binding upon the Parent Guarantor, and conclusive for the purpose of establishing the amount of the Guaranteed Obligations, absent manifest error. This Guaranty shall not be affected by the genuineness, validity, regularity or enforceability of the Guaranteed Obligations or any instrument or agreement evidencing any Guaranteed Obligations, or by the existence, validity, enforceability, perfection, non-perfection or extent of any collateral therefor, or by any fact or circumstance relating to the Guaranteed Obligations which might otherwise constitute a defense to the obligations of the Parent Guarantor under this Guaranty, and the Parent Guarantor hereby irrevocably waives any defenses it may now have or hereafter acquire in any way relating to any or all of the foregoing, other than a defense of payment or performance.

10.02 **Rights of Lenders.** The Parent Guarantor consents and agrees that the Administrative Agent, the L/C Issuers, the Lenders and any other Secured Party may, at any time and from time to time, without notice or demand, and without affecting the enforceability or continuing effectiveness hereof: (a) amend, extend, renew,

compromise, discharge, accelerate or otherwise change the time for payment or the terms of the Guaranteed Obligations or any part thereof; (b) take, hold, exchange, enforce, waive, release, fail to perfect, sell, or otherwise dispose of any security for the payment of this Guaranty or any Guaranteed Obligations; (c) apply such security and direct the order or manner of sale thereof as the Administrative Agent, the L/C Issuers, the Lenders and any other Secured Party in their sole discretion may determine; and (d) release or substitute one or more of any endorsers or other guarantors of any of the Guaranteed Obligations. Without limiting the generality of the foregoing, the Parent Guarantor consents to the taking of, or failure to take, any action which might in any manner or to any extent vary the risks of the Parent Guarantor under this Guaranty or which, but for this provision, might operate as a discharge of the Parent Guarantor.

10.03 Certain Waivers. The Parent Guarantor hereby irrevocably waives to the fullest extent permitted by law (a) any defenses (other than a defense of payment or performance) it may now have or hereafter acquire in any way relating to, any or all of the following: (i) any lack of validity or enforceability of any Loan Document or any agreement or instrument relating thereto; (ii) without limiting Section 10.02, any change in the time, manner or place of payment of, or in any other term of, all or any of the Guaranteed Obligations of any other Loan Party under or in respect of the Loan Documents, or any other amendment or waiver of or any consent to departure from any Loan Document, including any increase in the Guaranteed Obligations resulting from the extension of additional credit to any Loan Party or any of its Subsidiaries or otherwise; (iii) any disability or other defense of the Borrower or any other Loan Party or the cessation from any cause whatsoever of the liability of the Borrower or any other Loan Party; (iv) any claim that the Parent Guarantor's obligations exceed or are more burdensome than those of the Borrower or any other Loan Party; (v) any change, restructuring or termination of the corporate structure or existence of any Loan Party or any of its Subsidiaries; or (vi) without limiting Section 10.02, the failure of any other Person to Guarantee the Guaranteed Obligations as required pursuant to Section 6.11 or the release or reduction of liability of any other Guarantor; (b) the benefit of any statute of limitations affecting the Parent Guarantor's liability hereunder; (c) any right to proceed against the Borrower or any other Loan Party, proceed against or exhaust any security for the Guaranteed Obligations, or pursue any other remedy in the power of the Administrative Agent, any Lender, any L/C Issuer, the Swing Line Lender or any other Secured Party whatsoever; (d) any benefit of and any right to participate in any security now or hereafter held as security for the Guaranteed Obligations; (e) presentment or protest to, demand of or payment from the Borrower or any other Guaranter of any of the Guaranteed Obligations; (f) notice of acceptance of its guarantee and notice of protest for nonpayment; and (g) any and all other defenses or benefits that may be derived from or afforded by applicable law limiting the liability of or exonerating guarantors or sureties. The Parent Guarantor expressly waives all setoffs and counterclaims and all presentments, demands for payment or performance, notices of nonpayment or nonperformance, protests, notices of protest, notices of dishonor and all other notices or demands of any kind or nature whatsoever with respect to the Guaranteed Obligations, and all notices of acceptance of this Guaranty or of the existence, creation or incurrence of new or additional Guaranteed Obligations.

10.04 **Obligations Independent.** The obligations of the Parent Guarantor hereunder are those of primary obligor, and not merely as surety, and are independent of the Guaranteed Obligations and the obligations of any other guarantor, and a separate action may be brought against the Parent Guarantor to enforce this Guaranty whether or not the Borrower or any other Person is joined as a party.

10.05 **Subrogation.** The Parent Guarantor shall not exercise any right of subrogation, contribution, indemnity, reimbursement or similar rights with respect to any payments it makes under this Guaranty until Payment In Full. If any amounts are paid to the Parent Guarantor in violation of the foregoing limitation, then such amounts shall be held in trust for the benefit of the Administrative Agent, the Lenders, the L/C Issuers, the Swing Line Lender and any other Secured Party and shall forthwith be paid to the Administrative Agent to reduce the amount of the Guaranteed Obligations, whether matured or unmatured.

10.06 **Termination; Reinstatement.** This Guaranty is a continuing and irrevocable guaranty of all Guaranteed Obligations now or hereafter existing and shall remain in full force and effect until Payment In Full. Notwithstanding the foregoing, this Guaranty shall continue in full force and effect or be revived, as the case may be, if any payment by or on behalf of the Borrower or any other Loan Party is made, or any of the Administrative Agent, the Lenders, the L/C Issuers, the Swing Line Lender or any other Secured Party exercises its right of setoff, in respect of the Obligations and such payment or the proceeds of such setoff or any part thereof is subsequently invalidated, declared to be fraudulent or preferential, set aside or required (including pursuant to any settlement entered into by any of the Administrative Agent, the Lenders, the L/C Issuers, the Swing Line Lender or any other Secured Party in their discretion) to be repaid to a trustee, receiver or any other party, in connection with any proceeding under any applicable bankruptcy, insolvency, reorganization, moratorium or other laws affecting creditors' rights generally or otherwise, all as if such payment had not been made or such setoff had not occurred and whether or not any of the Administrative Agent, the Lenders, the L/C Issuers, the Swing Line Lender or any other Secured Party are in possession of or have released this Guaranty and regardless of any prior revocation, rescission, termination or reduction. The obligations of the Parent Guarantor under this Section shall survive termination of this Guaranty.

10.07 **Subordination.** The Parent Guarantor hereby subordinates the payment of all obligations and indebtedness of the Borrower or any other Loan Party owing to the Parent Guarantor, whether now existing or hereafter arising, including any obligation of the Borrower or any other Loan Party to the Parent Guarantor as subrogee of the Administrative Agent, the Lenders, the L/C Issuers, the Swing Line Lender and any other Secured Party or resulting from the Parent Guarantor's performance under this Guaranty, to the payment in full in cash of all Guaranteed Obligations (other than indemnities and other contingent obligations not then due and payable and as to which no claim has been made). So long as an Event of Default exists, if the Administrative Agent or the Required Lenders so request, any such obligation or indebtedness of the Borrower or any other Loan Party to the Parent Guarantor shall be enforced and performance received by the Parent Guarantor as trustee for the Administrative Agent, the Lenders, the L/C Issuers, the Swing Line Lender and any other Secured Party and the proceeds thereof shall be paid over to the Administrative Agent on account of the Guaranteed Obligations, but without reducing or affecting in any manner the liability of the Parent Guarantor under this Guaranty.

10.08 **Stay of Acceleration.** If acceleration of the time for payment of any of the Guaranteed Obligations is stayed in connection with any case commenced by or against the Parent Guarantor or any other Loan Party under any Debtor Relief Law or otherwise, all such amounts shall nonetheless be payable by the Parent Guarantor immediately upon demand by the Administrative Agent.

ARTICLE XI. MISCELLANEOUS

11.01 Amendments, Etc.

No amendment or waiver of any provision of this Agreement or any other Loan Document, and no consent to any departure by the Borrower therefrom, shall be effective unless in writing signed by the Required Lenders and the Borrower, and acknowledged by the Administrative Agent, and each such waiver or consent shall be effective only in the specific instance and for the specific purpose for which given; <u>provided</u>, <u>however</u>, that no such amendment, waiver or consent shall:

- (a) waive any condition set forth in Section 4.01 or Section 4.02 without the written consent of each Lender;
- (b) extend or increase the Commitment of any Lender (or reinstate any Commitment terminated pursuant to <u>Section 8.02</u>) without the written consent of such Lender;
- (c) postpone any date fixed by this Agreement or any other Loan Document for any payment or mandatory prepayment of principal, interest, fees or other amounts due to the Lenders (or any of them) hereunder or under any other Loan Document without the written consent of each Lender directly affected thereby:
- (d) reduce the principal of, or the rate of interest specified herein on, any Loan, Swing Line Loan or L/C Borrowing, or any fees or other amounts payable hereunder or under any other Loan Document, without the written consent of each Lender directly affected thereby; <u>provided</u>, <u>however</u>, that only the consent of the Required Lenders shall be necessary (i) to amend the definition of "<u>Default Rate</u>" or to waive any obligation of the Borrower to pay interest or Letter of Credit Fees at the Default Rate or (ii) to amend any financial covenant hereunder (or any defined term used therein) even if the effect of such amendment would be to reduce the rate of interest on any Loan, Swing Line Loan or L/C Borrowing or to reduce any fee payable hereunder;
- (e) change <u>Section 2.13</u> or <u>Section 8.03</u> in a manner that would alter the pro rata sharing of payments required thereby without the written consent of each Lender;
- (f) change any provision of this Section or the definition of "Required Lenders" or any other provision hereof specifying the number or percentage of Lenders required to amend, waive or otherwise modify any rights hereunder or make any determination or grant any consent hereunder, without the written consent of each Lender;
- (g) release (i) the Parent Guarantor from its Guaranty or (ii) all or substantially all of the value of the Subsidiary Guarantees under the Subsidiary Guaranty Agreement (other than as authorized pursuant to <u>Section 9.10</u>), without the consent of each Lender; or
 - (h) release all or substantially all of the Collateral in any transaction or series of transactions, without the consent of each Lender;

and, provided further, that (i) no amendment, waiver or consent shall affect the rights or duties of any L/C Issuer under this Agreement or any Letter of Credit Application relating to any Letter of Credit issued or to be issued by it unless in writing and signed by such L/C Issuer, in addition to the Lenders required above; (ii) no amendment, waiver or consent shall, unless in writing and signed by the Swing Line Lender in addition to the Lenders required above, affect the rights or duties of the Swing Line Lender under this

Agreement relating to any Swing Line Loan; (iii) no amendment, waiver or consent shall, unless in writing and signed by the Administrative Agent in addition to the Lenders required above, affect the rights or duties of the Administrative Agent under this Agreement or any other Loan Document; (iv) the Fee Letters may be amended, or rights or privileges thereunder waived, in a writing executed only by the parties thereto; and (v) the Administrative Agent and the Borrower (or other applicable Loan Party) shall be permitted to amend any provision of the Loan Documents (and such amendment shall become effective without any further action or consent of any other party to any Loan Document) if the Administrative Agent and the Borrower shall have jointly identified an obvious error or any error or omission of a technical or immaterial nature in any such provision. In accordance with Section 11.14, the Borrower may replace any Lender that does not consent to any amendment, waiver or other action that requires the consent of all Lenders and that has been consented to by the Required Lenders. Notwithstanding anything to the contrary herein, no Defaulting Lender shall have any right to approve or disapprove any amendment, waiver or consent hereunder, except that (w) the Commitment (and, where applicable, maturity date) of such Lender may not be increased or extended without the consent of such Lender, (x) the principal amount of any obligation, or the amount or rate of interest, owing to such Lender may not be reduced without the consent of such Lender, (y) the provisions of this sentence may not be revised in a manner that would be adverse to such Lender without the consent of such Lender, and (z) any amendment, waiver or consent that would materially adversely affect the rights and obligations of such Lender in a manner that is disproportional to the other Lenders will require the consent of such Lender.

Without the consent of any other Person, the applicable Loan Party or Loan Parties and the Administrative Agent may (in its or their respective sole discretion, or shall, to the extent required by any Loan Document) enter into any waiver, amendment or modification of any Loan Document, or enter into any new agreement or instrument, in each case to effect the granting, perfection, protection, expansion or enhancement of any security interest in any Collateral or additional property to become Collateral for the benefit of the Secured Parties, or as required by local law to give effect to, or protect any security interest for the benefit of the Secured Parties, in any property or so that the security interests therein comply with applicable Law.

11.02 Notices; Electronic Communications.

- (a) <u>Notices Generally</u>. Except in the case of notices and other communications expressly permitted to be given by telephone (and except as provided in <u>subsection (b)</u> below), all notices and other communications provided for herein shall be in writing and shall be delivered by hand or overnight courier service, mailed by certified or registered mail or sent by telecopier as follows, and all notices and other communications expressly permitted hereunder to be given by telephone shall be made to the applicable telephone number, as follows:
- (i) if to the Borrower, the Administrative Agent, or to Wells Fargo Bank, as Swing Line Lender or as a L/C Issuer, to the address, telecopier number, electronic mail address or telephone number specified for such Person on <u>Schedule 11.02</u>;
- (ii) if to any other L/C Issuer, to it at its address, telecopier number, electronic mail address or telephone number separately notified in writing by such L/C Issuer to the Borrower and the Administrative Agent; and
- (iii) if to any other Lender, to the address, telecopier number, electronic mail address or telephone number specified in its Administrative Questionnaire

Notices sent by hand or overnight courier service, or mailed by certified or registered mail, shall be deemed to have been given when received; notices sent by telecopier shall be deemed to have been given when sent (except that, if not given during normal business hours for the recipient, shall be deemed to have been given at the opening of business on the next business day for the recipient). Notices delivered through electronic communications to the extent provided in <u>subsection (b)</u> below, shall be effective as provided in such <u>subsection (b)</u>.

(b) <u>Electronic Communications</u>. Notices and other communications to the Lenders and the L/C Issuers hereunder may be delivered or furnished by electronic communication (including e-mail and Internet or intranet websites) pursuant to procedures approved by the Administrative Agent, <u>provided</u> that the foregoing shall not apply to notices to any Lender or L/C Issuer pursuant to <u>Article II</u> if such Lender or L/C Issuer, as applicable, has notified the Administrative Agent that it is incapable of receiving notices under such Article by electronic communication. The Administrative Agent or the Borrower may, in its discretion, agree to accept notices and other communications to it hereunder by electronic communications pursuant to procedures approved by it, <u>provided</u> that approval of such procedures may be limited to particular notices or communications.

Unless the Administrative Agent otherwise prescribes, (i) notices and other communications sent to an e-mail address shall be deemed received upon the sender's receipt of an acknowledgement from the intended recipient (such as by the "return receipt requested" function, as available, return e-mail or other written acknowledgement), <u>provided</u> that if such notice or other communication is not sent during the normal business hours of the recipient, such notice or communication shall be deemed to have been sent at the opening of business on the next business day for the recipient, and (ii) notices or communications posted to an Internet or intranet website shall be deemed received upon the deemed receipt by the intended recipient at its e-mail address as described in the foregoing clause (i) of notification that such notice or communication is available and identifying the website address therefor.

(c) The Platform.

- (i) Each Loan Party agrees that the Administrative Agent may, but shall not be obligated to, make the Borrower Materials available to the L/C Issuers and the Lenders by posting the Borrower Materials on the Platform.
- (ii) THE PLATFORM IS PROVIDED "AS IS" AND "AS AVAILABLE." THE AGENT PARTIES (AS DEFINED BELOW) DO NOT WARRANT THE ACCURACY OR COMPLETENESS OF THE BORROWER MATERIALS OR THE ADEQUACY OF THE PLATFORM, AND EXPRESSLY DISCLAIM LIABILITY FOR ERRORS IN OR OMISSIONS FROM THE BORROWER MATERIALS. NO WARRANTY OF ANY KIND, EXPRESS, IMPLIED OR STATUTORY, INCLUDING ANY WARRANTY OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, NON-INFRINGEMENT OF THIRD PARTY RIGHTS OR FREEDOM FROM VIRUSES OR OTHER CODE DEFECTS, IS MADE BY ANY AGENT PARTY IN CONNECTION WITH THE BORROWER MATERIALS OR THE PLATFORM. In no event shall the Administrative Agent or any of its Related Parties (collectively, the "Agent Parties") have any liability to the Borrower, any Lender or any other Person for losses, claims, damages, liabilities or expenses of any kind (whether in tort, contract or otherwise) arising out of the Borrower's or the Administrative Agent's transmission of Borrower Materials through the Internet, except to the extent that such losses, claims, damages, liabilities or expenses are determined by a court of competent jurisdiction by a final and nonappealable judgment to have resulted from the gross negligence or willful misconduct of such Agent Party; provided, however, that in no event shall any Agent Party have any liability to the Borrower, any Lender or any other Person for indirect, special, incidental, consequential or punitive damages (as opposed to direct or actual damages).

- (d) <u>Change of Address, Etc.</u> Each of the Borrower, the Administrative Agent and the L/C Issuers may change its address, telecopier or telephone number for notices and other communications hereunder by notice to the other parties hereto. Each Lender may change its address, telecopier or telephone number for notices and other communications hereunder by notice to the Administrative Agent. In addition, each Lender agrees to notify the Administrative Agent from time to time to ensure that the Administrative Agent has on record (i) an effective address, contact name, telephone number, telecopier number and electronic mail address to which notices and other communications may be sent and (ii) accurate wire instructions for such Lender. Furthermore, each Public Lender agrees to cause at least one individual at or on behalf of such Public Lender to at all times have selected the "Private Side Information" or similar designation on the content declaration screen of the Platform in order to enable such Public Lender or its delegate, in accordance with such Public Lender's compliance procedures and applicable Law, including United States Federal and state securities Laws, to make reference to Borrower Materials that are not made available through the "Public Side Information" portion of the Platform and that may contain material non-public information with respect to the Borrower or its securities for purposes of United States Federal or state securities laws.
- (e) <u>Reliance by Administrative Agent, L/C Issuers and Lenders</u>. The Administrative Agent, the L/C Issuers and the Lenders shall be entitled to rely and act upon any notices (including telephonic Loan Notices) purportedly given by or on behalf of the Borrower even if (i) such notices were not made in a manner specified herein, were incomplete or were not preceded or followed by any other form of notice specified herein, or (ii) the terms thereof, as understood by the recipient, varied from any confirmation thereof. The Borrower shall indemnify the Administrative Agent, each L/C Issuer, each Lender and the Related Parties of each of them from all losses, costs, expenses and liabilities resulting from the reliance by such Person on each notice purportedly given by or on behalf of the Borrower. All telephonic notices to and other telephonic communications with the Administrative Agent may be recorded by the Administrative Agent, and each of the parties hereto hereby consents to such recording.

11.03 **No Waiver; Cumulative Remedies; Enforcement.** No failure by any Lender or the Administrative Agent to exercise, and no delay by any such Person in exercising, any right, remedy, power or privilege hereunder shall operate as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privileges. The rights, remedies, powers and privileges herein provided are cumulative and not exclusive of any rights, remedies, powers and privileges provided by law.

Notwithstanding anything to the contrary contained herein or in any other Loan Document, the authority to enforce rights and remedies hereunder and under the other Loan Documents against the Borrower or any Subsidiary shall be vested exclusively in, and all actions and proceedings at law in connection with such enforcement shall be instituted and maintained exclusively by, the Administrative Agent in accordance with Section 8.02 for the benefit of all the Lenders and L/C Issuers; provided, however, that the foregoing shall not prohibit (a) the Administrative Agent from exercising on its own behalf the rights and remedies that inure to its benefit (solely in its capacity as Administrative Agent) hereunder and under the other Loan Documents, (b) any L/C Issuer from exercising the rights and remedies that inure to its benefit (solely in its capacity as L/C Issuer) hereunder and under the other Loan Documents, (c) any Lender from exercising setoff rights in accordance with Section 11.08 (subject to the terms of Section 2.13), or (d) any Lender from filing proofs of claim or appearing and filing pleadings on its own behalf during the pendency of a proceeding relative to the Borrower or any Subsidiary under any Debtor Relief Law; and provided, further, that if at any time there is no Person acting as Administrative Agent hereunder and under the other Loan Documents, then (i) the Required Lenders shall have the rights otherwise ascribed to the Administrative Agent pursuant to Section 8.02 and (ii) in addition to the matters set forth in clauses (b), (c) and (d) of the preceding proviso and subject to Section 2.13, any Lender may, with the consent of the Required Lenders, enforce any rights and remedies available to it and as authorized by the Required Lenders.

11.04 Expenses; Indemnity; Damage Waiver.

- (a) <u>Costs and Expenses</u>. The Borrower shall pay (i) all reasonable out of pocket expenses incurred by the Administrative Agent and its Affiliates (including the reasonable fees, charges and disbursements of one counsel for the Administrative Agent), in connection with the syndication of the credit facilities provided for herein, the preparation, negotiation, execution, delivery and administration of this Agreement and the other Loan Documents or any amendments, modifications or waivers of the provisions hereof or thereof (whether or not the transactions contemplated hereby or thereby shall be consummated), (ii) all reasonable out of pocket expenses incurred by any L/C Issuer in connection with the issuance, amendment, renewal or extension of any Letter of Credit or any demand for payment thereunder and (iii) all out of pocket expenses incurred by the Administrative Agent, any Lender or any L/C Issuer), in connection with the enforcement or protection of its rights (A) in connection with this Agreement and the other Loan Documents, including its rights under this Section, or (B) in connection with the Loans made or Letters of Credit issued hereunder, including all such out of pocket expenses incurred during any workout, restructuring or negotiations in respect of such Loans or Letters of Credit.
- (b) <u>Indemnification by the Borrower</u>. Except with respect to Taxes, which are addressed in <u>Section 3.01</u> hereof, the Borrower shall indemnify the Administrative Agent (and any sub-agent thereof), each Lender, each L/C Issuer and the Swing Line Lender, and each Related Party of any of the foregoing Persons (each such Person being called an "Indemnitee") against, and hold each Indemnitee harmless from, any and all losses, claims, damages, penalties, liabilities and related expenses (including the fees, charges and disbursements of any counsel for any Indemnitee), incurred by any Indemnitee or asserted against any Indemnitee by any third party or any Loan Party or any Subsidiary arising out of, in connection with, or as a result of (i) the execution or delivery of this Agreement, any other Loan Document or any agreement or instrument contemplated hereby or thereby, the performance by the parties hereto of their respective obligations hereunder or thereunder, the consummation of the transactions contemplated hereby or thereby, or, in the case of the Administrative Agent (and any sub-agent thereof) and its Related Parties only, the administration of this Agreement and the other Loan Documents (including in respect of any matters addressed in Section 3.01), (ii) any Loan or Letter of Credit or the use or proposed use of the proceeds therefrom (including any refusal by a L/C Issuer to honor a demand for payment under a Letter of Credit if the documents presented in connection with such demand do not strictly comply with the terms of such Letter of Credit), (iii) any actual or alleged presence or release of Hazardous Materials on or from any property owned or operated by any Loan Party or Subsidiary, or any Environmental Liability related in any way to any Loan Party or Subsidiary, or (iv) any actual or prospective claim, litigation, investigation or proceeding relating to any of the foregoing, whether based on contract, tort or any other theory, whether brought by a third party or by a Loan Party or Subsidiary, and regardless of whether any Indemnitee is a party thereto, IN ALL CASES, WHETHER OR NOT CAUSED BY OR ARISING, IN WHOLE OR IN PART, OUT OF THE COMPARATIVE, CONTRIBUTORY OR SOLE NEGLIGENCE OF THE INDEMNITEE; provided that such indemnity shall not, as to any Indemnitee, be available to the extent that such losses, claims, damages, liabilities or related expenses (x) are determined by a court of competent jurisdiction by final and nonappealable judgment to have resulted from the gross negligence or willful misconduct of such Indemnitee or (y) result from a claim brought by a Loan Party or any Subsidiary against an Indemnitee for breach in bad faith of such Indemnitee's obligations hereunder or under any other Loan Document, if such Loan Party or such Subsidiary has obtained a final and nonappealable judgment in its favor on such claim as determined by a court of competent jurisdiction.

- (c) <u>Reimbursement by Lenders</u>. To the extent that the Borrower for any reason fails to indefeasibly pay any amount required under <u>subsection (a)</u> or <u>(b)</u> of this Section to be paid by it to the Administrative Agent (or any sub-agent thereof), any L/C Issuer, the Swing Line Lender or any Related Party of any of the foregoing, each Lender severally agrees to pay to the Administrative Agent (or any such sub-agent), such L/C Issuer, the Swing Line Lender or such Related Party, as the case may be, such Lender's Pro Rata Share (determined as of the time that the applicable unreimbursed expense or indemnity payment is sought) of such unpaid amount, <u>provided</u> that the unreimbursed expense or indemnified loss, claim, damage, liability or related expense, as the case may be, was incurred by or asserted against the Administrative Agent (or any such sub-agent), such L/C Issuer or the Swing Line Lender in its capacity as such, or against any Related Party of any of the foregoing acting for the Administrative Agent (or any such sub-agent), such L/C Issuer or the Swing Line Lender in connection with such capacity. The obligations of the Lenders under this <u>subsection (c)</u> are subject to the provisions of <u>Section 2.12(e)</u>.
- (d) <u>Waiver of Consequential Damages</u>, <u>Etc</u>. To the fullest extent permitted by applicable law, no Loan Party shall assert, and each Loan Party hereby waives, any claim against any Indemnitee, on any theory of liability, for special, indirect, consequential or punitive damages (as opposed to direct or actual damages) arising out of, in connection with, or as a result of, this Agreement, any other Loan Document or any agreement or instrument contemplated hereby, the transactions contemplated hereby or thereby, any Loan or Letter of Credit or the use of the proceeds thereof. No Indemnitee referred to in <u>subsection</u> (<u>b</u>) above shall be liable for any damages arising from the use by unintended recipients of any information or other materials distributed to such unintended recipients by such Indemnitee through telecommunications, electronic or other information transmission systems in connection with this Agreement or the other Loan Documents or the transactions contemplated hereby or thereby other than for direct or actual damages resulting from the gross negligence or willful misconduct of such Indemnitee as determined by a final and nonappealable judgment of a court of competent jurisdiction.
 - (e) Payments. All amounts due under this Section shall be payable not later than ten Business Days after demand therefor.
- (f) <u>Survival</u>. The agreements in this Section shall survive the resignation of the Administrative Agent and any L/C Issuer, the replacement of any Lender, the termination of the Aggregate Commitments and the repayment, satisfaction or discharge of the Obligations.

11.05 **Payments Set Aside.** To the extent that any payment by or on behalf of any Loan Party is made to the Administrative Agent or any Lender, or the Administrative Agent or any Lender exercises its right of set-off, and such payment or the proceeds of such set-off or any part thereof is subsequently invalidated, declared to be fraudulent or preferential, set aside or required (including pursuant to any settlement entered into by the Administrative Agent or such Lender in its discretion) to be repaid to a trustee, receiver or any other party, in connection with any proceeding under any Debtor Relief Law or otherwise, then (a) to the extent of such recovery, the obligation or part thereof originally intended to be satisfied shall be revived and continued in full force and effect as if such payment had not been made or such set-off had not occurred, and (b) each Lender severally agrees to pay to the Administrative Agent upon demand its applicable share of any amount so recovered from or repaid by the Administrative Agent, plus interest thereon from the date of such demand to the date such payment is made at a rate per annum equal to the Federal Funds Rate from time to time in effect.

11.06 Successors and Assigns.

- (a) <u>Successors and Assigns Generally</u>. The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns permitted hereby, except that the Borrower may not assign or otherwise transfer any of its rights or obligations hereunder without the prior written consent of the Administrative Agent and each Lender and no Lender may assign or otherwise transfer any of its rights or obligations hereunder except (i) to an assignee in accordance with the provisions of <u>subsection (b)</u> of this Section, (ii) by way of participation in accordance with the provisions of <u>subsection (d)</u> of this Section, or (iii) by way of pledge or assignment of a security interest subject to the restrictions of <u>subsection (f)</u> of this Section, and any other attempted assignment or transfer by any party hereto shall be null and void. Nothing in this Agreement, expressed or implied, shall be construed to confer upon any Person (other than the parties hereto, their respective successors and assigns permitted hereby, Participants to the extent provided in <u>subsection (d)</u> of this Section and, to the extent expressly contemplated hereby, the Indemnitees) any legal or equitable right, remedy or claim under or by reason of this Agreement.
- (b) <u>Assignments by Lenders</u>. Any Lender may at any time assign to one or more assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of its Commitment and the Loans (including for purposes of this <u>subsection (b)</u>, participations in L/C Obligations) at the time owing to it); <u>provided</u> that any such assignment shall be subject to the following conditions:

(i) Minimum Amounts.

- (A) in the case of an assignment of the entire remaining amount of the assigning Lender's Commitment and the Loans at the time owing to it or in the case of an assignment to a Lender, an Affiliate of a Lender or an Approved Fund, no minimum amount need be assigned; and
- (B) in any case not described in <u>Section 11.06(b)(i)(A)</u>, the aggregate amount of the Commitment (which for this purpose includes Loans outstanding thereunder) or, if the Commitment is not then in effect, the principal outstanding balance of the Loans of the assigning Lender subject to each such assignment, determined as of the date the Assignment and Assumption with respect to such assignment is delivered to the Administrative Agent or, if "Trade Date" is specified in the Assignment and Assumption, as of the Trade Date, shall not be less than \$5,000,000 unless each of the Administrative

Agent and, so long as no Event of Default has occurred and is continuing, the Borrower otherwise consents (each such consent not to be unreasonably withheld or delayed and such consent of the Borrower being deemed to have been given by the Borrower unless it objects thereto within five Business Days after having received notice thereof); <u>provided</u>, however, that concurrent assignments to members of an Assignee Group and concurrent assignments from members of an Assignee Group to a single Eligible Assignee (or to an Eligible Assignee and members of its Assignee Group) will be treated as a single assignment for purposes of determining whether such minimum amount has been met.

- (ii) <u>Proportionate Amounts</u>. Each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement with respect to the Loans or the Commitment assigned.
 - (iii) Required Consents. No consent shall be required for any assignment except to the extent required by Section 11.06(b)(i)(B) and, in addition:
- (A) the consent of the Borrower (such consent not to be unreasonably withheld or delayed and such consent of the Borrower being deemed to have been given by the Borrower unless it objects thereto within five Business Days after having received notice thereof) shall be required unless (1) an Event of Default has occurred and is continuing at the time of such assignment or (2) such assignment is to a Lender, an Affiliate of a Lender or an Approved Fund;
- (B) the consent of the Administrative Agent (such consent not to be unreasonably withheld or delayed) shall be required if such assignment is to a Person that is not a Lender, an Affiliate of such Lender or an Approved Fund with respect to such Lender; and
- (C) the consent of each L/C Issuer and the Swing Line Lender (each such consent not to be unreasonably withheld or delayed) shall be required for any assignment.
- (iv) <u>Assignment and Assumption</u>. The parties to each assignment shall execute and deliver to the Administrative Agent an Assignment and Assumption, together with a processing and recordation fee in the amount of \$3,500; <u>provided</u>, <u>however</u>, that the Administrative Agent may, in its sole discretion, elect to waive such processing and recordation fee in the case of any assignment. The assignee, if it is not a Lender, shall deliver to the Administrative Agent an Administrative Questionnaire.
- (v) No Assignment to Certain Persons. No such assignment shall be made to (A) the Borrower or any of the Borrower's Affiliates or Subsidiaries or (B) to any Defaulting Lender or any of its Subsidiaries, or any Person who, upon becoming a Lender hereunder, would constitute any of the foregoing Persons described in this <u>clause (B)</u>.
 - (vi) No Assignment to Natural Persons. No such assignment shall be made to a natural person.
- (vii) <u>Certain Additional Payments</u>. In connection with any assignment of rights and obligations of any Defaulting Lender hereunder, no such assignment shall be effective unless and until, in addition to the other conditions thereto set forth herein, the parties to the assignment shall make such additional payments to the Administrative Agent in an aggregate amount sufficient, upon distribution thereof as appropriate (which may be outright payment, purchases by the

assignee of participations or subparticipations, or other compensating actions, including funding, with the consent of the Borrower and the Administrative Agent, the applicable pro rata share of Loans previously requested but not funded by the Defaulting Lender, to each of which the applicable assignee and assignor hereby irrevocably consent), to (A) pay and satisfy in full all payment liabilities then owed by such Defaulting Lender to the Administrative Agent, the L/C Issuers, the Swing Line Lender and each other Lender hereunder (and interest accrued thereon), and (B) acquire (and fund as appropriate) its full pro rata share of all Loans and participations in Letters of Credit and Swing Line Loans in accordance with its Pro Rata Share. Notwithstanding the foregoing, in the event that any assignment of rights and obligations of any Defaulting Lender hereunder shall become effective under applicable law without compliance with the provisions of this paragraph, then the assignee of such interest shall be deemed to be a Defaulting Lender for all purposes of this Agreement until such compliance occurs.

Subject to acceptance and recording thereof by the Administrative Agent pursuant to <u>subsection (c)</u> of this Section, from and after the effective date specified in each Assignment and Assumption, the assignee thereunder shall be a party to this Agreement and, to the extent of the interest assigned by such Assignment and Assumption, have the rights and obligations of a Lender under this Agreement, and the assigning Lender thereunder shall, to the extent of the interest assigned by such Assignment and Assumption, be released from its obligations under this Agreement (and, in the case of an Assignment and Assumption covering all of the assigning Lender's rights and obligations under this Agreement, such Lender shall cease to be a party hereto) but shall continue to be entitled to the benefits of <u>Sections 3.01</u>, 3.04, 3.05, and 11.04 with respect to facts and circumstances occurring prior to the effective date of such assignment; <u>provided</u>, that except to the extent otherwise expressly agreed by the affected parties, no assignment by a Defaulting Lender will constitute a waiver or release of any claim of any party hereunder arising from that Lender's having been a Defaulting Lender. Upon request, the Borrower (at its expense) shall execute and deliver a Note to the assignee Lender. Any assignment or transfer by a Lender of rights or obligations under this Agreement that does not comply with this subsection shall be treated for purposes of this Agreement as a sale by such Lender of a participation in such rights and obligations in accordance with <u>subsection (d)</u> of this Section.

- (c) <u>Register</u>. The Administrative Agent, acting solely for this purpose as a non-fiduciary agent of the Borrower, shall maintain at one of its offices in Charlotte, North Carolina, a copy of each Assignment and Assumption delivered to it and a register for the recordation of the names and addresses of the Lenders, and the Commitments of, and principal amounts (and stated interest) of the Loans and L/C Obligations owing to, each Lender pursuant to the terms hereof from time to time (the "<u>Register</u>"). The entries in the Register shall be conclusive absent manifest error, and the Borrower, the Administrative Agent and the Lenders shall treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement, notwithstanding notice to the contrary. The Register shall be available for inspection by the Borrower and any Lender, at any reasonable time and from time to time upon reasonable prior notice.
- (d) <u>Participations</u>. Any Lender may at any time, without the consent of, or notice to, the Borrower, the Administrative Agent, the L/C Issuers or the Swing Line Lender, sell participations to any Person (other than a natural person or the Borrower or any of the Borrower's Affiliates or Subsidiaries) (each, a "<u>Participant</u>") in all or a portion of such Lender's rights and/or obligations under this Agreement (including all or a portion of its Commitment and/or the Loans (including such Lender's participations in L/C Obligations) owing to it); <u>provided</u> that (i) such Lender's obligations under this Agreement shall remain unchanged, (ii) such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations and (iii) the Borrower, the Administrative Agent, the Lenders and the L/C Issuers shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement.

Any agreement or instrument pursuant to which a Lender sells such a participation shall provide that such Lender shall retain the sole right to enforce this Agreement and to approve any amendment, modification or waiver of any provision of this Agreement; provided that such agreement or instrument may provide that such Lender will not, without the consent of the Participant, agree to any amendment, waiver or other modification described in the first proviso to Section 11.01 that affects such Participant. Subject to subsection (e) of this Section, the Borrower agrees that each Participant shall be entitled to the benefits of Sections 3.01 (subject to the requirements and limitations therein, including the requirements under Section 3.01(g)), 3.04 and 3.05 to the same extent as if it were a Lender and had acquired its interest by assignment pursuant to subsection (b) of this Section. To the extent permitted by law, each Participant also shall be entitled to the benefits of Section 11.08 as though it were a Lender, provided such Participant agrees to be subject to Section 2.13 as though it were a Lender.

- (e) <u>Limitations upon Participant Rights</u>. A Participant shall not be entitled to receive any greater payment under <u>Section 3.01</u> or <u>3.04</u> than the applicable Lender would have been entitled to receive with respect to the participation sold to such Participant, unless the sale of the participation to such Participant is made with the Borrower's prior written consent. A Participant that would be a Foreign Lender if it were a Lender shall not be entitled to the benefits of <u>Section 3.01</u> unless the Borrower is notified of the participation sold to such Participant and such Participant agrees, for the benefit of the Borrower, to comply with <u>Section 3.01(g)</u> as though it were a Lender.
- (f) Participant Register. Each Lender that sells a participation shall, acting solely for this purpose as a non-fiduciary agent of the Borrower, maintain a register on which it enters the name and address of each Participant and the principal amounts (and stated interest) of each Participant's interest in the Loans or other obligations under the Loan Documents (the "Participant Register"); provided that no Lender shall have any obligation to disclose all or any portion of the Participant Register (including the identity of any Participant or any information relating to a Participant's interest in any commitments, loans, letters of credit or its other obligations under any Loan Document) to any Person except to the extent that such disclosure is necessary to establish that such commitment, loan, letter of credit or other obligation is in registered form under Section 5f.103-1(c) of the United States Treasury Regulations. The entries in the Participant Register shall be conclusive absent manifest error, and such Lender shall treat each Person whose name is recorded in the Participant Register as the owner of such participation for all purposes of this Agreement notwithstanding any notice to the contrary. For the avoidance of doubt, the Administrative Agent (in its capacity as Administrative Agent) shall have no responsibility for maintaining a Participant Register.
- (g) <u>Certain Pledges</u>. Any Lender may at any time pledge or assign a security interest in all or any portion of its rights under this Agreement (including under its Note, if any) to secure obligations of such Lender, including any pledge or assignment to secure obligations to a Federal Reserve Bank; <u>provided</u> that no such pledge or assignment shall release such Lender from any of its obligations hereunder or substitute any such pledgee or assignee for such Lender as a party hereto.
 - (h) As used herein, the following terms have the following meanings:

"<u>Eligible Assignee</u>" means (a) a Lender; (b) an Affiliate of a Lender; (c) an Approved Fund; and (d) any other Person (other than a natural person) approved by (i) the Administrative Agent and each L/C Issuer, and (ii) unless an Event of Default has occurred and is continuing, the Borrower (each such approval not to be unreasonably withheld or delayed); <u>provided</u> that notwithstanding the foregoing, "Eligible Assignee" shall not include the Borrower or any of the Borrower's Affiliates or Subsidiaries.

"Fund" means any Person (other than a natural person) that is (or will be) engaged in making, purchasing, holding or otherwise investing in commercial loans and similar extensions of credit in the ordinary course of its business.

"Approved Fund" means any Fund that is administered or managed by (a) a Lender, (b) an Affiliate of a Lender or (c) an entity or an Affiliate of an entity that administers or manages a Lender.

(i) Resignation as L/C Issuer and Swing Line Lender after Assignment. Notwithstanding anything to the contrary contained herein, if at any time Wells Fargo Bank assigns all of its Commitment and Loans pursuant to subsection (b) above, Wells Fargo Bank may, upon 30 days' notice to the Borrower and the Lenders, resign as L/C Issuer and Swing Line Lender. In the event of any such resignation as L/C Issuer and Swing Line Lender, the Borrower shall be entitled to appoint from among the Lenders a successor L/C Issuer and Swing Line Lender hereunder; provided, however, that no failure by the Borrower to appoint any such successor shall affect the resignation of Wells Fargo Bank as L/C Issuer and Swing Line Lender. If Wells Fargo Bank resigns as L/C Issuer and Swing Line Lender, it shall retain all the rights, powers, privileges and duties of a L/C Issuer and Swing Line Lender hereunder with respect to all Letters of Credit and Swing Line Loans outstanding as of the effective date of its resignation as L/C Issuer and Swing Line Lender (including the right to require the Lenders to make Base Rate Loans or fund risk participations in Unreimbursed Amounts pursuant to Section 2.03(c) and Section 2.04). Upon the appointment of a successor L/C Issuer and Swing Line Lender, (a) such successor shall succeed to and become vested with all of the rights, powers, privileges and obligations of the retiring L/C Issuer and Swing Line Lender and (b) the successor L/C Issuer shall issue letters of credit in substitution for the Letters of Credit, if any, outstanding at the time of such succession or make other arrangements satisfactory to Wells Fargo Bank to effectively assume the obligations of Wells Fargo Bank with respect to such Letters of Credit.

11.07 Confidentiality. Each of the Administrative Agent and the Lenders agrees to maintain the confidentiality of the Information (as defined below), except that Information may be disclosed (a) to its Affiliates, and to its and its Affiliates' respective partners, directors, officers, employees, agents, advisors and representatives (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of such Information and instructed to keep such Information confidential), (b) to the extent requested by any regulatory authority purporting to have jurisdiction over it (including any self-regulatory authority, such as the National Association of Insurance Commissioners), (c) to the extent required by applicable laws or regulations or by any subpoena or similar legal process, (d) to any other party hereto, (e) in connection with the exercise of any remedies hereunder, under any other Loan Document or under any Secured Hedge Agreement or Secured Cash Management Agreement, or any action or proceeding relating to this Agreement, any other Loan Document or any Secured Hedge Agreement or Secured Cash Management Agreement, or the enforcement of rights hereunder or thereunder, (f) subject to an agreement containing provisions substantially the same as those of this Section, to (i) any assignee of or Participant in, or any prospective assignee of or Participant in, any of its rights or obligations under this Agreement or (ii) any actual or prospective counterparty (or its Related Parties) to any swap, derivative or other agreement under which payments are to be made by reference to the Borrower or any Subsidiary and its or their obligations, this Agreement, or payments hereunder, (g) with the consent of the Borrower, (h) on a confidential basis to (x) any rating agency in connection with rating the Parent Guarantor or its Subsidiaries or the credit facility provided hereunder, or (y) the CUSIP Service Bureau or any similar agency in connection with the issuance and monitoring of CUSIP numbers with respect to the credit facility provided hereunder, or (i) to the extent such Information (x) becomes publicly available other than as a result of a breach of this Section or (y) becomes available to the Administrative Agent or any Lender on a nonconfidential basis from a source other than the Borrower.

For purposes of this Section, "<u>Information</u>" means all information received from the Borrower or any of its Subsidiaries relating to the Borrower or any Subsidiary or any of their respective businesses, other than any such information that is available to the Administrative Agent or any Lender on a nonconfidential basis prior to disclosure by the Borrower or any Subsidiary, <u>provided</u> that, in the case of information received from the Borrower or any Subsidiary after the date hereof, such information is clearly identified at the time of delivery as confidential. Any Person required to maintain the confidentiality of Information as provided in this Section shall be considered to have complied with its obligation to do so if such Person has exercised the same degree of care to maintain the confidentiality of such Information as such Person would accord to its own confidential information.

Each of the Administrative Agent and the Lenders acknowledges that (a) the Information may include material non-public information concerning the Borrower or a Subsidiary, as the case may be, (b) it has developed compliance procedures regarding the use of material non-public information and (c) it will handle such material non-public information in accordance with applicable Law, including United States Federal and state securities Laws.

11.08 **Set-off.** In addition to any rights and remedies of the Lenders provided by law, upon the occurrence and during the continuance of any Event of Default, each Lender is authorized at any time and from time to time, without prior notice to any Loan Party or any Subsidiary, any such notice being waived by the Borrower and the other Loan Parties (each on its own behalf and on behalf of each Subsidiary) to the fullest extent permitted by law, to set off and apply any and all deposits (general or special, time or demand, provisional or final) at any time held by, and other obligations at any time owing by, such Lender to or for the credit or the account of the Borrower or other Loan Party against any and all Obligations owing to such Lender hereunder or under any other Loan Document, now or hereafter existing, irrespective of whether or not the Administrative Agent or such Lender shall have made demand under this Agreement or any other Loan Document and although such Obligations may be contingent or unmatured or denominated in a currency different from that of the applicable deposit or indebtedness or are owed to a branch, office or Affiliate of a Lender different from the branch, office or Affiliate holding such deposit or obligated on such obligation. Each Lender agrees promptly to notify the Borrower and the Administrative Agent after any such set-off and application made by such Lender; provided, however, that the failure to give such notice shall not affect the validity of such set-off and application.

11.09 **Interest Rate Limitation**. Notwithstanding anything to the contrary contained in any Loan Document, the interest paid or agreed to be paid under the Loan Documents shall not exceed the maximum rate of non-usurious interest permitted to a Lender by applicable Law (the "<u>Maximum Rate</u>"). If the Administrative Agent or any Lender shall receive interest in an amount that exceeds the Maximum Rate, the excess interest shall be applied to the principal of the Loans or, if it exceeds such unpaid principal, refunded to the Borrower. In determining whether the interest contracted for, charged, or received by the Administrative Agent or a Lender exceeds the Maximum Rate, such Person may, to the extent permitted by applicable Law, (a) characterize any payment that is not principal as an expense, fee, or premium rather than interest, (b) exclude voluntary prepayments and the effects thereof, and (c) amortize, prorate, allocate, and spread in equal or unequal parts the total amount of interest throughout the contemplated term of the Obligations hereunder.

- 11.10 **Counterparts.** This Agreement may be executed in counterparts (and by different parties hereto in different counterparts), each of which shall constitute an original, but all of which when taken together shall constitute a single contract. Delivery of an executed counterpart of a signature page of this Agreement or any other Loan Document or other document or writing delivered in connection therewith by facsimile or in electronic (i.e., "pdf" or "tif") format shall be effective as delivery of a manually executed counterpart of this Agreement.
- 11.11 **Integration.** This Agreement, together with the other Loan Documents, comprises the complete and integrated agreement of the parties on the subject matter hereof and thereof and supersedes all prior agreements, written or oral, on such subject matter. In the event of any conflict between the provisions of this Agreement and those of any other Loan Document, the provisions of this Agreement shall control; <u>provided</u> that the inclusion of supplemental rights or remedies in favor of the Administrative Agent or the Lenders in any other Loan Document shall not be deemed a conflict with this Agreement. Each Loan Document was drafted with the joint participation of the respective parties thereto and shall be construed neither against nor in favor of any party, but rather in accordance with the fair meaning thereof.
- 11.12 **Survival of Representations and Warranties.** All representations and warranties made hereunder and in any other Loan Document or other document delivered pursuant hereto or thereto or in connection herewith or therewith shall survive the execution and delivery hereof and thereof. Such representations and warranties have been or will be relied upon by the Administrative Agent and each Lender, regardless of any investigation made by the Administrative Agent or any Lender or on their behalf and notwithstanding that the Administrative Agent or any Lender may have had notice or knowledge of any Default at the time of any Credit Extension, and shall continue in full force and effect as long as any Loan or any other Obligation hereunder shall remain unpaid or unsatisfied or any Letter of Credit shall remain outstanding.
- 11.13 **Severability.** If any provision of this Agreement or the other Loan Documents is held to be illegal, invalid or unenforceable, (a) the legality, validity and enforceability of the remaining provisions of this Agreement and the other Loan Documents shall not be affected or impaired thereby and (b) the parties shall endeavor in good faith negotiations to replace the illegal, invalid or unenforceable provisions with valid provisions the economic effect of which comes as close as possible to that of the illegal, invalid or unenforceable provisions. The invalidity of a provision in a particular jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.
- 11.14 **Replacement of Lenders.** Under any circumstances set forth herein providing that the Borrower shall have the right to replace a Lender as a party to this Agreement, the Borrower may, upon notice to such Lender and the Administrative Agent, replace such Lender by requiring such Lender to assign its Commitment (with the assignment fee to be paid by the Borrower in such instance) pursuant to (and subject to the consents required by) Section 11.06(b) to one or more Eligible Assignees; provided that such assignment does not conflict with applicable Laws and such Lender shall have received payment of an amount equal to the outstanding principal of its Loans and L/C Advances, accrued interest thereon, accrued fees and all other amounts payable to it hereunder and under the other Loan Documents (including any amounts under

Section 3.05) from the assignee (to the extent of such outstanding principal and accrued interest and fees) or the Borrower (in the case of all other amounts); and provided, further, that if the Borrower elects to exercise such right with respect to any Lender pursuant to Section 3.06(b), it shall be obligated to replace all Lenders that have made similar requests for compensation pursuant to Section 3.01 or 3.04. Upon the making of any such assignment, the Borrower shall (x) pay in full any amounts payable pursuant to Section 3.05 and (y) provide appropriate assurances and indemnities (which may include letters of credit) to the L/C Issuers as each may reasonably require with respect to any continuing obligation to fund participation interests in any L/C Obligations then outstanding.

11.15 Governing Law.

- (a) <u>GOVERNING LAW</u>. THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS AND ANY CLAIMS, CONTROVERSY, DISPUTE OR CAUSE OF ACTION (WHETHER IN CONTRACT OR TORT OR OTHERWISE) BASED UPON, ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT (EXCEPT, AS TO ANY OTHER LOAN DOCUMENT, AS EXPRESSLY SET FORTH THEREIN) AND THE TRANSACTIONS CONTEMPLATED HEREBY AND THEREBY SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAW OF THE STATE OF NEW YORK, WITHOUT REGARD TO CONFLICTS OF LAWS PRINCIPLES.
- (b) JURISDICTION. THE BORROWER AND EACH OTHER LOAN PARTY IRREVOCABLY AND UNCONDITIONALLY AGREES THAT IT WILL NOT COMMENCE ANY ACTION, LITIGATION OR PROCEEDING OF ANY KIND OR DESCRIPTION, WHETHER IN LAW OR EQUITY, WHETHER IN CONTRACT OR IN TORT OR OTHERWISE, AGAINST THE ADMINISTRATIVE AGENT, ANY LENDER, ANY L/C ISSUER, OR ANY RELATED PARTY OF THE FOREGOING IN ANY WAY RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT OR THE TRANSACTIONS RELATING HERETO OR THERETO, IN ANY FORUM OTHER THAN THE COURTS OF THE STATE OF NEW YORK SITTING IN THE BOROUGH OF MANHATTAN IN NEW YORK CITY, AND OF THE UNITED STATES DISTRICT COURT OF THE SOUTHERN DISTRICT OF NEW YORK, AND ANY APPELLATE COURT FROM ANY THEREOF, AND EACH OF THE PARTIES HERETO IRREVOCABLY AND UNCONDITIONALLY SUBMITS TO THE JURISDICTION OF SUCH COURTS AND AGREES THAT ALL CLAIMS IN RESPECT OF ANY SUCH ACTION, LITIGATION OR PROCEEDING MAY BE HEARD AND DETERMINED IN SUCH NEW YORK STATE COURT OR, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, IN SUCH FEDERAL COURT. EACH LOAN PARTY AND EACH OF THE PARTIES HERETO AGREES THAT A FINAL JUDGMENT IN ANY SUCH ACTION, LITIGATION OR PROCEEDING SHALL BE CONCLUSIVE AND MAY BE ENFORCED IN OTHER JURISDICTIONS BY SUIT ON THE JUDGMENT OR IN ANY OTHER MANNER PROVIDED BY LAW. NOTHING IN THIS AGREEMENT OR IN ANY OTHER LOAN DOCUMENT SHALL AFFECT ANY RIGHT THAT THE ADMINISTRATIVE AGENT, ANY LENDER OR ANY L/C ISSUER MAY OTHERWISE HAVE TO BRING ANY ACTION OR PROCEEDING RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT AGAINST THE BORROWER OR ANY OTHER LOAN PARTY OR ITS PROPERTIES IN THE COURTS OF ANY JURISDICTION.
- (c) <u>WAIVER OF VENUE</u>. THE BORROWER AND EACH OTHER LOAN PARTY IRREVOCABLY AND UNCONDITIONALLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY OBJECTION THAT IT MAY NOW OR HEREAFTER HAVE TO THE LAYING OF VENUE OF ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT IN ANY COURT REFERRED TO IN PARAGRAPH (B) OF THIS SECTION. EACH LOAN PARTY AND EACH OF THE PARTIES HERETO HEREBY IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, THE DEFENSE OF AN INCONVENIENT FORUM TO THE MAINTENANCE OF SUCH ACTION OR PROCEEDING IN ANY SUCH COURT.

- (d) <u>SERVICE OF PROCESS</u>. EACH LOAN PARTY AND EACH PARTY HERETO IRREVOCABLY CONSENTS TO SERVICE OF PROCESS IN THE MANNER PROVIDED FOR NOTICES IN <u>SECTION 11.02</u>. NOTHING IN THIS AGREEMENT WILL AFFECT THE RIGHT OF ANY PARTY HERETO TO SERVE PROCESS IN ANY OTHER MANNER PERMITTED BY APPLICABLE LAW.
- 11.16 Waiver of Right to Trial by Jury. EACH LOAN PARTY AND EACH PARTY HERETO HEREBY IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN ANY LEGAL PROCEEDING DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY (WHETHER BASED ON CONTRACT, TORT OR ANY OTHER THEORY). EACH LOAN PARTY AND EACH PARTY HERETO (A) CERTIFIES THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PERSON HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PERSON WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER AND (B) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HERETO HAVE BEEN INDUCED TO ENTER INTO THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION.
- 11.17 **No Advisory or Fiduciary Responsibility.** In connection with all aspects of each transaction contemplated hereby (including in connection with any amendment, waiver or other modification hereof or of any other Loan Document), each Loan Party acknowledges and agrees, and acknowledges its Affiliates' understanding, that: (i) (A) the arranging and other services regarding this Agreement provided by the Administrative Agent and the Arrangers are arm's-length commercial transactions between the Borrower and its Affiliates, on the one hand, and the Administrative Agent and the Arrangers, on the other hand, (B) the Borrower has consulted its own legal, accounting, regulatory and tax advisors to the extent it has deemed appropriate, and (C) the Borrower is capable of evaluating, and understands and accepts, the terms, risks and conditions of the transactions contemplated hereby and by the other Loan Documents; (ii) (A) the Administrative Agent, each Arranger and each Lender is and has been acting solely as a principal and, except as expressly agreed in writing by the relevant parties, has not been, is not, and will not be acting as an advisor, agent or fiduciary for the Borrower or any of its Affiliates, or any other Person and (B) neither the Administrative Agent, nor any Arranger or any Lender has any obligation to the Borrower or any of its Affiliates with respect to the transactions contemplated hereby except those obligations expressly set forth herein and in the other Loan Documents; and (iii) the Administrative Agent, the Arrangers and the Lenders and their respective Affiliates may be engaged in a broad range of transactions that involve interests that differ from those of the Borrower or its Affiliates, and neither the Administrative Agent nor any Arranger nor any Lender has any obligation to disclose any of such interests to the Borrower or its Affiliates. To the fullest extent permitted by law, each Loan Party hereby waives and releases any claims that it may have against the Administrative Agent, the

11.18 **Electronic Execution of Assignments and Certain Other Documents.** The words "execution," "signed," "signature," and words of like import in any Assignment and Assumption or in any amendment or other modification hereof (including waivers and consents) shall be deemed to include electronic signatures or the keeping of records in electronic form, each of which shall be of the same legal effect, validity or enforceability as a manually executed signature or the use of a paper-based recordkeeping system, as the case may be, to the extent and as provided for in any applicable law, including the Federal Electronic Signatures in Global and National Commerce Act, the New York State Electronic Signatures and Records Act, or any other similar state laws based on the Uniform Electronic Transactions Act.

11.19 **USA PATRIOT Act Notice.** Each Lender that is subject to the Act (as hereinafter defined) and the Administrative Agent (for itself and not on behalf of any Lender) hereby notifies the Borrower and the other Loan Parties that pursuant to the requirements of the USA Patriot Act (Title III of Pub. L. 107,56 (signed into law October 26, 2001)) (the "Act"), it is required to obtain, verify and record information that identifies the Loan Parties, which information includes the name and address of each Loan Party and other information that will allow such Lender or the Administrative Agent, as applicable, to identify the Loan Parties in accordance with the Act. The Borrower and the Parent Guarantor shall, and shall cause their Subsidiaries to, promptly following a request by the Administrative Agent or any Lender, provide such documentation and other information that the Administrative Agent or such Lender reasonably requests in order to comply with its ongoing obligations under applicable "know your customer" and anti-money laundering rules and regulations, including the Act.

11.20 **ENTIRE AGREEMENT.** THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS REPRESENT THE FINAL AGREEMENT AMONG THE PARTIES AND MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES. THERE ARE NO UNWRITTEN ORAL AGREEMENTS AMONG THE PARTIES.

[SIGNATURE PAGES FOLLOW]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the date first above written.

QEP MIDSTREAM PARTNERS OPERATING, LLC, as Borrower

By: QEP MIDSTREAM PARTNERS, LP, its sole member

By: QEP MIDSTREAM PARTNERS GP, LLC, its general partner

By: /s/ Richard J. Doleshek

Name: Richard J. Doleshek

Title: Executive Vice President and Chief

Financial Officer

QEP MIDSTREAM PARTNERS, LP, as Parent Guarantor

By: QEP MIDSTREAM PARTNERS GP, LLC, its general partner

By: /s/ Richard J. Doleshek

Name: Richard J. Doleshek

Title: Executive Vice President and Chief

Financial Officer

WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent, L/C Issuer, Swing Line Lender, and a Lender

By: /s/ Leanne S. Phillips

Name: Leanne S. Phillips

Title: Director

CITIBANK, N.A., as a Lender and a L/C Issuer

By: /s/ Andrew Sidford

Name: Andrew Sidford Title: Vice President

By: /s/ Robert Mendoza

Name: Robert Mendoza Title: Vice President

By: /s/ Justin M. Alexander

Name: Justin M. Alexander Title: Senior Vice President

DEUTSCHE BANK AG NEW YORK BRANCH, as a Lender and a L/C Issuer

By: /s/ Marcus M. Tarkington
Name: Marcus M. Tarkington
Title: Director

By: /s/ Dusan Lazarov

Name: Dusan Lazarov

Title: Director

By: /s/ Kelly Chin

Name: Kelly Chin

Title: Authorized Signatory

By: /s/ Kevin Utsey

Name: Kevin Utsey Title: Director

CANADIAN IMPERIAL BANK OF COMMERCE, NEW YORK AGENCY, as a Lender

By:	/s/ Trudy Nelson
_	Name: Trudy Nelson
	Title: Managing Director
By:	/s/ Jonathan Kim
_	Name: Jonathan Kim

Signature Page to the QEP Midstream Partners Operating, LLC Credit Agreement

Title: Authorized Signatory

CAPITAL ONE, NATIONAL ASSOCIATION,

as a Lender

By: /s/ Kristin Oswald

Name: Kristin Oswald Title: Vice President

By: /s/ Dorothy Marchand

Name: Dorothy Marchand Title: Executive Vice President

CREDIT AGRICOLE CORPORATE AND INVESTMENT BANK, as a Lender

By:	/s/ Sharada Manne		
•	Name: Sharada Manne		
	Title: Managing Director		
By:	/s/ Michael Willis		
•	Name: Michael Willis		
	Title: Managing Director		

DNB BANK ASA, GRAND CAYMAN BRANCH, as a Lender

By:	/s/ Kristie Li		
_	Name: Kristie Li		
	Title: First Vice President		
By:	/s/ Philip F. Kurpiewski		
_	Name: Philip F. Kurpiewski		

Signature Page to the QEP Midstream Partners Operating, LLC Credit Agreement

Title: Senior Vice President

By: /s/ Helen Wiggins

Name: Helen Wiggins Title: Assistant Vice President

By: /s/ Brett R. Schweikle

Name: Brett R. Schweikle Title: Senior Vice President

By: /s/ Shannon Juhan

Name: Shannon Juhan Title: Vice President

By: /s/ Sherwin Brandford

Name: Sherwin Brandford Title: Vice President

By: /s/ Masood Fikree

Name: Masood Fikree Title: Authorized Signatory

By: /s/ Kevin Donaldson

Name: Kevin Donaldson Title: Senior Vice Predient

BRANCH BANKING AND TRUST COMPANY, as a Lender

By: /s/ James Giordano

Name: James Giordano Title: Vice President

By: /s/ V. Mark Fuqua

Name: V. Mark Fuqua Title: Senior Vice President

EXPORT DEVELOPMENT CANADA, as a Lender

By: /s/ Anna Kane
Name: Anna Kane
Title: Senior Associate

By: /s/ Joanne Tognarelli
Name: Joanne Tognarelli

Signature Page to the QEP Midstream Partners Operating, LLC Credit Agreement

Title: Senior Financing Manager

By: /s/ Mark Walton

Name: Mark Walton

Title: Authorized Signatory

SUMITOMO MITSUI BANKING CORPORATION, NY BRANCH, as a Lender

By: /s/ Shuji Yabe

Name: Shuji Yabe Title: Managing Director

COMMITMENTS AND PRO RATA SHARES

QEP Midstream Partners Operating, LLC Credit Agreement

Lender	Commitment	Pro Rata Share
Wells Fargo Bank, National Association	\$ 28,500,000	5.700%
Citibank, N.A.	\$ 27,500,000	5.500%
JPMorgan Chase Bank, N.A.	\$ 27,500,000	5.500%
U.S. Bank National Association	\$ 27,500,000	5.500%
Deutsche Bank AG New York Branch	\$ 26,000,000	5.200%
Morgan Stanley Bank, N.A.	\$ 26,000,000	5.200%
BMO Harris Bank N.A.	\$ 23,000,000	4.600%
Canadian Imperial Bank of Commerce, New York Agency	\$ 23,000,000	4.600%
Capital One, National Association	\$ 23,000,000	4.600%
Compass Bank	\$ 23,000,000	4.600%
Credit Agricole Corporate and Investment Bank	\$ 23,000,000	4.600%
DNB Bank ASA, Grand Cayman Branch	\$ 23,000,000	4.600%
Fifth Third Bank	\$ 23,000,000	4.600%
PNC Bank, National Association	\$ 23,000,000	4.600%
SunTrust Bank	\$ 23,000,000	4.600%
The Bank of Tokyo-Mitsubishi UFJ, Ltd.	\$ 23,000,000	4.600%
Toronto Dominion (New York) LLC	\$ 23,000,000	4.600%
Amegy Bank National Association	\$ 14,000,000	2.800%
Branch Banking and Trust Company	\$ 14,000,000	2.800%
Comerica Bank	\$ 14,000,000	2.800%
Export Development Canada	\$ 14,000,000	2.800%
Goldman Sachs Bank USA	\$ 14,000,000	2.800%
Sumitomo Mitsui Banking Corporation, NY Branch	\$ 14,000,000	2.800%
Total	\$ 500,000,000.00	100.000000000%

Schedule 2.01

SWING LINE RATE CALCULATION

The rate of interest for a Swing Line Loan shall be (a) the "ASK" rate for Federal Funds appearing on Page 5 of the Dow Jones Market Service (or on any successor or substitute page of such Service, or any successor to or substitute for such Service, providing rate quotations comparable to those currently provided on such page of such Service, as determined by the Administrative Agent from time to time for purposes of providing quotations of the offer rates applicable to Federal Funds for a term of one Business Day) at the time reviewed by the Administrative Agent plus (b) the Applicable Rate for Eurodollar Rate Loans. In the event that part (a) of such rate is not available at such time for any reason, then part (a) of such rate will be the rate agreed to between the Administrative Agent and the Borrower. The Borrower understands and agrees that the rate quoted from Page 5 of the Dow Jones Market Service is a real-time rate that changes from time to time. The rate quoted by the Administrative Agent and used for the purpose of setting the interest rate for a Swing Line Loan will be the rate on the screen of the Administrative Agent at the time of setting the rate and will not be an average or composite of rates for that day.

Schedule 2.04

LABOR DISPUTES AND ACTS OF GOD

None.

Schedule 5.09

SUBSIDIARIES, JOINT VENTURES AND OTHER EQUITY INVESTMENTS

Subsidiaries.

- QEP Midstream Partners Operating, LLC (100% owned by QEP Midstream Partners, LP)
- QEPM Gathering I, LLC (100% owned by QEP Midstream Partners Operating, LLC)
- Rendezvous Pipeline Company, LLC (100% owned by QEPM Gathering I, LLC) Rendezvous Gas Services, L.L.C. (78% owned by QEPM Gathering I, LLC)

Other Equity Interests.

Three Rivers Gathering, LLC (50% owned by QEP Midstream Partners Operating, LLC) Schedule 5.12

INSURANCE

See attached.

Schedule 5.19



INSURANCE SUMMARY 2013

COVERAGE	LIMITS	DEDUCTIBLES/RETENTIONS
Automobile Liability National Union Fire Insurance Company of Pittsburgh, PA Best Rating: A XV Policy No. CA1574461 Policy Term: October 1, 2012 – 2013	\$1,000,000 Liability – Any Auto (Symbol 1) \$ 3,000 Utah – Personal Injury Protection (Medical Expense) \$ 1,500 Utah - Funeral Expenses \$ 3,000 Utah - Survivor Loss	\$1,000 Collision/Comprehensive Deductible for Hired Autos Only (Symbol 8)
Workers Compensation/Employers Liability New Hampshire Insurance Company Best Rating: A XV Policy No. WC025889584 Policy Term: October 1, 2012 - 2013	Workers Compensation Statutory Employers Liability \$1,000,000 Bodily Injury by Accident; Each Accident \$1,000,000 Bodily Injury by Disease; Policy Limit \$1,000,000 Bodily Injury by Disease; Each Employee	\$100,000 Each Accident or Each Person for Disease \$100,000 Each Accident or Each Person for Disease \$100,000 Each Accident or Each Person for Disease
Commercial General Liability National Union Fire Insurance Company of Pittsburgh, PA Best Rating: A XV Policy No. GL1949962 Policy Term: October 1, 2012 - 2013	\$ 1,000,000 Each Occurrence Limit \$ 1,000,000 Damage to Premises Rented to You Limit; Any One Premise \$ 5,000 Medical Expense Limit; Any One Person \$ 1,000,000 Personal & Advertising Injury Limit; Any One Person or Organization \$ 10,000,000 General Aggregate Limit Products/Completed Operations Aggregate Limit	Nil \$1,000 Employee Benefits Liability



Schedule 5.19



COVERAGE	LIMITS	DEDUCTIBLES/RETENTIONS	
\$25MM Umbrella Liability American Guarantee & Liability Insurance Company Best Rating: A+ XV Policy No. UMB6692611 02 Policy Term: October 1, 2012 - 2013	\$25,000,000 Occurrence \$25,000,000 Other Aggregate \$25,000,000 Products/Completed Operations Aggregate \$250,000 Casualty Business Crisis Aggregate Limit	Commercial General Liability \$ 1,000,000 Each Occurrence \$ 2,000,000 Forducts-Completed Operations Aggregate \$ 2,000,000 General Aggregate \$ 1,000,000 Personal Injury Advertising Injury \$ 1,000,000 Employee Benefits Liability – Each Claim \$ 1,000,000 Employee Benefits Liability – Aggregate Commercial Auto Liability \$ 1,000,000 Bodily Injury & Property Damage Combined Single Limit Employers Liability \$ 1,000,000 Bodily Injury By Accident; Each Accident \$ 1,000,000 Bodily Injury by Disease; Policy Limit \$ 1,000,000 Bodily Injury by Disease; Each Employee \$ 10,000 Retained Limit; Occurrence	
1st Layer Excess Liability - \$25MM xs \$25MM Alterra America Insurance Company Best Rating: A XV Policy No. MAX6XL0000395 Policy Term: October 1, 2012 - 2013	\$25,000,000 Each Occurrence Limit \$25,000,000 Aggregate (where applicable)	Underlying Limits of Insurance: \$25,000,000 Each Occurrence \$25,000,000 Aggregate (where applicable)	





COVERAGE		LIMITS	DEDUCTIBLES/RETENTIONS	
2 nd Layer Excess Liability - \$25MM xs \$50MM Ironshore Specialty Insurance Company Best Rating: A- XIII Policy No. 000734502 Policy Term: October 1, 2012 – 2013	\$25,000,000 \$25,000,000	Loss (as in Controlling underlying Policy)	Limits of Underlying Policy \$50,000,000 Per Occurrence, Per Claim or Per Loss \$50,000,000 Aggregate (where applicable)	
3 rd Excess Liability - \$25MM xs \$75MM Starr Surplus Lines Insurance Company Best Rating: A XV Policy No. SLSLXNR03001912 Policy Term: October 1, 2012 - 2013	\$25,000,000 \$25,000,000		First Underlying Insurance Policy(ies) Lead Umbrella \$25,000,000	





COVERAGE	LIMITS	DEDUCTIBLES/RETENTIONS
4th Excess Liability - \$25MM xs \$100MM Arch Insurance Company Best Rating: A+ XV Policy No. EEO 0040824 02 Policy Term: October 1, 2012 - 2013	\$25,000,000 Policy General Aggregate \$25,000,000 Products/Completed Operations Aggregate \$25,000,000 Each Occurrence Limit	Controlling Underlying Insurance Lead Umbrella \$25,000,000
		Other Underlying Insurance 1 st Excess \$25,000,000





COVERAGE	LIMITS		DEDUCTIBLES/RETENTIONS	
6th Layer Excess Liability - \$25MM xs \$125MM Gotham Insurance Company through Prosight Specialty Insurance Best Rating: A IX Policy No. ML10169312 Policy Term: October 1, 2012 - 2013	\$25,000,000 \$25,000,000	Limit in all in respect of each occurrence Limit in the aggregate for each annual period	Schedule of Underlying Insurances Primary \$ 1,000,000	
6th Laver Excess Liability - \$25MM xs \$150MM Axis Surplus Insurance Company through Swett & Crawford of Georgia, Inc. Best Rating: A XV Policy No. EAU755348012012 Policy Term: October 1, 2012 - 2013	\$25,000,000 \$25,000,000 \$25,000,000	Each Occurrence Limit General Aggregate Limit Products/Completed Operations Aggregate Limit	Schedule of Underlying Insurances \$25,000,000 Umbrella Liability \$25,000,000 Excess Liability	
Th Laver Excess Liability - \$25MM xs \$175MM Lexington Insurance Company through Risk Specialists Companies Insurance Agency, Inc. Best Rating: A XV Policy No. 021375503 Policy Term: October 1, 2012 - 2013	\$25,000,000 \$25,000,000 \$25,000,000	Each Occurrence Limit General Aggregate Limit Products/Completed Operations Aggregate Limit	Schedule of Retained Amounts: Followed Policy: American Guarantee Primary Umbrella October 1, 2012 – 2013 TOTAL OCCURRENCE LIMITS OF INSURANCE OF ALL UNDERLYING POLICIES IN EXCESS OF WHICH THIS POLICY APPLIES: 3176,000,000. Sunset Clause Excluded.	





Property - Domestic	COVERAGE	LIMITS	DEDUCTIBLES/RETENTIONS
	Policy Term: April 1, 2013 - October 1, 2014	Sublimits: S 50,000,000	\$ 2,500,000 Per Occurrence, but \$ 1,000,000 Builders Risk/Course of Construction Projects valued under \$25MM





COVERAGE		LIMITS	DEDU	ICTIBLES/RETENTIONS
Property – London Policy Term: April 1, 2013 – October 1, 2014 Physical Damage#8I 33% 61.5% Policy No. Property E1307870 Terrorism E1307871 Terrorism E1307871 Companies Lloyd's Syndicate AFB 2623 Lloyd's Syndicate AFB 0623 Swiss Re International SE Lloyd's Syndicate AFB 0623 Swiss Re International SE Lloyd's Syndicate AFB 0623 Lloyd's Syndicate AFB 0623 Swiss Re International SE Lloyd's Syndicate AFB 0623 Lloyd's Syndicate AFB 0623 Lloyd's Syndicate AFB 0623 Swiss Re International SE Lloyd's Syndicate AFB 0623 Lloyd's Syndicate AFB 0623 Lloyd's Syndicate AFB 0623	\$300,000,000 Sublimits: \$150,000,000 \$150,000,000 \$ 5,000,000 \$ 1,000,000 \$ 1,000,000 \$ 5,000,000 \$ 1,000,000 \$ 1,000,000 \$ 1,000,000 \$ 1,000,000 \$ 10,000,000 \$ 10,000,000 \$ 10,000,000 \$ 10,000,000 \$ 10,000,000 \$ 5,000,000 \$ 5,000,000 \$ 5,000,000 \$ 5,000,000 \$ 5,000,000 \$ 5,000,000 \$ 5,000,000 \$ 5,000,000 \$ 5,000,000 \$ 5,000,000 \$ 5,000,000 \$ 5,000,000 \$ 5,000,000 \$ 5,000,000 \$ 5,000,000 \$ 5,000,000 \$ 5,000,000 \$ 1,000,000 \$ 1,000,000 \$ 1,000,000 \$ 1,000,000 \$ 1,000,000 \$ 1,000,000 \$ 1,000,000 \$ 1,000,000	Any One Accident or Occurrence Flood Zones A & V Aggregate Terrorism Transit Electrical Transmission & Distribution Lines Professional Fees Seepage and Pollution Aggregate Debris Removal Fire Fighting Materials Course of Construction Bylaws and Regulations Automatic Acquisitions Valuable Papers and Records Fine Arts CB1 - Direct Named Customers/Suppliers CB1 - Unnamed Customers/Suppliers Advance Loss of Profits Rental Value Leasehold Interest Off Premises Power Extra Expenses EXPEQUIPMENT and Media Miscellaneous Unnamed Locations Land Improvements Pipelines – Rivers Crossings, PD Only Pipelines – Underground PD Only	Property \$ 2,500,000 \$ 1,000,000 Transit \$ 250,000	Per Occurrence, but Per Occurrence in respect of all Builders Risk Projects (excluding Ocean Marine)





COVERAGE	LIMITS	DEDUCTIBLES/RETENTIONS
Excess Property - \$50MMx\$300MM Policy Term: April 1, 2013 - October 1, 2014 Company AEGIS (100%) P05265101P	\$ 50,000,000 Any One Accident or Occurrence Excess of \$300,000,000 Any One Accident or Occurrence Sublimits apply as per primary layer	Deductibles apply as per primary layer

This summary is intended for information purposes only. Please refer to policy for actual terms, conditions and exclusions.



Schedule 5.19

INVESTMENTS ON THE EXECUTION DATE AND THE CLOSING DATE

See Schedule 5.12.

Schedule 7.05

ADMINISTRATIVE AGENT'S OFFICE, CERTAIN ADDRESSES FOR NOTICES

QEP MIDSTREAM PARTNERS OPERATING, LLC:

QEP Midstream Partners Operating, LLC 1050 17th Street

Suite 500

Denver, CO 80265

Attention: Richard J. Doleshek, Chief Financial Officer

Email: Richard.doleshek@qepres.com

Fax: (303) 308-3639 Phone: (303) 640-4242

With a copy to: Rod H. Larsen Email: rod.larsen@qepres.com

Fax: (303) 308-3639 Phone: (303) 640-4261

ADMINISTRATIVE AGENT:

Administrative Agent's Office

(for payments and Requests for Borrowings): Wells Fargo Bank, National Association

1525 W WT Harris Blvd

1st Floor

Charlotte, North Carolina 28262-8522

Attention: Keisha Dockery Telephone: 704-590-2770 Facsimile: 704-590-2782

Electronic Mail: keisha.m.dockery@wellsfargo.com

Other Notices as Administrative Agent: Wells Fargo Bank, National Association

1525 W WT Harris Blvd

1st Floor

Charlotte, North Carolina 28262-8522

Attention: Keisha Dockery Telephone: 704-590-2770 Facsimile: 704-590-2782

Electronic Mail: keisha.m.dockery@wellsfargo.com

WELLS FARGO BANK AS L/C ISSUER:

Wells Fargo Bank, National Association

1525 W WT Harris Blvd

1st Floor

Charlotte, North Carolina 28262-8522

Attention: Keisha Dockery

Schedule 11.02

Telephone: 704-590-2770 Facsimile: 704-590-2782

Electronic Mail: keisha.m.dockery@wellsfargo.com

SWING LINE LENDER:

Wells Fargo Bank, National Association 1525 W WT Harris Blvd

1st Floor

Charlotte, North Carolina 28262-8522

Attention: Keisha Dockery Telephone: 704-590-2770 Facsimile: 704-590-2782

Electronic Mail: keisha.m.dockery@wellsfargo.com

Schedule 11.02

Date: _____,___

FORM OF LOAN NOTICE

To: Wells Fargo Bank, National Association, as Administrative Agent

La	idies and Gen	tlemen:	
LL	writing from LC, a Delawa	time to time, the " <u>Agreement</u> ;" the re limited liability company (the "	reement, dated as of August 14, 2013 (as amended, restated, extended, supplemented or otherwise modified e terms defined therein being used herein as therein defined), among QEP Midstream Partners Operating, Borrower "), QEP Midstream Partners, LP, a Delaware limited partnership, each Lender and L/C Issuer from National Association, as Administrative Agent, L/C Issuer and Swing Line Lender.
Th	ne undersigne	d hereby requests (select one):	
		A Borrowing of Loans ¬ A conversion or continuation of Loans	
1	On		
	(a Busi	ness Day).	
2	In the a	mount of	
3	Comprised of	 [Type of Loan requested	11
4	Loans: Period of the company of the	[month(s) /]].	vith the proviso to the first sentence of <u>Section 2.01</u> of the Agreement.
Q]	EP MIDSTR	REAM PARTNERS OPERATIN	G, LLC
Ву	y:	QEP MIDSTRE	EAM PARTNERS, LP, its sole member
Ву	y:	QEP MIDSTRE	EAM PARTNERS GP, LLC, its general partner
Ву	y:		
		Name:	
		Title:	
			Exhibit A-1

FORM OF SWING LINE LOAN NOTICE

		Date:,
To: W	Wells Fargo Bank, National Association, as Administrative Agent	
Ladies	es and Gentlemen:	
in writ	Reference is made to that certain Credit Agreement, dated as of August 14, 2013 (as amended, reiting from time to time, the "Agreement;" the terms defined therein being used herein as therein of a Delaware limited liability company (the "Borrower"), QEP Midstream Partners, LP, a Delawar to time party thereto, and Wells Fargo Bank, National Association, as Administrative Agent, L/C	lefined), among QEP Midstream Partners Operating, e limited partnership, each Lender and L/C Issuer from
The u	undersigned hereby requests a Swing Line Loan:	
1	On(a Business Day).	
2	In the amount of \$	
3	With a Swing Line Rate equal to [the Base Rate plus the Applicable Rate for Base Rate Loans] [the rate established as provided on Schedule 2.04]. The Swing Line Borrowing requested herein complies with the requirements of the provisos to the provisor to	ne first sentence of <u>Section 2.04(a)</u> of the Agreement.
QEP	P MIDSTREAM PARTNERS OPERATING, LLC	
By:	QEP MIDSTREAM PARTNERS, LP, its sole member	
By:	QEP MIDSTREAM PARTNERS GP, LLC, its general partner	
By:		
ŭ	Name:	
	Title:	

Exhibit A-2 Page 1

FORM OF NOTE

[], 20[]

FOR VALUE RECEIVED, the undersigned (the "Borrower") hereby promises to pay to or registered assigns (the "Lender"), in accordance with the provisions of the Agreement (as hereinafter defined), the principal amount of each Loan from time to time made by the Lender to the Borrower under that certain Credit Agreement, dated as of August 14, 2013 (as amended, restated, extended, supplemented or otherwise modified in writing from time to time, the "Agreement;" the terms defined therein being used herein as therein defined), among the Borrower, QEP Midstream Partners, LP, each Lender and L/C Issuer from time to time party thereto, and Wells Fargo Bank, National Association, as Administrative Agent, L/C Issuer and Swing Line Lender.

The Borrower promises to pay interest on the unpaid principal amount of each Loan from the date of such Loan until such principal amount is paid in full, at such interest rates and at such times as provided in the Agreement. All payments of principal and interest shall be made to the Administrative Agent for the account of the Lender in Dollars in immediately available funds at the Administrative Agent's Office. If any amount is not paid in full when due hereunder, such unpaid amount shall bear interest, to be paid upon demand, from the due date thereof until the date of actual payment (and before as well as after judgment) computed at the per annum rate set forth in the Agreement.

This Note is one of the Notes referred to in the Agreement, is entitled to the benefits thereof and may be prepaid in whole or in part subject to the terms and conditions provided therein. Upon the occurrence and continuation of one or more of the Events of Default specified in the Agreement, all amounts then remaining unpaid on this Note shall become, or may be declared to be, immediately due and payable all as provided in the Agreement. Loans made by the Lender shall be evidenced by one or more loan accounts or records maintained by the Lender in the ordinary course of business. The Lender may also attach schedules to this Note and endorse thereon the date, amount and maturity of its Loans and payments with respect thereto.

The Borrower, for itself, its successors and assigns, hereby waives diligence, presentment, protest and demand and notice of protest, demand, dishonor and non-payment of this Note.

Exhibit B Page 1 THIS NOTE SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK, WITHOUT REGARD TO CONFLICTS OF LAWS PRINCIPLES.

QEP MIDSTREAM PARTNERS OPERATING, LLC

Ву:	QEP MIDSTREAM PARTNERS, LP, its sole	member	
Ву:	QEP MIDSTREAM PARTNERS GP, LLC, its	general partner	
By:			
Dy.	Name:Title:		
		Exhibit B	

FORM OF COMPLIANCE CERTIFICATE

Financial Statement Date:	
Tillaliciai Statellielit Date.	

To: Wells Fargo Bank, National Association, as Administrative Agent

Ladies and Gentlemen:

Reference is made to that certain Credit Agreement, dated as of August 14, 2013 (as amended, restated, extended, supplemented or otherwise modified in writing from time to time, the "<u>Agreement</u>;" the terms defined therein being used herein as therein defined), among QEP Midstream Partners Operating, LLC, a Delaware limited liability company (the "<u>Borrower</u>"), QEP Midstream Partners, LP (the "<u>Parent Guarantor</u>"), each Lender and L/C Issuer from time to time party thereto, and Wells Fargo Bank, National Association, as Administrative Agent, L/C Issuer and Swing Line Lender.

The undersigned Responsible Officer of the Parent Guarantor and Responsible Officer of the Borrower hereby certifies as of the date hereof that he/she is the of each of the Parent Guarantor and the Borrower, and that, as such, he/she is authorized to execute and deliver this Certificate to the Administrative Agent on the behalf of the Parent Guarantor and behalf of the Borrower, and that:

[Use following paragraph 1 for fiscal year-end financial statements]

1. Attached hereto as <u>Schedule 1</u> are the year-end audited financial statements required by <u>Section 6.01(a)</u> and the reconciliation, if required, by <u>Section 6.01(d)(i)</u> of the Agreement for the fiscal year of the Parent Guarantor ended as of the above date, together with the report and opinion of an independent certified public accountant required by such section.

[Use following paragraph 1 for fiscal quarter-end financial statements]

- 1. Attached hereto as <u>Schedule 1</u> are the unaudited financial statements required by <u>Section 6.01(b)</u> and the reconciliation, if required, by <u>Section 6.01(d)(i)</u> of the Agreement for the fiscal quarter of the Parent Guarantor ended as of the above date. Such financial statements fairly present, in all material respects, the financial condition, results of operations and cash flows of the Parent Guarantor and its Subsidiaries in accordance with GAAP as at such date and for such period, subject only to normal year-end adjustments and the absence of footnotes.
- 2. The undersigned has reviewed and is familiar with the terms of the Agreement and has made, or has caused to be made under his/her supervision, a detailed review of the transactions and condition (financial or otherwise) of the Borrower and the Parent Guarantor during the accounting period covered by the attached financial statements.
- 3. A review of the activities of the Parent Guarantor and the Borrower during such fiscal period has been made under the supervision of the undersigned with a view to determining whether during such fiscal period the Borrower and the Parent Guarantor performed and observed all their Obligations under the Loan Documents, and

[select one:]

Exhibit C

[to the best knowledge of the undersigned	, there is no condition or	event at the end of such	fiscal period or at th	e time of this certifi	cate which
constit	utes a Default.]					

-or-

[to the best knowledge of the undersigned, at the end of such fiscal period, or on the date of this certificate, the following conditions and events constitute a Default and their nature and period of existence are specified below:]

- 4. The financial covenant analyses and information set forth on <u>Schedule 2</u> attached hereto are true and accurate on and as of the date of this Certificate.
- 5. [Attached hereto as <u>Schedule 3</u> are updating changes to the Schedules to the Security Agreement.] The Schedules to the Security Agreement, [as amended by <u>Schedule 3</u>] attached hereto, are accurate and complete in all material respects.
 - 6. During the fiscal quarter ending on the above date,

[no Loan Party or Material Subsidiary acquired any Material Real Property on which a Lien is required to be created pursuant to Section 6.11(c) of the Credit Agreement.]

-or-

[Material Real Property was acquired by a Loan Party or a Material Subsidiary, consisting of [include general description].]

Exhibit C

Name: Title:

By: QEP Midstream Partners GP, LLC, its general partner By: QEP Midstream Partners GP, LLC, its general partner Name: Title: QEP MIDSTREAM PARTNERS OPERATING, LLC By: QEP MIDSTREAM PARTNERS, LP, its sole member By: QEP MIDSTREAM PARTNERS GP, LLC, its general partner By: QEP MIDSTREAM PARTNERS GP, LLC, its general partner

Exhibit C Page 3

FORM OF SCHEDULE 1 TO COMPLIANCE CERTIFICATE

SCHEDULE 1 TO COMPLIANCE CERTIFICATE

Financial Statements

[To be attached]

Schedule 1 to Compliance Certificate

FORM OF SCHEDULE 2 TO COMPLIANCE CERTIFICATE

SCHEDULE 2 TO COMPLIANCE CERTIFICATE

For the Quarter/Year ended , 20 (" $\underline{\text{Statement Date}}$ ") (\$\in 000's)

[To be attached]

Schedule 2 to Compliance Certificate

FORM OF SCHEDULE 3 TO COMPLIANCE CERTIFICATE

SCHEDULE 3 TO COMPLIANCE CERTIFICATE

Updates to Schedules to the Security Agreement

[To be attached]

Schedule 3 to Compliance Certificate

FORM OF ASSIGNMENT AND ASSUMPTION

This Assignment and Assumption (this "Assignment and Assumption") is dated as of the Effective Date set forth below and is entered into by and between [Insert name of Assignor] (the "Assignor") and [Insert name of Assignee] (the "Assignee"). Capitalized terms used but not defined herein shall have the meanings given to them in the Credit Agreement identified below (the "Credit Agreement"), receipt of a copy of which is hereby acknowledged by the Assignee. The Standard Terms and Conditions set forth in Annex 1 attached hereto are hereby agreed to and incorporated herein by reference and made a part of this Assignment and Assumption as if set forth herein in full.

For an agreed consideration, the Assignor hereby irrevocably sells and assigns to the Assignee, and the Assignee hereby irrevocably purchases and assumes from the Assignor, subject to and in accordance with the Standard Terms and Conditions and the Credit Agreement, as of the Effective Date inserted by the Administrative Agent as contemplated below (i) all of the Assignor's rights and obligations as a Lender under the Credit Agreement and any other documents or instruments delivered pursuant thereto to the extent related to the amount and percentage interest identified below of all of such outstanding rights and obligations of the Assignor under the credit facility under the Credit Agreement (including, without limitation, the Letters of Credit) and (ii) to the extent permitted to be assigned under applicable law, all claims, suits, causes of action and any other right of the Assignor (in its capacity as a Lender) against any Person, whether known or unknown, arising under or in connection with the Credit Agreement, any other documents or instruments delivered pursuant thereto or the loan transactions governed thereby or in any way based on or related to any of the foregoing, including, but not limited to, contract claims, tort claims, malpractice claims, statutory claims and all other claims at law or in equity related to the rights and obligations sold and assigned pursuant to clause (i) above (the rights and obligations sold and assigned pursuant to clauses (i) and (ii) above being referred to herein collectively as, the "Assigned Interest"). Such sale and assignment is without recourse to the Assignor and, except as expressly provided in this Assignment and Assumption, without representation or warranty by the Assignor.

1

Assignor: ___

2.	Assignee:[identify Lender]	[and is an Affiliate/Approved Fund of
3.	Borrower: QEP Midstr	ream Partners Operating, LLC
4.	Administrative Agent:	Wells Fargo Bank, National Association, as the administrative agent under the Credit Agreement
5.		dit Agreement, dated as of August 14, 2013, among QEP Midstream Partners Operating, LLC, QEP Midstream Partners, L ssuer from time to time party thereto, and Wells Fargo Bank, National Association, as Administrative Agent, L/C Issuer and
6.	Assigned Interest:	
	ant of Commitment/Loans	
Amou	int of Commitment/Loans	Assigned:
Perce	ntage Assigned of Commit	ment:
CUSI	P Number :	

Exhibit D Page 1

[7.	Trade Date:	_]
Effectiv RECOR		, 20 [TO BE INSERTED BY ADMINISTRATIVE AGENT AND WHICH SHALL BE THE EFFECTIVE DATE OF ANSFER IN THE REGISTER THEREFOR.]
	The terms set	forth in this Assignment and Assumption are hereby agreed to:
		ASSIGNOR [NAME OF ASSIGNOR]
		By:
		ASSIGNEE [NAME OF ASSIGNEE]
		By:
[Conser	nted to and] Accep	ted:
	FARGO BANK, inistrative Agent	NATIONAL ASSOCIATION,
By: Title:		
[Conse	nted to:]	
QEP M	IDSTREAM PART	TNERS OPERATING, LLC
By: QE	P MIDSTREAM P	PARTNERS, LP, its sole member
By: QE	P MIDSTREAM P	PARTNERS GP, LLC, its general partner
By: Name: Title:		
		Exhibit D Page 2

STANDARD TERMS AND CONDITIONS FOR

ASSIGNMENT AND ASSUMPTION

1. Representations and Warranties.

- 1.1. <u>Assignor</u>. The Assignor (a) represents and warrants that (i) it is the legal and beneficial owner of the Assigned Interest, (ii) the Assigned Interest is free and clear of any lien, encumbrance or other adverse claim, (iii) it has full power and authority, and has taken all action necessary; to execute and deliver this Assignment and Assumption and to consummate the transactions contemplated hereby and (iv) it is [not] a Defaulting Lender; and (b) assumes no responsibility with respect to (i) any statements, warranties or representations made in or in connection with the Credit Agreement or any other Loan Document, (ii) the execution, legality, validity, enforceability, genuineness, sufficiency or value of the Loan Documents or any collateral thereunder, (iii) the financial condition of the Borrower, any of its Subsidiaries or Affiliates or any other Person obligated in respect of any Loan Document or (iv) the performance or observance by the Borrower, any of its Subsidiaries or Affiliates or any other Person of any of their respective obligations under any Loan Document.
- 1.2. Assignee. The Assignee (a) represents and warrants that (i) it has full power and authority, and has taken all action necessary, to execute and deliver this Assignment and Assumption and to consummate the transactions contemplated hereby and to become a Lender under the Credit Agreement, (ii) it meets all requirements of an Eligible Assignee under the Credit Agreement (subject to receipt of such consents as may be required under the Credit Agreement), (iii) from and after the Effective Date, it shall be bound by the provisions of the Credit Agreement as a Lender thereunder and, to the extent of the Assigned Interest, shall have the obligations of a Lender thereunder, (iv) it is sophisticated with respect to decisions to acquire assets of the type represented by the Assigned Interest and either it, or the Person exercising discretion in making its decision to acquire the Assigned Interest, is experienced in acquiring assets of such type, (v) it has received a copy of the Credit Agreement, together with copies of the most recent financial statements delivered pursuant to Section 6.01 thereof, as applicable, and such other documents and information as it has deemed appropriate to make its own credit analysis and decision to enter into this Assignment and Assumption and to purchase the Assigned Interest on the basis of which it has made such analysis and decision independently and without reliance on the Administrative Agent or any other Lender, and (vi) if it is a Foreign Lender, attached hereto is any documentation required to be delivered by it pursuant to the terms of the Credit Agreement, duly completed and executed by the Assignee; and (b) agrees that (i) it will, independently and without reliance on the Administrative Agent, the Assignor or any other Lender, and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under the Loan Documents, and (ii) it will perform in accordance with their terms all of

Annex I to Exhibit D

- 2. <u>Payments</u>. From and after the Effective Date, the Administrative Agent shall make all payments in respect of the Assigned Interest (including payments of principal, interest, fees and other amounts) to the Assignor for amounts which have accrued to but excluding the Effective Date and to the Assignee for amounts which have accrued from and after the Effective Date.
- 3. <u>General Provisions</u>. This Assignment and Assumption shall be binding upon, and inure to the benefit of, the parties hereto and their respective successors and assigns. This Assignment and Assumption may be executed in any number of counterparts, which together shall constitute one instrument. Delivery of an executed counterpart of a signature page of this Assignment and Assumption by telecopy shall be effective as delivery of a manually executed counterpart of this Assignment and Assumption. This Assignment and Assumption shall be governed by, and construed in accordance with, the law of the State of New York, without regard to conflicts of laws principles.

Annex I to Exhibit D

(For Foreign Lenders That Are Not Partnerships For U.S. Federal Income Tax Purposes)

Reference is hereby made to the Credit Agreement dated as of August 14, 2013 (as amended, supplemented or otherwise modified from time to time, the "Credit Agreement"), among QEP MIDSTREAM PARTNERS OPERATING, LLC, as Borrower, QEP MIDSTREAM PARTNERS, LP, as Guarantor, WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent, L/C Issuer and Swing Line Lender, and each Lender and L/C Issuer from time to time party thereto.

Pursuant to the provisions of Section 3.01 of the Credit Agreement, the undersigned hereby certifies that (i) it is the sole record and beneficial owner of the Loan(s) (as well as any Note(s) evidencing such Loan(s)) in respect of which it is providing this certificate, (ii) it is not a bank within the meaning of Section 881(c)(3)(A) of the Code, (iii) it is not a ten percent shareholder of the Borrower within the meaning of Section 871(h)(3)(B) of the Code and (iv) it is not a controlled foreign corporation related to the Borrower as described in Section 881(c)(3)(C) of the Code.

The undersigned has furnished the Administrative Agent and the Borrower with a certificate of its non-U.S. Person status on IRS Form W-8BEN. By executing this certificate, the undersigned agrees that (1) if the information provided on this certificate changes, the undersigned shall promptly so inform the Borrower and the Administrative Agent, and (2) the undersigned shall have at all times furnished the Borrower and the Administrative Agent with a properly completed and currently effective certificate in either the calendar year in which each payment is to be made to the undersigned, or in either of the two calendar years preceding such payments.

Unless otherwise defined herein, terms defined in the Credit Agreement and used herein shall have the meanings given to them in the Credit Agreement.

OF LENDER]				
Name:				
Title:				
Date:	_, 20[]		
		_		

Exhibit E-1 Page 1

(For Foreign Participants That Are Not Partnerships For U.S. Federal Income Tax Purposes)

Reference is hereby made to the Credit Agreement dated as of August 14, 2013 (as amended, supplemented or otherwise modified from time to time, the "Credit Agreement"), among QEP MIDSTREAM PARTNERS OPERATING, LLC, as Borrower, QEP MIDSTREAM PARTNERS, LP, as Guarantor, WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent, L/C Issuer and Swing Line Lender, and each Lender and L/C Issuer from time to time party thereto.

Pursuant to the provisions of Section 3.01 of the Credit Agreement, the undersigned hereby certifies that (i) it is the sole record and beneficial owner of the participation in respect of which it is providing this certificate, (ii) it is not a bank within the meaning of Section 881(c)(3)(A) of the Code, (iii) it is not a ten percent shareholder of the Borrower within the meaning of Section 871(h)(3)(B) of the Code, and (iv) it is not a controlled foreign corporation related to the Borrower as described in Section 881(c)(3)(C) of the Code.

The undersigned has furnished its participating Lender with a certificate of its non-U.S. Person status on IRS Form W-8BEN. By executing this certificate, the undersigned agrees that (1) if the information provided on this certificate changes, the undersigned shall promptly so inform such Lender in writing, and (2) the undersigned shall have at all times furnished such Lender with a properly completed and currently effective certificate in either the calendar year in which each payment is to be made to the undersigned, or in either of the two calendar years preceding such payments.

Unless otherwise defined herein, terms defined in the Credit Agreement and used herein shall have the meanings given to them in the Credit Agreement.

[NAM	E OF PARTICIPANT]	
By:		
	Name:	
	Title:	
	Date:, 20[]

Exhibit E-2 Page 1

(For Foreign Participants That Are Partnerships For U.S. Federal Income Tax Purposes)

Reference is hereby made to the Credit Agreement dated as of August 14, 2013 (as amended, supplemented or otherwise modified from time to time, the "Credit Agreement"), among QEP MIDSTREAM PARTNERS OPERATING, LLC, as Borrower, QEP MIDSTREAM PARTNERS, LP, as Guarantor, WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent, L/C Issuer and Swing Line Lender, and each Lender and L/C Issuer from time to time party thereto.

Pursuant to the provisions of Section 3.01 of the Credit Agreement, the undersigned hereby certifies that (i) it is the sole record owner of the participation in respect of which it is providing this certificate, (ii) its direct or indirect partners/members are the sole beneficial owners of such participation, (iii) with respect such participation, neither the undersigned nor any of its direct or indirect partners/members is a bank extending credit pursuant to a loan agreement entered into in the ordinary course of its trade or business within the meaning of Section 881(c)(3)(A) of the Code, (iv) none of its direct or indirect partners/members is a ten percent shareholder of the Borrower within the meaning of Section 871(h)(3)(B) of the Code and (v) none of its direct or indirect partners/members is a controlled foreign corporation related to the Borrower as described in Section 881(c)(3)(C) of the Code.

The undersigned has furnished its participating Lender with IRS Form W-8IMY accompanied by one of the following forms from each of its partners/members that is claiming the portfolio interest exemption: (i) an IRS Form W-8BEN or (ii) an IRS Form W-8IMY accompanied by an IRS Form W-8BEN from each of such partner's/member's beneficial owners that is claiming the portfolio interest exemption. By executing this certificate, the undersigned agrees that (1) if the information provided on this certificate changes, the undersigned shall promptly so inform such Lender and (2) the undersigned shall have at all times furnished such Lender with a properly completed and currently effective certificate in either the calendar year in which each payment is to be made to the undersigned, or in either of the two calendar years preceding such payments.

Unless otherwise defined herein, terms defined in the Credit Agreement and used herein shall have the meanings given to them in the Credit Agreement.

[NAME OF PARTICIPANT]	
By:	
Name:	
Title:	
Date:	, 20[]

Exhibit E-3

(For Foreign Lenders That Are Partnerships For U.S. Federal Income Tax Purposes)

Reference is hereby made to the Credit Agreement dated as of August 14, 2013 (as amended, supplemented or otherwise modified from time to time, the "<u>Credit Agreement</u>"), among QEP MIDSTREAM PARTNERS OPERATING, LLC, as Borrower, QEP MIDSTREAM PARTNERS, LP, as Guarantor, WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent, L/C Issuer and Swing Line Lender, and each Lender and L/C Issuer from time to time party thereto.

Pursuant to the provisions of Section 3.01 of the Credit Agreement, the undersigned hereby certifies that (i) it is the sole record owner of the Loan(s) (as well as any Note(s) evidencing such Loan(s)) in respect of which it is providing this certificate, (ii) its direct or indirect partners/members are the sole beneficial owners of such Loan(s) (as well as any Note(s) evidencing such Loan(s)), (iii) with respect to the extension of credit pursuant to this Credit Agreement or any other Loan Document, neither the undersigned nor any of its direct or indirect partners/members is a bank extending credit pursuant to a loan agreement entered into in the ordinary course of its trade or business within the meaning of Section 881(c)(3)(A) of the Code, (iv) none of its direct or indirect partners/members is a ten percent shareholder of the Borrower within the meaning of Section 871(h)(3)(B) of the Code and (v) none of its direct or indirect partners/members is a controlled foreign corporation related to the Borrower as described in Section 881(c)(3)(C) of the Code.

The undersigned has furnished the Administrative Agent and the Borrower with IRS Form W-8IMY accompanied by one of the following forms from each of its partners/members that is claiming the portfolio interest exemption: (i) an IRS Form W-8BEN or (ii) an IRS Form W-8IMY accompanied by an IRS Form W-8BEN from each of such partner's/member's beneficial owners that is claiming the portfolio interest exemption. By executing this certificate, the undersigned agrees that (1) if the information provided on this certificate changes, the undersigned shall promptly so inform the Borrower and the Administrative Agent, and (2) the undersigned shall have at all times furnished the Borrower and the Administrative Agent with a properly completed and currently effective certificate in either the calendar year in which each payment is to be made to the undersigned, or in either of the two calendar years preceding such payments.

Unless otherwise defined herein, terms defined in the Credit Agreement and used herein shall have the meanings given to them in the Credit Agreement.

[NAI	ME OF LENDER]	
By:		
	Name:	
	Title:	
	Date:, 20[]

Exhibit E-4 Page 1

QEP MIDSTREAM PARTNERS, LP 2013 LONG-TERM INCENTIVE PLAN

SECTION 1. Purpose of the Plan.

This QEP Midstream Partners, LP 2013 Long-Term Incentive Plan (the "Plan") has been adopted by QEP Midstream Partners GP, LLC, a Delaware limited liability company (the "Company"), the general partner of QEP Midstream Partners, LP, a Delaware limited partnership (the "Partnership"). The Plan is intended to promote the interests of the Partnership and the Company by providing incentive compensation awards denominated in or based on Units to Employees, Consultants and Directors to encourage superior performance. The Plan is also intended to enhance the ability of the Partnership, the Company and their Affiliates to attract and retain the services of individuals who are essential for the growth and profitability of the Partnership, the Company and their Affiliates and to encourage them to devote their best efforts to advancing the business of the Partnership, the Company and their Affiliates.

SECTION 2. Definitions.

As used in the Plan, the following terms shall have the meanings set forth below:

- "Affiliate" means, with respect to any Person, any other Person that directly or indirectly through one or more intermediaries controls, is controlled by or is under common control with, the Person in question. As used herein, the term "control" means the possession, direct or indirect, of the power to direct or cause the direction of the management and policies of a Person, whether through ownership of voting securities, by contract or otherwise.
 - "ASC Topic 718" means Accounting Standards Codification Topic 718, Compensation Stock Compensation, or any successor accounting standard.
- "Award" means an Option, Restricted Unit, Phantom Unit, DER, Substitute Award, Unit Appreciation Right, Unit Award or Profits Interest Unit granted under the Plan.
 - "Award Agreement" means the written or electronic agreement between a Participant and the Company that sets forth the terms of the Award.
 - "Board" means the board of directors or board of managers, as the case may be, of the Company.
 - "Change in Control" means, and shall be deemed to have occurred upon one or more of the following events:
- (i) any "person" or "group" within the meaning of Sections 13(d) and 14(d)(2) of the Exchange Act, other than the Company, QEP or an Affiliate of the Company or QEP (as determined immediately prior to such event), shall become the beneficial owner, by way of merger, acquisition, consolidation, recapitalization, reorganization or otherwise, of 50% or more of the combined voting power of the equity interests in the Company or the Partnership;
 - (ii) the limited partners of the Partnership approve, in one or a series of transactions, a plan of complete liquidation of the Partnership;
- (iii) the sale or other disposition by either the Company or the Partnership of all or substantially all of its assets in one or more transactions to any Person other than the Company, the Partnership, QEP or an Affiliate of the Company the Partnership or QEP;
- (iv) a transaction resulting in a Person other than QEP, the Company or an Affiliate of QEP or the Company (as determined immediately prior to such event) being the sole general partner of the Partnership; or
- (v) a change in control of QEP, as defined under Section 15 of the QEP Resources, Inc. 2010 Stock Incentive Plan or as may be defined pursuant to the terms of any successor equity compensation plan of QEP.

Notwithstanding the foregoing, if a Change in Control constitutes a payment event with respect to any Award which provides for the deferral of compensation and is subject to Section 409A, the transaction or event described in subsection (i), (ii), (iii) or (iv) above with respect to such Award must also constitute a "change in control event," as defined in Treasury Regulation §1.409A-3(i)(5), and as relates to the holder of such Award, to the extent required to comply with Section 409A.

"Code" means the Internal Revenue Code of 1986, as amended.

"Committee" means the Board, except that it shall mean such committee of the Board as is appointed by the Board to administer the Plan.

"Consultant" means an individual who renders consulting services to the Company, the Partnership or any of their Affiliates.

"<u>DER</u>" means a distribution equivalent right, representing a contingent right to receive an amount in cash, Units, Restricted Units and/or Phantom Units equal in value to the distributions made by the Partnership with respect to a Unit during the period such Award is outstanding.

"<u>Director</u>" means a member of the board of directors or board of managers, as the case may be, of the Company, the Partnership or any of their Affiliates who is not an Employee or a Consultant (other than in that individual's capacity as a Director).

"<u>Disability</u>" means a total and permanent disability within the meaning of Section 22(e)(3) of the Code; provided, however, that if a Disability constitutes a payment event with respect to any Award which provides for the deferral of compensation and is subject to Section 409A, then, to the extent required to comply with Section 409A, the Participant must also be considered "disabled" within the meaning of Section 409A(a)(2)(C) of the Code. A determination of Disability may be made by a physician selected or approved by the Committee and, in this respect, Participants shall submit to an examination by such physician upon request by the Committee.

"Employee" means an employee of the Company, the Partnership or any of their Affiliates.

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

"Fair Market Value" means, as of any given date, the closing sales price on such date during normal trading hours (or, if there are no reported sales on such date, on the last date prior to such date on which there were sales) of the Units on the New York Stock Exchange or, if not listed on such exchange, on any other national securities exchange on which the Units are listed or on an inter-dealer quotation system, in any case, as reported in such source as the Committee shall select. If there is no regular public trading market for the Units, the Fair Market Value of the Units shall be determined by the Committee in good faith and, to the extent applicable, in compliance with the requirements of Section 409A.

"Option" means an option to purchase Units granted pursuant to Section 6(a) of the Plan.

"Other Unit-Based Award" means an award granted pursuant to Section 6(f) of the Plan.

"Participant" means an Employee, Consultant or Director granted an Award under the Plan and any authorized transferee of such individual.

"Partnership Agreement" means the Agreement of Limited Partnership of the Partnership, as it may be amended or amended and restated from time to time.

"Person" shall have the meaning ascribed to such term in Section 3(a)(9) of the Exchange Act and used in Sections 13(d) and 14(d) thereof, including a "group" as defined in Section 13(d) thereof.

"Phantom Unit" means a notional interest granted under the Plan that, to the extent vested, entitles the Participant to receive a Unit or an amount of cash equal to the Fair Market Value of a Unit, as determined by the Committee in its discretion.

"Profits Interest Unit" means to the extent authorized by the Partnership Agreement, an interest in the Partnership that is intended to constitute a "profits interest" within the meaning of the Code, Treasury Regulations promulgated thereunder, and any published guidance by the Internal Revenue Service with respect thereto.

"QEP" means QEP Resources, Inc., a Delaware corporation, or any successor thereto.

"Restricted Period" means the period established by the Committee with respect to an Award during which the Award remains subject to forfeiture and is either not exercisable by or payable to the Participant, as the case may be.

"Restricted Unit" means a Unit granted pursuant to Section 6(b) of the Plan that is subject to a Restricted Period.

"Securities Act" means the Securities Act of 1933, as amended.

"SEC" means the Securities and Exchange Commission, or any successor thereto.

"Section 409A" means Section 409A of the Code and the Department of Treasury Regulations and other interpretive guidance issued thereunder, including without limitation any such regulations or other guidance that may be issued after the Effective Date (as defined in Section 9 below).

"Service" means service as an Employee, Consultant or Director. The Committee, in its sole discretion, shall determine the effect of all matters and questions relating to terminations of Service, including, without limitation, the questions of whether and when a termination of Service occurred and/or resulted from a discharge for cause (as such term may be defined in an Award Agreement or, otherwise, as may be determined by the Committee), and all questions of whether particular changes in status or leaves of absence constitute a termination of Service. The Committee, in its sole discretion, subject to the terms of any applicable Award Agreement, may determine that a termination of Service has not occurred in the event of (a) a termination where there is simultaneous commencement by the Participant of a relationship with the Partnership, the Company or any of their Affiliates as an Employee, Director or Consultant or (b) a termination which results in a temporary severance of the service relationship.

"Substitute Award" means an award granted pursuant to Section 6(g) of the Plan.

"Unit" means a Common Unit of the Partnership.

"<u>Unit Appreciation Right</u>" or "<u>UAR</u>" means a contingent right that entitles the holder to receive the excess of the Fair Market Value of a Unit on the exercise date of the UAR over the exercise price of the UAR.

" $\underline{\text{Unit Award}}$ " means an award granted pursuant to Section 6(d) of the Plan.

SECTION 3. Administration.

- (a) The Plan shall be administered by the Committee, subject to subsection (b) below; provided, however, that in the event that the Board is not also serving as the Committee, the Board, in its sole discretion, may at any time and from time to time exercise any and all rights and duties of the Committee under the Plan. The governance of the Committee shall be subject to the charter, if any, of the Committee as approved by the Board. Subject to the terms of the Plan and applicable law, and in addition to other express powers and authorizations conferred on the Committee by the Plan, the Committee shall have full power and authority to: (i) designate Participants; (ii) determine the type or types of Awards to be granted to a Participant; (iii) determine the number of Units to be covered by Awards; (iv) determine the terms and conditions of any Award; (v) determine whether, to what extent, and under what circumstances Awards may be settled, exercised, canceled, or forfeited; (vi) interpret and administer the Plan and any instrument or agreement relating to an Award made under the Plan; (vii) establish, amend, suspend, or waive such rules and regulations and appoint such agents as it shall deem appropriate for the proper administration of the Plan; and (viii) make any other determination and take any other action that the Committee deems necessary or desirable for the administration of the Plan. The Committee may correct any defect or supply any omission or reconcile any inconsistency in the Plan or an Award Agreement in such manner and to such extent as the Committee deems necessary or appropriate. Unless otherwise expressly provided in the Plan, all designations, determinations, interpretations, and other decisions under or with respect to the Plan or any Award shall be within the sole discretion of the Committee, may be made at any time and shall be final, conclusive, and binding upon all Persons, including the Company, the Partnership, any of their Affiliates, any Participant and any beneficiary of any Particip
- (b) To the extent permitted by applicable law and the rules of any securities exchange on which the Units are listed, quoted or traded, the Board or Committee may from time to time delegate to a committee of one or more members of the Board or one or more officers of the Company the authority to grant or amend Awards or to take other administrative actions pursuant to Section 3(a); provided, however, that in no event shall an officer of the Company be delegated the authority to grant awards to, or amend awards held by, the following individuals: (i) individuals who are subject to Section 16 of the Exchange Act, or (ii) officers of the Company (or Directors) to whom authority to grant or amend Awards has been delegated hereunder; provided, further, that any delegation of administrative authority shall only be permitted to the extent that it is permissible under applicable provisions of the Code and applicable securities laws and the rules of any securities exchange on which the Units are listed, quoted or traded. Any delegation hereunder shall be subject to such restrictions and limitations as the Board or Committee, as applicable, specifies at the time of such delegation, and the Board or Committee, as applicable, may at any time rescind the authority so delegated or appoint a new delegatee. At all times, the delegatee appointed under this Section 3(b) shall serve in such capacity at the pleasure of the Board and the Committee.

SECTION 4. Units.

(a) <u>Limits on Units Deliverable</u>. Subject to adjustment as provided in Section 4(c), the number of Units that may be delivered with respect to Awards under the Plan is Five Million Three Hundred Forty-One Thousand (5,341,000). If any Award is forfeited, cancelled, exercised, paid, or otherwise terminates or expires without the actual delivery of Units pursuant to such Award (for the avoidance of doubt, the grant of Restricted Units is not a delivery of Units for this purpose unless and until such Restricted Units vest and any restrictions placed upon them under the Plan lapse), the Units subject to such Award shall again be available for Awards under the Plan. To the extent permitted by applicable law and securities exchange rules, Substitute Awards and Units issued in assumption of, or in substitution for, any outstanding awards of any entity acquired in any form of combination by the Partnership or any Affiliate thereof shall not be counted against the Units available for issuance pursuant to the Plan. There shall not be any limitation on the number of Awards that may be paid in cash.

(b) <u>Sources of Units Deliverable Under Awards</u>. Any Units delivered pursuant to an Award shall consist, in whole or in part, of Units acquired in the open market, from the Partnership, any Affiliate thereof or any other Person, or Units otherwise issuable by the Partnership, or any combination of the foregoing, as determined by the Committee in its discretion.

(c) Anti-dilution Adjustments.

- (i) Equity Restructuring. With respect to any "equity restructuring" event that could result in an additional compensation expense to the Company or the Partnership pursuant to the provisions of ASC Topic 718 if adjustments to Awards with respect to such event were discretionary, the Committee shall equitably adjust the number and type of Units covered by each outstanding Award and the terms and conditions, including the exercise price and performance criteria (if any), of such Award to equitably reflect such event and shall adjust the number and type of Units (or other securities or property) with respect to which Awards may be granted under the Plan after such event. With respect to any other similar event that would not result in an ASC Topic 718 accounting charge if the adjustment to Awards with respect to such event were subject to discretionary action, the Committee shall have complete discretion to adjust Awards and the number and type of Units (or other securities or property) with respect to which Awards may be granted under the Plan in such manner as it deems appropriate with respect to such other event.
- (ii) Other Changes in Capitalization. In the event of any non-cash distribution, Unit split, combination or exchange of Units, merger, consolidation or distribution (other than normal cash distributions) of Partnership assets to unitholders, or any other change affecting the Units of the Partnership, other than an "equity restructuring," the Committee may make equitable adjustments, if any, to reflect such change with respect to (A) the aggregate number and kind of Units that may be issued under the Plan; (B) the number and kind of Units (or other securities or property) subject to outstanding Awards; (C) the terms and conditions of any outstanding Awards (including, without limitation, any applicable performance targets or criteria with respect thereto); and (D) the grant or exercise price per Unit for any outstanding Awards under the Plan.

SECTION 5. Eligibility.

Any Employee, Consultant or Director shall be eligible to be designated a Participant and receive an Award under the Plan.

SECTION 6. Awards.

- (a) Options and UARs. The Committee shall have the authority to determine the Employees, Consultants and Directors to whom Options and/or UARs shall be granted, the number of Units to be covered by each Option or UAR, the exercise price therefor, the Restricted Period and other conditions and limitations applicable to the exercise of the Option or UAR, including the following terms and conditions and such additional terms and conditions, as the Committee shall determine, that are not inconsistent with the provisions of the Plan. Options which are intended to comply with Treasury Regulation Section 1.409A-1(b)(5)(i)(B) or, in each case, any successor regulation, may be granted only if the requirements of Treasury Regulation Section 1.409A-1(b)(5)(iii), or any successor regulation, are satisfied. Options and UARs that are otherwise exempt from or compliant with Section 409A may be granted to any eligible Employee, Consultant or Director.
- (i) Exercise Price. The exercise price per Unit purchasable under an Option or subject to a UAR shall be determined by the Committee at the time the Option or UAR is granted but, except with respect to a Substitute Award, may not be less than the Fair Market Value of a Unit as of the date of grant of the Option or UAR.
- (ii) <u>Time and Method of Exercise</u>. The Committee shall determine the exercise terms and any applicable Restricted Period with respect to an Option or UAR, which may include, without limitation, provisions for accelerated vesting upon the achievement of specified performance goals and/or other events, and the method or methods by which payment of the exercise price with respect to an Option or UAR may be made or deemed to have been made, which may include, without limitation, cash, check acceptable to the Company, withholding Units having a Fair Market Value on the exercise date equal to the relevant exercise price from the Award, a "cashless" exercise through procedures approved by the Company, or any combination of the foregoing methods.
- (iii) Exercise of Options and UARs on Termination of Service. Each Option and UAR Award Agreement shall set forth the extent to which the Participant shall have the right to exercise the Option or UAR following a termination of the Participant's Service. Unless otherwise determined by the Committee, if the Participant's Service is terminated for cause (as such term may be defined in an Award Agreement or, otherwise, as may be determined by the Committee), the Participant's right to exercise the Option or UAR shall terminate as of the start of business on the effective date of the Participant's termination. Unless otherwise determined by the Committee, to the extent the Option or UAR is not vested and exercisable as of the termination of Service, the Option or UAR shall terminate when the Participant's Service terminates.
- (iv) <u>Term of Options and UARs</u>. The term of each Option and UAR shall be stated in the Award Agreement, *provided*, that the term shall be no more than ten (10) years from the date of grant thereof.

- (b) <u>Restricted Units and Phantom Units</u>. The Committee shall have the authority to determine the Employees, Consultants and Directors to whom Restricted Units and/or Phantom Units shall be granted, the number of Restricted Units or Phantom Units to be granted to each such Participant, the applicable Restricted Period, the conditions under which the Restricted Units or Phantom Units may become vested or forfeited and such other terms and conditions, including, without limitation, restrictions on transferability, as the Committee may establish with respect to such Awards.
- (i) <u>Payment of Phantom Units</u>. The Committee shall specify, or permit the Participant to elect in accordance with the requirements of Section 409A, the conditions and dates or events upon which the cash or Units underlying an award of Phantom Units shall be issued, which dates or events shall not be earlier than the date on which the Phantom Units vest and become nonforfeitable and which conditions and dates or events shall be subject to compliance with Section 409A (unless the Phantom Units are exempt therefrom).
- (ii) <u>Vesting of Restricted Units</u>. Upon or as soon as reasonably practicable following the vesting of each Restricted Unit, subject to satisfying the tax withholding obligations of Section 8(b), the Participant shall be entitled to have the restrictions removed from his or her Unit certificate (or book-entry account, as applicable) so that the Participant then holds an unrestricted Unit.
- (c) <u>DERs</u>. The Committee shall have the authority to determine the Employees, Consultants and/or Directors to whom DERs are granted, whether such DERs are tandem or separate Awards, whether the DERs shall be paid directly to the Participant, be credited to a bookkeeping account (with or without interest in the discretion of the Committee), any vesting restrictions and payment provisions applicable to the DERs, and such other provisions or restrictions as determined by the Committee in its discretion, all of which shall be specified in the applicable Award Agreements. Distributions in respect of DERs shall be credited as of the distribution dates during the period between the date an Award is granted to a Participant and the date such Award vests, is exercised, is distributed or expires, as determined by the Committee. Such DERs shall be converted to cash, Units, Restricted Units and/or Phantom Units by such formula and at such time and subject to such limitations as may be determined by the Committee. Tandem DERs may be subject to the same or different vesting restrictions as the tandem Award, or be subject to such other provisions or restrictions as determined by the Committee in its discretion. Notwithstanding the foregoing, DERs shall only be paid in a manner that is either exempt from or in compliance with Section 409A.
- (d) <u>Unit Awards</u>. Awards of Units may be granted under the Plan (i) to such Employees, Consultants and/or Directors and in such amounts as the Committee, in its discretion, may select, and (ii) subject to such other terms and conditions, including, without limitation, restrictions on transferability, as the Committee may establish with respect to such Awards.
- (e) <u>Profits Interest Units</u>. Any Award consisting of Profits Interest Units may be granted to an Employee, Consultant or Director for the performance of services to or for the benefit of the Partnership (i) in the Participant's capacity as a partner of the Partnership, (ii) in anticipation of the Participant becoming a partner of the Partnership, or (iii) as otherwise determined by the Committee. At the time of grant, the Committee shall specify the date or dates on which the Profits Interest Units shall vest and become nonforfeitable, and may specify such conditions to vesting as it deems appropriate. Profits Interest Units shall be subject to such restrictions on transferability and other restrictions as the Committee may impose.
- (f) Other Unit-Based Awards. Other Unit-Based Awards may be granted under the Plan to such Employees, Consultants and/or Directors as the Committee, in its discretion, may select. An Other Unit-Based Award shall be an award denominated or payable in, valued in or otherwise based on or related to Units, in whole or in part. The Committee shall determine the terms and conditions of any Other Unit-Based Award. Upon vesting, an Other Unit-Based Award may be paid in cash, Units (including Restricted Units) or any combination thereof as provided in the Award Agreement.
- (g) <u>Substitute Awards</u>. Awards may be granted under the Plan in substitution of similar awards held by individuals who become Employees, Consultants or Directors as a result of a merger, consolidation or acquisition by the Partnership or an Affiliate of another entity or the assets of another entity. Such Substitute Awards that are Options or UARs may have exercise prices less than the Fair Market Value of a Unit on the date of the substitution if such substitution complies with Section 409A and other applicable laws and securities exchange rules.

(h) General.

- (i) <u>Forfeitures</u>. Except as otherwise provided in the terms of an Award Agreement, upon termination of a Participant's Service for any reason during an applicable Restricted Period, all outstanding, unvested Awards held by such Participant shall be automatically forfeited by the Participant. The Committee may, in its discretion, waive in whole or in part such forfeiture with respect to any such Award; *provided*, that any such waiver shall be effective only to the extent that such waiver will not cause any Award intended to satisfy the requirements of Section 409A to fail to satisfy such requirements.
- (ii) <u>Awards May Be Granted Separately or Together</u>. Awards may, in the discretion of the Committee, be granted either alone or in addition to, in tandem with, or in substitution for any other Award granted under the Plan or any award granted under any other plan of the Company or any Affiliate. Awards granted in addition to or in tandem with other Awards or

awards granted under any other plan of the Company or any Affiliate may be granted either at the same time as or at a different time from the grant of such other Awards or awards.

(iii) Limits on Transfer of Awards.

- (A) Except as provided in paragraph (C) below, each Option and UAR shall be exercisable only by the Participant during the Participant's lifetime, or by the person to whom the Participant's rights shall pass by will or the laws of descent and distribution.
- (B) Except as provided in paragraph (C) below, no Award and no right under any such Award may be assigned, alienated, pledged, attached, sold or otherwise transferred or encumbered by a Participant other than by will or the laws of descent and distribution and any such purported assignment, alienation, pledge, attachment, sale, transfer or encumbrance shall be void and unenforceable against the Company, the Partnership or any Affiliate.
- (C) The Committee may provide in an Award Agreement that an Award may, on such terms and conditions as the Committee may from time to time establish, be transferred by a Participant without consideration to any "family member" of the Participant, as defined in the instructions to use of the Form S-8 Registration Statement under the Securities Act, as applicable, or any other transferee specifically approved by the Committee after taking into account any state, federal, local or foreign tax and securities laws applicable to transferable Awards. In addition, vested Units may be transferred to the extent permitted by the Partnership Agreement and not otherwise prohibited by the Award Agreement or any other agreement restricting the transfer of such Units.
- (iv) <u>Term of Awards</u>. Subject to Section 6(a)(iv) above, the term of each Award, if any, shall be for such period as may be determined by the Committee.
- (v) <u>Unit Certificates</u>. Unless otherwise determined by the Committee or required by any applicable law, rule or regulation, neither the Company nor the Partnership shall deliver to any Participant certificates evidencing Units issued in connection with any Award and instead such Units shall be recorded in the books of the Partnership (or, as applicable, its transfer agent or equity plan administrator). All certificates for Units or other securities of the Partnership delivered under the Plan and all Units issued pursuant to book entry procedures pursuant to any Award or the exercise

thereof shall be subject to such stop-transfer orders and other restrictions as the Committee may deem advisable under the Plan or the rules, regulations, and/or other requirements of the SEC, any securities exchange upon which such Units or other securities are then listed, and any applicable federal or state laws, and the Committee may cause a legend or legends to be inscribed on any such certificates or book entry to make appropriate reference to such restrictions.

- (vi) <u>Consideration for Grants</u>. To the extent permitted by applicable law, Awards may be granted for such consideration, including services, as the Committee shall determine.
- (vii) <u>Delivery of Units or other Securities and Payment by Participant of Consideration</u>. Notwithstanding anything in the Plan or any Award Agreement to the contrary, subject to compliance with Section 409A, the Company shall not be required to issue or deliver any certificates or make any book entries evidencing Units pursuant to the exercise or vesting of any Award, unless and until the Board or the Committee has determined, with advice of counsel, that the issuance of such Units is in compliance with all applicable laws, regulations of governmental authorities and, if applicable, the requirements of any securities exchange on which the Units are listed or traded, and the Units are covered by an effective registration statement or applicable exemption from registration. In addition to the terms and conditions provided herein, the Board or the Committee may require that a Participant make such reasonable covenants, agreements, and representations as the Board or the Committee, in its discretion, deems advisable in order to comply with any such laws, regulations, or requirements. Without limiting the generality of the foregoing, the delivery of Units pursuant to the exercise or vesting of an Award may be deferred for any period during which, in the good faith determination of the Committee, the Company is not reasonably able to obtain or deliver Units pursuant to such Award without violating applicable law or the applicable rules or regulations of any governmental agency or authority or securities exchange. No Units or other securities shall be delivered pursuant to any Award until payment in full of any amount required to be paid pursuant to the Plan or the applicable Award Agreement (including, without limitation, any exercise price or tax withholding) is received by the Company.

SECTION 7. Amendment and Termination; Certain Transactions.

Except to the extent prohibited by applicable law:

(a) <u>Amendments to the Plan</u>. Except as required by applicable law or the rules of the principal securities exchange, if any, on which the Units are traded and subject to Section 7(b) below, the Board or the Committee may amend, alter, suspend, discontinue, or terminate the Plan in any manner without the consent of any partner, Participant, other holder or beneficiary of an Award, or any other Person. The Board shall obtain securityholder approval of any Plan amendment to the extent necessary to comply with applicable law or securities exchange listing standards or rules.

- (b) <u>Amendments to Awards</u>. Subject to Section 7(a) above, the Committee may waive any conditions or rights under, amend any terms of, or alter any Award theretofore granted, provided that no change, other than pursuant to Section 7(c) below, in any Award shall materially reduce the rights or benefits of a Participant with respect to an Award without the consent of such Participant.
- (c) <u>Actions Upon the Occurrence of Certain Events</u>. Upon the occurrence of a Change in Control, any transaction or event described in Section 4(c) above, any change in applicable laws or regulations affecting the Plan or Awards hereunder, or any change in accounting principles affecting the financial statements of the Company or the Partnership, the Committee, in its sole discretion, without the consent of any Participant or holder of an Award, and on such terms and conditions as it deems appropriate, may take any one or more of the following actions:
- (i) provide for either (A) the termination of any Award in exchange for a payment in an amount, if any, equal to the amount that would have been attained upon the exercise of such Award or realization of the Participant's rights under such Award (and, for the avoidance of doubt, if as of the date of the occurrence of such transaction or event, the Committee determines in good faith that no amount would have been payable upon the exercise of such Award or realization of the Participant's rights, then such Award may be terminated by the Company without payment) or (B) the replacement of such Award with other rights or property selected by the Committee in its sole discretion having an aggregate value not exceeding the amount that could have been attained upon the exercise of such Award or realization of the Participant's rights had such Award been currently exercisable or payable or fully vested;
- (ii) provide that such Award be assumed by the successor or survivor entity, or a parent or subsidiary thereof, or be exchanged for similar options, rights or awards covering the equity of the successor or survivor, or a parent or subsidiary thereof, with appropriate adjustments as to the number and kind of equity interests and prices;
- (iii) make adjustments in the number and type of Units (or other securities or property) subject to outstanding Awards, the number and kind of outstanding Awards, the terms and conditions of (including the exercise price), and/or the vesting and performance criteria included in, outstanding Awards;
- (iv) provide that such Award shall vest or become exercisable or payable, notwithstanding anything to the contrary in the Plan or the applicable Award Agreement; and
 - (v) provide that the Award cannot be exercised or become payable after such event and shall terminate upon such event.

Notwithstanding the foregoing, (i) with respect to an above event that constitutes an "equity restructuring" that would be subject to a compensation expense pursuant ASC Topic 718, the provisions in Section 4(c) above shall control to the extent they are in conflict with the discretionary provisions of this Section 7, *provided*, *however*, that nothing in this Section 7(c) or Section 4(c) above shall be construed as providing any Participant or any beneficiary of an Award any rights with respect to the "time value," "economic opportunity" or "intrinsic value" of an Award or limiting in any manner the Committee's actions that may be taken with respect to an Award as set forth in this Section 7 or in Section 4(c) above; and (ii) no action shall be taken under this Section 7 which shall cause an Award to result in taxation under Section 409A, to the extent applicable to such Award.

SECTION 8. General Provisions.

- (a) No Rights to Award. No Person shall have any claim to be granted any Award under the Plan, and there is no obligation for uniformity of treatment of Participants, including the treatment upon termination of Service. The terms and conditions of Awards need not be the same with respect to each recipient.
- (b) <u>Tax Withholding</u>. Unless other arrangements have been made that are acceptable to the Company, the Company or any Affiliate thereof is authorized to deduct or withhold, or cause to be deducted or withheld, from any Award, from any payment due or transfer made under any Award, or from any compensation or other amount owing to a Participant the amount (in cash or Units, including Units that would otherwise be issued pursuant to such Award or other property) of any applicable taxes payable in respect of an Award, including its grant, its exercise, the lapse of restrictions thereon, or any payment or transfer thereunder or under the Plan, and to take such other action as may be necessary in the opinion of the Company to satisfy its withholding obligations for the payment of such taxes. In the event that Units that would otherwise be issued pursuant to an Award are used to satisfy such withholding obligations, the number of Units which may be so withheld or surrendered shall be limited to the number of Units which have a Fair Market Value on the date of withholding equal to the aggregate amount of such liabilities based on the minimum statutory withholding rates for federal, state, local and foreign income tax and payroll tax purposes that are applicable to such supplemental taxable income.
- (c) No Right to Employment or Services. The grant of an Award shall not be construed as giving a Participant the right to be retained in the employ of the Company, the Partnership or any of their Affiliates, continue consulting services or to remain on the Board, as applicable. Furthermore, the Company, the Partnership and/or an Affiliate thereof may at any time dismiss a

Participant from employment or consulting free from any liability or any claim under the Plan, unless otherwise expressly provided in the Plan, any Award Agreement or other written agreement between any such entity and the Participant.

- (d) No Rights as Unitholder. Except as otherwise provided herein, a Participant shall have none of the rights of a unitholder with respect to Units covered by any Award unless and until the Participant becomes the record owner of such Units.
- (e) Section 409A. To the extent that the Committee determines that any Award granted under the Plan is subject to Section 409A, the Award Agreement evidencing such Award shall include the terms and conditions required by Section 409A. To the extent applicable, the Plan and Award Agreements shall be interpreted in accordance with Section 409A. Notwithstanding any provision of the Plan to the contrary, in the event that following the Effective Date (as defined in Section 9 below), the Committee determines that any Award may be subject to Section 409A, the Committee may adopt such amendments to the Plan and the applicable Award Agreement, adopt other policies and procedures (including amendments, policies and procedures with retroactive effect), and/or take any other actions that the Committee determines are necessary or appropriate to preserve the intended tax treatment of the Award, including without limitation, actions intended to (i) exempt the Award from Section 409A, or (ii) comply with the requirements of Section 409A; provided, however, that nothing herein shall create any obligation on the part of the Committee, the Partnership, the Company or any of their Affiliates to adopt any such amendment, policy or procedure or take any such other action, nor shall the Committee, the Partnership, the Company or any of their Affiliates have any liability for failing to do so. Notwithstanding any provision in the Plan to the contrary, the time of payment with respect to any Award that is subject to Section 409A shall not be accelerated, except as permitted under Treasury Regulation Section 1.409A-3(j)(4).
- (f) <u>Lock-Up Agreement</u>. Each Participant shall agree, if so requested by the Company or the Partnership and any underwriter in connection with any public offering of securities of the Partnership or any Affiliate, not to directly or indirectly offer, sell, contract to sell, sell any option or contract to purchase, purchase any option or contract to sell, grant any option, right or warrant for the sale of or otherwise dispose of or transfer any Units held by it for such period, not to exceed one hundred eighty (180) days following the effective date of the relevant registration statement filed under the Securities Act in connection with such public offering, as such underwriter shall specify reasonably and in good faith. The Company or the Partnership may impose stop-transfer instructions with respect to securities subject to the foregoing restrictions until the end of such 180-day period. Notwithstanding the foregoing, the 180-day period may be extended in the discretion of the Company for up to such number of additional days as is deemed necessary by such underwriter or the Company or Partnership to continue coverage by research analysts in accordance with FINRA Rule 2711 or any successor or other applicable rule.
- (g) Compliance with Laws. The Plan, the granting and vesting of Awards under the Plan and the issuance and delivery of Units and the payment of money under the Plan or under Awards granted or awarded hereunder are subject to compliance with all applicable federal, state, local and foreign laws, rules and regulations (including but not limited to state, federal and foreign securities law and margin requirements), the rules of any securities exchange or automated quotation system on which the Units are listed, quoted or traded, and to such approvals by any listing, regulatory or governmental authority as may, in the opinion of counsel for the Company or the Partnership, be necessary or advisable in connection therewith. Any securities delivered under the Plan shall be subject to such restrictions, and the Person acquiring such securities shall, if requested by the Company or the Partnership, provide such assurances and representations to the Company or the Partnership as the Company or the Partnership may deem necessary or desirable to assure compliance with all applicable legal requirements. To the extent permitted by applicable law, the Plan and Awards granted or awarded hereunder shall be deemed amended to the extent necessary to conform to such laws, rules and regulations. In the event an Award is granted to or held by a Participant who is employed or providing services outside the United States, the Committee may, in its sole discretion, modify the provisions of the Plan or of such Award as they pertain to such Participant to comply with applicable foreign law or to recognize differences in local law, currency or tax policy. The Committee may also impose conditions on the grant, issuance, exercise, vesting, settlement or retention of Awards in order to comply with such foreign law and/or to minimize the Company's or the Partnership's obligations with respect to tax equalization for Participants employed outside their home country.
- (h) <u>Governing Law</u>. The validity, construction, and effect of the Plan and any rules and regulations relating to the Plan shall be determined in accordance with the laws of the State of Delaware without regard to its conflicts of laws principles.
- (i) <u>Severability</u>. If any provision of the Plan or any Award is or becomes, or is deemed to be, invalid, illegal, or unenforceable in any jurisdiction or as to any Person or Award, or would disqualify the Plan or any Award under any law deemed applicable by the Committee, such provision shall be construed or deemed amended to conform to the applicable law or, if it cannot be construed or deemed amended without, in the determination of the Committee, materially altering the intent of the Plan or the Award, such provision shall be stricken as to such jurisdiction, Person or Award and the remainder of the Plan and any such Award shall remain in full force and effect.
- (j) Other Laws. The Committee may refuse to issue or transfer any Units or other consideration under an Award if, in its sole discretion, it determines that the issuance or transfer of such Units or such other consideration might violate any applicable law or regulation, the rules of the principal securities exchange on which the Units are then traded, or entitle the Partnership or an Affiliate to recover the same under Section 16(b) of the Exchange Act, and any payment tendered to the Company by a

Participant, other holder or beneficiary in connection with the exercise of such Award shall be promptly refunded to the relevant Participant, holder or beneficiary.

- (k) No Trust or Fund Created. Neither the Plan nor any Award shall create or be construed to create a trust or separate fund of any kind or a fiduciary relationship between the Company, the Partnership or any of their Affiliates, on the one hand, and a Participant or any other Person, on the other hand. To the extent that any Person acquires a right to receive payments pursuant to an Award, such right shall be no greater than the right of any general unsecured creditor of the Partnership or any participating Affiliate of the Partnership.
- (l) <u>No Fractional Units</u>. No fractional Units shall be issued or delivered pursuant to the Plan or any Award, and the Committee shall determine whether cash, other securities, or other property shall be paid or transferred in lieu of any fractional Units or whether such fractional Units or any rights thereto shall be canceled, terminated, or otherwise eliminated.
- (m) <u>Headings</u>. Headings are given to the Sections and subsections of the Plan solely as a convenience to facilitate reference. Such headings shall not be deemed in any way material or relevant to the construction or interpretation of the Plan or any provision hereof.
- (n) No Guarantee of Tax Consequences. None of the Board, the Committee, the Company or the Partnership provides or has provided any tax advice to any Participant or any other Person or makes or has made any assurance, commitment or guarantee that any federal, state or local tax treatment will (or will not) apply or be available to any Participant or other Person.
- (o) <u>Clawback</u>. To the extent required by applicable law or any applicable securities exchange listing standards, or as otherwise determined by the Committee, Awards and amounts paid or payable pursuant to or with respect to Awards shall be subject to the provisions of any clawback policy implemented by the Company, which clawback policy may provide for forfeiture, repurchase and/or recoupment of Awards and amounts paid or payable pursuant to or with respect to Awards. Notwithstanding any provision of this Plan or any Award Agreement to the contrary, the Company reserves the right, without the consent of any Participant, to adopt any such clawback policies and procedures, including such policies and procedures applicable to this Plan or any Award Agreement with retroactive effect.
- (p) <u>Facility Payment</u>. Any amounts payable hereunder to any Person under legal disability or who, in the judgment of the Committee, is unable to manage properly his or her financial affairs, may be paid to the legal representative of such Person, or may be applied for the benefit of such Person in any manner that the Committee may select, and the Partnership, the Company and all of their Affiliates shall be relieved of any further liability for payment of such amounts.

SECTION 9. Term of the Plan.

The Plan shall be effective on the date on which the Plan is adopted by the Board (the "<u>Effective Date</u>") and shall continue until the date terminated by the Board. However, any Award granted prior to such termination, and the authority of the Board or the Committee to amend, alter, adjust, suspend, discontinue, or terminate any such Award or to waive any conditions or rights under such Award, shall extend beyond such termination date. The Plan shall, within twelve (12) months after the date of the Board's initial adoption of the Plan, be submitted for approval by a majority of the outstanding Units of the Partnership entitled to vote.

QEP MIDSTREAM PARTNERS, LP 2013 LONG-TERM INCENTIVE PLAN PHANTOM UNIT AGREEMENT

Pursuant to this Phantom Unit Agreement, dated as of [], 2013 (this "Agreement"), QEP Midstream Partners GP, LLC (the "Company"), as the general partner of QEP Midstream Partners, LP (the "Partnership"), hereby grants to [] (the "Participant") the following award of Phantom Units ("Phantom Units"), pursuant and subject to the terms and conditions of this Agreement and the QEP Midstream Partners, LP 2013 Long-Term Incentive Plan (the "Plan"), the terms and conditions of which are hereby incorporated into this Agreement by reference. Each Phantom Unit shall constitute a Phantom Unit under the terms of the Plan and is hereby granted in tandem with a corresponding DER, as further detailed in Section 3 below. Except as otherwise expressly provided herein, all capitalized terms used in this Agreement, but not defined, shall have the meanings provided in the Plan.

GRANT NOTICE

Subject to the terms and conditions of this Agreement, the principal features of this Award are as follows:

Number of Phantom Units: [] Phantom Units

Grant Date: [], 2013

Vesting of Phantom Units: [Insert time-based vesting schedule].

Forfeiture of Phantom Units: In the event of a cessation of the Participant's Service for any reason, all Phantom Units that have not vested prior to or in connection with such cessation of Service shall, subject to <u>Section 4(b)</u> below, thereupon automatically be forfeited by the Participant without further action and for no consideration.

Payment of Phantom Units: Vested Phantom Units shall be paid to the Participant in the form of Units as set forth in Section 5 below.

DERs: Each Phantom Unit granted under this Agreement shall be issued in tandem with a corresponding dividend equivalent right, or DER, which shall entitle the Participant to receive payments in an amount equal to Partnership distributions in accordance with <u>Section 3</u> below.

TERMS AND CONDITIONS OF PHANTOM UNITS

- 1. <u>Grant</u>. The Company hereby grants to the Participant, as of the Grant Date, an award of [] Phantom Units, subject to all of the terms and conditions contained in this Agreement and the Plan.
- 2. <u>Phantom Units</u>. Subject to <u>Section 4</u> below, each Phantom Unit that vests shall represent the right to receive payment, in accordance with <u>Section 5</u> below, in the form of one (1) Unit. Unless and until a Phantom Unit vests, the Participant will have no right to payment in respect of such Phantom Unit. Prior to actual payment in respect of any vested Phantom Unit, such Phantom Unit will represent an unsecured obligation of the Partnership, payable (if at all) only from the general assets of the Partnership.
- 3. <u>Grant of Tandem DER</u>. Each Phantom Unit granted hereunder is hereby granted in tandem with a corresponding DER, which shall remain outstanding from the Grant Date until the earlier of the payment or forfeiture of the related Phantom Unit. Each DER shall entitle the Participant to receive payments, subject to and in accordance with this Agreement, in an amount equal to any distributions made by the Partnership in respect of the Unit underlying the Phantom Unit to which such DER relates, regardless of whether or not such Phantom Unit is vested.
 - 4. Vesting and Forfeiture.
- (a) *Vesting*. Subject to <u>Section 4(c)</u> below, the Phantom Units shall vest in such amounts and at such times as are set forth in the Grant Notice above.
- (b) Accelerated Vesting. Subject to Section 4(c) below, the Phantom Units shall vest in full upon the occurrence of any of the following events: [Insert circumstances resulting in accelerated vesting, if any].
- (c) *Forfeiture*. Notwithstanding the foregoing, in the event of a cessation of the Participant's Service for any reason, all Phantom Units that have not vested prior to or in connection with such cessation of Service shall thereupon automatically be forfeited by the Participant without further action and without payment of consideration therefor. No portion

of the Phantom Units which has not become vested at the date of the Participant's cessation of Service shall thereafter become vested.

(d) *Payment*. Vested Phantom Units shall be subject to the payment provisions set forth in Section 5 below.

5. Payment of Phantom Units and DERs.

- (a) *Phantom Units*. Unpaid, vested Phantom Units shall be paid to the Participant in the form of Units in a lump-sum as soon as reasonably practical, but not later than sixty (60) days, following the date on which such Phantom Units vest. Payments of any Phantom Units that vest in accordance herewith shall be made to the Participant (or in the event of the Participant's death, to the Participant's estate) in whole Units in accordance with this Section 5. In lieu of the foregoing, the Committee may elect at its discretion to pay the Phantom Units in cash equal to the Fair Market Value of the Units that would otherwise be distributed as of the date of vesting.
- (b) *DERs*. Payments in respect of DERs shall be made to the Participant promptly upon the making of the corresponding distributions by the Partnership to its common unitholders.
- (c) *Potential Delay.* Notwithstanding anything to the contrary in this Agreement, no amounts payable under this Agreement shall be paid to the Participant prior to the expiration of the six (6)-month period following his "separation from service" (within the meaning of Treasury Regulation Section 1.409A-1(h)) (a "**Separation from Service**") to the extent that the Company determines that paying such amounts prior to the expiration of such six (6)-month period would result in a prohibited distribution under Section 409A(a)(2)(B)(i) of the Code. If the payment of any such amounts is delayed as a result of the previous sentence, then on the first business day following the end of the applicable six (6)-month period (or such earlier date upon which such amounts can be paid under Section 409A of the Code without resulting in a prohibited distribution, including as a result of the Participant's death), such amounts shall be paid to the Participant.
- 6. <u>Tax Withholding</u>. The Company and/or its Affiliates shall have the authority and the right to deduct or withhold, or to require the Participant to remit to the Company and/or its Affiliates, an amount sufficient to satisfy all applicable federal, state and local taxes (including the Participant's employment tax obligations) required by law to be withheld with respect to any taxable event arising in connection with the Phantom Units and the DERs. In satisfaction of the foregoing requirement, unless otherwise determined by the Committee, the Company and/or its Affiliates shall withhold Units otherwise issuable in respect of such Phantom Units having a Fair Market Value equal to the sums required to be withheld. In the event that Units that would otherwise be issued in payment of the Phantom Units are used to satisfy such withholding obligations, the number of Units which shall be so withheld shall be limited to the number of Units which have a Fair Market Value (which, in the case of a broker-assisted transaction, shall be determined by the Committee, consistent with applicable provisions of the Code) on the date of withholding equal to the aggregate amount of such liabilities based on the minimum statutory withholding rates for federal, state, local and foreign income tax and payroll tax purposes that are applicable to such supplemental taxable income.
- 7. <u>Rights as Unit Holder</u>. Neither the Participant nor any person claiming under or through the Participant shall have any of the rights or privileges of a holder of Units in respect of any Units that may become deliverable hereunder unless and until certificates representing such Units shall have been issued or recorded in book entry form on the records of the Partnership or its transfer agents or registrars, and delivered in certificate or book entry form to the Participant or any person claiming under or through the Participant.
- 8. Non-Transferability. Neither the Phantom Units nor any right of the Participant under the Phantom Units may be assigned, alienated, pledged, attached, sold or otherwise transferred or encumbered by the Participant (or any permitted transferee) other than by will or the laws of descent and distribution and any such purported assignment, alienation, pledge, attachment, sale, transfer or encumbrance shall be void and unenforceable against the Company, the Partnership and any of their Affiliates.
- 9. <u>Distribution of Units</u>. Unless otherwise determined by the Committee or required by any applicable law, rule or regulation, neither the Company nor the Partnership shall deliver to the Participant certificates evidencing Units issued pursuant to this Agreement and instead such Units shall be recorded in the books of the Partnership (or, as applicable, its transfer agent or equity plan administrator). All certificates for Units issued pursuant to this Agreement and all Units issued pursuant to book entry procedures hereunder shall be subject to such stop transfer orders and other restrictions as the Company may deem advisable under the Plan or the rules, regulations, and other requirements of the Securities Exchange Commission, any stock exchange upon which such Units are then listed, and any applicable federal or state laws, and the Company may cause a legend or legends to be inscribed on any such certificates or book entry to make appropriate reference to such restrictions. In addition to the terms and conditions provided herein, the Company may require that the Participant make such covenants, agreements, and representations as the Company, in its sole discretion, deems advisable in order to comply with any such laws, regulations, or requirements. No fractional Units shall be issued or delivered pursuant to the Phantom Units and the Committee shall determine whether cash, other securities, or other property shall be paid

or transferred in lieu of fractional Units or whether such fractional Units or any rights thereto shall be canceled, terminated, or otherwise eliminated.

- 10. <u>Partnership Agreement</u>. Units issued upon payment of the Phantom Units shall be subject to the terms of the Plan and the Partnership Agreement. Upon the issuance of Units to the Participant, the Participant shall, automatically and without further action on his or her part, (i) be admitted to the Partnership as a Limited Partner (as defined in the Partnership Agreement) with respect to the Units, and (ii) become bound, and be deemed to have agreed to be bound, by the terms of the Partnership Agreement.
- 11. No Effect on Service. Nothing in this Agreement or in the Plan shall be construed as giving the Participant the right to be retained in the employ or service of the Company or any Affiliate thereof. Furthermore, the Company and its Affiliates may at any time dismiss the Participant from employment or consulting free from any liability or any claim under the Plan or this Agreement, unless otherwise expressly provided in the Plan, this Agreement or any other written agreement between the Participant and the Company or an Affiliate thereof.
- 12. <u>Severablility</u>. If any provision of this Agreement is or becomes or is deemed to be invalid, illegal, or unenforceable in any jurisdiction, such provision shall be construed or deemed amended to conform to the applicable law or, if it cannot be construed or deemed amended without, in the determination of the Committee, materially altering the intent of this Agreement, such provision shall be stricken as to such jurisdiction, and the remainder of this Agreement shall remain in full force and effect.
- 13. <u>Tax Consultation</u>. None of the Board, the Committee, the Company nor the Partnership has made any warranty or representation to Participant with respect to the income tax consequences of the issuance of the Phantom Units, the DERs, the Units or the transactions contemplated by this Agreement, and the Participant represents that he or she is in no manner relying on such entities or their representatives for tax advice or an assessment of such tax consequences. The Participant understands that the Participant may suffer adverse tax consequences in connection with the Phantom Units and DERs granted pursuant to this Agreement. The Participant represents that the Participant has consulted with any tax consultants that the Participant deems advisable in connection with the Phantom Units and DERs.
- 14. <u>Amendments, Suspension and Termination</u>. To the extent permitted by the Plan, this Agreement may be wholly or partially amended or otherwise modified, suspended or terminated at any time or from time to time by the Board or the Committee. Except as provided in the preceding sentence, this Agreement cannot be modified, altered or amended, except by an agreement, in writing, signed by both the Partnership and the Participant.
- 15. <u>Lock-Up Agreement</u>. The Participant shall agree, if so requested by the Company or the Partnership and any underwriter in connection with any public offering of securities of the Partnership or any Affiliate thereof, not to directly or indirectly offer, sell, contract to sell, sell any option or contract to purchase, purchase any option or contract to sell, grant any option, right or warrant for the sale of or otherwise dispose of or transfer any Units held by him or her for such period, not to exceed one hundred eighty (180) days following the effective date of the relevant registration statement filed under the Securities Act in connection with such public offering, as such underwriter shall specify reasonably and in good faith. The Company or the Partnership may impose stop-transfer instructions with respect to securities subject to the foregoing restrictions until the end of such 180-day period. Notwithstanding the foregoing, the 180-day period may be extended in the discretion of the Company for up to such number of additional days as is deemed necessary by such underwriter or the Company or Partnership to continue coverage by research analysts in accordance with FINRA Rule 2711 or any successor or other applicable rule.
- 16. <u>Conformity to Securities Laws</u>. The Participant acknowledges that the Plan and this Agreement are intended to conform to the extent necessary with all provisions of the Securities Act and the Exchange Act, any and all regulations and rules promulgated by the Securities and Exchange Commission thereunder, and all applicable state securities laws and regulations. Notwithstanding anything herein to the contrary, the Plan shall be administered, and the Phantom Units and DERs are granted, only in such a manner as to conform to such laws, rules and regulations. To the extent permitted by applicable law, the Plan and this Agreement shall be deemed amended to the extent necessary to conform to such laws, rules and regulations.
- 17. <u>Code Section 409A</u>. None of the Phantom Units, the DERs or any amounts paid pursuant to this Agreement are intended to constitute or provide for a deferral of compensation that is subject to Section 409A of the Code. Nevertheless, to the extent that the Committee determines that the Phantom Units or DERs may not be exempt from (or compliant with) Section 409A of the Code, the Committee may (but shall not be required to) amend this Agreement in a manner intended to comply with the requirements of Section 409A of the Code or an exemption therefrom (including amendments with retroactive effect), or take any other actions as it deems necessary or appropriate to (a) exempt the Phantom Units or DERs from Section 409A of the Code and/or preserve the intended tax treatment of the benefits provided with respect to the Phantom Units or DERs, or (b) comply with the requirements of Section 409A of the Code. To the extent applicable, this Agreement shall be

interpreted in accordance with the provisions of Section 409A of the Code. Notwithstanding anything in this Agreement to the contrary, to the extent that any payment or benefit hereunder constitutes non-exempt "nonqualified deferred compensation" for purposes of Section 409A of the Code, and such payment or benefit would otherwise be payable or distributable hereunder by reason of the Participant's cessation of Service, all references to the Participant's cessation of Service shall be construed to mean a Separation from Service, and the Participant shall not be considered to have a cessation of Service unless such cessation constitutes a Separation from Service with respect to the Participant.

- 18. <u>Adjustments; Clawback</u>. The Participant acknowledges that the Phantom Units are subject to modification and forfeiture in certain events as provided in this Agreement and Section 7 of the Plan. The Participant further acknowledges that the Phantom Units, DERs and Units issuable hereunder, whether vested or unvested and whether or not previously issued, are subject to clawback as provided in Section 8(o) of the Plan.
- 19. <u>Successors and Assigns</u>. The Company or the Partnership may assign any of its rights under this Agreement to single or multiple assignees, and this Agreement shall inure to the benefit of the successors and assigns of the Company and the Partnership. Subject to the restrictions on transfer contained herein, this Agreement shall be binding upon the Participant and his or her heirs, executors, administrators, successors and assigns.
- 20. <u>Governing Law</u>. The validity, construction, and effect of this Agreement and any rules and regulations relating to this Agreement shall be determined in accordance with the laws of the State of Delaware without regard to its conflicts of laws principles.
- 21. <u>Headings</u>. Headings are given to the sections and subsections of this Agreement solely as a convenience to facilitate reference. Such headings shall not be deemed in any way material or relevant to the construction or interpretation of this Agreement or any provision hereof.

[Signature page follows]

The Participant's signature below indicates the Participant's agreement with and understanding that this award is subject to all of the terms and conditions contained in the Plan and in this Agreement, and that, in the event that there are any inconsistencies between the terms of the Plan and the terms of this Agreement, the terms of the Plan shall control. The Participant further acknowledges that the Participant has read and understands the Plan and this Agreement, which contains the specific terms and conditions of this grant of Phantom Units. The Participant hereby agrees to accept as binding, conclusive and final all decisions or interpretations of the Committee upon any questions arising under the Plan or this Agreement.

By:
Name:
Title:

QEP Midstream Partners, LP
a Delaware limited partnership

By:
QEP Midstream Partners GP, LLC
Its:
General Partner

By:
Title:

QEP Midstream Partners GP, LLC

"PARTICIPANT"
[Name]

OMNIBUS AGREEMENT

This Omnibus Agreement ("Agreement") is entered into on, and effective as of, August 14, 2013 (the "Closing Date") among QEP Resources, Inc., a Delaware corporation ("QEP"), QEP Field Services Company, a Delaware corporation ("Field Services"), QEP Midstream Partners, LP, a Delaware limited partnership (the "Partnership"), QEP Midstream Partners GP, LLC, a Delaware limited liability company and the general partner of the Partnership (the "General Partner"), and QEP Midstream Partners Operating, LLC, a Delaware limited liability company and wholly owned subsidiary of the Partnership (the "Operating Company" and, together with QEP, Field Services, the Partnership and the General Partner, the "Parties" and each a "Party").

RECITALS

- 1. The Parties desire by their execution of this Agreement to evidence their understanding, as more fully set forth in <u>Article II</u>, with respect to certain indemnification obligations of the Parties to each other.
- 2. The Parties desire by their execution of this Agreement to evidence their understanding, as more fully set forth in Article III, with respect to the amount to be paid by the Partnership for the centralized corporate services to be performed by QEP and its Affiliates for and on behalf of the Partnership Group.
- 3. The Parties desire by their execution of this Agreement to evidence their understanding, as more fully set forth in <u>Article IV</u>, with respect to the granting of certain licenses between the Parties.

In consideration of the premises and the covenants, conditions, and agreements contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties agree as follows:

ARTICLE I

Definitions and Interpretation

- 1.1 <u>Definitions.</u> In addition to the terms defined in the introductory paragraph and the recitals of this Agreement, for purposes hereof, the capitalized terms used herein and not otherwise defined shall have the meanings set forth in <u>Appendix A</u>.
- 1.2 <u>Rules of Construction.</u> Unless expressly provided for elsewhere in this Agreement, this Agreement shall be interpreted in accordance with the following provisions:
 - (a) If a word or phrase is defined, its other grammatical forms have a corresponding meaning.
 - (b) The headings contained in this Agreement are for reference purposes only and shall not affect the meaning or interpretation of this Agreement.

- (c) A reference to any Party to this Agreement or another agreement or document includes the Party's successors and assigns.
- (d) The words "hereof," "herein" and "hereunder" and words of similar import when used in this Agreement shall refer to this Agreement as a whole and not to any particular provision of this Agreement, and article, section, subsection and schedule references are to this Agreement unless otherwise specified.
 - (e) The words "including," "include," "includes" and all variations thereof shall mean "including without limitation."
 - (f) The word "or" shall have the inclusive meaning represented by the phrase "and/or."
 - (g) The words "shall" and "will" have equal force and effect.
 - (h) The schedules identified in this Agreement are incorporated herein by reference and made a part of this Agreement.
 - (i) References to "\$" or to "dollars" shall mean the lawful currency of the United States of America.

ARTICLE II Indemnification

2.1 Environmental Indemnification.

- (a) Each of QEP and Field Services, jointly and severally, shall indemnify, defend and hold harmless each Group Member from and against any Losses suffered or incurred by such Group Member, directly or indirectly, by reason of or arising out of:
- (i) any violation of Environmental Laws as in effect prior to the Closing Date and such violation commenced, occurred or existed before the Closing Date;
- (ii) any environmental event, condition or matter associated with or arising from the ownership or operation of the Assets that commenced, occurred or existed before the Closing Date (including the presence of Hazardous Substances on, under, about or migrating to or from the Assets or the disposal or the release of Hazardous Substances generated by operation of the Assets at non-Asset locations), including (A) the cost and expense of any investigation, assessment, evaluation, monitoring, containment, cleanup, repair, restoration, remediation, risk-based closure activities, or other corrective action required or necessary under Environmental Laws and (B) the cost and expense of the preparation and implementation of any closure, remedial, corrective action, or other plans required or necessary under Environmental Laws as in effect prior to the Closing Date; and

(iii) any environmental event, condition or matter associated with or arising from the Retained Assets, whether occurring before, on or after the Closing Date and whether occurring under Environmental Laws as in effect prior to, at or after the Closing Date (clauses (i), (ii) and (iii) being referred to collectively as "Covered Environmental Losses");

provided, *however*, that QEP and Field Services will be obligated to indemnify such Group Member for a Covered Environmental Loss described in Section 2.1(a)(i) or Section 2.1(a)(ii) only to the extent that QEP is notified in writing of such violation, event, condition or environmental matter prior to the Identification Deadline.

- (b) The Partnership shall indemnify, defend and hold harmless QEP and Field Services from and against any Losses suffered or incurred by any of the QEP Entities, directly or indirectly, by reason of or arising out of:
- (i) any violation of Environmental Laws as in effect on or after the Closing Date and such violation is associated with or arises from the ownership or operation of the Assets on or after the Closing Date; and
- (ii) any environmental event, condition or matter associated with or arising from the ownership or operation of the Assets on or after the Closing Date (including the presence of Hazardous Substances on, under, about or migrating to or from the Assets or the disposal or the release of Hazardous Substances generated by operation of the Assets at Asset locations) including (A) the cost and expense of any investigation, assessment, evaluation, monitoring, containment, cleanup, repair, restoration, remediation, risk-based closure activities, or other corrective action required or necessary under Environmental Laws in effect on or after the Closing Date, and (B) the cost or expense of the preparation and implementation of any closure, remedial, corrective action, or other plans required or necessary under Environmental Laws as in effect on or after the Closing Date;

and regardless of whether such violation under <u>Section 2.1(b)(ii)</u> or such environmental event, condition or matter included under <u>Section 2.1(b)(ii)</u> occurred before or after the Closing Date, in each case, to the extent that any of the foregoing are not Covered Environmental Losses (without giving effect to the Environmental Deductible).

2.2 <u>Right of Way Indemnification</u>. Each of QEP and Field Services, jointly and severally, shall indemnify, defend and hold harmless each Group Member from and against any Losses suffered or incurred by such Group Member by reason of or arising out of (a) the failure of such Group Member to be the owner of such valid and indefeasible easement rights or fee ownership or leasehold interests in and to the lands on which any of the Assets conveyed or contributed to such Group Member on the Closing Date is located as of the Closing Date, and such failure renders such Group Member liable to a third party or unable to use or operate the Assets in substantially the same manner that the Assets were used and operated as of immediately prior to the Closing Date; (b) the failure of such Group Member to have the consents, licenses and permits necessary to allow (1) any pipeline included in the Assets to cross the roads, waterways, railroads and other areas upon which any such pipeline is located as of the Closing Date, or (2) the transfer of any of the Assets to the Partnership Group, in each case, where such failure renders the Partnership Group liable to a third party or unable to use or operate the Assets in substantially the same manner that the Assets were used and operated as of

immediately prior to the Closing Date; and (c) the cost of curing any condition set forth in <u>Section 2.2(a)</u> or (b) that does not allow any Asset to be operated in accordance with prudent industry practice, in each case to the extent that QEP and Field Services are notified in writing of any of the foregoing prior to the Identification Deadline.

2.3 Additional Indemnification.

- (a) Each of QEP and Field Services, jointly and severally, shall indemnify, defend, and hold harmless each Group Member from and against any Losses suffered or incurred by such Group Member by reason of or arising out of:
- (i) the consummation of the transactions contemplated by the Contribution Agreement. For the avoidance of doubt, the Parties agree that, subject to the qualifications set forth on Schedule A, each Group Member shall be entitled to indemnification by QEP and Field Services under this Section 2.3(a)(i) for those litigation matters listed on Schedule A;
 - (ii) events and conditions associated with the Retained Assets, whether occurring before, on or after the Closing Date;
- (iii) all federal, state and local tax liabilities attributable to the ownership or operation of the Assets on or prior to the Closing Date, including under Treasury Regulation Section 1.1502-6, as it may be amended (or any similar provision of state or local law), and any such tax liabilities that may result from the consummation of the formation transactions for the Partnership Group and the General Partner occurring prior to the Closing Date or from the consummation of the transactions contemplated by the Contribution Agreement;
- (iv) the failure of any Group Member to have on the Closing Date any consent, license, permit or approval necessary to allow such Group Member to own or operate the Assets in substantially the same manner that the Assets were owned or operated immediately prior to the Closing Date; and
- (v) events and conditions in the ordinary course of business that result in the release of a Hazardous Substance from, or failures in the integrity of, the Mesa Condensate Trunk Line, including any cost and expense incurred by a Group Member with respect to any investigation, assessment, evaluation, monitoring, containment, cleanup, repair, restoration, replacement, remediation, risk-based closure activities, or other corrective action required or necessary pursuant to (A) applicable laws, including Environmental Laws, or (B) the Partnership's good faith determination as a reasonably prudent operator; *provided*, *however*, that such events or conditions that resulted in the release of a Hazardous Substance from, or caused a failure in the integrity of, the Mesa Condensate Trunk Line commenced, occurred or existed prior to the Identification Deadline; *provided*, *further* that the indemnification provided by QEP and Field Services in this Section 2.3(a)(y) shall terminate upon the earlier of (X) the Identification Deadline and (Y) in the event that QEP or Field Services elects, in their sole discretion, to replace the Mesa Condensate Trunk Line in its entirety, the date of such replacement.

(b) The Partnership shall indemnify, defend, and hold harmless QEP and Field Services from and against any Losses suffered or incurred by any of the QEP Entities by reason of or arising out of events and conditions to the extent associated with the ownership or operation of the Assets and occurring after the Closing Date (other than Covered Environmental Losses which are provided for under Section 2.1(a) and Losses for which the Partnership is indemnifying QEP under Section 2.1(b)), unless such indemnification would not be permitted by any Group Member under the Partnership Agreement.

2.4 Indemnification Procedures.

- (a) The Indemnified Party agrees that within a reasonable period of time after it becomes aware of facts giving rise to a claim for indemnification under this <u>Article II</u>, it will provide notice thereof in writing to the Indemnifying Party, specifying the nature of and specific basis for such claim.
- (b) The Indemnifying Party shall have the right to control all aspects of the defense of (and any counterclaims with respect to) any claims brought against the Indemnified Party that are covered by the indemnification under this <u>Article II</u>, including, without limitation, the selection of counsel, determination of whether to appeal any decision of any court and the settling of any such claim or any matter or any issues relating thereto; *provided*, *however*, that no such settlement for only the payment of money shall be entered into without the consent of the Indemnified Party unless it includes a full release of the Indemnified Party from such claim; *provided further*, that no such settlement containing any form of injunctive or similar relief shall be entered into without the prior written consent of the Indemnified Party, which consent shall not be unreasonably delayed or withheld.
- (c) The Indemnified Party agrees to cooperate in good faith and in a commercially reasonable manner with the Indemnifying Party, with respect to all aspects of the defense of and pursuit of any counterclaims with respect to any claims covered by the indemnification under this Article II, including, without limitation, the prompt furnishing to the Indemnifying Party of any correspondence or other notice relating thereto that the Indemnified Party may receive, permitting the name of the Indemnified Party to be utilized in connection with such defense and counterclaims, the making available to the Indemnified Party that the Indemnifying Party considers relevant to such defense and counterclaims, the making available to the Indemnifying Party of any employees of the Indemnified Party and the granting to the Indemnifying Party of reasonable access rights to the properties and facilities of the Indemnified Party; provided, however, that in connection therewith the Indemnifying Party agrees to use reasonable efforts to minimize the impact thereof on the operations of the Indemnified Party and further agrees to maintain the confidentiality of all files, records, and other information furnished by the Indemnified Party pursuant to this Section 2.4. The obligation of the Indemnified Party to cooperate with the Indemnifying Party as set forth in the immediately preceding sentence shall not be construed as imposing upon the Indemnified Party an obligation to hire and pay for counsel in connection with the defense of and pursuit of any counterclaims with respect to any claims covered by the indemnification set forth in this Article II; provided, however, that the Indemnified Party may, at its own option, cost and expense, hire and pay for counsel in connection with any such defense and counterclaims. The Indemnifying Party agrees to keep any such counsel hired by the Indemnified Party informed as

to the status of any such defense or counterclaim, but the Indemnifying Party shall have the right to retain sole control over such defense and counterclaims so long as the Indemnified Party is still seeking indemnification hereunder.

(d) In determining the amount of any loss, cost, damage or expense for which the Indemnified Party is entitled to indemnification under this Agreement, the gross amount of the indemnification will be reduced by (i) any insurance proceeds realized by the Indemnified Party, and such correlative insurance benefit shall be net of any incremental insurance premium that becomes due and payable by the Indemnified Party as a result of such claim and (ii) all amounts recovered by the Indemnified Party under contractual indemnities from third Persons.

2.5 <u>Limitations Regarding Indemnification</u>.

- (a) With respect to Covered Environmental Losses under Section 2.1(a)(i) or Section 2.1(a)(ii) that arise out of an event, condition or matter that is first discovered after the Closing Date, neither QEP nor Field Services shall be obligated to indemnify, defend and hold harmless any Group Member until such time as the total aggregate amount of Losses incurred by the Partnership Group for such Covered Environmental Losses exceeds \$500,000 (the "Environmental Deductible"), at which time QEP and Field Services shall be obligated to indemnify the Partnership Group for the excess of such Covered Environmental Losses over the Environmental Deductible. For the avoidance of doubt, it is agreed that the Environmental Deductible shall not apply to any Covered Environmental Losses incurred by any Group Member attributable to those locations identified on Schedule B.
- (b) For the avoidance of doubt, there is no deductible or cap with respect to the indemnification owed by any Indemnifying Party under any portion of this <u>Article II</u> other than that described in <u>Section 2.5(a)</u>.
- (c) For the avoidance of doubt, the obligation of QEP and Field Services to indemnify any Group Member, as specified in Section 2.1, Section 2.2 and Section 2.3 shall be limited to the extent of the Losses incurred by the Partnership with respect to its direct or indirect ownership interest in such Group Member.
- (d) NOTWITHSTANDING ANYTHING HEREIN TO THE CONTRARY, IN NO EVENT SHALL ANY PARTY'S INDEMNIFICATION OBLIGATION HEREUNDER COVER OR INCLUDE CONSEQUENTIAL, INDIRECT, INCIDENTAL, PUNITIVE, EXEMPLARY, SPECIAL OR SIMILAR DAMAGES OR LOST PROFITS (INCLUDING ANY DIMINUTION IN VALUE OF ANY PARTY'S RESPECTIVE INVESTMENT IN THE PARTNERSHIP) SUFFERED, DIRECTLY OR INDIRECTLY, BY ANY OTHER PARTY ENTITLED TO INDEMNIFICATION UNDER THIS AGREEMENT, EXCEPT AS A REIMBURSEMENT FOR ANY SUCH DAMAGES AS ARE PAID TO A GOVERNMENTAL AUTHORITY OR OTHER THIRD PARTY.

ARTICLE III General and Administrative Services

3.1 <u>General</u>. QEP agrees to provide, and agrees to cause its Affiliates to provide, to the General Partner, for the Partnership Group's benefit, the centralized general and administrative services that QEP and its Affiliates have traditionally provided in connection with the ownership and operation of the Assets, which consist of the services set forth on <u>Schedule C</u> (the "General and Administrative Services"). Absent the written agreement of the Partnership Group, at the General Partner, for the benefit of the Partnership Group, at the General Partner's principal place of business.

3.2 Administrative Fee.

- (a) As consideration for QEP's and its Affiliates' provision of the General and Administrative Services, the Partnership Group will pay to QEP an annual fee that will reflect the costs incurred by QEP and its Affiliates in providing such General and Administrative Services (other than those costs for which QEP and its Affiliates are entitled to reimbursement pursuant to Section 3.3), as determined in good faith by QEP in accordance with Schedule C (the "Administrative Fee"). The Parties acknowledge and agree that it is the intent of the Parties that the General and Administrative Services be provided based on an arm's-length standard, and that the Administrative Fee is intended to reflect such standard. For the avoidance of doubt, the Parties further acknowledge and agree that the Administrative Fee will cover the fully burdened cost of the General and Administrative Services provided by QEP and its Affiliates to the Partnership Group, as well as any third party costs actually incurred by QEP and its Affiliates on behalf of the Partnership Group in providing such General and Administrative Services (other than those costs for which QEP and its Affiliates are entitled to reimbursement pursuant to Section 3.3), including the following:
- (i) the compensation and employee benefits of employees of QEP or its Affiliates (and any withholding or payroll taxes related thereto), to the extent, but only to the extent, such employees perform General and Administrative Services for the Partnership Group's benefit. For the avoidance of doubt, the Administrative Fee shall include any withholding and payroll related taxes paid by QEP and its Affiliates in connection with any long-term incentive plan of the General Partner or the Partnership Group. With respect to employees that do not devote all of their business time to the Partnership Group, such compensation and employee benefits (and any withholding or payroll taxes related thereto) shall be allocated to the Partnership Group based on the annual weighted average of time spent and number of employees devoting services to the Partnership Group;
- (ii) any expenses incurred or payments made by QEP or its Affiliates on behalf of the Partnership Group for insurance coverage with respect to the Assets or the business of the Partnership Group;
- (iii) all expenses and expenditures incurred by QEP or its Affiliates on behalf of the Partnership Group as a result of the Partnership becoming and continuing as a publicly traded entity, including, but not limited to, costs associated with annual, quarterly and current reporting; tax return and Schedule K-1 preparation and distribution expenses; Sarbanes-Oxley compliance expenses; expenses associated with listing on the NYSE; independent auditor fees; legal fees; investor relations expenses; registrar and transfer agent fees, outside director fees and director and officer insurance expenses; and

- (iv) all sales, use, excise, value added or similar taxes, if any, that may be applicable from time to time with respect to the services provided by QEP and its Affiliates to the Partnership Group pursuant to Section 3.1.
- (b) As part of the Administrative Fee, the Partnership Group shall pay to QEP a fixed fee, in the amount shown on <u>Schedule C</u>, in consideration for the services of certain employees of QEP and its Affiliates in their capacities as officers of the General Partner and the Group Members.
- (c) The Parties acknowledge and agree that the Administrative Fee may change each calendar year, as determined by QEP in good faith, to accurately reflect the degree and extent of the General and Administrative Services provided to the Partnership Group and may be adjusted to reflect, among other things, the contribution, acquisition or disposition of assets to or by the Partnership Group or to reflect any change in the cost of providing General and Administrative Services to the Partnership Group due to changes in any law, rule or regulation applicable to the QEP Entities or the Partnership Group, including any interpretation of such laws, rules or regulations.
- (d) On or prior to January 1 of each calendar year during the term of this Agreement, QEP will notify the General Partner of the estimated amount of the Administrative Fee (including both the fixed and variable portions of the Administrative Fee as described in Schedule C) to be paid by the Partnership Group for such calendar year. For the calendar year in which the Closing Date occurs, such estimate shall be made on or prior to the Closing Date and shall pertain only to the remainder of such calendar year. Commencing with the first full month following the Closing Date, the Administrative Fee shall be invoiced and paid as follows:
- (i) Within 20 days following the end of each month during the term of this Agreement, QEP will submit to the Partnership Group an invoice of the amounts due for such month for the Administrative Fee. Each invoice will contain reasonably satisfactory support of such amounts and such other supporting detail as the General Partner may reasonably require.
- (ii) The Partnership Group will pay the Administrative Fee within 10 days after the receipt of the invoice therefor. The Partnership Group shall not offset any amounts owing to it by QEP or any of its Affiliates against the Administrative Fee payable hereunder.

3.3 Reimbursement of Expenses.

- (a) In addition to the Administrative Fee payable under <u>Section 3.2</u>, the Partnership Group will reimburse QEP and its Affiliates for any additional out-of-pocket costs and expenses actually incurred by QEP and its Affiliates in providing the General and Administrative Services, as well as any other out-of-pocket expenses incurred on behalf of the Partnership Group.
- (b) The Partnership Group will reimburse QEP and its Affiliates for any costs and expenses incurred by QEP and its Affiliates under Section 3.3(a) as incurred on a monthly basis.

ARTICLE IV Licenses of Marks

- 4.1 <u>Grant of QEP License</u>. Upon the terms and conditions set forth in this <u>Article IV</u>, QEP hereby grants and conveys to the Partnership and each of the entities currently or hereafter comprising a part of the Partnership Group a nontransferable, nonexclusive, royaltyfree right and license (the "**QEP License**") to use the "**QEP**" logo and trademark and the other trademarks and tradenames owned by QEP (collectively, the "**QEP Marks**").
- 4.2 Ownership and Quality of QEP Marks. The Partnership, on behalf of itself and the other Group Members, agrees that ownership of the QEP Marks and the goodwill relating thereto shall remain vested in QEP during the term of the QEP License and thereafter. The Partnership agrees, and agrees to cause the other Group Members, to the fullest extent permitted by applicable law, never to challenge, contest or question the validity of QEP's ownership of the QEP Marks or any registration thereof by QEP. In connection with the use of the QEP Marks, the Partnership and any other Group Member shall not in any manner represent that they have any ownership in the QEP Marks or registration thereof. The Partnership, on behalf of itself and the other Group Members, acknowledges that the use of the QEP Marks shall not create any right, title or interest in or to the QEP Marks, and all use of the QEP Marks by the Partnership or any other Group Member shall inure to the benefit of QEP. The Partnership agrees, and agrees to cause the other Group Members, to use the QEP Marks in accordance with such quality standards established by QEP and communicated to the Partnership Group from time to time, it being understood that the products and services offered by the Group Members as of the Closing Date are of a quality that is acceptable to QEP.
- 4.3 <u>Grant of the Partnership License</u>. Upon the terms and conditions set forth in this <u>Article IV</u>, the General Partner, for the benefit of the Partnership, hereby grants and conveys to QEP and its Affiliates a nontransferable, nonexclusive, royalty-free right and license ("**Partnership License**") to use the "QEPM" logo and trademark and the other trademarks and tradenames owned by the General Partner for the benefit of the Partnership (collectively, the "**Partnership Marks**").
- 4.4 Ownership and Quality of the Partnership Marks. QEP agrees, on behalf of itself and the other QEP Entities, that ownership of the Partnership Marks and the goodwill relating thereto shall remain vested in the General Partner, for the benefit of the Partnership, during the term of the Partnership License and thereafter. QEP agrees, and agrees to cause the other QEP Entities, to the fullest extent permitted by applicable law, never to challenge, contest or question the validity of the General Partner's ownership of the Partnership Marks or any registration thereof by the General Partner or the Partnership. In connection with the use of the Partnership Marks, neither QEP nor any of the other QEP Entities shall in any manner represent that they have any ownership in the Partnership Marks or registration thereof. QEP, on behalf of itself and the other QEP Entities, acknowledges that the use of the Partnership Marks shall not create any right, title or interest in or to the Partnership Marks, and all use of the Partnership Marks by QEP or any of the other QEP Entities shall inure to the benefit of the General Partner and the Partnership. QEP agrees, and agrees to cause the other QEP Entities, to use the Partnership Marks in accordance with such quality standards established by the General Partner, on behalf of and for the benefit of the Partnership, and communicated to QEP from time to time.

4.5 <u>Termination</u>. The QEP License and the Partnership License shall each terminate upon the termination of this Agreement pursuant to <u>Section 5.5</u>.

ARTICLE V Miscellaneous

5.1 Confidentiality.

- (a) From and after the Closing Date, each of the Parties shall hold, and shall cause their respective Subsidiaries and Affiliates and its and their directors, officers, employees, agents, consultants, advisors, and other representatives (collectively, "Representatives") to hold all Confidential Information in strict confidence, with at least the same degree of care that applies to such Party's confidential and proprietary information and shall not use such Confidential Information and shall not release or disclose such Confidential Information to any other Person, except its Representatives or except as required by applicable law. Each Party shall be responsible for any breach of this section by any of its Representatives.
- (b) If a Party receives a subpoena or other demand for disclosure of Confidential Information received from any other Party or must disclose to a Governmental Authority any Confidential Information received from such other Party in order to obtain or maintain any required governmental approval, the receiving Party shall, to the extent legally permissible, provide notice to the providing Party before disclosing such Confidential Information. Upon receipt of such notice, the providing Party shall promptly either seek an appropriate protective order, waive the receiving Party's confidentiality obligations hereunder to the extent necessary to permit the receiving Party to respond to the demand, or otherwise fully satisfy the subpoena or demand or the requirements of the applicable Governmental Authority. If the receiving Party is legally compelled to disclose such Confidential Information or if the providing Party does not promptly respond as contemplated by this section, the receiving Party may disclose that portion of Confidential Information covered by the notice or demand.
- (c) Each Party acknowledges that the disclosing Party would not have an adequate remedy at law for the breach by the receiving Party of any one or more of the covenants contained in this <u>Section 5.1</u> and agrees that, in the event of such breach, the disclosing Party may, in addition to the other remedies that may be available to it, apply to a court for an injunction to prevent breaches of this <u>Section 5.1</u> and to enforce specifically the terms and provisions of this <u>Section 5.1</u>. Notwithstanding any other section hereof, to the extent permitted by applicable law, the provisions of this <u>Section 5.1</u> shall survive the termination of this Agreement.

5.2 Choice of Law; Mediation; Submission to Jurisdiction.

(a) This Agreement shall be subject to and governed by the laws of the State of Delaware, excluding any conflicts-of-law rule or principle that might refer the construction or interpretation of this Agreement to the laws of another state. EACH OF THE PARTIES HERETO AGREES THAT THIS AGREEMENT INVOLVES AT LEAST U.S. \$100,000.00 AND THAT THIS AGREEMENT HAS BEEN ENTERED INTO IN EXPRESS RELIANCE UPON 6 Del. C. § 2708. EACH OF THE PARTIES HERETO IRREVOCABLY AND

UNCONDITIONALLY AGREES (i) TO BE SUBJECT TO THE JURISDICTION OF THE COURTS OF THE STATE OF DELAWARE AND OF THE FEDERAL COURTS SITTING IN THE STATE OF DELAWARE, AND (ii) TO THE EXTENT SUCH PARTY IS NOT OTHERWISE SUBJECT TO SERVICE OF PROCESS IN THE STATE OF DELAWARE, TO APPOINT AND MAINTAIN AN AGENT IN THE STATE OF DELAWARE AS SUCH PARTY'S AGENT FOR ACCEPTANCE OF LEGAL PROCESS AND TO NOTIFY THE OTHER PARTY OF THE NAME AND ADDRESS OF SUCH AGENT.

- (b) If the Parties cannot resolve any dispute or claim arising under this Agreement, then no earlier than 10 days nor more than 60 days following written notice to the other Parties, any Party may initiate mandatory, non-binding mediation hereunder by giving a notice of mediation (a "Mediation Notice") to the other Parties to the dispute or claim. In connection with any mediation pursuant to this Section 5.2, the mediator shall be jointly appointed by the Parties to the dispute or claim and the mediation shall be conducted in Denver, Colorado unless otherwise agreed to by the Parties to the dispute or claim. All costs and expenses of the mediator appointed pursuant to this section shall be shared equally by the Parties to the dispute or claim. The then-current Model ADR Procedures for Mediation of Business Disputes of the Center for Public Resources, Inc., either as written or as modified by mutual agreement of the Parties to the dispute or claim, shall govern any mediation pursuant to this section. In the mediation, each Party to the dispute or claim shall be represented by one or more senior representatives who shall have authority to resolve any disputes. If a dispute or claim has not been resolved within 30 days after the receipt of the Mediation Notice by a Party, then any Party to the dispute or claim may refer the resolution of the dispute or claim to litigation.
- (c) Subject to Section 5.2(b), each Party agrees that it shall bring any action or proceeding in respect of any claim arising out of or related to this Agreement, whether in tort or contract or at law or in equity, exclusively in any federal or state courts located in Delaware and (i) irrevocably submits to the exclusive jurisdiction of such courts, (ii) waives any objection to laying venue in any such action or proceeding in such courts, (iii) waives any objection that such courts are an inconvenient forum or do not have jurisdiction over it and (iv) agrees that, to the fullest extent permitted by law, service of process upon it may be effected by mailing a copy thereof by registered or certified mail (or any substantially similar form of mail), postage prepaid, to it at its address specified in Section 5.3. The foregoing consents to jurisdiction and service of process shall not constitute general consents to service of process in the State of Delaware for any purpose except as provided herein and shall not be deemed to confer rights on any Person other than the Parties.
- 5.3 Notice. All notices or requests or consents provided for by, or permitted to be given pursuant to, this Agreement must be in writing and must be given by e-mail or United States mail, addressed to the Person to be notified, postpaid, and registered or certified with return receipt requested or by delivering such notice in person or by facsimile to such Party. Notice given by personal delivery or mail shall be effective upon actual receipt. Notice given by e-mail or facsimile shall be effective upon actual receipt if received during the recipient's normal business hours or at the beginning of the recipient's next business day after receipt if not received during the recipient's normal business hours. All notices to be sent to a Party pursuant to this Agreement shall be sent to or made at the address set forth below or at such other address as such Party may stipulate to the other Parties in the manner provided in this Section 5.3.

If to QEP or Field Services:

QEP Resources, Inc. 1050 17thStreet, Suite 500 Denver, Colorado 80265 Attn: General Counsel Facsimile: (303) 295-2190

E-mail: Chris.Woosley@qepres.com

E-mail: Chris. Woosley@qepres.com

If to any Group Member:

QEP Midstream Partners, LP c/o QEP Midstream Partners GP, LLC, its General Partner 1050 17th Street, Suite 500 Denver, Colorado 80265 Attn: General Counsel Facsimile: (303) 295-2190

5.4 Entire Agreement. This Agreement constitutes the entire agreement of the Parties relating to the matters contained herein, superseding all prior contracts or agreements, whether oral or written, relating to the matters contained herein.

- 5.5 <u>Termination of Agreement</u>. This Agreement, other than the provisions set forth in <u>Article II</u> hereof, may be terminated (a) by the written agreement of all of the Parties or (b) by QEP or the Partnership immediately upon a Partnership Change of Control by written notice given to the other Parties to this Agreement. For the avoidance of doubt, the Parties' indemnification obligations under <u>Article II</u> shall, to the fullest extent permitted by law, survive the termination of this Agreement in accordance with their respective terms.
- 5.6 <u>Amendment or Modification</u>. This Agreement may be amended or modified from time to time only by the written agreement of all the Parties. Each such instrument shall be reduced to writing and shall be designated on its face an "Amendment" or an "Addendum" to this Agreement.
- 5.7 <u>Assignment</u>. No Party shall have the right to assign its rights or obligations under this Agreement without the consent of the other Parties; *provided*, *however*, that the Partnership Group may make a collateral assignment of this Agreement solely to secure financing for the Partnership Group.
- 5.8 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts with the same effect as if all signatory parties had signed the same document and shall be construed together and shall constitute one and the same instrument.
- 5.9 <u>Severability</u>. If any provision of this Agreement shall be held invalid or unenforceable by a court or regulatory body of competent jurisdiction, the remainder of this Agreement shall remain in full force and effect.

- 5.10 <u>Further Assurances</u>. In connection with this Agreement and all transactions contemplated by this Agreement, each signatory party hereto agrees to execute and deliver such additional documents and instruments and to perform such additional acts as may be necessary or appropriate to effectuate, carry out and perform all of the terms, provisions and conditions of this Agreement and all such transactions.
- 5.11 <u>Rights of Limited Partners</u>. The provisions of this Agreement are enforceable solely by the Parties to this Agreement, and no Limited Partner or other interest holder of the Partnership shall have the right, separate and apart from the Partnership, to enforce any provision of this Agreement or to compel any Party to this Agreement to comply with the terms of this Agreement.

[Remainder of page intentionally left blank.]

IN WITNESS WHEREOF, the Parties have executed this Agreement on, and effective as of, the Closing Date.

QEP Resources, Inc.

By: /s/ Richard J. Doleshek

Name: Richard J. Doleshek

Title: Executive Vice President, Chief Financial Officer and Treasurer

QEP Field Services Company

By: /s/ Richard J. Doleshek

Name: Richard J. Doleshek

Title: Executive Vice President and Chief Financial Officer

QEP Midstream Partners GP, LLC

By: /s/ Richard J. Doleshek

Name: Richard J. Doleshek

Title: Executive Vice President and Chief Financial Officer

QEP Midstream Partners, LP

By: QEP Midstream Partners GP, LLC, its general partner

By: /s/ Richard J. Doleshek

Name: Richard J. Doleshek

Title: Executive Vice President and Chief Financial Officer

QEP Midstream Partners Operating, LLC

By: /s/ Richard J. Doleshek

Name: Richard J. Doleshek

Title: Executive Vice President and Chief Financial Officer

[Signature page to Omnibus Agreement]

APPENDIX A

Attached to and made part of that certain Omnibus Agreement, dated as of August 14, 2013, by and among QEP Resources, Inc., a Delaware corporation, QEP Field Services Company, a Delaware corporation, QEP Midstream Partners, LP, a Delaware limited partnership (the "Partnership"), QEP Midstream Partners GP, LLC, a Delaware limited liability company and the general partner of the Partnership, and QEP Midstream Partners Operating, LLC, a Delaware limited liability company and wholly owned subsidiary of the Partnership.

"1993 Gathering Agreement" means that certain gas gathering agreement, dated September 1, 1993, between Questar Gas Company (f/k/a Mountain Fuel Supply Company) and QEP Field Services Company (f/k/a Questar Pipeline Company), as amended.

- "Administrative Fee" is defined in Section 3.2(a).
- "Agreement" has the meaning set forth in the preamble.
- "Affiliate" means, with respect to any Person, any other Person that directly or indirectly through one or more intermediaries controls, is controlled by or is under common control with, the Person in question. As used herein, the term "control" means the possession, direct or indirect, of the power to direct or cause the direction of the management and policies of a Person, whether through ownership of voting securities, by contract or otherwise.
- "Assets" means the equity interests in the entities being conveyed, contributed or otherwise transferred to any Group Member pursuant to the Contribution Agreement and any gathering pipelines, transportation pipelines, compressor stations, pump stations, metering stations, vehicles, related equipment, offices, real estate, contracts and other assets, or portions thereof owned by, leased by or necessary for the operation of the business of any Group Member as of the Closing Date.

"Closing Date" has the meaning set forth in the preamble.

"Confidential Information" means any proprietary or confidential information that is competitively sensitive material or otherwise of value to a Party or its Affiliates and not generally known to the public, including trade secrets, scientific or technical information, design, invention, process, procedure, formula, improvements, product planning information, marketing strategies, financial information, information regarding operations, consumer and/or customer relationships, consumer and/or customer identities and profiles, sales estimates, business plans, and internal performance results relating to the past, present or future business activities of a Party or its Affiliates and the consumers, customers, clients and suppliers of any of the foregoing. Confidential Information includes such information as may be contained in or embodied by documents, substances, engineering and laboratory notebooks, reports, data, specifications, computer source code and object code, flow charts, databases, drawings, pilot plants or demonstration or operating facilities, diagrams, specifications, bills of material, equipment, prototypes and models, and any other tangible manifestation (including data in computer or other digital format) of the foregoing; provided, however, that Confidential Information does not include information that a receiving Party can show (A) has been published

or has otherwise become available to the general public as part of the public domain without breach of this Agreement, (B) has been furnished or made known to the receiving Party without any obligation to keep it confidential by a third party under circumstances which are not known to the receiving Party to involve a breach of the third party's obligations to a Party or (C) was developed independently of information furnished or made available to the receiving Party as contemplated under this Agreement.

"Contribution Agreement" means that certain Contribution, Conveyance and Assumption Agreement, dated as of the Closing Date, among Field Services, the General Partner, the Partnership and the Operating Company, together with the additional conveyance documents and instruments contemplated or referenced thereunder, as such may be amended, supplemented or restated from time to time.

"Covered Environmental Losses" is defined in Section 2.1.

"*Departing General Partner*" means a former General Partner from and after the effective date of any withdrawal or removal of such former General Partner pursuant to Section 11.1 or Section 11.2 of the Partnership Agreement.

"Environmental Deductible" is defined in Section 2.5.

"Environmental Defect" is defined Section 2.5.

"Environmental Laws" means all federal, state, and local laws, statutes, rules, regulations, orders, judgments, ordinances, codes, injunctions, decrees, Environmental Permits and other legally enforceable requirements and rules of common law relating to pollution or protection of human health, natural resources, wildlife and the environment or workplace health or safety including, without limitation, the federal Comprehensive Environmental Response, Compensation, and Liability Act of 1980, as amended, 42 U.S.C. §§9601 et seq., the Resource Conservation and Recovery Act of 1976, as amended, 42 U.S.C. §§7401 et seq., the Federal Water Pollution Control Act, as amended, 33 U.S.C. §§1251 et seq., the Toxic Substances Control Act, as amended, 15 U.S.C. §§2601 et seq., the Oil Pollution Act of 1990, 33 U.S.C. §§2701 et seq., the Safe Drinking Water Act of 1974, as amended, 42 USC §§300f et seq., the Hazardous Materials Transportation Act of 1994, as amended, 49 U.S.C. §§ 5101 et seq., and other environmental conservation and protection laws and the Occupational Safety and Health Act of 1970, 29 U.S.C. §§ 651 et seq, and the regulations promulgated pursuant thereto, and any state or local counterparts, each as amended from time to time.

"Environmental Permit" means any permit, approval, identification number, license, registration, certification, consent, exemption, variance or other authorization required under or issued pursuant to any applicable Environmental Law, including applications for renewal of such permits in which the application allows for continued operation under the terms of an expired permit.

"Field Services" has the meaning set forth in the preamble.

"General and Administrative Services" is defined in Section 3.1.

"General Partner" means QEP Midstream Partners GP, LLC, a Delaware limited liability company, and its successors and permitted assigns that are admitted to the Partnership as general partner of the Partnership, in its capacity as general partner of the Partnership (except as the context otherwise requires).

"Governmental Authority" means any federal, state, tribal, foreign or local governmental entity, authority, department, court or agency, including any political subdivision thereof, exercising, or entitled to exercise, any administrative, executive, judicial, legislative, police, regulatory or taxing authority or power of any nature, and including any arbitrating body, commission or quasi-governmental authority or self-regulating organization of competent authority exercising or enlisted to exercise similar power or authority.

"Group Member" means a member of the Partnership Group.

"Hazardous Substance" means (a) any substance, whether solid, liquid, gaseous, semi-solid, or any combination thereof, that is designated, defined or classified as a hazardous waste, solid waste, hazardous material, pollutant, contaminant or toxic or hazardous substance, or terms of similar meaning, or that is otherwise regulated under any Environmental Law, including, without limitation, any hazardous substance as defined under the Comprehensive Environmental Response, Compensation, and Liability Act, as amended, and including asbestos and lead-containing paints or coatings, and (b) petroleum, oil, gasoline, natural gas, fuel oil, motor oil, waste oil, diesel fuel, jet fuel, and other refined petroleum hydrocarbons.

"Identification Deadline" means the third anniversary of the Closing Date.

"Indemnified Party" means the Party entitled to indemnification in accordance with Article II.

"Indemnifying Party" means the Party from whom indemnification may be sought in accordance with Article II.

"Limited Partner" means, unless the context otherwise requires, each Person that becomes a Limited Partner pursuant to the terms of the Partnership Agreement and any Departing General Partner upon the change of its status from General Partner to Limited Partner pursuant to Section 11.3 of the Partnership Agreement, in each case, in such Person's capacity as a limited partner of the Partnership; provided, however, that when the term "Limited Partner" is used herein in the context of any vote or other approval, such term shall not, solely for such purpose, include any holder of any Incentive Distribution Right (solely with respect to its Incentive Distribution Rights and not with respect to any other limited partner interest held by such Person) except as may otherwise be required by law.

"Losses" means any losses, damages, liabilities, claims, demands, causes of action, judgments, settlements, fines, penalties, costs and expenses (including, without limitation, court costs and reasonable attorney's and expert's fees) of any and every kind or character, known or unknown, fixed or contingent.

"Mediation Notice" is defined in Section 5.2(b).

- "*Mesa Condensate Trunk Line*" means that certain section of 6 ⁵/8" OD pipe beginning in Sublette County, Wyoming in the Center T33N R109W S18 Latitude 42.69205036 Longitude -109.8102758 traversing in a southeasterly direction approximately 68,000 linear feet and ending in the SW 1/4 NW 1/4 T31N R109W S2 Latitude 42.82930891 Longitude -109.9076653.
 - "Operating Company" has the meaning set forth in the preamble.
 - "Partnership" has the meaning set forth in the preamble.
 - "Partnership Agreement" means the First Amended and Restated Agreement of Limited Partnership of the Partnership, dated as of the Closing Date.
- "Partnership Change of Control" means QEP ceases to control, directly or indirectly, the General Partner. For purposes of this definition, "control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of the General Partner, whether through ownership of voting securities, by contract, or otherwise.
 - "Partnership Group" means, collectively, the Partnership and its Subsidiaries.
 - "Partnership License" is defined in Section 4.3.
 - "Partnership Marks" is defined in Section 4.3.
 - "Party" has the meaning set forth in the preamble.
- "*Person*" means an individual or a corporation, firm, limited liability company, partnership, joint venture, trust, unincorporated organization, association, government agency or political subdivision thereof or other entity.
 - "QEP" has the meaning set forth in the preamble.
 - "QEP Entities" means QEP and each of its Affiliates, other than the General Partner and the Group Members.
 - "QEP License" is defined in Section 4.1.
 - "QEP Marks" is defined in Section 4.1.
 - "Representatives" is defined in Section 5.1(a).
- "Retained Assets" means all pipelines, processing facilities, treating assets, vehicles, other midstream infrastructure, offices and related equipment, real estate, contracts and other related assets, or ownership interests or portions thereof owned by Field Services that were not directly or indirectly conveyed, contributed or otherwise transferred to the Partnership Group pursuant to the Contribution Agreement or the other documents referenced in the Contribution Agreement.

"Subsidiary" means, with respect to any Person, (a) a corporation of which more than 50% of the voting power of shares entitled (without regard to the occurrence of any contingency) to vote in the election of directors or other governing body of such corporation is owned, directly or indirectly, at the date of determination, by such Person, by one or more Subsidiaries of such Person or a combination thereof, (b) a partnership (whether general or limited) in which such Person or a Subsidiary of such Person is, at the date of determination, a general or limited partner of such partnership, but only if more than 50% of the partnership interests of such partnership (considering all of the partnership interests of the partnership as a single class) is owned, directly or indirectly, at the date of determination, by such Person, by one or more Subsidiaries of such Person, or a combination thereof; or (c) any other Person (other than a corporation or a partnership) in which such Person, one or more Subsidiaries of such Person, or a combination thereof, directly or indirectly, at the date of determination, has (i) at least a majority ownership interest or (ii) the power to elect or direct the election of a majority of the directors or other governing body of such Person.

Schedule A Pre-Closing Litigation

Matter Name	Matter Type	Party	Matter Description	Case/Docket #	Court/Agency	State
	Litigation	QEP Field	Contract Dispute	120,902,969	Third Judicial	UT
Questar Gas Company v.		Services			District Court	
OEP Field Services Company ¹		Company				

QEP Resources, Inc. and QEP Field Services Company shall not be obligated to indemnify QEP Midstream Partners, LP or any other Group Member for Losses incurred as a result of a decrease in the annual gathering rate charged under the 1993 Gathering Agreement subsequent to the date that this litigation is settled or resolved by a final, binding and non-appealable court order.

Schedule A-1

Schedule B Environmental Remediation Locations

Responsible Party	Description
Field Services	South Baxter Groundwater Monitoring (Rock Springs, Wyoming)
Field Services	Mesa Condensate Trunk Line

Schedule B-1

Schedule C General and Administrative Services

Pursuant to Section 3.1

- (1) Management services of QEP and its Affiliates (other than the General Partner) provided by employees who devote less than 50% of their business time to the business and affairs of the Partnership. This cost includes QEP-stock based compensation expense.
- (2) Financial and administrative services (including treasury and accounting)
- (3) Information technology services—professional services
- (4) Legal services
- (5) Health, environmental, safety and security services (including third party security services)
- (6) Human resources services
- (7) Tax services
- (8) Procurement services
- (9) Investor relations; Government & public affairs services
- (10) Analytical & engineering services
- (11) Business development services

Pursuant to Section 3.2

The fixed portion of the Administrative Fee for calendar year 2014, as described in <u>Section 3.2</u>, will be \$1,400,000. For the avoidance of doubt, the fixed portion of the Administrative Fee for the remainder of calendar year 2013 will be the same annual amount as calendar year 2014 pro-rated based on the number of days remaining in 2013 from the Closing Date.

The portion of the Administrative Fee attributable to any marketing and transportation engineering services, information technology services, administrative/office services, and public company expenses will be a variable amount based on the costs actually incurred by QEP and its Affiliates on behalf of the Partnership Group (other than any costs for which QEP and its Affiliates are reimbursed pursuant to Section 3.3). The portion of the variable amount of the Administrative Fee attributable to any marketing and transportation engineering services described in the preceding sentence will be based on the costs incurred by QEP and its Affiliates on behalf of the Partnership Group (other than any costs for which QEP and its Affiliates are reimbursed pursuant to Section 3.3).

Pursuant to Section 3.2(b)

The fixed portion of the Administrative Fee for calendar year 2014 includes as a part thereof, the following amounts attributable to services provided by officers of the Partnership Group:

(1) Charles B. Stanley, Chairman of the Board, President and Chief	
ExecutiveOfficer	\$ 650,000
Richard J. Doleshek, Executive Vice President and Chief Financial	
Officer	\$ 350,000
Perry H. Richards, Senior Vice President and General Manager	\$ 400,000
Total	\$ 1,400,000

Schedule C-1

INDEMNIFICATION AGREEMENT

This Indemnification Agreement (this "Agreement"), dated as of _	, 20, is made by and between QEP RESOURCES, INC.,
Delaware corporation (the "Company") and [NAME] (the "Indemnitee").	

RECITALS

- A. The Company recognizes that competent and experienced persons are increasingly reluctant to serve or to continue to serve as directors or officers of corporations unless they are protected by comprehensive liability insurance and indemnification, due to increased exposure to litigation costs and risks resulting from their service to such corporations, and because the exposure frequently bears no reasonable relationship to the compensation of such directors and officers:
- B. The statutes and judicial decisions regarding the duties of directors and officers are often difficult to apply to particular facts, can be ambiguous or conflicting, and may fail to provide such directors and officers with adequate, reliable knowledge of legal risks to which they are exposed or information regarding the proper course of action to take;
 - C. The Company and Indemnitee recognize that the risk of litigation is significant;
- D. The Company does not believe its directors and officers should bear the risk of judgments and other expenses that may occur in cases in which the director or officer had no improper personal interest and in cases where the director or officer was not culpable;
- E. The Company believes that the interests of the Company and its shareholders would best be served by a combination of liability insurance and indemnification provided by the Company for the benefit of the directors and officers of the Company;
- F. The Company's Bylaws require the Company to indemnify its directors and officers to the fullest extent permitted by the Delaware General Corporation Law (the "DGCL"). The Bylaws expressly provide that the indemnification provisions set forth therein are not exclusive and contemplate that contracts may be entered into between the Company and its directors and officers with respect to indemnification;
- G. Section 145 of the DGCL ("Section 145"), under which the Company is organized, empowers the Company to indemnify its officers, directors, employees and agents by agreement and to indemnify persons who serve, at the request of the Company, as the directors, officers, employees or agents of other corporations or enterprises, and expressly provides that the indemnification provided by Section 145 is not exclusive;
- H. Section 102(b)(7) of the DGCL allows a corporation to include in its certificate of incorporation a provision limiting or eliminating the personal liability of a director for monetary damages in respect of claims by shareholders and corporations for breach of certain fiduciary duties, and the Company has provided in its Certificate of Incorporation that each Director shall be exculpated from such liability to the maximum extent permitted by law;
- I. The Board of Directors of the Company (the "Board of Directors") has determined that contractual indemnification as set forth herein is not only reasonable and prudent but also promotes the best interests of the Company and its shareholders;
- J. The Company desires and has requested Indemnitee to serve or continue to serve as a director or officer of the Company free from undue concern for unwarranted claims for damages arising out of or related to services to the Company; and

K. Indemnitee is willing to serve, continue to serve or to provide additional service for or on behalf of the Company on the condition that he or she is furnished the indemnity provided for herein.

AGREEMENT

In consideration of the mutual covenants and agreements set forth below, and other good and valuable consideration, the receipt and adequacy of which are hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

Section 1. <u>Employment Rights and Duties</u>. Subject to any other obligations imposed on either of the parties by contract or by law, and with the understanding that this Agreement is not intended to confer employment rights on either party which they did not possess on the date of its execution, Indemnitee agrees to serve as a director or officer so long as Indemnitee is duly appointed or elected and qualified in accordance with the applicable provisions of the Certificate of Incorporation and Bylaws of the Company or any subsidiary of the Company and until such time as Indemnitee resigns or fails to stand for election or until Indemnitee's employment, if any, terminates. Indemnitee may from time to time also perform other services at the request of, or for the convenience of, or otherwise benefiting, the Company. Indemnitee may at any time and for any reason resign or be removed from such position (subject to any other contractual obligation or other obligation imposed by operation of law), in which event the right to indemnification shall continue only as set forth in this Agreement.

Section 2. Indemnification Generally. To the fullest extent permitted by the laws of the State of Delaware:

- (a) The Company shall indemnify Indemnitee if Indemnitee was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative, by reason of the fact that Indemnitee is or was or has agreed to serve at the request of the Company as a director, officer, employee or agent of the Company ("Position with the Company"), or while serving as a director or officer of the Company, is or was serving or has agreed to serve at the request of the Company as a director, officer, employee or agent (which, for purposes hereof, shall include a trustee, partner, manager or similar capacity) of another corporation, partnership, joint venture, trust, employee benefit plan or other enterprise ("Service at the Request of the Company"), or by reason of any action alleged to have been taken or omitted in such capacity. For the avoidance of doubt, the foregoing indemnification obligation includes, without limitation, claims for monetary damages against Indemnitee in respect of an alleged breach of fiduciary duties, to the fullest extent permitted under Section 102(b)(7) of the DGCL as in existence on the date hereof. Indemnitee will be deemed a party to a proceeding for all purposes hereof if Indemnitee is named as a defendant or respondent in a complaint or petition for relief in that proceeding, regardless of whether Indemnitee is ever served with process or makes an appearance in that proceeding.
- (b) The indemnification provided by this Section 2 shall be from and against Expenses and judgments, fines and amounts paid in any settlement made in accordance with Section 10 ("Judgments"), incurred by Indemnitee or on Indemnitee's behalf in connection with such action, suit or proceeding and any appeal therefrom, but shall not be provided if it is finally determined that the Indemnitee acted in bad faith.
- (c) Notwithstanding the foregoing provisions of this Section 2, in the case of any threatened, pending or completed action or suit by or in the right of the Company to procure a judgment in its favor by reason of Indemnitee's Position with the Company or Service at the Request of the Company, no indemnification shall be made in respect of any claim, issue or matter as to which Indemnitee shall have been finally adjudged to be liable to the Company. Indemnification may be made, however, to the extent that the Delaware Court of Chancery or the court in which such action or suit was brought shall determine upon application that, despite the adjudication of liability but in view of all the circumstances of the case, Indemnitee is fairly and reasonably entitled to indemnity for such expenses which the Delaware Court of Chancery or such other court shall deem proper.
- (d) The termination of any action, suit or proceeding by judgment, order, settlement, conviction, or upon a plea of *nolo contendere* or its equivalent, shall not, of itself, create a presumption that Indemnitee did not act in good faith and in a manner which Indemnitee reasonably believed to be in or not opposed to the best interests of the Company, and, with respect to any criminal action or proceeding, had reasonable cause to believe that Indemnitee's conduct was unlawful.
- (e) For purposes of Section 2, the meaning of the phrase "to the fullest extent permitted by the laws of the State of Delaware" shall include, but not be limited to:

- (i) to the fullest extent permitted by the provision of the DGCL that authorizes or contemplates additional indemnification by agreement, or the corresponding provision of any amendment to or replacement of the DGCL, and
- (ii) to the fullest extent authorized or permitted by any amendments to or replacements of the DGCL adopted after the date of this Agreement that increase the extent to which a corporation may indemnify its officers and directors.
- Section 3. Indemnification for Expenses of a Party Who is Wholly or Partly Successful. To the extent that Indemnitee has been successful on the merits or otherwise in defense of any action, suit or proceeding referred to in Section 2 hereof or in defense of any claim, issue or matter therein (in whole or in part), Indemnitee shall be indemnified against all Expenses incurred in connection therewith. If Indemnitee is not wholly successful in such action, suit or proceeding, but is successful on the merits or otherwise, as to one or more but less than all claims, issues or matters in such action, suit or proceeding, Indemnitee shall be indemnified against Expenses incurred in connection with each successfully resolved claim, issue or matter. If Indemnitee is not wholly successful in such action, suit or proceeding, Indemnitee shall also be indemnified against all Expenses reasonably incurred in connection with a claim, issue or matter related to any claim issue or matter on which Indemnitee was successful. For purposes of this Section and without limitation, the termination of any claim, issue or matter in such an action, suit or proceeding by dismissal, with or without prejudice, shall be deemed to be a successful result as to such claim, issue or matter.

Section 4. Notification of Claims; Determination That Indemnification Is Proper.

- (a) Promptly after receipt by Indemnitee of notice of the commencement of any action, suit or proceeding, Indemnitee shall, if a claim thereof is to be made against the Company hereunder, notify the Company of the commencement thereof. The failure to promptly notify the Company of the commencement of the action, suit or proceeding, or Indemnitee's request for indemnification, will not relieve the Company from any liability that it may have to Indemnitee hereunder, except to the extent the Company is prejudiced in its defense of such action, suit or proceeding as a result of such failure.
- (b) Any indemnification hereunder shall (unless otherwise ordered by a court) be made by the Company unless a determination is made in the specific instance that indemnification of such person is not proper in the circumstances because he or she has not met the applicable standard of conduct set forth in Section 2 hereof. Any such determination shall be made (i) if no Change in Control has occurred, (A) by a majority vote of the directors who are not parties to the action, suit or proceeding in question ("disinterested directors"), even if less than a quorum, (B) by a majority vote of a committee of disinterested directors designated by majority vote of disinterested directors, even if less than a quorum, and (C) if there are no disinterested directors or if the disinterested directors so direct, by Independent Counsel or by the Delaware Chancery Court, and (ii) if a Change in Control has occurred, or at the election of the Indemnitee at anytime, by Independent Counsel.
- In the event the determination of entitlement to indemnification is to be made by Independent Counsel pursuant to Section 4(b) hereof, the Independent Counsel shall be selected by the Indemnitee, and the Indemnitee shall give written notice to the Company advising it of the identity of the Independent Counsel so selected. The Company may, within seven days after such written notice of selection shall have been given, deliver to the Indemnitee a written objection to such selection. Such objection may be asserted only on the ground that the Independent Counsel so selected does not meet the requirements of "Independent Counsel" as defined in Section 8 of this Agreement, and the objection shall set forth with particularity the factual basis of such assertion. If such written objection is timely made, the Independent Counsel so selected may not serve as Independent Counsel unless and until a court has determined that the Independent Counsel meets such requirements. If, within 20 days after submission by the Company of an objection to the selection of the Independent Counsel, the Indemnitee has not proposed an alternate Independent Counsel or submitted the matter to a court of competent jurisdiction, the Company may petition the Court of Chancery of the State of Delaware or other court of competent jurisdiction for resolution of any objection that shall have been made by the Company to the Indemnitee's selection of Independent Counsel and/or for the appointment as Independent Counsel of a person selected by the court or by such other person as the court shall designate, and the person with respect to whom an objection is so resolved or the person so appointed shall act as Independent Counsel under Section 4(b) hereof. The Company shall pay any and all reasonable fees and expenses incurred by such Independent Counsel in connection with acting pursuant to Section 4(b) hereof, and the Company shall pay all reasonable fees and expenses incident to the procedures of this Section 4(c), regardless of the manner in which such Independent Counsel was selected or appointed. Upon the due commencement of any judicial proceeding or arbitration pursuant to Section 6 of this Agreement, Independent Counsel shall be discharged and relieved of any further responsibility in such capacity (subject to the applicable standards of professional conduct then prevailing).

- (d) Indemnitee and the Company shall provide to the person, persons or entity making such determination with respect to Indemnitee's entitlement to indemnification, documentation or information related to Indemnitee's claim for indemnification that is not privileged or otherwise protected from disclosure and that is reasonably available to Indemnitee or the Company, as applicable, and reasonably necessary to such determination and shall otherwise cooperate with such person, persons or entity making such determination. Any Expenses incurred by Indemnitee in providing documentation in response to inquiries from the person, persons or entity making such determination, or in so cooperating, shall be borne by the Company (irrespective of the determination as to Indemnitee's entitlement to indemnification), and the Company hereby indemnifies and agrees to hold Indemnitee harmless therefrom.
- (e) The Indemnitee shall be presumed to be entitled to indemnification under this Agreement upon submission of a request for indemnification pursuant to this Section 4, and the Company shall have the burden of proof in overcoming that presumption by clear and convincing evidence in reaching a determination contrary to that presumption.
- (f) Any Independent Counsel, member of the Board of Directors or court shall act reasonably and in good faith in making a determination regarding Indemnitee's entitlement to indemnification under this Agreement. The determination whether to grant Indemnitee's indemnification request shall be made promptly, and in any event, with respect to indemnification for Judgments, within 60 days following receipt of a request for indemnification for Judgments. Advances of Expenses shall be paid within 20 days following receipt of a request for advancement of Expenses pursuant to Section 5(a).

Section 5. <u>Advance Payment of Expenses; Defense of Claim.</u>

- (a) To the fullest extent permitted by the laws of the State of Delaware, Expenses incurred by Indemnitee in defending a threatened or pending civil, criminal, administrative or investigative action, suit or proceeding, or in connection with an enforcement action pursuant to Section 6(a), shall be paid by the Company in advance of the final disposition of such action, suit or proceeding within twenty (20) days after receipt by the Company of a statement or statements from Indemnitee requesting such advance or advances from time to time (which shall include invoices received by Indemnitee in connection with such expenses but, in the case of invoices in connection with legal services, any references to legal work performed or to expenditures made that may cause Indemnitee to waive any privilege accorded by applicable law need not be included with the invoice). Advances shall be unsecured and interest-free and shall be made without reference to the financial ability of Indemnitee to make such repayment. By executing this Agreement, Indemnitee hereby provides an undertaking to repay any advance or advances if, and to the extent that, it shall ultimately be determined that Indemnitee is not entitled to be indemnified by the Company as authorized by this Agreement or otherwise.
- (b) In the event the Company shall be obligated to make advance payment of the Expenses of Indemnitee with respect to an action, suit or proceeding, as provided in this Agreement, the Company, if appropriate, shall be entitled to assume the defense of such action, suit or proceeding, with counsel reasonably acceptable to Indemnitee, upon the delivery to Indemnitee of written notice of its election to do so. After delivery of such notice, approval of such counsel by Indemnitee and the retention of such counsel by the Company, the Company will not be liable to Indemnitee under this Agreement for any fees of counsel subsequently incurred by Indemnitee with respect to the same action, suit or proceeding, provided that (i) Indemnitee shall have the right to employ Indemnitee's own counsel in such action, suit or proceeding at Indemnitee's expense and (ii) if (A) the employment of counsel by Indemnitee has been previously authorized in writing by the Company, (B) counsel to the Company or Indemnitee shall have reasonably concluded that there may be a conflict of interest or position, or reasonably believes that a conflict is likely to arise, on any significant issue between the Company and Indemnitee in the conduct of any such defense, or (C) the Company shall not, in fact, have employed counsel to assume the defense of such action, suit or proceeding, then the actual and reasonable fees and expenses of Indemnitee's counsel shall be at the expense of the Company, except as otherwise expressly provided by this Agreement. The Company shall not be entitled, without the consent of Indemnitee, to assume the defense of any claim brought by or in the right of the Company or as to which counsel for the Company or Indemnitee shall have reasonably made the conclusion provided for in clause (ii)(B) above.
- (c) To the fullest extent permitted by the laws of the State of Delaware, if Indemnitee is, by reason of Indemnitee's Position with the Company or Service at the Request of the Company, a witness, including an expert witness, or otherwise participates in any action, suit or proceeding at a time when Indemnitee is not a party in the action, suit or proceeding, the Company shall advance payment to Indemnitee against all Expenses incurred by Indemnitee or on Indemnitee's behalf in connection therewith.
 - (d) This Section 5 shall not apply to any claim made by Indemnitee for which indemnity is excluded pursuant to Section 9.

Section 6. <u>Appeal by Indemnitee of Denial of Claim for Indemnification.</u>

- (a) The right to indemnification as granted by Section 2 of this Agreement and to advancement of Expenses as provided in Section 5 of this Agreement shall be enforceable by Indemnitee in the Delaware Court of Chancery if the Company denies such request, in whole or in part, or fails to respond within the 60-day period specified in Section 4(f), with respect to a claim for indemnification for Judgments, or the 20-day period specified in Section 4(f), with respect to a claim for indemnification for Judgments, or the 20-day period specified in Section 4(f), with respect to a claim for indemnification for Judgments, or the 20-day period specified in Section 4(f), with respect to a claim for indemnification for Judgments, or the 20-day period specified in Section 4(f), with respect to a claim for indemnification for Judgments, or the 20-day period specified in Section 4(f), with respect to a claim for indemnification for Judgments, or the 20-day period specified in Section 4(f), with respect to a claim for indemnification for Judgments, or the 20-day period specified in Section 4(f), with respect to a claim for indemnification for Judgments, or the 20-day period specified in Section 4(f), with respect to a claim for indemnification Rules of the American Arbitration Association. The arbitration itself, shall remain confidential. Indemnitee shall commence such proceeding seeking an adjudication or an award in arbitration within 180 days following the date on which Indemnitee first has the right to commence such proceeding pursuant to this Section 6(a). If Indemnitee elects arbitration pursuant to Section 6(a) and the parties are unable to agree on an arbitrator, the parties shall provide the American Arbitration Association ("AAA") with a statement of the nature of the dispute and the desired qualifications of the arbitrator. AAA shall then provide a list of three available arbitrators. Each party may strike one of the names on the list, and the remaining person shall serve as the arbitrator. If both parties strike th
- (b) It shall be a defense to any such action (other than an action brought to enforce a claim for the advance of Expenses under Section 5 hereof, which action has not been concluded) that Indemnitee has not met the standard of conduct set forth in Section 2 hereof, but the burden of proving such defense by clear and convincing evidence shall be on the Company. Any adjudication or arbitration shall be conducted as a de novo trial, or arbitration, on the merits and neither the failure of the Company (including the Board of Directors or one of its committees), Independent Counsel or the Delaware Chancery Court to have made a determination prior to the commencement of such action that indemnification of Indemnitee is proper in the circumstances because Indemnitee has met the applicable standard of conduct set forth in Section 2 hereof, nor the fact that there has been an actual determination by the Company (including the Board of Directors or one of its committees), Independent Counsel or the Delaware Chancery Court that Indemnitee has not met such applicable standard of conduct, shall be a defense to the action or create a presumption that Indemnitee has or has not met the applicable standard of conduct. A determination by the relevant party that the Indemnitee has met the applicable standard set forth in Section 2 hereof shall be binding on the Company in any such proceeding. The Indemnitee's Expenses incurred in connection with successfully establishing Indemnitee's right to indemnification, in whole or in part, in any such proceeding or otherwise shall also be indemnified by the Company.
- (c) For purposes of any determination of good faith, Indemnitee shall be deemed to have acted in good faith if Indemnitee's action is based on the records or books of account of the Company, including financial statements, or on information supplied to Indemnitee by the officers of the Company in the course of their duties, or on the advice of legal counsel for the Company or on information or records given or reports made to the Company by an independent certified public accountant or by an appraiser or other expert selected with the reasonable care by the Company. The provisions of this Section 6(c) shall not be deemed to be exclusive or to limit in any way the other circumstances in which the Indemnitee may be deemed to have met the applicable standard of conduct set forth in this Agreement.
- (d) The knowledge and/or actions, or failure to act, of any director, officer, agent or employee of the Company shall not be imputed to Indemnitee for purposes of determining the right to indemnification under this Agreement.

Section 7. <u>Insurance and Subrogation</u>.

- (a) The Company may purchase and maintain insurance on behalf of any person (including Indemnitee) by virtue of such person's Position with the Company or Service at the Request of the Company against any liability asserted against, and incurred by, such person or on such person's behalf in any such capacity, or arising out of such person's status as such, whether or not the Company would have the power to indemnify such person against such liability under the provisions of this Agreement.
- (b) To the extent the Company maintains an insurance policy or policies providing liability insurance for directors, officers, employees or agents of the Company or for other persons for Service at the Request of the Company, Indemnitee shall be covered by such policy or policies in accordance with its or their terms, to the maximum extent of the coverage available for any director, officer, employee or agent under such policy or policies.
- (c) If the Company has such insurance in effect at the time the Company receives from Indemnitee any notice of the commencement of a proceeding, the Company shall give prompt notice of the

commencement of such proceeding to the insurer(s) at the times and in accordance with the procedures set forth in such policy or policies. The Company shall thereafter take all actions that the Company believes, in its discretion, are reasonably necessary or desirable to cause such insurer(s) to pay, on behalf of the Indemnitee, all amounts payable as a result of such proceeding in accordance with the terms of such policy or policies.

- (d) In the event of any payment by the Company under this Agreement, the Company shall be subrogated to the extent of such payment to all of the rights of recovery of Indemnitee with respect to any insurance policy, and Indemnitee shall execute all papers required and take all action necessary to secure such rights, including execution of such documents as are necessary to enable the Company to bring suit to enforce such rights in accordance with the terms of such insurance policy. The Company shall pay or reimburse all expenses actually and reasonably incurred by Indemnitee in connection with such subrogation.
- (e) The Company shall not be liable under this Agreement to make any payment of amounts otherwise indemnifiable hereunder (including, but not limited to, Judgments) if and to the extent that Indemnitee has otherwise actually received such payment under this Agreement or any insurance policy, contract, agreement or otherwise.

Section 8. <u>Certain Definitions</u>. For purposes of this Agreement, the following definitions shall apply:

- (a) The term "action, suit or proceeding" shall be broadly construed and shall include, without limitation, the investigation, preparation, prosecution, defense, settlement, arbitration and appeal of, and the giving of testimony in, any potential, threatened, pending or completed claim, action, suit or proceeding, whether civil, criminal, administrative or investigative (including any internal investigation conducted or initiated by the Company), without regard to whether such potential, threatened, pending or completed claim, action, suit or proceeding specifically names the Indemnitee as a defendant, respondent or target thereof.
- (b) The terms "by reason of Indemnitee's Position with the Company" and "Service at the Request of the Company" shall be broadly construed and shall include, without limitation, any actual or alleged act or omission to act by the Company, the Indemnitee or any other officer, director, employee or agent of the Company.
- (c) The term "Expenses" shall be broadly and reasonably construed and shall include, without limitation, all direct and indirect costs of any type or nature whatsoever (including, without limitation, all attorneys' fees and related disbursements, expert witness fees, appeal bonds, other out-of-pocket costs and reasonable compensation for time spent by Indemnitee for which Indemnitee is not otherwise compensated by the Company or any third party), actually and reasonably incurred by Indemnitee in connection with either the investigation, defense or appeal of an action, suit or proceeding or establishing or enforcing a right to indemnification under this Agreement, Section 145 of the DGCL or otherwise.
- (d) The term "Judgments" shall be broadly construed and shall include, without limitation, all direct and indirect payments of any type or nature whatsoever (including, without limitation, all penalties and amounts required to be forfeited or reimbursed to the Company, as well as any penalties or excise taxes assessed on a person with respect to an employee benefit plan).
- (e) The term "Company" shall include, without limitation and in addition to the resulting corporation in any business combination to which the Company is a party, any constituent corporation (including any constituent of a constituent) absorbed in a consolidation or merger which, if its separate existence had continued, would have had power and authority to indemnify its directors, officers, employees or agents, so that any person who is or was a director, officer, employee or agent of such constituent corporation, or is or was serving at the request of such constituent corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust, employee benefit plan or other enterprise, shall stand in the same position under the provisions of this Agreement with respect to the resulting or surviving corporation as he or she would have with respect to such constituent corporation if its separate existence had continued.
 - (f) The term "other enterprises" shall include, without limitation, employee benefit plans.
- (g) The term "Service at the Request of the Company" shall include, without limitation, any service as a director, officer, employee or agent of the Company that imposes duties on, or involves services by, such director, officer, employee or agent with respect to an employee benefit plan, its participants or beneficiaries.
- (h) A person who acted in good faith and in a manner such person reasonably believed to be in the interest of the participants and beneficiaries of an employee benefit plan shall be deemed to have acted in a manner "not opposed to the best interests of the Company" as referred to in this Agreement.
- (i) The term "Independent Counsel" means a law firm, or a member of a law firm, that is experienced in matters of corporation law and trial work, and neither presently is, nor in the past five years has been, retained to represent: (i) the Company or Indemnitee in any matter material to either such party (other than as Independent

Counsel under this Agreement or similar agreements), or (ii) any other party to the proceeding giving rise to a claim for indemnification hereunder. Notwithstanding the foregoing, the term "Independent Counsel" shall not include any person who, under the applicable standards of professional conduct then prevailing, would have a conflict of interest in representing either the Company or Indemnitee in an action to determine Indemnitee's rights under this Agreement.

A "Change in Control" of the Company shall be deemed to have occurred if (i) any individual, entity, or group (within the meaning of Section 13(d)(3) or 14(d)(2) of the Securities Exchange Act of 1934 (the "Exchange Act")) other than a trustee or other fiduciary holding securities under an employee benefit plan of the Company, is or becomes the beneficial owner (as such term is used in Rule 13d-3 under the Exchange Act) of securities of the Company representing 30 percent or more of the combined voting power of the Company; (ii) the following individuals cease for any reason to constitute a majority of the number of directors then serving: individuals who, as of the date of this Agreement, constitute the Board of Directors and any new director (other than a director whose initial assumption of office is in connection with an actual or threatened election contest, including but not limited to a consent solicitation, relating to the election of directors of the Company) whose appointment or election by the Board of Directors or nomination for election by the Company's stockholders was approved or recommended by a vote of at least two-thirds of the directors then still in office who either were directors on the date of this Agreement, or whose appointment, election or nomination for election was previously so approved or recommended; (iii) there is consummated a merger or consolidation of the Company or any direct or indirect subsidiary of the Company with any corporation, other than a merger or consolidation that would result in the voting securities of the Company outstanding immediately prior to such merger or consolidation continuing to represent (either by remaining outstanding or by being converted into voting securities of the surviving entity or any parent thereof) at least 60 percent of the combined voting power of the securities of the Company or such surviving entity or its parent outstanding immediately after such merger or consolidation, or a merger or consolidation effected to implement a recapitalization of the Company (or similar transaction) in which no person is or becomes the beneficial owner, directly or indirectly, of securities of the Company representing 30 percent or more of the combined voting power of the Company's then outstanding securities; or (iv) the Company's stockholders approve a plan of complete liquidation or dissolution of the Company or there is consummated the sale or disposition by the Company of all or substantially all of the Company's assets, other than a sale or disposition by the Company of all or substantially all of the Company's assets to an entity, at least 60 percent of the combined voting power of the voting securities of which are owned by the stockholders of the Company in substantially the same proportions as their ownership of the Company immediately prior to such sale.

Section 9. <u>Limitation on Indemnification</u>. Notwithstanding any other provision herein to the contrary, the Company shall not be obligated pursuant to this Agreement:

- (a) <u>Claims Initiated by Indemnitee</u>. To indemnity Indemnitee for Judgments or advance Expenses to Indemnitee with respect to an action, suit or proceeding (or part thereof) initiated by Indemnitee, including by way of cross-claim, counter-claim or the like, except with respect to an action, suit or proceeding brought to establish or enforce a right to indemnification (which shall be governed by the provisions of Section 6 and Section 9(b) of this Agreement) or a compulsory counter-claim, unless such action, suit or proceeding (or part thereof) was authorized to by a majority vote of the disinterested directors.
- (b) Action for Indemnification. To indemnify Indemnitee for any Expenses incurred by Indemnitee with respect to any action, suit or proceeding instituted by Indemnitee to enforce or interpret this Agreement, unless Indemnitee is successful in establishing Indemnitee's right to indemnification in such action, suit or proceeding, in whole or in part, or unless and to the extent that the court in such action, suit or proceeding shall determine that, despite Indemnitee's failure to establish his or her right to indemnification, Indemnitee is entitled to indemnity for such Expenses; provided, however, that nothing in this Section 9(b) is intended to limit the Company's obligation with respect to the advancement of Expenses to Indemnitee in connection with any such action, suit or proceeding instituted by Indemnitee to enforce or interpret this Agreement, as provided in Section 6 hereof.
- (c) <u>Section 16 Violations</u>. To indemnify Indemnitee for Judgments or advance Expenses to Indemnitee for payment or an accounting of profits arising from the purchase or sale by Indemnitee of securities in violation of Section 16(b) of the Securities Exchange Act of 1934, as amended, or any similar successor statute.
- (d) <u>Non-compete and Non-disclosure</u>. To indemnify Indemnitee for Judgments or advance Expenses to Indemnitee in connection with proceedings or claims involving the enforcement of non-compete and/or non-disclosure agreements or the non-compete and/or non-disclosure provisions of employment, consulting or similar agreements to which Indemnitee may be a party with the Company's parent or any subsidiary of the Company or, at the Company's instance, with any other foreign or domestic corporation, partnership, joint venture, trust or other enterprise.

- (e) <u>Clawback</u>. To indemnify Indemnitee for Judgments as a consequence of any action involving reimbursement of the Company by the Indemnitee of any bonus or other incentive-based or equity-based compensation or of any profits realized by the Indemnitee from the sale of securities of the Company, as required in each case under the federal securities laws or stock exchange requirements (including, but not limited to, any such reimbursements that arise from an accounting restatement of the Company pursuant to Section 304 of the Sarbanes-Oxley Act of 2002 (the "Sarbanes-Oxley Act"), or the payment to the Company of profits arising from the purchase and sale by Indemnitee of securities in violation of Section 306 of the Sarbanes-Oxley Act).
- (f) <u>Fraud or Personal Profit</u>. To indemnify Indemnitee for Judgments or advance Expenses to Indemnitee if it is proved by final judgment in a court of law or other final adjudication that Indemnitee was guilty of fraud or gained any personal profit or advantage to which Indemnitee was not legally entitled.
- (g) <u>Prohibited Payment</u>. To indemnify Indemnitee for Judgments or advance Expenses to Indemnitee for any Judgment that the Company is prohibited by applicable law from paying as indemnity, provided, however, nothing herein shall be deemed to restrict the ability of Indemnitee to obtain benefits under any insurance policy or policies or the Company's obligations under Section 7 to support Indemnitee's efforts to obtain such insurance benefits from the insurer(s).
- (h) Payment by Insurer. To indemnify Indemnitee for Judgments or advance Expenses to Indemnitee to the extent that Indemnitee is reimbursed pursuant to any such insurance as may exist for Indemnitee's benefit, including but not limited to insurance for directors, officers, employees or agents of the Company or for other persons for Service at the Request of the Company. Notwithstanding the availability of such other insurance, Indemnitee also may claim advancement of Expenses and indemnification from the Company pursuant to this Agreement by assigning to the Company any claims under such insurance if and to the extent Indemnitee is paid by the Company. Indemnitee shall reimburse the Company for any sums Indemnitee receives as indemnification from other sources to the extent of any amount paid to Indemnitee for that purpose by the Company.
- Section 10. <u>Certain Settlement Provisions</u>. The Company shall have no obligation to indemnify Indemnitee under this Agreement for amounts paid in settlement of any action, suit or proceeding without the Company's prior written consent, which shall not be unreasonably withheld. The Company shall not settle any action, suit or proceeding in any manner that would impose any fine, penalty or other obligation or limitation on Indemnitee without Indemnitee's prior written consent, which shall not be unreasonably withheld.
- Section 11. <u>Savings Clause</u>. If any provision or provisions of this Agreement shall be invalidated on any ground by any court of competent jurisdiction, then the Company shall nevertheless indemnify Indemnitee as to Expenses and Judgments in any action, suit or proceeding, whether civil, criminal, administrative or investigative, including an action by or in the right of the Company, to the full extent permitted by any applicable portion of this Agreement that shall not have been invalidated and to the full extent permitted by applicable law.
- Section 12. <u>Contribution</u>. In order to provide for just and equitable contribution in circumstances in which the indemnification provided for herein is held by a court of competent jurisdiction to be unavailable to Indemnitee in whole or in part, including by reason of applicable law or public policy, it is agreed that, in such event, in lieu of indemnification, the Company shall, to the fullest extent permitted by law, contribute to the payment of Indemnitee's Expenses and Judgments in any action, suit or proceeding, whether civil, criminal, administrative or investigative, in connection with any claim relating to a claim for indemnification under this Agreement. Such contribution payment shall be in the proportion deemed by any court of competent jurisdiction in which the proceeding is brought as fair and reasonable in light of all of the circumstances of such action, suit or proceeding in order to reflect (i) the relative benefits received by the Company and Indemnitee as a result of the event(s) and/or transaction(s) giving cause to such action, suit or proceeding and/or (ii) the relative fault of the Company (and its directors, officers, employees and agents) and Indemnitee in connection with such event(s) and/or transaction(s), taking into account, among other things, contributions by other directors and officers of the Company or others pursuant to indemnification agreements or otherwise. Without limiting the generality of the foregoing, the contribution contemplated by this Section 12 shall not be required where such holding by the court is due to (i) the failure of Indemnitee to meet the standard of conduct set forth in Section 2 hereof, or (ii) any limitation on indemnification set forth in Section 7(e), 9 or 10 hereof.
- Section 13. <u>Form and Delivery of Communications</u>. Any notice, request or other communication required or permitted to be given to the parties under this Agreement shall be in writing and either delivered in person or sent by e-mail, overnight mail or courier service, or certified or registered mail, return receipt requested, postage prepaid, to the parties at the following addresses (or at such other addresses for a party as shall be specified by like notice):

QEP Resources, Inc.
1050 17th Street, Suite 500
Denver, Colorado 80205
Attn: General Counsel

If to Indemnitee:

If to the Company:

Section 14. <u>Subsequent Legislation</u>. If the DGCL is amended after adoption of this Agreement to expand further the indemnification permitted to directors or officers, then the Company shall indemnify Indemnitee to the fullest extent permitted by the DGCL, as so amended.

Section 15. Nonexclusivity. The provisions for indemnification and advancement of Expenses set forth in this Agreement shall not be deemed exclusive of any other rights which Indemnitee may have under any provision of law, the Company's Certificate of Incorporation or Bylaws, or from action taken by a majority vote of the Company's shareholders or disinterested directors, other agreements or otherwise. Indemnitee's rights hereunder shall continue after Indemnitee has ceased acting as an agent of the Company and shall inure to the benefit of the heirs, executors and administrators of Indemnitee. However, no amendment or alteration of the Company's Certificate of Incorporation or Bylaws or any other agreement entered into after the date hereof shall adversely affect the rights provided to Indemnitee under this Agreement.

Section 16. <u>Enforcement</u>. The Company shall be precluded from asserting in any judicial proceeding that the procedures and presumptions of this Agreement are not valid, binding and enforceable. The Company agrees that its execution of this Agreement shall constitute a stipulation by which it shall be irrevocably bound in any court of competent jurisdiction in which a proceeding by Indemnitee for enforcement of his or her rights hereunder shall have been commenced, continued or appealed, that its obligations set forth in this Agreement are unique and special, and that failure of the Company to comply with the provisions of this Agreement will cause irreparable and irremediable injury to Indemnitee, for which a remedy at law will be inadequate. As a result, in addition to any other right or remedy Indemnitee may have at law or in equity with respect to breach of this Agreement, Indemnitee shall be entitled to injunctive or mandatory relief directing specific performance by the Company of its obligations under this Agreement.

Section 17. <u>Interpretation of Agreement</u>. It is understood that the parties hereto intend this Agreement to be interpreted and enforced so as to provide indemnification to Indemnitee to the fullest extent now or hereafter permitted by law. This Agreement is intended to be retroactive and to apply to events occurring prior to the date of this Agreement.

Section 18. <u>Entire Agreement</u>. This Agreement and the documents expressly referred to herein constitute the entire agreement between the parties hereto with respect to the matters covered hereby, and any other prior or contemporaneous oral or written understandings or agreements with respect to the matters covered hereby are expressly superseded by this Agreement.

Section 19. <u>Modification and Waiver</u>. No supplement, modification or amendment of this Agreement shall be binding unless executed in writing by both of the parties hereto. No waiver of any of the provisions of this Agreement shall be deemed or shall constitute a waiver of any other provision hereof (whether or not similar) nor shall such waiver constitute a continuing waiver.

Section 20. <u>Successor and Assigns</u>. All of the terms and provisions of this Agreement shall be binding upon, shall inure to the benefit of and shall be enforceable by the parties hereto and their respective successors, assigns, heirs, executors, administrators and legal representatives. The Company shall require and cause any direct or indirect successor (whether by purchase, merger, consolidation or otherwise) to all or substantially all of the business or assets of the Company expressly to assume and agree to perform this Agreement in the same manner and to the same extent that the Company would be required to perform if no such succession had taken place.

Section 21. <u>Service of Process and Venue</u>. Except with respect to any arbitration commenced by Indemnitee pursuant to <u>Section 6</u>, the Company and Indemnitee hereby irrevocably and unconditionally: (a) agree that

any action or proceeding arising out of or in connection with this Agreement shall be brought only in the Delaware Chancery Court, and not in any other state or federal court in the United States of America or any court in any other country; (b) consent to submit to the exclusive jurisdiction of the Delaware Chancery Court for purposes of any action or proceeding arising out of or in connection with this Agreement; (c) waive any objection to the laying of venue of any such action or proceeding in the Delaware Chancery Court; and (d) waive, and agree not to plead or to make, any claim that any such action or proceeding brought in the Delaware Chancery Court has been brought in an improper or inconvenient forum, or is subject (in whole or in part) to a jury trial.

Section 22. Governing Law. This Agreement shall be governed exclusively by and construed according to the laws of the State of Delaware, as applied to contracts between Delaware residents entered into and to be performed entirely within Delaware. If a court of competent jurisdiction shall make a final determination that the provisions of the law of any state other than Delaware govern indemnification by the Company of its officers and directors, then the indemnification provided under this Agreement shall in all instances be enforceable to the fullest extent permitted under such law, notwithstanding any provision of this Agreement to the contrary.

Section 23. <u>Employment or Board Rights.</u> Nothing in this Agreement is intended to create in Indemnitee any right to employment or continued employment or right to continued service on the Board of Directors.

Section 24. <u>Counterparts</u>. This Agreement may be executed in two or more counterparts, each of which shall be deemed to be an original and all of which together shall be deemed to be one and the same instrument, notwithstanding that both parties are not signatories to the original or same counterpart.

Section 25. <u>Headings</u>. The section and subsection headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement.

Section 26. <u>Disclosure</u>. In certain instances, applicable law (including applicable federal law that may preempt or override applicable state law) or public policy may prohibit the Company from indemnifying the directors and officers of the Company under this Agreement or otherwise. For example, the U.S. Securities and Exchange Commission has taken the position that indemnification of directors, officers and controlling persons of the Company for liabilities arising under federal securities laws is against public policy and, therefore, unenforceable. The Company has undertaken or may be required in the future to undertake with the Securities and Exchange Commission to submit the question of indemnification to a court in certain circumstances for a determination of the Company's right under public policy to indemnify Indemnitee. In addition, federal law prohibits indemnification for certain violations of the Employee Retirement Income Security Act of 1974, as amended.

Section 27. <u>Rights Continued</u>. The rights of indemnification and to receive advancement of Expenses as provided by this Agreement shall continue as to Indemnitee even though Indemnitee may have ceased to be a director or officer of the Company and shall inure to the benefit of Indemnitee's personal or legal representatives, executors, administrators, successors, heirs, distributees, devisees and legatees.

IN WITNESS WHEREOF, this Agreement has been duly executed and delivered to be effective as of the date first above written.

QEP RESOURCES, INC.
Зу
Name:
Title:
NDEMNITEE:
Name:

CERTIFICATION

I, Charles B. Stanley, certify that:

- 1. I have reviewed this Form 10-Q of QEP Resources, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

November 5, 2013

/s/ Charles B. Stanley
Charles B. Stanley
Chairman, President and Chief Executive Officer

CERTIFICATION

I, Richard J. Doleshek, certify that:

- 1. I have reviewed this Form 10-Q of QEP Resources, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

November 5, 2013

/s/ Richard J. Doleshek

Richard J. Doleshek

Executive Vice President, Chief Financial Officer and Treasurer

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with this report of QEP Resources, Inc. (the Company) on Form 10-Q for the period ended September 30, 2013, as filed with the Securities and Exchange Commission on the date hereof (the Report), Charles B. Stanley, Chairman, President and Chief Executive Officer of the Company, and Richard J. Doleshek, Executive Vice President, Chief Financial Officer and Treasurer of the Company, each hereby certifies, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that, to the best of his knowledge:

- (1) The Report fully complies with the requirements of Section 13(a) or Section 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

QEP RESOURCES, INC.

November 5, 2013

/s/ Charles B. Stanley
Charles B. Stanley

Charles B. Stainey

Chairman, President and Chief Executive Officer

November 5, 2013

/s/ Richard J. Doleshek
Richard J. Doleshek
Executive Vice President,
Chief Financial Officer and Treasurer